THE EFFECTS OF A VIRTUAL DISRUPTION ON MOTOR CONTROL AND MOTOR ADAPTATION STUDIED WITH TRANSCRANIAL MAGNETIC STIMULATION (TMS)

P. MOHAJER SHOJAII

Ph.D.

THE EFFECTS OF A VIRTUAL DISRUPTION ON MOTOR CONTROL AND MOTOR ADAPTATION STUDIED WITH TRANSCRANIAL MAGNETIC STIMULATION (TMS)

PEGAH MOHAJER SHOJAII

A thesis submitted in partial fulfilment of the requirements of the University of East London for the degree of Doctor of Philosophy

September 2019

Abstract

The main aim of this thesis was to explore the neural and behavioural responses underpinning upper-limb motor control in a novel (force-field) robot-mediated reaching task using a non-invasive brain stimulation method known as transcranial magnetic stimulation (TMS). A new TMS-based network mapping technique was used to target different regions of the motor circuit (i.e. network nodes) using a 'virtual disruption' approach.

Seven cortical regions including the left and right primary motor cortex (M1), the supplementary motor area (SMA), the left and right posterior parietal cortex (PPC) and the left and right dorsal pre-motor cortex (PMC) were targeted with TMS at nine different time points during the preparation phase of upper-limb reaching towards a north-west target (i.e. reaching away from the body). Both neural mechanisms (corticospinal excitability with left M1 stimulation) and kinematic (behavioural) responses such as, movement onset, movement offset, maximum velocity, movement duration, summed error (reaching errors quantified by the calculating the difference between the subject's reaching trajectory and the ideal reaching trajectory) and maximum force were explored offline. When exploring the impact of TMS on each cortical region individually, the results demonstrated a behavioural effect on reaching responses because 1) TMS caused a significant disruption in reaching trajectories during motor adaptation compared to normal reaching (no force-field) at most time points and 2) TMS caused a significant delay in movement onset, particularly during motor adaptation. As well as exploring the effect of TMS on each region separately, it was important to determine the network of regions that may play a more functional role in novel reaching. Therefore a comparative analysis was performed between all stimulated regions for each kinematic parameter. The comparative analysis revealed a region specific relative influence on summed error. More specifically, the left M1 and left PPC were the principle structures that were involved in novel reaching because TMS to these structures resulted in significantly greater reaching trajectory errors. Based on this finding, it can be concluded that the left M1 and left PPC play a pivotal role in the preparation phase of upper-limb novel reaching compared to other regions in the motor network, including the right M1, SMA, left and right dPMC and right PPC.

Overall, the findings from this project can not only help 1) refine our understanding of the mechanistic elements that operate during reaching and 2) gain an insight into the functional role of the different regions that are involved in novel reaching, but they also have a wide range of applications, ranging from brain machine interfaces (BMI) to neurocomputational models where data-based virtual lesions have been introduced into models of stroke patients.

Declaration

I declare that the work reported in this thesis was carried out in accordance with the regulations of the University of East London.

Pegah Mohajer Shojaii

Abstract.....iii Declaration.....v List of Accompanying Material xv Acknowledgements: xviii 1.1 1.2 1.2.1 1.2.21.3 1.3.1 Principles of TMS and its advantages of TMS compared to other imaging modalities 8 TMS protocols 12 1.3.2 1.4 1.4.1 1.4.2 The human motor system...... 19 1.5 1.5.1 1.5.2 1.5.2.2.1 1.5.2.2.2 1.5.2.2.3 1.5.2.2.4 1.6 1.6.11.7 1.7.1 1.8 1.8.1 Other regions of the motor network (the prefrontal cortex [dorsolateral] and the 1.9 2 2.1 2.2 3

Contents

| 3.1 Eth | ics | 57 |
|------------------|---|----------|
| 3.2 Par | ticipants | |
| 3.3 Tra 3.3.1 | Electromyography (EMG) recordings for MEPs: | 61 61 |
| 3.3.2 | TMS Protocol: | |
| 3.3.2.1 | Cortical targeting with TMS: | 67 |
| 3.3.2.2 | Cortical targeting of regions without an experimental output and coil pos | sitions |
| employe | :d: | |
| 3.3.2.2.1 | Left PPC location and coil position: | 68 |
| 3.3.2.2.2 | 2 SMA location and coil position: | 68 |
| 3.3.2.2.3 | Left dPMC location and coil position: | 68 |
| 3.3.2.2.4 | Right PPC location and coil position: | 69 |
| 3.3.2.2.5 | Right dPMC location and coil position: | 69 |
| 3.3.2.3 | Ensuring accurate targeting: | 69 |
| 3.4 Rol | botics | |
| 3.4.1 | Reaching task: motor reaching and motor adaptation: | |
| 3.4.2 | Reaching task set up and instructions: | 77 |
| 3.5 Syr | nchronisation | 80 |
| 3.6 Dat | ta acquisition and analysis | |
| 2.6.2 | MED _a | 00 02 |
| 3.0.2 2.7 Sta | | |
| 3.7 Sta 3.7.1 | Kinematic data (experiments 1-8): | 86 86 |
| 3.7.2 | MEP data (experiment 1 and 2) | 87 |
| 3.7.3 | Additional statistical testing: | |
| Chanter 4 | · · · · · · · · · · · · · · · · · · · | 89 |
| A Evnorin | nant 1 | 20 |
| The neurol | l and helesviewed mechanisms medicing right arm reaching mehod w | |
| delivered t | o the contralateral (left) and ipsilateral (right) primary motor cortex | |
| 4.1 Intr | roduction | |
| 4.2 Me | thodology | |
| 4.3 Dat | ta acquisition | |
| 4.3.1 | MEPS recorded with EMG: | |
| 4.3.2 | Kinematics recorded with the robot: | |
| 4.4 Sta | tistical analysis | |
| 4.4.1 | MEPs: | |
| 4.4.2 | Kinematics: | |
| 4.5 Res | MED results: Controlatoral M1 stimulation only. | |
| 4.3.1 | Where results: Contralateral W1 sumulation only: | |
| 4.5.2 | Kinematics: contralateral and ipsilateral stimulation: | |
| 4.6 Dis | cussion | |

| | 4.6.1 | MEP responses: | |
|--------|------------------|--|------------|
| | 4.6.2 | 2 Kinematics: | |
| 2 | 4.7 | Chapter conclusions: | 101 |
| Ch | apter : | 5 | 102 |
| 5 | Expe | eriment 2 | 102 |
| - | The net | ural and behavioural mechanisms mediating right arm reaching in a force-fiel | d studied |
| V | with sir | ngle-pulse TMS to the left primary motor cortex (M1) | 102 |
| 4 | 5.1 | Introduction | 102 |
| - | 5.2 | Data acquisition: MEPs and Kinematics | 103 |
| - | 5.5 5.4 | Statistical analysis: MEPs and Kinematics | 104 |
| 4 | 5.5 | Results | 104 |
| | 5.5.1 | MEPs: | 104 |
| | 5.5.2 | 2 Kinematics: | 105 |
| 4 | 5.6 | Discussion | 112 |
| - | 5.6.1 | MEP responses: | 112 |
| | 5.6.2 | 2 Kinematics: | 113 |
| | 5.7 | Chapter conclusions: | 114 |
| Ch | anter (| 6 | 116 |
| 6 | Evna | arimant 3 | 116 |
| U , | п. | | ······ 110 |
| | I he imp | pact of SP-1MS to the left posterior parietal cortex (PPC) during right arm re | aching in |
| 6 | 1 1110101 5.1 | Introduction | 110 |
| (| 5.2 | Methodology | 118 |
| (| 5.3 | Data acquisition: MEPs and Kinematics | 118 |
| (| 5.4 | Statistical analysis: Kinematics | 118 |
| (| 5.5 | Results | 119 |
| | 6.5.1 | Kinematics - parameters and trajectories of reaching: | 119 |
| (| 5.6 | Discussion | 125 |
| | 6.6.1 | Kinematics - The shift in movement onset: | 125 |
| | 6.6.2 | 2 Kinematics - Impact of TMS on trajectories of reaching (summed error) | and |
| | reach | hing velocity: | 126 |
| (| 5.7 | Chapter conclusions: | 127 |
| Ch | apter | 7 | 128 |
| 7 | Expe | eriment 4 | 128 |
|] | Explori | ing the impact of SP-TMS to the supplementary motor area (SMA) during : | right arm |
| 1 | eachin | g in a novel motor learning protocol | 128 |
| | 7.1 | Introduction | 128 |
| | 1.2 | Methodology | 130 |
| | 7.3 | Data acquisition: MEPs and Kinematics | 130 |
| , , | /.4 7.5 | Statistical analysis: Kinematics | 130 |
| | 7.5.1 | Kinematics - Repeated measures ANOVA and post-hoc testing | 130 |
| | 7.6 | Discussion | 136 |
| | 1.0 | 21040001011 | |

| 7.6.1 | 1 Kinematics | 136 |
|--------------|---|------------|
| 7.6.1 | 1.2 Shift in movement onset and offset: | 136 |
| 7.6.1 | 1.3 Impact of SMA stimulation on reaching trajectories (SE): | 136 |
| 7.7 | Chapter conclusions: | 138 |
| Chapter | 8 | 139 |
| 8 Exp | eriment 5 | 139 |
| The im | npact of SP-TMS to the right primary motor cortex (M1) during right arm reaching | in a |
| novel r | motor learning protocol | 139 |
| 8.1 | Introduction | 139 |
| 8.2 8.3 | Data acquisition: MEPs and Kinematics | 140 |
| 8.4 | Statistical analysis: Kinematics | 141 |
| 8.5 | Results | 141 |
| 8.5.1 | 1 Kinematics: Repeated measures ANOVA and post-hoc testing: | 141 |
| 8.6 | Discussion | 147 |
| 8.6.1 | Kinematics | 147 |
| 8.6. | 1.2 The shift in movement onset and offset: | 147 |
| 8.6.1 | 1.3 Right M1 connectivity in facilitating novel reaching: | 147 |
| 8.7 | Chapter conclusions: | 149 |
| Chapter | 9 | 151 |
| 9 Exp | eriment 6 | 151 |
| Explor | ring impact of SP-TMS to the left dorsal pre-motor cortex (dPMC) during right a | arm |
| reachir | ng in a novel motor learning protocol | 151 |
| 9.1 | Introduction | 151 |
| 9.2 | Data acquisition: MEPs and Kinematics | 153 |
| 9.4 | Statistical analysis: Kinematics | 153 |
| 9.5 | Results | 153 |
| 9.5.1 | 1 Kinematics: | 153 |
| 9.6 | Discussion: | 158 |
| 9.6.1 | Kinematics | 158 |
| 9.6. | 1 The role of the dPMC in reaching preparation: | 158 |
| 9.6.2 | 2 Disrupted novel reaching with dPMC stimulation: | 159 |
| 9.7 | Chapter conclusions: | 160 |
| Chapter | 10 | 162 |
| 10 E | xperiment 7 | 162 |
| Explor | ring the impact of SP-TMS to the right posterior parietal cortex (PPC) during right a | arm |
| reachir | ng in a novel motor learning protocol | 162 |
| 10.1 10.2 | Introduction | 162 164 |
| 10.2 | D. t MED 11/2 | 104 |
| 10.3 10.4 | Data acquisition: MEPs and Kinematics | 104 164 |
| 10.4 | Results | 164 |
| 10.5.1 | Kinematics | 164 |

| 10.6 | Discussion | 170 |
|---------|--|-----|
| 10.6.1 | Kinematics | 170 |
| 10.6 | 6.1.2The shift in movement onset and movement offset: | 170 |
| 10.6 | 6.1.3 Errors in reaching with right PPC TMS: | 172 |
| 10.7 | Chapter conclusions: | 173 |
| Chapter | · 11 | 174 |
| 11 E | Experiment 8 | 174 |
| Explo | ring the impact of SP-TMS to right dorsal pre-motor cortex (dPMC) during right a | arm |
| reachi | ng in a novel motor learning protocol | 174 |
| 11.1 | Introduction | 174 |
| 11.2 | Methodology | 176 |
| 11.3 | Data acquisition: MEPs and Kinematics | 176 |
| 11.4 | Statistical analysis: Kinematics | 176 |
| 11.5 | Kesults | 176 |
| 11 | 5.1 Kinematics: | 170 |
| 11.6 | Discussion | 182 |
| 11.6.1 | Kinematics | 182 |
| 11.0 | 5.1.2 The role of the right dPMC regarding motor preparation: | 182 |
| 11.6 | 6.1.3 The impact of right dPMC TMS on reaching trajectories: | 183 |
| 11.7 | Conclusion: | 184 |
| Chapter | · 12 | 186 |
| 12.1 | Introduction | 186 |
| 12.2 | Methodology | 188 |
| 12.3 | Data acquisition: Kinematics | 188 |
| 12.4 | Statistical analysis: Kinematics | 188 |
| 12.5 | Results | 189 |
| 12.5 | 5.1 Kinematics | 189 |
| 12.5 | 5.1.1 Mixed ANOVA findings: | 189 |
| 12.5 | 5.1.2 One way ANOVA and post-hoc testing findings for REGION: | 190 |
| 12.5 | 5.1.3 One way ANOVA and post-hoc testing findings for CONDITION: | 191 |
| 12.5 | 5.1.4 One way ANOVA and post-hoc testing findings for TIME: | 191 |
| 12.5 | 5.1.5 Significant interaction effects: | 191 |
| 12.6 | Discussion: | 199 |
| 12.6.1 | Kinematics: | 199 |
| 12.6 | 6.1.2 Region specific relative influence defined by summed error | 199 |
| 12.6 | 6.1.3 Evidence for hemispheric asymmetries during motor adaptation | 203 |
| 12.6 | 6.1.4 Regions and conditions impacted for maximum velocity | |
| (RE | GION*CONDITION) | 203 |
| 12.6 | 6.1.5 Limitations to consider | 204 |
| 12.7 | Conclusion: | 206 |
| Chapter | : 13 | 207 |
| Final o | conclusion and Future directions | 207 |
| REFER | ENCES | 212 |

| Appendix 1: Ethical approval. | . 266 |
|--|-------|
| Appendix 2: Participant information sheet and consent form | 268 |
| Appendix 3: Medical questionnaire | 270 |

|--|

| Chapter 1: | |
|---|----------|
| Figure 1.1: Previous imaging methods | 3 |
| Figure 1.2: TMS components | 11 |
| Figure 1.3: The corticospinal tract | 23 |
| Figure 1.4: Layers of the cortex and motor network projections: | 24 |
| Figure 1.5: The primary motor cortex | 28 |
| Figure 1.6: TMS to the human motor cortex | 29 |
| Figure 1.7: The effect of coil directionality on responses | 30 |
| Figure 1.8: Patient MEP responses | 35 |
| Figure 1.9: PPC location and a schematic presentation of neural pathways between motor | |
| regions and their contribution to motor behaviours | 40 |
| Figure 1.10: SMA location and coil position for targeting this region | 14 |
| Figure 1.11: PMC location | 18 |
| Figure 1.12: DLPFC and cerebellum location | 52 |
| Chapter 3: | |
| Figure 3.1: Muscles of interest chosen for electrode placement (outlined in red) | 54 |
| Figure 3.2: Visualisation of surface EMG recordings of MEPs | 55 55 |
| Figure 3.3: FEG locations for stimulated regions and optimal coil positions for the different | ,,, |
| locations targeted with TMS | 72 |
| Figure 3.4: Graphical illustration of the paradigm implemented for experiment 1 | 75 |
| Figure 3.5: Graphical illustration of the reaching paradigm implemented for experiments 2-8: 7 | 76 |
| Figure 3.6: Experimental set up and reaching task | 78 |
| Figure 3.7: Calculation for summed error | 22 |
| Figure 3.8: Protocol for MEP analysis | 32 |
| Chapter 4 | |
| Figure 4 1[A-C]: Graphical presentation of the post-hoc testing results (based on table 4.3) for | |
| the significant kinematic variables: | 97 |
| Chapter 5: | . , |
| Figure 5 1[A-F]: Graphical presentation of the post-hoc testing results (based on table 5 3) for | |
| the significant kinematic variables: | 10 |
| Chapter 6 | |
| Figure 6.1: Reaching block trajectories with TMS delivered to the left PPC in a single- | |
| narticinant | 20 |
| Figure 6 2[A-F]: Graphical presentation of the post-hoc testing results (based on table 6.2) for | 20 |
| the significant kinematic variables: | 23 |
| Chapter 7: | |
| Figure 7.1 [A-E] Graphical presentation of the post-hoc testing results (based on table 7.2) for | |
| the significant kinematic variables: | 34 |
| Chapter 8: | , , |
| Figure 8.1 [A-E]: Graphical presentation of the post-hoc testing results (based on table 8.2) for | |
| the significant kinematic variables: | 15 |
| Chapter 9. | 1.5 |
| Figure 9.1 [A-D] Graphical presentation of the post-hoc testing results (based on table 9.2) for | r |
| the significant kinematic variables: | 57 |
| Chapter 10: | - 1 |
| Figure 10.1 [A-E] Graphical presentation of the post-hoc testing results (based on table 10.2) | |
| for the significant kinematic variables: | 58 |
| | |

Chapter 11:

| Figure 11.1 [A-E]: Graphical presentation of the post-hoc testing results (based on table 11.2) |
|---|
| for the significant kinematic variables: |
| Chapter 12: |
| Figure 12.2 [C-E]: A graphical illustration of post-hoc testing results for summed error TMS |
| responses between regions for different reaching conditions |
| Figure 12.4 [A-C].Graphical illustration of REGION*CONDITION interaction effects in TMS |
| blocks: |
| Figure 12.5 [A-B]. Graphical illustration of REGION*TIME interaction effects in TMS blocks: |
| |

List of Tables

| <u>Chapter 3:</u> | |
|---|------|
| Table 3.1: Inclusion and exclusion criteria for participation | . 59 |
| Table 3.2: Participant demographics and experimental details for each region targeted in this | |
| thesis | . 60 |
| Chapter 4: | |
| Table 4.1: Average peak-to-peak MEP amplitudes during FAM reaching at different time | |
| points, and results of RMANOVA testing: | . 94 |
| Table 4.2. Two-way RMANOVA results for kinematic variables during contralateral and | |
| ipsilateral M1 stimulation | . 95 |
| Table 4.3. Post-hoc testing following significant RMANOVA findings: | . 96 |
| Chapter 5: | |
| Table 5.1. Results of three-factorial RMANOVA for peak-to-peak MEP responses, followed | by |
| post hoc testing: | 107 |
| Table 5.2. Results of the two-way RMANOVA for kinematics: | 108 |
| Table 5.3. Post-hoc testing results | .09 |
| Chapter 6: | |
| Table 6.1. Results of the two-way RMANOVA: | 121 |
| Table 6.2. Post-hoc testing results | 122 |
| Chapter 7: | |
| Table 7.1. Results of the two-way RMANOVA: | 132 |
| Table 7.2. Results following post-hoc testing | 133 |
| Chapter 8: | |
| Table 8.1. Results of the two-way RMANOVA: | 143 |
| Table 8.2. Post-hoc testing results: | 144 |
| Chapter 9: | |
| Table 9.1. Results of the two-way RMANOVA: | 155 |
| Table 9.2. Post-hoc testing results: | 156 |
| Chapter 10: | |
| Table 10.1. Results of the two-way RMANOVA: | 166 |
| Table 10.2. Post-hoc testing results | 167 |
| Chapter 11: | |
| Table 11.1. Results of the two-way RMANOVA: | 178 |
| Table 11.2. Post-hoc testing results: | 179 |
| Chapter 12: | |
| Table 12.1. Results of the three-factorial mixed ANOVA for TMS blocks of reaching: | 192 |
| Table 12.2A. REGION specific summed error values (mean $[\pm SEM]$) for the different reach | ing |
| conditions, followed by a one-way between subjects ANOVA for summed error (collapsed for | or |
| TIME): | 193 |
| Table 12.2B: Post hoc testing results for summed error TMS (collapsed for TIME) (REGION | 1 |
| main factor): | 194 |
| | |

List of Accompanying Material

This page details the accepted publications during the PhD period prior to thesis submission.

- Pizzamiglio, S., Desowska, A., Shojaii, P., Taga, M. and Turner, D. L. (2017) 'Muscle co-contraction patterns in robot-mediated force field learning to guide specific muscle group training', *NeuroRehabilitation*, 41(1), pp. 17-29. doi: 10.3233/NRE-171453.
- Shojaii, P. and Turner, D. L. (2019) 'The posterior parietal cortex has a greater role than the supplementary motor area in novel motor behaviour: A TMS-based virtual disruption study', *Brain Stimulation: Brain, Translational, and Clinical Research in Neuromodulation*, 12(2), pp. 432 433. doi: 10.1016/j.brs.2018.12.402.

Abbreviations

- ANOVA Analysis of Variance
- AP Anterior Posterior
- BB Biceps Brachii
- **BMI** Brain Machine Interfaces
- BOLD Blood Oxygen Level Dependent
- CED Cambridge Electronic Design
- CNS Central Nervous System
- **CSE** Corticospinal Excitability
- CSF Cerebrospinal Fluid
- CST Corticospinal Tract
- CT Computed Tomography
- DAI Diffused Axonal Injury
- DLPFC Dorsolateral Prefrontal Cortex
- dPMC Dorsolateral premotor Cortex
- **DTI** Diffusion Tensor Imaging
- ECR Extensor Carpi Radialis
- EEG Electroencephalography
- **EMG** Electromyography
- FAM Familiarisation
- FCR Flexor Carpi Radialis
- FDI First Dorsal Interosseous
- $FF-Force\mbox{-field}$
- fMRI Functional Magnetic Resonance Imaging
- GABA_A gamma-aminobutyric acid A
- GABA_b gamma-aminobutyric acid B
- ICF -- Intracortical Facilitation
- IHI Interhemispheric Inhibition
- $\label{eq:ISI-Interstimulus intervals} \textbf{ISI}-\textbf{Interstimulus intervals}$
- LICI Long-interval Intracortical Inhibition
- LIHI Long Latency Interhemispheric Inhibition
- LM Lateral Medial
- M1 Primary Motor Cortex
- MA Motor Adaptation
- $\boldsymbol{MEP}-\boldsymbol{Motor}\;\boldsymbol{Evoked}\;\boldsymbol{Potential}$

- MEP Motor Evoked Potential
- MIT Manus Interactive Motion Technologies
- MNI Montreal Neurological Institute
- MRI Magnetic Resonance Imaging
- MRS Magnetic Resonance Spectroscopy
- MS Multiple Sclerosis
- mV Millivolt
- NFF Null-field Reaching
- NHS National Health Service
- PA Posterior Anterior
- PEG Pneumoencephalography
- PET Positron Emission Tomography
- **PMC** Premotor Cortex
- **PPC** Posterior Parietal Cortex
- **PPR** Posterior Parietal Reach
- **PP-TMS** Paired Pulse TMS
- rCBF Regional Cerebral Blood Flow
- RMANOVA Repeated Measures Analysis of Variance
- RMT Resting Motor Threshold
- rTMS Repetitive Transcranial Magnetic Stimulation
- SEF Supplementary Eye-field
- SEM Standard Error of the Mean
- SENIAM Surface Electromyography for the Non-invasive Assessment of Muscles
- SICF Short Intracortical Facilitation
- SICI Short-interval Intracortical Inhibition
- SIHI Short Latency Interhemispheric Inhibition
- SMA Supplementary Motor Area
- SP-TMS Single Pulse TMS
- SPSS Statisical Package for Social Sciences
- TB Triceps Brachii
- TBI Traumatic Brain Injury
- TMS Transcranial Magnetic Stimulation
- TTL-Transistor-Transistor-Logic
- VBM Voxel Based Morphometry
- vPMC Ventral Premotor Cortex
- WO- Washout

Acknowledgements:

I would like to take this opportunity to give my sincerest gratitude to all of the people who have helped me during my studies, in particular I would like to thank my supervisor Professor Duncan Turner for his valulable guidance, support, encouragement and advice throughout my PhD. I would also like to thank my second supervisors, Dr. Azmy Faisal and Dr. Arietta Spinou for their advice, time and help during my studies.

There are not enough words for me to express my graditude to my family. A huge thank you to my mum, dad, brother Pej and my aunty Lisa for their love and support, they raised me to never give up, provided me with the most valuable advice and comforted me in times of joy and stress! I would also like to thank my friends (for the lunch-breaks and gym-session breaks!), they have always kept an eye on me and provided me with words of encouragement when I needed it the most, and I thank you for your companionship along this journey.

A big thanks to all of the participants who kindly gave up their time to part-take in my research project, and the staff at the school of Health, Sport and Bioscience (University of East London) for their support and guidance throughout my studies.

Chapter 1: Introduction

1.1 Brief overview

The main aim of this thesis was to explore neural mechanisms and behaviours underpinning upper-limb motor control in a novel (force-field) robot-mediated reaching task using transcranial magnetic stimulation (TMS). A TMS-based network mapping technique was used to target different regions of the motor circuit (i.e. network nodes) using a 'virtual disruption' approach.

In order to study the neural mechanisms and behavioural responses of upper-limb reaching a laboratory-based set up was developed. A robot-mediated task was used for the reaching paradigm (i.e. to explore reaching performance), and a single-pulse TMS protocol was used to explore whether a disruption in neural activity can affect reaching performance. Generally, the neural mechanisms of reaching were investigated by acquiring electromyography (EMG) signals and the behavioural performance of reaching were investigated by acquiring kinematic data. The use of TMS and robotic reaching in an open-based environment in this project overcomes the general challenges when exploring reaching behaviours with other imaging modalities. For example, techniques such as functional magnetic resonance imaging (fMRI) and positron emission tomography (PET) have been confined to exploring reaching-to-point tasks and reaching-to-grasp tasks (as explored further on in this thesis).

Overall, this project will aim to provide novel findings into the effects of single-pulse TMS during preparation for novel reaching, which can help; 1) refine our understanding of the mechanistic elements that operate during reaching, and 2) gain an insight into the functional role of the different regions that are involved in novel reaching (i.e. whether one region in the motor circuit plays a greater functional role in novel reaching compared to another region). Findings from this project have a wide range of applications, ranging brain machine interfaces (BMI) to neurocomputational models where data-based virtual lesions have been introduced into patient models.

Literature Review

1.2 History of neuroimaging

Neuroimaging has been defined as 'methods of brain imaging that help answer questions about brain structure and function' (Bandettini, 2009). Neuroimaging methods have developed over the last century from invasive methods such as ventriculography and pneumoencephalography (PEG) (Hoeffner *et al.*, 2012) to non-invasive methods, such as magnetic resonance imaging (functional [fMRI] and structural), magnetic

resonance spectroscopy (MRS), positron emission tomography (PET), electroencephalography (EEG) and transcranial magnetic stimulation (TMS - which was employed in this thesis).

1.2.1 Invasive methods

1.2.1.2 Ventriculography and PEG

The discovery of X-rays by Roentgen in the mid 1890s (Sansare, Khanna and Karjodkar, 2011) which were first used to explore bone structure led to the development of invasive imaging methods, including ventriculography and pneumoencephalography (PEG) to explore brain structure (Tubiana, 1996; Hoeffner *et al.*, 2012). Although X-rays successfully illustrate bone deterioration, as an imaging method it is poor in terms of demonstrating changes in intracranial tissue (Schuller, 1912). However, the invention of air ventriculography by Dandy during the 1910s enabled scientists to overcome this limitation. The technique was based on injecting air (via lumbar puncture) into the ventricles in order to outline intracranial pathologies on an X-ray image (Hoeffner *et al.*, 2012) (figure 1.1A & B – Dandy, 1918). Although ventriculography methods increased the rate of tumour diagnosis, its invasiveness was associated with various after-effects, ranging from headaches to nausea (Hoeffner *et al.*, 2012).

Similar to ventriculography in terms of its invasiveness, PEG was a method that drained cerebrospinal fluid (CSF) via lumbar puncture and replaced it with air for structural diagnostic purposes (Tondreau, 1985). The technique enabled an early diagnoses of hydrocephalous but could be fatal in terms of the various complications it could lead to (Bohn, 1937; Moseley, Loh and du Boulay, 1977; Tondreau, 1985). Figure 1.1C demonstrates the various types of symptoms that were reported in over 40 patients who had undergone PEG (White, Bell and Mellick, 1973). Although effective in terms of their diagnosis of structural pathology, the detrimental after-effects of ventriculography and PEG discouraged both patients and neurosurgeons and as a result both methods were abandoned. Nonetheless, its invasive nature helped improved radiographic techniques and build the foundations for non-invasive neuroimaging methods, including computerised tomography (CT).





(A) Drill used for puncture creation (Dandy, 1918), (B) Air ventriculography needle (Dandy, 1918) and (C) Symptoms reported following PEG procedures in over 40 patients (based on data from White, Bell and Mellick, 1973).

Ventriculogaphy includes:

- (A) Lumbar puncture with a drill (Dandy, 1918)
- (B) Followed by the use of a needle to remove CSF in the anterior fontanelle and replace it with air (Dandy, 1918).
- (C) The various symptoms exhibited following PEG is demonstrated, whilst headaches and irregular heart rates were the most common symptoms reported, other symptoms included fevers, stiff neck, vomiting, neurological symptoms and changes in consciousness (based on data from White, Bell and Mellick, 1973).

1.2.2 Non-invasive methods

1.2.2.1 Computed tomography (CT) imaging

CT was developed by Hounsfield during the early 1970s, the method collects readings from the skull at multiple angles via a rotating X-ray tube (Hounsfield, 1973; Hounsfield 1980; Wellington and Vinegar, 1987; Kotwaliwale *et al.*, 2011). Image slices of the cranium are then generated from these readings via a back-projection formula (Cunningham and Judy, 2000; Wellington and Vinegar, 1987). The main advantage of CT compared to conventional X-rays is its ability to explore anatomy through different angles as opposed to only one (conventional X-ray) (Wellington and Vinegar, 1987).

Despite CT scans being less physically invasive (compared to PEG) as no incisions are made during scanning, researchers have nonetheless argued that the method poses a radiation risk (Brenner and Hall, 2007; Berrington De Gonzalez *et al.*, 2009). Berrington De Gonzalez *et al.*, (2009) explored the adverse effects of CT radiation, including the risk of cancer and concluded that roughly 2% of future cancer diagnoses could be a result of radiation from CT scanning. Despite these concerns, it should be taken into account that such findings are general and not specific enough, i.e. they do not report the quantity of doses that can lead to the risk of cancer (McCollough, Guimarães and Fletcher, 2009). Additionally, radiographers as well as equipment manufacturers have employed different strategies, including ALARA (as low as reasonably achievable) to keep radiation doses low (McCollough, Guimarães and Fletcher, 2009; Mayo-Smith *et al.*, 2014) and by following such principles, participant radiation exposure during CT scans can be reduced (Mayo-Smith *et al.*, 2014).

The overall positive impact that CT scanning has had within the neuroimaging field has helped increase the rates of clinical diagnoses (McCollough, Guimarães and Fletcher, 2009). It has also helped pave the way for other imaging methods, including PET as both are usually combined (Townsend, 2008; Papathanasiou *et al.*, 2011).

1.2.2.2 PET imaging

PET is a non-invasive imaging technique and was first used in humans during the mid-1970s to explore physiological functioning (Hoffman *et al.*, 1976; Berger, 2003). The first stage of PET scanning involves the injection of radio-tracers into the participant's peripheral vein (Berger, 2003; National Health Service, 2018). The tracer that is used depends on what is being studied. For example, ¹⁸F-fluorodeoxyglucose (FDG) is commonly used to diagnose brain tumours (Kosaka *et al.*, 2008). The pronounced glucose metabolism of tumor cells can result in an increase of FDG uptake during PET scanning and thus help diagnose and characterise tumours (Shukla and Kumar, 2006).

PET imaging studies have also helped demonstrate regions of the brain associated with upper-limb reaching. For example, during visually guided arm-reaching in non-human primates, Picard and Strick (2003) found significantly increased 2-deoxyglucose (2DG) uptake in the supplementary motor area (SMA) and the primary motor cortex (M1), but not in dorsal and ventral regions of the cingulate motor area. PET studies exploring the reaching preparation phase have revealed increased metabolic activity in the prefrontal cortex and areas of the parietal lobe (Decety *et al.*, 1992). PET techniques during novel reaching (i.e. force-field paradigms) have also illustrated greater regional cerebral blood flow (an indicator of metabolic activity) in the cerebellum during the initial stages of reaching adaptation, however this was reduced following repeated reaching exposure (i.e. as reaching errors reduced, activity in the cerebellum also reduced; Nezafat, Shadmehr and Holcomb, 2001).

Despite PETs clinical efficacy and use in the research setting, there are some limitations to the technique. For instance, those who do undergo PET imaging should avoid contact with pregnant women as well as children as they can still remain radioactive for up to a couple of hours following scanning (National Health Service, 2018). The expensiveness of the technique also makes it difficult for some patients to have access to (Griffeth, 2005; Bateman, 2012; Akbari Sari *et al.*, 2013). Nonetheless, PET remains to be effective in terms of 1) high quality imaging, 2) increasing the precision of diagnoses and 3) having shorter experimental protocols, thus making it a more patient friendly method (Bateman, 2012).

1.2.2.3 MRI imaging (structural and functional)

MRI brain imaging (both structural and functional [fMRI]) is a common non-invasive method used in both research and clinical fields. MRI functions by using the body's magnetic elements to create in depth images of organs, including the brain (Berger, 2002). MRI machines employ a magnetic field which can alter the ways in which the body's protons react (specifically hydrogen nuclei protons). The powerful MRI magnet leads to the alignment of protons, which in turn creates a magnetic vector along the axis of an MRI scanner (Berger, 2002). When additional radio-wave energy is added to the machine, the aligned magnetic vector gets deflected and the protons resonate, i.e. they

spin out of an equilibrium state. When the radiofrequency is turned off a signal is released and the protons return back to a resting state, MR images are then created from this emitted signal (Berger, 2002).

Structural MRIs have helped enhance the anatomical understanding of brain matter and regions and have been used as a form of clinical assessment among patients (Frisoni *et al.*, 2010; Chen, Jiao and Herskovits, 2011). For example, diagnostic markers such as reduced hippocampal volume and medial-temporal atrophy have been consistently shown in structural MRIs and have been hypothesised to be the initial structural indicators of first stage Alzheimer's disease (AD) (Kantarci and Jack, 2003; de Leon *et al.*, 2004; Devanand *et al.*, 2012; Vijayakumar and Vijayakumar, 2013). Such imaging used for a clinical purpose can help delay the onset of diseases at an early stage, for example, through pharmacological interventions (e.g. balancing out the neurochemical disruption with drugs for AD; Yiannopoulou and Papageorgiuo, 2012).

As opposed to structural MRI, fMRI places greater emphasis on exploring brain function and activation patterns. Functional MRI methods have been used in cognitive, behavioural and motor and clinical studies (to observe changes in brain function in neurological diseases) (Glover, 2011). With regards to its clinical applications, surgeons have used the technique to map out and determine eloquent brain locations (e.g. lingual and motor regions) that may reside next to brain tumours or lesions in order to preserve their functions during neurosurgery (Beers and Federico, 2012; Mahdavi *et al.*, 2015; Nadkarni *et al.*, 2015).

When a brain region is activated by a particular task, it results in greater neuronal activity and blood flow to the region that is facilitating the activity (Buxton, 2013), and this can be demonstrated online and offline with fMRI. Therefore, fMRI methods are used to explore changes in blood oxygen level-dependent (BOLD) signals that occur when the brain state is altered e.g. from rest to a task (Gore, 2003; Glover, 2011). For example, in an fMRI motor observation task conducted by Buccino *et al.*, (2001) participants were instructed to observe motor behaviours performed by other individuals, including arm reaching, hand grasping and mouth motions. Their findings revealed varied neural activity (i.e. cerebral blood blow) in different regions of the premotor cortex (PMC) depending on the movement that was performed. For instance, arm reaching and hand grasping movement led to greater neural activity in the dorsal regions of the PMC, whereas mouth actions resulted in increased neural activity in the

ventral PMC. Other activated regions during reaching includes the M1. For instance, in an fMRI reaching-outward task using non-ferromagnetic equipment, Eisenberg *et al.*, (2011) reported increased BOLD related-activity in the M1. Increased M1 activity during reaching has been argued to be related to its functional role in encoding movement parameters, such as the direction of movement, arm position and movement speed (Ashe and Georgopoulous, 1994; Fu *et al.*, 1995; Moran and Schwartz, 1999; Teka *et al.*, 2017). fMRI studies have also helped identify cortical regions that are involved in other aspects of reaching, such as planning and preparing movements. For instance, Andersen and Buneo (2002) reported increased posterior parietal cortex (PPC) neural activity that was associated with movement planning of a reach during a memory-guided reaching task. Increased BOLD activity has also been noted in parietal regions during reaching: 1) observation, 2) imagery and 3) execution (Filimon *et al.*, 2015).

Both fMRI and PET studies have illustrated that a large network of regions are involved in reaching preparation, planning and execution, with neural connections ranging from the parietal lobe (e.g. the PPC) to the frontal lobe (e.g. M1, PMC and SMA) (Rizzolatti and Luppino, 2001; Begliomini et al., 2014). Although fMRI and PET techniques have been useful in identifying cortical regions that are involved in reaching, there are limitations to consider. For instance, signal dropouts may occur during scanning which can affect the degree of BOLD signal found in fMRI studies (Glover, 2011). Furthermore, the confined parameters of an MRI scanner can cause set-backs in studying reaching behaviours, and can also result in movement artefacts (Culham, Cavina-Pratesi and Singhal, 2006). As a result, reaching fMRI and PET studies have mainly focused on 'point' reaching and 'grasp' reaching paradigms compared to armreaching paradigms (Culham, Cavina-Pratesi and Singhal, 2006). It can be argued that upper-limb reaching tasks may lead to novel findings in terms of identifying areas of cortex that are functionally related to reaching. This thesis will therefore introduce a new TMS mapping technique by targeting different regions of the motor circuit via a 'virtual disruption' approach during upper-limb reaching. The high temporal resolution of TMS and it use in disrupting regional functioning sets it apart from other noninvasive imaging methods, including fMRI and PET which are more correlation based and lack temporal resolution (Hallett, 2000).

In this thesis, TMS was used to explore neural behavioural responses during upper limb reaching. Section 1.3 below outlines the method in depth, in terms of: 1) its principles, 2) types of TMS coils, 3) types of TMS protocols, 4) how it has been used in both the clinical and research field, and 5) how it will be employed in this thesis.

1.3 TMS

1.3.1 Principles of TMS and its advantages of TMS compared to other imaging modalities

The non-invasive brain imaging method that this thesis is based on is TMS. TMS hardware includes a coil(s) which is connected to a stimulator. The stimulator contains; 1) capacitors that produce pulses, 2) voltage sources that produce the magnetic field, 3) switches to turn currents on and off, and 4) a pulse circuit to determine the type of TMS pulse shape that is administered (e.g. monophasic [employed in this thesis] or biphasic) (Wagner, Valero-Cabre and Pascual-Leone, 2007; Farzan *et al.*, 2016) (these components are illustrated in Figure 1.2A - Farzan *et al.*, 2016).

The key component of TMS is the coil that is used for stimulation, which operates on the principle of Faraday's law of electromagnetic induction (Eldaief, Press and Pascual-Leone 2013; Farzan *et al.*, 2016). The principle is based on the notion that the pulse of the current which passes through copper wires in a TMS coil results in a magnetic field that is perpendicular to the plane of the coil and as a result this causes a secondary induced electrical current in cortical regions that are parallel to the orientation of the coil (Hallett, 2000; Hallett, 2007; Ridding and Rothwell, 2007; Bolognini and Ro, 2010; Eldaief, Press and Pascual-Leone, 2013; Farzan *et al.*, 2016) (see figure 1.2B - Hallett 2000; Hallett, 2007). This induced TMS-current can briefly modulate neuronal activity in structures that generate action potentials and can result in either enhancing or inhibiting neural excitability (Kobayashi and Pascual-Leone, 2003; Eldaief, Press and Pascual-Leone, 2016). Section 1.5.2.1 highlights specific details with regards to how TMS works and the cortical components activated during stimulation.

It should be noted that the effect of the induced secondary current does not only rely on the TMS coil and the stimulation intensity, but also other factors including cerebral matter (Hallett, 2000; Lefaucheur *et al.*, 2014; Klomjai, Katz and Lackmy-Vallée, 2015). For example, it has been suggested that grey matter has greater levels of TMS resistance/impedance compared to white matter (Lefaucheur *et al.*, 2014; Klomjai, Katz and Lackmy-Vallée, 2015). Based on this notion, the electrical currents that are induced in grey matter structures are weaker and as a result TMS does not activate deep-cortical grey matter structures including the thalamus, but on the other hand activates superficial cerebral white matter layers of the cortex, such as the primary motor cortex (M1) (Lefaucheur et al., 2014; Klomjai, Katz and Lackmy-Vallée, 2015). There are also action at distance limitations to consider with TMS, which is based on the notion that the strength of the electrical field induced with TMS differs depending on the distance between the coil and the area being targeted (Deng, Lisanby and Peterchev, 2013). Action to distance limitations can however be overcome through the use of different coil types. For example, more recently designed TMS coils can be used to target deeper cortical structures. The round and figure-8 shaped coil (used in this thesis) provides a focal and localized current flow to explore regions on the surface of the cortex such as the M1. This coil has been used in different single pulse TMS protocols to investigate motor function in reaching (e.g. Hunter, Sacco and Turner, 2011; Orban de Xivry et al., 2013). The double cone coil on the other hand is able to penetrate into deeper cortical structures and has been used to target structures including the anterior cingulate, the cerebellum and the middle cingulate cortex (Ueno, Tashiro and Harada, 1988; Ugawa et al., 1995; Hallet, 2007; Deng, Lisanby and Peterchev, 2013; Hardwick, Lesage and Miall, 2014; D'Agata et al., 2015). Additional TMS coils includes the batwing coil. Due to its geometric features, it has been employed in a similar way to the double cone coil for the stimulation of areas deep within the cortex (Cai et al., 2011). Researchers have used this coil to target deeper muscle representations in Penfield's homunculus, such as lower limb areas including the leg (Roy and Gorassini, 2008). The H-coil has also been used in TMS experiments, namely repetitive (r) TMS (rTMS) protocols to stimulate areas that are deep within the temporal lobe (Gersner et al., 2016; Tendler et al., 2017). The "depth-focality trade off" described by Deng, Lisanby and Peterchev (2013) is a limitation that should be considered when using deeper coils to study neural responses. Deng, Lisanby and Peterchev (2013) argued that different coils have different electrical field features, and coils used to target deeper structures have limited focality. The figure-8 shaped coil, has greater focality and the electrical field induced with this coil is confined to targeting superficial areas (Deng, Lisanby and Peterchev, 2013). In this thesis, the depth-focality trade off is less of a concern considering that only superficial targets were explored with a figure-8 coil. The availability and development of various

TMS coils has helped researchers gain a deeper understanding into the depth-focality trade off, which has led to the design of novel coils to study human behaviour (Deng, Lisanby and Peterchev, 2013).

As well as the availability of various TMS coils, an additional advantage of TMS as method is based on its safety principles which has been noted in studies among healthy subjects. For example, Machii et al., (2006) applied rTMS to a range of cortical regions, such as the cerebellum, dorsolateral prefrontal cortex, the occipital lobe and areas of the parietal cortex in 200 healthy subjects. Their findings revealed that the most common negative effect of TMS included headaches, with the pain being very mild (as quantified by ratings on a questionnaire). Neck pains were also common, however this depended on the region that was stimulated (the cerebellum was associated with greater neck pain compared to areas such as the parietal cortex). In addition to this, they reported no siezures in the subjects that were tested (Machii et al., 2006). The thermal impact of TMS is also a factor to consider as some protocols may cause tissue heating (Rossi et al., 2009; Sabouni, Honrath and Khamechi, 2017). Tissue heating can become a cause for concern among individuals who have implanted electrodes (Roth et al., 1992; Rossi et al., 2009). It should be noted that in this thesis, only SP-TMS was delivered and studies have revealed that tissue heating with such protocols are minimal (Ruohonen and Ilmoniemi, 2002 cited in Pascual-Leone et al., 2002; Rossi et al., 2009). TMS has also been reported to be a pain-free method, particularly when compared to other stimulation methods, such as transcranial electric stimulation (TES) (Rossi et al., 2009). During TES the electrical charge projects to the scalp and flows through the skull (Rossi et al., 2009) and due to the the skull having low levels of conductivity, TES currents are commonly set to a high density level to stimulate neurons, which can therefore cause pain (Rossi et al., 2009). With TMS however, current density in the scalp to- current density brain is lower, which results in less pain during stimulation (Rossi et al., 2009). Ethical and safety principles are consistently growing in the field of TMS research in order to identify the possible side effects of the method and factors to consider during stimulation, whether this be in healthy controls or clinical populations. Generally, the non-invasive nature of TMS compared to other methods such as TES, and its safety record has made it a popular technique to use in research and in the treatment of psychiatric conditions (Padberg et al., 2002; Rossi et al., 2009).



B.

Figure 1.2: TMS components

А.

- (A) The different components in a stimulation unit includes; a coil (L), a voltage device (V) as well as a switch (S) and a capacitor (C). Other elements include a Thyristor (T) and a resistor (R) (figure from Farzan *et al.*, 2016)
- (B) During stimulation, a pulse is created in copper wires that are located in the magnetic coil, this produces a magnetic field with lines of a magnetic flux that pass perpendicularly to the coil. Following this, an electrical field is produced which is perpendicular to the magnetic field (figure from Hallett, 2000).

1.3.2 TMS protocols

When TMS is delivered to one hemisphere of the M1 (e.g. left), a corticospinal neural volley is elicited and this can be recorded as a motor evoked potential (MEPs) (Hallett, 2007) in muscles in the contralateral limb (e.g. right) (section 1.5.2.1 outlines the procedure in full - figure 1.6: Klomjai, Katz and Lackmy-Vallée, 2015). When MEPs are elicited by TMS, corticospinal excitability can be explored. Corticospinal excitability (CSE) is based on the affects that different excitatory and inhibitory mechanisms have on the corticospinal tract (CST) (Hallett, 2007; Cirillo, Todd and Semmler, 2011; Hunter, Sacco and Turner, 2011; Rossini et al., 2015). Such mechanisms include, short-interval intracortical inhibition (SICI), short intracortical facilitation (SICF) and IHI (interhemispheric inhibition [short latency - SIHI; and long latency LIHI]) and these can be studied with different TMS protocols. TMS protocols include single-pulse TMS (SP-TMS), paired-pulse TMS (PP-TMS) and repetitive TMS (rTMS). Whilst SP stimulation denotes a type of TMS that is not repetitive, during rTMS stimulation a train of single pulses are administered to a specific brain region at a specific frequency (Mishra et al., 2011). Low frequency rTMS stimulation inhibits neuronal firing, whereas high frequency rTMS stimulation facilitates neuronal firing (Mishra et al., 2011). rTMS protocols have been used in the treatment of psychiatric conditions, ranging from depression to schizophrenia (Padberg et al., 2002; Aleman, Sommer and Kahn, 2007; Rossi et al., 2009; Farzan et al., 2012; George, Taylor and Short, 2013).

PP-TMS can function either through the same coil at one cortical location or alternatively through two coils at two cerebral loci. PP-TMS protocols have enabled researchers to examine cortical networks in the brain, such as SICI, SICF and IHI (Rossini *et al.*, 2015). There is a greater understanding of SICI, ICF and SIHI compared to LIHI. SICI suppresses cortical cell firing in the M1 (i.e. cortical inhibition; Wagle-Shukla *et al.*, 2009; Rossini *et al.*, 2015) and has also been linked to neurotransmitters in the M1 such as gamma-aminobutyric acid-A (GABA₄) (Di Lazzaro *et al.*, 2006). ICF on the other hand, is based on excitatory cortical processes and has been linked to glutamate receptors with excitatory functions (Noda *et al.*, 2017). For example, increases in glutaminergic activity has been associated with an increase in SICF (Smith et al., 1999; Kapogiannis and Wassermann, 2009; Wagle-Shukla *et al.*, 2009).

With regards to IHI, one hemisphere of the brain suppresses the functioning of the other hemisphere (Ni et al., 2009) and this can also be studied with a PP-TMS protocol. Ferbert et al., (1992) were the first researchers to report IHI. They found that TMS applied to the ipsilateral M1, results in a reduced MEP in the contralateral M1 (i.e. inhibited). IHI has also been reported between different cortical regions. For instance, Mochizuki, Huang and Rothwell (2004) demonstrated that a conditioning pulse over the dorsal premotor cortex (dPMC) led to a reduced MEP response in the contralateral M1. This illustrates that IHI is not region specific, but rather a widely distributed phenomena found in different cortical structures (Ni et al., 2009). There are two IHI phases; short latency IHI (SIHI) and long latency IHI (LIHI), which occur when using different interstimulus intevals (ISIs). An ISI of approximately 10 milliseconds between the conditioning pulse and test pulse results in SIHI, whereas an ISI of 40-50 milliseconds between the conditioning pulse and the test pulse leads to LIHI (Ni et al., 2009). Although the mechanistic elements of SIHI are understood, for example, it has been reported to inhibit unwanted reflex actions, the role of LIHI is yet to be determined (Hubers, Orekhov and Ziemann, 2008; Morishita et al., 2014).

These neural mechanisms have also been studied with TMS during motor tasks, i.e. upper limb reaching and motor adaptation. Before considering how these mechanisms respond to motor tasks, it is important to define what is meant by motor reaching and motor adaptation.

1.4 Motor behaviours

1.4.1 Upper limb reaching

Motor behaviours fall into one of two categories; discrete or continuous. Upper limb reaching has been classed within the discrete category as it has a defined beginning and end point (Muratori *et al.*, 2013). The concept of 'fine' and 'gross' motor skills should also be considered when defining reaching. While fine motor skills refer to actions that use small muscles (i.e. hands), gross motor skills are actions that use larger muscles (i.e. upper arm and the trunk of the body; Muratori *et al.*, 2013). Motor reaching can involve both fine and gross motor skills. For instance, if someone is reaching for a cup whilst standing, both small (e.g. hand and fingers) and large muscles (e.g. torso and upper limb) will be used to carry out the action (Muratori *et al.*, 2013). Carr and Shepherd (2000) noted that the environment in which motor skills are performed can be classed as

either open-based or closed-based. They defined skills in the closed-based category as actions that can begin and stop at any given time, whilst the environment remains constant. During open-based motor performance however, it is essential for the individual carrying out the task to adapt to the changes in their environment to be successful. Such skills can be seen when someone is catching a ball, i.e. they must adjust their body to the time at which the ball is arriving (Carr and Shephard, 2000; Muratori *et al.*, 2013).

In this thesis, upper limb reaching will involve both gross and fine muscles as the hand will be used to grasp the joystick on the robotic manipulandum and upper limb muscles will be used to carry out the reaching motion. Also, the environment reflects open-based motor reaching, as the participant will have to adjust their reaching speed to avoid 'late' and 'early' reaching responses (see general methodology, section 3.4.1). The role of the corticospinal tract (CST) is vital during reaching. This has been shown in both human studies (such as stroke patients) and animal models, whereby damage to the CST resulted in impaired motor function and poor reaching accuracy (Martin and Ghez 199; Maraka *et al.*, 2014).

The act of upper limb reaching begins with preparatory neural activity within pre-motor and motor cortices before actual movement of the arm (Jones, 2012). These regions (i.e. M1 and premotor cortices) help regulate timing and outputs of motor behaviour (Halsband et al., 1993; Overduin, Richardson and Bizzi, 2009; Chang et al., 2015; Panouilleres et al., 2015). Motor output during reaching takes place due to neural communication between different brain regions (Figure 1.4B demonstrates the different projections and neural connections between regions of the motor circuitry - adapted from Briggs et al., 2018). The extent of projections from different motor areas to faciliciate motor functions were highlighted by Dum and Strick (1991). They reported that the M1 receives projections from 3 main regions: the SMA, the premotor cortices (specifically the arcuate sulcus) and the frontal lobe, particularly the cingulate motor areas. Different regional projections contribute to distinct motor behaviours and this has been shown in non-human primate lesion studies. For example, Pavlides, Miyashita and Asanuma (1993) noted that damage to somatosensory cortex in monkeys impaired the acquisition of novel motor skills but not previously learned motor skills. It was therefore concluded that cortical projections between the somatosensory cortex and the M1 facilitates the learning of a novel motor skill (Pavlides, Miyashita and Asanuma,

1993). As well as complex interactions and projections between different motor regions, there are also complex mechanisms that underlie motor responses elicited by SP-TMS during reaching, and these are explained below.

1.4.1.1 TMS during reaching preparation and reaching execution

TMS protocols have been used to explore whether changes in CST function occur during reaching. TMS motor responses during reaching have been consistently demonstrated with surface electrodes placed on the muscles of interest (Hunter, Sacco and Turner 2011; Groppa *et al.*, 2012; Orban de Xivry *et al.*, 2013). Responses investigated during the preparation phase and execution phase of motor reaching include both physiological changes (e.g. MEPs) as well as movement parameter changes (e.g. movement velocity and reaction time).

SP-TMS delivered to the M1 in reaching paradigms have shown changes in MEP amplitude. For example, during the preparation phase of right-arm reaching (towards the body), Hunter, Sacco and Turner (2011) revealed that the biceps brachii (BB) exhibited larger MEP responses compared to the triceps brachii (TB). However, reaching away from the body resulted in no differences in MEP amplitude between the BB and TB. Time specific differences related to MEP amplitude have also been found. For example, MEP responses in the BB was larger compared to the TB when TMS was delivered closer to movement onset. PP-TMS studies have also shown changes in physiological responses during preparation for reaching, including a reduction in SICI and an increase in ICF prior to movement onset (Floeter and Rothwell, 1999; Reynolds and Ashby, 1999; Nikolova et al., 2006; Koch, et al., 2008a; Hunter, Sacco and Turner, 2011). Therefore, changes in the balance of SICI and ICF mechanisms in the M1 can occur before the onset of movement, even if the MEP amplitude appears not to significantly change. rTMS protocols targeting the left and right M1 during motor preparation have also demonstrated increased motor cortical excitability and decreased inhibition at approximately 140ms before movement execution (Gilio et al., 2008; Massie et al., 2014).

Other physiological changes that have been explored with TMS protocols during reaching preparation includes the cortical silent-period. As previously mentioned, TMS to the M1 results in muscle contraction and MEPs occur (Klomjai, Katz and Lackmy-Vallée, 2015). However it should be noted that following the MEP there is a pause in EMG activity which is known as the cortical silent-period (Ahonen *et al.*, 1998).

Therefore, whilst MEPs are based on excitatory responses, the cortical silent-period is based on inhibitory responses facilitated by GABAb interneurons synapsing on pyramidal neurons (Inghilleri *et al.*, 1996; Ahonen *et al.*, 1998; Poston *et al.*, 2012). Studies have shown that TMS delivered during cue presentation in reaction time tasks can delay contralateral arm responses (Day *et al.*, 1989; Hannah *et al.*, 2018). Additionally, Ibanez *et al.*, (2018) reported that TMS delivered 30ms, 60ms and 200ms before a subject's average movement led to a delayed response, and researchers have attributed such delays to the silent-period phenomenon (Day *et al.*, 1989; Hannah *et al.*, 2018; Ibanez *et al.*, 2018). Based on such findings, it can be argued that TMS can cause a disruption to movement preparation in the studies conducted in this thesis by generating the cortical silent-period.

Whilst the physiological changes associated with upper-limb reaching have mainly focused on motor preparation responses, behavioural changes have focused on execution of motor responses involving reaction time, peak velocity and errors made during reaching (e.g. trajectory errors). Reaction time has been reported to be affected with TMS stimulation delivered during movement preparation. For example, Busan et al., (2012) found that TMS to the parietal lobe led to an increase in reaction time when TMS was delivered at the starting phase of reaching preparation, but a decrease in reaction time when TMS was delivered mid-way through the preparation phase of a reach. Similar findings have been noted when stimulating parietal regions near the intraparietal sulcus. For instance, Busan et al., (2009) found a decrease in reaction time when TMS was administered at approximately half of the mean reaction time among subjects. With regards to motor execution, Hunter, Sacco and Turner (2011) found that TMS to the M1 during reaching towards different targets (135° and 270° on a visually displayed "dart-board") did not lead to any significant changes in normal reaching trajectories. Studies that have targeted other regions with TMS have also demonstrated similar results. For example, Della-Maggiore et al., (2004) revealed no significant changes in reaching trajectories during normal reaching when TMS was delivered to both the posterior parietal cortex (PPC) and the occipital lobe following movement onset. However, it is not known if using TMS to target other regions of the wider motor network during reaching preparation and execution may lead to different behavioural findings.

1.4.2 Motor adaptation

TMS protocols have been used in both motor skill learning and motor adaptation studies. However, before exploring this it is important to consider the differences between 'motor skill learning' and 'motor adaptation' because the two often fall under the general term 'motor learning', and this project concerns motor adaptation only.

Motor adaptation is based on modifying a motor skill through an error driven process (Martin *et al.*, 1996; Malone, Vasudevan and Bastian, 2011) compared to motor skill learning which is based on learning new information to acquire a new motor skill (Luft and Buitrago, 2005). Motor adaptation is also studied with force-field paradigms where individuals gradually learn to adjust their motor behaviours towards forces administered by a robotic device (Stockinger, Focke and Stein, 2014). On the other hand, motor skill learning has been investigated with motor sequence paradigms, which are tasks associated with learning motor information in a particular arrangement (Morin *et al.*, 2008; Weiermann, Cock and Meier, 2010). During motor adaptation, behaviour can be unlearned (i.e. disappears when the perturbation disappears) and therefore has short-term effects (Reisman, Block and Bastian, 2005; Krakauer and Mazzoni, 2011) whereas motor skill learning has long term-effects (Luft and Buitrago, 2005).

1.4.2.1 TMS during motor adaptation

Corticospinal excitability has been investigated with TMS protocols (SP and PP) in motor adaptation paradigms. Studies have mainly explored differences in MEP amplitude in force-field reaching compared to normal reaching. For example, the BB muscle has been found to exhibit larger MEP responses during motor adaptation compared to normal reaching (Hunter, Sacco and Turner, 2011). Similarly, Orban de Xivry *et al.*, (2013) revealed significant increases in peak-to-peak MEP amplitudes during force-field reaching in the BB compared to the TB.

The direction of a reach in force-field paradigms has also been shown to have an impact on physiological responses in different muscles. For example, the results from Orban de Xivry *et al.*, (2013) concluded that reaching towards a south-east direction led to increased MEPs in BB, whereas reaching towards a north-west direction (135° target) increased activity in the TB and deltoid muscles. Therefore, there is direction-specific tuning of the corticospinal excitability (Orban de Xivry *et al.*, 2013). PP-TMS paradigms of motor adaptation have also illustrated changes in physiological mechanisms. For example, SICI has been found to be reduced in the BB and TB nearer to movement onset following motor adaptation (Hunter, Sacco and Turner, 2011). This finding suggests that SICI focuses the excitatory drive to specific muscles before movement in the force. SICI and ICF have also been found to be modified with regards to the factor of time. For instance, following motor adaptation in a north-west direction (135° target) a significant reduction in SICI was found at 190ms, whereas ICF was significantly increased at 160ms during reaching preparation (Hunter, Sacco and Turner, 2011).

Behavioural changes have also been noted during motor adaptation. For example, errors during reaching are increased during the initial stages of motor adaptation, however this gradually reduces trial by trial following additional blocks of force-field reaching (Hunter, Sacco and Turner, 2011; Pizzamiglio *et al.*, 2017a; Pizzamiglio *et al.*, 2017b). Errors made in motor adaptation when TMS has been delivered to regions of the motor network have also been studied. For example, Della-Maggiore *et al.*, (2004) found significant errors in reaching trajectories during motor adaptation when SP-TMS was delivered to the left posterior parietal cortex (PPC). They also reported no errors in reaching trajectories when an area of the visual cortex was stimulated during motor adaptation. This was important as it suggested that the PPC had a specific role in motor adaptation but not in normal un-perturbed reaching (see earlier). Other kinematic measures such as movement onset and offset of reaching, movement velocity and movement duration have not been studied extensively with TMS paradigms.

A key aim of this thesis is to not only to explore neural mechanisms of reaching during motor adaptation, but to also investigate how TMS delivered to different cortical regions affects the kinematic behavioural responses of simple reaching and motor adaptation. Several cortical regions play a role in both reaching and motor adaptation. These regions include areas surrounding the M1 (e.g. the PPC, the SMA and the PMC) (Halsband *et al.*, 1993; Krebs *et al.*, 1998; Baizer, Kralj-Hans and Glickstein, 1999; Landi, Baguear, and Della-Maggiore, 2011; Shum *et al.*, 2011; D'Angelo and Casali, 2012; Panouilleres *et al.*, 2015; Borich *et al.*, 2015; Chang *et al.*, 2015). Other nodes in the motor network include associative cortical and subcortical regions (e.g. prefrontal cortex, basal ganglia and the cerebellum) (Goldman-Rakic 1987; Shima *et al.*, 2007; Overduin, Richardson and Bizzi, 2009; D'Angelo and Casali, 2012). Before addressing the roles of specific cortical areas, it is important to outline the structure of the cortex (e.g. different cortical layers) and the ways in which the motor system governs motor
output. Section 1.5 therefore outlines the 1) human motor system, 2) other motor-related cortical regions that contribute to reaching behaviours, 3) the effects of TMS to these regions, and 4) the ways in which responses are altered when the motor network is impaired.

1.5 The human motor system

Numerous cortical regions make up the human motor system, including the posterior parietal cortices (PPC), the supplementary motor area (SMA), the pre-motor cortices (PMC), the cerebellum as well as the pre-frontal cortex, and regions of the basal ganglia (Dum, Levinthal and Strick, 2016). However, the key pivotal structure within the motor system is the primary motor cortex (M1) which is situated in the dorsal region of the frontal lobe (Miyachi *et al.*, 2005). To move voluntarily, the M1 sends complex descending signals via the CST to motor neurons that innervate skeletal muscles (Drew, Prentice and Schepens, 2004). The corticospinal pathway crosses at the brainstem level of the neuraxis as it descends, so that the right side of our brain is responsible for moving the left side of our body and vice versa (Sun and Walsh, 2006) (figure 1.3 - adapted from Welniarz *et al.*, 2017).

The human cortex contains different cortical layers, including layer I (the molecular layer), II (the external granular layer), III (external pyramidal layer), IV (internal granular layer), V (internal pyramidal layer) and VI (multiform later) (figure 1.4A figure adapted from Mitchell and Patterson 1954 cited in, Crossman and Neary, 2015). These layers contain different types of neurons that have different projections (e.g. afferent and efferent) and different functions (e.g. excitatory and inhibitory) (Briggs, 2010). Layer I is made up of axonal projections (from local regions), dendrites and fusiform cells. Layers II and III on the other hand contain pyramidal neurons (with the majority being found in layer III). Pyramidal neurons are glutamatergic, have an excitatory function, and communicate with other cortical structures via axonal projections (DeFelipe and Fariñas, 1992; Spruston, 2008; Tjia et al., 2017). Pyramidal neurons in layer II and III enable both local and distant communication between different cortical regions via long axonal projections (Fame, MacDonald and Macklis, 2010; Tjia et al., 2017). For example, it has been found that pyramidal neurons in layer III contain axons that project to regions such as the parietal cortices and the temporal gyrus (Goldman-Rakic, 1987; Pierri et al., 2001). Axonal projections in layer III end at the thalamic nucleus. It has been suggested that the thalamic nucleus facilitates the

excitatory functions of pyramidal neurons in layer III (Giguere and Goldman-Rakic, 1998; Pierri *et al.*, 2001). Axons from regions such as the somatosensory and premotor areas also contribute to CST functions, more specifically their projections facilitate different phases of motor behaviours (Ueno *et al.*, 2018). For example, Ueno *et al.*, (2018) reported that both sensory and motor responses are combined to enable grasping and retrieving during reaching, as outlined in section 1.4.1 and figure 1.4B (adapted from Briggs *et al.*, 2018).

Layer IV contains pyramidal and stellate cells (stellate cells are also excitatory) which are similarly distributed within the layer (Meyer et al., 1992; Schubert et al., 2003). The two cells do however have different properties with regards to their connecting pathways (Alonso and Klink, 1993; Alexander and Hasselmo, 2018). For example, stellate cells have been reported to have direct connections with various subhippocampal regions, however the same has not been noted regarding pyramidal cells (Alexander and Hasselmo, 2018). Studies have also revealed that while the excitatory functions of stellate neurons rely on monosynaptic excitatory inputs within the same layer, pyramidal neurons receive further excitatory inputs from other layers of the cortex (Schubert et al., 2003). Layer V contains large pyramidal neurons and the most extending axonal projections to different cortical regions, including the pons (Harris and Shepherd, 2015; Tjia et al., 2017). Studies have revealed that pyramidal neuronal dendrites in layer V extend to the pia matter, however this is not the case for pyramidal neurons in other layers, such as II and III (Spruston, 2008; Tjia et al., 2017). The pyramidal neurons in layer V also have higher action potential firing thresholds compared to other layers of the cortex and obtain strong monosynaptic inputs from stellate neurons in layer IV (Feldmeyer, Roth and Sakmann, 2005; Petersen and Crochet, 2013; Tjia et al., 2017).

Layer VI contains pyramidal neurons and non-pyramidal neurons, including stellate cells and fusiform neurons (Briggs, 2010). These neurons play different roles in receiving and transmitting signals, for instance pyramidal neurons project to other layers of the cortex (e.g. layer IV and V), whereas non-pyramidal neurons such as stellate and fusiform cells mainly have local projections that are confined to layer IV (Briggs, 2010). Inhibitory cells are present in layer VI, and these are also found in layer IV. The neurons found in layer VI have distinct functions, all of which play an

important role in regulating and enabling cortical processing between different structures (Briggs, 2010).

In the M1, cortical layers differ in terms of 1) cell distribution, and 2) axonal projections (see section 1.5.1 below). As this thesis is concerned with the motor network, it is important to highlight the ways in which the different cortical layers of the M1 enable motor output (specifically reaching) to take place.

1.5.1 Cortical layers of the M1

Excitatory responses in the M1 are a result of pyramidal neurons that project onto various cortical structures which enable the processing of 1) sensori-motor information (via cortico-cortical projections), 2) cerebellar information (via thalamo-cortical projections) and 3) modulatory information (via neuro-modulatory projections) (Donoghue and Wise, 1982; Weiler et al., 2008). Cortical layers in the M1 have varied cell distribution levels (Weiler et al., 2008; Castro-Alamancos, 2013; Harris and Shepherd, 2015; Tjia et al., 2017). Layer I of the M1 does not contain many pyramidal neurons, however it does have a horizontal axonal system which has resulted in the formation of dense synapses within its layer (Douglas and Martin, 2004; Harms et al., 2008; Weiler et al., 2008). The connecting inputs from other regions into layer I facilitates neural communication between different cortical regions, and also aids the learning process (Cauller, 1995; Sanes and Donoghue, 2000; Harms et al., 2008). For example, Harms et al., (2008) found that 7 days of repetitive motor learning in rats resulted in significantly greater synaptic strength in layer I of the forelimb motor cortex representation, as measured by slice recordings (Harms et al., 2008). Synaptic connections in the M1 rely on descending pathways in layers II and III, which converge onto both corticospinal neurons and pyramidal neurons in layer V (Kaneko et al., 2000; Weiler et al., 2008).

Additionally, layers II and III in the M1 have both efferent and afferent projections which facilitate neural communication in the motor circuitry (Kim *et al.*, 2016a). The existence of layer IV in the M1 has been questioned because staining methods have only illustrated small sets of neurons in its layer (García-Cabezas and Barbas, 2014). However, it has been suggested that densely packed neurons in layers III and V has resulted in the hidden visibility of layer IV (García-Cabezas and Barbas, 2014). Labelling neuronal tracers (via markers) in layers III and V in primates has provided

evidence for the existence of M1 layer IV (García-Cabezas and Barbas, 2014). Moreover, studies have shown that the connecting neural pathways from the thalamus (which facilitates thalamic motor output) extend to layer IV, and this has provided further evidence for its presence (McFarland and Haber, 2002; García-Cabezas and Barbas, 2014).

Furthermore, layer V in the M1 has been argued to contain normal pyramidal neurons, and Betz cells (large pyramidal neurons) that extend into layer I (Betz, 1874; Castro-Alamancos, 2013). Electrical stimulation studies have shown that M1 current flow oscillations are facilitated by cells in layer V, with the strongest oscillation propagations found in its cellular dendrites (Castro-Alamancos and Rigas, 2002; Castro-Alamancos 2013). Layer VI in the M1 contains efferent neurons (Swadlow, 1994; Beloozerova, Sirota and Swadlow, 2003) that project into the sub-regions of the thalamus and this aids different types of motor activity including locomotion (Beloozerova, Sirota and Swadlow, 2003; Marlinski *et al.*, 2012). Overall, the neuronal differences between these layers have led to specialised M1 functions which facilitate motor output (Castro-Alamancos, 2013).



Figure 1.3: The corticospinal tract

- (A) A schematic presentation demonstrating the crossing of the corticospinal tract (as labelled in blue) in healthy controls which enables movement execution (i.e. the right-cortex controls left-sided motor execution, and the left-cortex controls right-sided motor execution) (figure adapted from Welniarz *et al.*, 2017).
- (B) Diffusion MRI tractography illustrating the fibres of the corticospinal tract (figure adapted from Dalamagka *et al.*, 2019).



Β.



Figure 1.4: Layers of the cortex and motor network projections:

- (A) The ways in which cells are presented within and across cortical layers (I, III, IV, V and VI) are shown (figure adapted from Mitchell and Patterson 1954 cited in, Crossman and Neary, 2014)
- (B) Projections in the motor network are illustrated with diffusion tensor images various interconnections can be seen between regions such as the M1, SMA, PMC, the insula, the anterior cingulate cortex as well as the middle frontal gyri (I [sagittal plane] and II [coronal plane]). Additional projections of sub-regions in the SMA (III) (e.g. the supplementary cingulate eyefield [SCEF], 6MA and 6MP) and ventral PMC (IV) (e.g. 3a, 3b, 4 and 6v) are also shown (figure adapted from Briggs *et al.*, 2018).

1.5.2 The primary motor cortex (M1)

Cortical and electrical stimulation studies led to the discovery of the homunculus which is a map of the body's muscle representations that is contained in the M1 (Catani, 2017; Penfield and Rasmussen, 1950) (see figure 1.5B – Murphy et al., 2008). Penfield's homunculus has however been questioned because although studies have shown that muscles can be stimulated individually, they have also shown that muscle responses are not only restricted to the M1 (Purves et al., 2001; Catani, 2017). For example, responses have also been evoked in the postcentral and precentral gyrus (Catani, 2017). Stimulation studies have also found that muscle representations can overlap (see figure 1.5C - Catani, 2017; Penfield and Boldrey, 1937). Despite the controversies surrounding the homunculus, its discovery played a key role in the field of cortical stimulation and has aided surgical mapping procedures (Catani, 2017). TMS studies use specific coil positionings in order to target muscle(s) of interest and record their activity with surface electromyography (EMG) (figure 1.7B and 1.7C - Davidson, Bolic and Tremblay, 2016 and Abdalla, 2011). Recordings can therefore be visualised and quantified in order to accurately identify the muscle(s) (region of the homunculus) that has been stimulated.

As well as containing muscle representations, the main role of the M1 is to regulate the control and output of movements (Sanes and Donoghue 2000; Chang *et al.*, 2015). It does this by directing movement signals (related to movement timing) towards the spinal cord which leads to an interaction with neuronal motor circuits, resulting in motor output (Alexander and Crutcher, 1990; Desmurget and Grafton, 2000; Scott, 2004; Teka *et al.*, 2017). This is important in motor reaching paradigms that have time requirements to reach a specific target (Hunter, Sacco and Turner, 2011; Pizzamiglio *et al.*, 2017a; Pizzamiglio, *et al.*, 2017b) Therefore, simple movements already require a complex interaction between different systems including the nervous and musculoskeletal systems (Teka *et al.*, 2017). Furthermore, the role of the M1 in motor execution has been attributed to its connections to various motor related regions, including the SMA, PPC, PMC and the cerebellum (Tanji, 1994; Lotze *et al.*, 1999; Whitlock, 2017). Different neural responses can be facilitated with M1 stimulation, and this is outlined below.

1.5.2.1 TMS delivered to the human M1

When TMS is delivered to the M1 of one hemisphere, a resulting corticospinal neural volley can be recorded as an MEP (Hallet, 2007) in muscles in the contralateral limb. The human motor cortex contains some of the largest pyramidal neurons which have numerous inputs and multiple functions (Kaas, 2000; Young, Collins and Kaas, 2013). Pyramidal neurons have excitatory functions in the cerebral cortex and are known as projection neurons (Bekkers, 2011). In layer V of the M1, these neurons transmit axons along the spinal cord for muscle function.

When TMS is delivered to the M1 (i.e. to a specific muscle representation in Penfield's Homunculus) it causes neuronal activity in a plane that is parallel to the coil and the surface area of the brain (Klomjai, Katz and Lackmy-Vallée, 2015). M1 TMS results in the transynaptic activation of pyramidal neurons, leading to the production of descending volleys (from pyramidal axons) which project onto the CST (Klomjai, Katz and Lackmy-Vallée, 2015). This causes an activation of motor neurons, leading to a contraction in the muscle of interest - i.e. an MEP is produced (Klomjai, Katz and Lackmy-Vallée, 2015). The MEP can be recorded using surface electromyographic (EMG) electrodes and corticospinal excitability can be quantified by measuring the peak-to-peak MEP amplitude (Summers, Chen, Kimberley, 2017). This process is graphically illustrated in figure 1.6 (Klomjai, Katz and Lackmy-Vallée, 2015). The degree of corticospinal excitability that one exhibits depends on a range of physiological factors. For example, glutamate transmittors have been argued to have an excitatory influence on connections between cortico-cortical axons and corticospinal neurons, and have been reported to play an important role in enabling rapid cortical synaptic transmission (Douglas and Martin 1998 cited in, Shepherd, 2004; Klomjai, Katz and Lackmy-Vallée, 2015). On the other hand, transmittors ranging from GABA and serotonin have been associated with inhibitory functions and have been found to reduce peak-to-peak MEP responses (Klomjai, Katz and Lackmy-Vallée, 2015).

The physiological activity evoked with TMS differs from TES activity (Klomjai, Katz and Lackmy-Vallée, 2015). TMS causes an indirect activation of corticospinal neurons via synaptic inputs. This has been illustrated by the different types of waves that are induced when a corticospinal volley is evoked with TMS, by which indirect waves (I-waves) as opposed to direct waves (D-waves) are first observed with stimulation (Di Lazzaro *et al.*, 1998; Hallett, 2000). D-waves are based on direct responses from

stimulated axons, whereas I-waves occur due to stimulated corticospinal neurons (Rossini et al., 2015) and are more indicative of TMS evoked activity (Terao and Ugawa, 2002). Contrastingly with TES, the corticospinal volleys observed include Dwaves, followed by late I-waves, and early I-waves. Based on this observation it has been argued that TES activates neurons in a plane vertical to the brains surface area, which differs from TMS which activates neurons in a plane that is parallel to the surface area (Klomjai, Katz and Lackmy-Vallée, 2015). Different factors can effect the types of waves that are recruited, for instance in some participants TMS at a high intensity can result in D-waves being observed prior to the I-waves (Klomjai, Katz and Lackmy-Vallée, 2015). However, in most participants, the I1 wave is recruited first, followed by the I2 and I3 wave (Klomjai, Katz and Lackmy-Vallée, 2015). Furthermore, there are various types of TMS coil orientations that can induce different electrical currents in the brain and can lead to different physiological mechanisms being observed (see figure 1.7C – Abdalla, 2011). For example, a lateral-medial (LM) induced current delivered to M1 results in late I-waves, whereas a posterior-anterior (PA) induced current first elicits an I1 wave, followed by late I-waves when stimulation intensity is increased (Di Lazzaro and Ziemann, 2013). On the other hand, an anterior-posterior (AP) induced current results in smaller and postponed I-waves (see figure 1.7A - Di Lazzaro and Ziemann, 2013). Figure 1.7A illustrates extrapyramidal recordings, although not usually elicited during surface EMG recording of MEPs, they have helped demonstrate the extent of indirect presynaptic activity and direct cortical axonal activity (i.e. I-waves and D-waves) (Di Lazzaro and Ziemann, 2013).

Differences in physiological responses such as the peak-to-peak MEP amplitude have also been observed with varying coil directionalities (Mills, Boniface and Schubert, 1992; Abdalla, 2011; Hallett, 2007). For example, Mills, Boniface and Schubert (1992) and Abdalla (2011) found that the largest MEP amplitude from the M1 (for the FDI muscle – Mills *et al.*, 1992, and for the BB muscle - Abdalla, 2001) was evoked with PA direction and the coil oriented 45° away from the midline (figure 1.7B – Figure adapted from Abdalla, 2011). Other cortical regions targeted with TMS do not have the same experimental output as the M1 (i.e. an MEP; Janssen, Oostendorp and Stegeman, 2015). Sections 4.1, 5.1 and 8.1 describe the neural mechanisms and behavioural responses that have been observed when TMS is delivered to the M1 (left and right) during both normal (experiment 1) and novel (experiment 2) reaching.



Figure 1.5: The primary motor cortex

- (A) The first image in the panel is a canonical figure of the M1 and its location in the rostral segment of the frontal lobe (green) is demonstrated (figures adapted from the Florida Institute for Neurologic Rehabilitation, Inc: Atlas of Brain Injury and Anatomy: <u>http://www.finr.net/</u>).
- (B) Penfield's homunculus is shown and the arrangement of the different muscles within the motor cortex can be seen ranging from the leg to the hip and trunk muscles (from Murphy *et al.*, 2018).
- (C) The motor responses elicited in different areas of the M1 are shown, demonstrating that muscle representations are not confined to one particular segment of the M1 (i.e. there are overlapping muscle representation responses during stimulation (data based on Penfield and Boldrey, 1937, figure adapted from Catani, 2017).



Figure 1.6: TMS to the human motor cortex

The mechanisms involved in producing MEPs with TMS to the human motor cortex is shown - when TMS is applied to the M1, pyramidal neurons are activated and this produces descending volleys that project onto the corticospinal tract. This activates motor neurons and leads to an MEP which can be quantified by measuring its peak-to-peak amplitude (figure adapted from Klomjai, Katz and Lackmy-Vallée, 2015).



Figure 1.7: The effect of coil directionality on responses

- (A) As previously mentioned in section 1.5.2.1, these extrapyramidal recordings demonstrate the different I-wave responses with varying coil orientations (figure adapted from Di Lazzaro and Ziemann, 2013). An LM induced current leads to late Iwaves, compared to a PA current which can result in later I-waves when stimulation intensity is increased. The AP induced current on the other hand leads to both smaller and delayed I-waves (Di Lazzaro and Ziemann, 2013).
- (B) Different types of coil orientations are demonstrated (figure adapted from Davidson, Bolic and Tremblay, 2016) (Canonical brain figures adapted from an MRIcron template – Rorden and Brett, 2000).
- (C) Changes in the BB muscle MEP responses with different coil orientations are illustrated - the largest MEP from the BB was obtained with PA inducted current compared to an LM induced current, with the coil positioned 45° away from the midline (figure adapted from Abdalla, 2011).

1.5.2.2 Factors effecting MEP responses

There are different factors that can influence the output of MEP responses which are important to take into account. These include brain injury (either traumatic or caused by neurological impairments), age and gender (Chistyakov *et al.*, 2001; McAllister, 2011; Prins *et al.*, 2013).

1.5.2.2.1 Traumatic brain injury (TBI)

Brain injury can either be caused by external factors or by damage that has occurred due to neurological disease (McAllister, 2011; Prins *et al.*, 2013). An example of brain injury caused by external factors includes traumatic brain injury (TBI) (McAllister, 2011). In cases of TBI, contact occurs between cortices and the bony ridges of the skull (McAllister, 2011). In severe cases, TBI can lead to permanent irreversible damage and its neurological impairments include white matter atrophy, demyelination and apoptosis (Newcombe *et al.*, 2011; Stocchetti and Zanier, 2016).

Abnormal MEP responses have been reported among TBI patients. For example, researchers have demonstrated that TBI causes less excitatory MEP responses, particularly among patients with extreme brain trauma compared to minor head trauma (Chistyakov et al., 2001; Bernabeu et al., 2009). The reduced MEP amplitude has been suggested to be a result of neuronal and axonal damage which effects both motor excitability and motor conduction (Chistyakov et al., 2001). Changes in corticospinal excitability as a result of brain injury depend on the extent of the injury (Bernabeu et al., 2009). Although synaptic re-organisation can occur following TBI, this may not be the case for patients with extreme TBI because cortical damage can lead to impaired surrounding neural networks which can halt the recovery process (Blumbergs, Jones and North, 1989; Povlishock and Katz, 2005; Bernabeu et al., 2009; Castellanos et al., 2011). Figure 1.8A (Bernabeu et al., 2009) illustrates varying MEP amplitudes among TBI patients with different levels of diffused axonal injury (DAI). Other types of brain injury that can also impair physiological functioning and effect inhibitory and excitatory mechanisms, include those that arise from neurological conditions, such as multiple sclerosis (MS), and stroke.

1.5.2.2.2 Multiple sclerosis (MS)

MS is an inflammatory condition which effects the central nervous system (CNS) and results in de-myelinated plaques (Calabresi, 2004; Goldenberg, 2012). These plaques have been found in both white and grey matter and have been characterised by impaired

myelin sheaths and degenerated axons (Brück *et al.*, 1996). The demyelination that occurs is caused by impaired immunological cellular responses, as well as glial cell death (glial cells are the primary cells responsible for myelin formation) (Brück *et al.*, 1996; Ghasemi, Razavi and Nikzad, 2017). The condition results in slower impulse transmission between neural networks (Ghasemi, Razavi and Nikzad 2017). Researchers have demonstrated white matter structural impairments in MS patients, particularly in the corticospinal tract (Raz *et al.*, 2010; Han *et al.*, 2017).

MEP responses have been reported to be disrupted among MS patients. For example, Brum, Cabib and Valls-Solé (2016) noted that MS patients had a longer MEP duration at rest (FDI muscle) compared to healthy controls. Researchers have also reported a decrease in peak-to-peak MEP amplitude, whereas others have noted no MEP response during cortical stimulation in MS patients (see figure 1.8B - Fernandez *et al.*, 2013) (Fernandez et al., 2013; Kale *et al.*, 2014).

The different types of MEP responses that have been found in MS patients, have been suggested to correspond to the different types of disruption caused by MS. For instance decreased MEP amplitude has been associated with axonal damage, whereas increased MEP conduction time has been linked to de-myelination (Fernandez *et al.*, 2013). Overall, the damage caused to white matter structures in MS detrimentally affects the corticospinal tract and its projections which facilitate healthy MEP responses (Diehl *et al.*, 2004).

1.5.2.2.3 Stroke

Stroke patients demonstrate abnormal responses to cortical stimulation (Pennisi *et al.*, 1999; Chae *et al.*, 2002), however the degree of abnormality exhibited does depend on the nature of the lesion caused by the stroke. The main types of stroke include ischaemic stroke and haemorrhagic stroke (intracerebral or subarachnoid), with ischaemic stroke being the most common of the two (Musuka *et al.*, 2015; Hui, Taddi and Patti, 2019). In an ischaemic stroke, there is a sudden reduced blood flow in the brain which causes destruction to various neurons and glial cells, resulting in focal damage and cellular death (Xing *et al.*, 2012). The stages of a haemorrhagic stroke initially begin with clotted blood tissues, followed by oedema and cellular death (Kitago and Ratan, 2017). An intracerebral haemorrhagic stroke causes bleeding within the brain, whilst a subarachnoid haemorrhagic stroke causes bleeding between the pia mater and arachnoid space (Naidech, 2011). Physical impairments caused by a stroke include weakened

and/or loss of motor function (Tatemichi *et al.*, 1994; Raghavan, 2015). Whishaw, Alaverdashvili and Kolb (2008) reported that limb dysfunction caused by a stroke can result in either; a) learned non-use of the limb, b) learned bad-use of the limb, whereby participants learn to use the paralysed limb in a non-typical way, which can result in poor movement accuracy, and c) failure to remember motor behaviours.

The damage caused to the motor network due to a stroke is vital when exploring cortical stimulation responses among stroke patients (Darling, Pizzimenti and Morecraft, 2011). The effects of lesions within surrounding motor regions can result in an inhibition in signal transmission from the M1 to the spinal cord which is involved in motor output behaviour (Raghavan, 2015). The impaired communication between the M1 and spinal cord can lead to delayed responses, which has been shown experimentally with cortical stimulation to the M1 (Pennisi *et al.*, 1999; Chae *et al.*, 2002). For example, Pennisi *et al.*, (1999) investigated MEP responses in patients following an ischaemic stroke; 48 hours post-stroke, and 1 year post-stroke. Their findings revealed an absence in MEP responses in patients, although the responses were both delayed and small in amplitude.

MEPs have been used as a measure of functional recovery post-stroke (Pennisi *et al.*, 1999) (figure 1.8C – Stinear, 2017 based on data from Byblow *et al.*, 2015). For instance, larger MEP amplitudes have been illustrated among patients with improved recovery post-stroke (Escudero *et al.*, 1998; Kim *et al.*, 2015; Kim *et al.*, 2016b). Network re-organisation following a stroke can contribute to the recovery process, which can lead to improved MEP responses during cortical stimulation (Nudo and Friel, 1999; Byrnes *et al.*, 1999; Thickbroom, et al., 2002). More specifically, plasticity can occur in motor regions other than the motor cortex, ranging from the SMA to the PMC, which can facilitate corticospinal projection among patients and thus aid the rehabilitation process (Dum and Strick, 1991; Thickbroom *et al.*, 2002; Thickbroom *et al.*, 2004).

1.5.2.2.4 Age and gender

Demographic factors have an effect on physiological responses during stimulation. During healthy aging, atrophy occurs due to diminished synaptic development which leads to cell death (Scahill *et al.*, 2003; Peters, 2006; Pascual-Leone *et al.*, 2011). This can lead to a general decline in cognitive and physical functions, including poor motor skills (Buchman and Bennett, 2011). For example, in a go-task, Bedard *et al.*, (2002)

found slower reaction times in older adults compared to younger adults. Similar results were obtained in a PEG-board task (a paradigm designed to assess fine motor control), whereby older-aged participants performed more slowly than middle-aged participants (Hamilton *et al.*, 2017).

Paired pulse TMS protocols have been used to explore the impact of age on corticospinal excitability (McGinley *et al.*, 2010; Opie and Semmler, 2014) For example, Opie and Semmler (2014) found that during muscle activation, short-interval intracortical inhibition (SICI) was reduced in older-aged subjects compared to younger subjects. They also found a decrease in long-interval intracortical inhibition (LICI) responses during rest among the older subjects. The decrease in LICI has been linked to age-related changes that occur in neurotransmitters (namely GABA_b) which can influence inhibition processes (Werhahn *et al.*, 1999). On the other hand, McGinley *et al.*, (2010) noted an increase in LICI among older subjects compared to younger subjects. Therefore mixed findings have been noted, but this can be attributed to different factors, including the intervals used in the paired pulse TMS protocols and the muscles explored (Opie and Semmler, 2014). Individual subject differences could also be a reason for the inconsistency in findings (Opie and Semmler, 2014).

TMS studies exploring gender differences in MEP responses have also reported mixed findings between males and females. For example, Livingston, Goodkin and Ingersoll (2010) found that gender did not have an influence on three key MEP features; motor threshold, peak-to-peak amplitude and motor conduction time. On the other hand, Tobimatsu *et al.*, (1998) reported a difference between males and females with regards to MEP latency and motor conduction time, but only in specific muscles (leg as opposed to hand). Surgical experiments which require motor cortex mapping have similarly demonstrated this difference - with females exhibiting shorter latencies than males (Picht, *et al.*, 2012). These findings have been argued to be a result of physical differences between males and females, such as height (Säisänen *et al.*, 2008; Sollmann *et al.*, 2017). For example, Sollmann *et al.*, (2017) found that matching physical features such as, height and upper limb length, led to no significant differences in MEP responses between males and females.



Figure 1.8: Patient MEP responses

- (A) MEP responses from TBI patients with varying levels of DAI: 1) healthy controls, 2) patients with mild DAI, without motor function impairment, 3) patients with severe DAI without motor function impairment, and 4) patients with severe DAI with motor function impairment. The MEP responses are less excitatory in severe TBI cases (4), compared to mild and healthy cases (figure adapted from Bernabeu *et al.*, 2009).
- (B) Peak to peak MEPs responses from the lower limb in MS patients are shown (outlined in red). A smaller MEP amplitude was found in the right lower limb compared to the left lower limb whereby no MEP was elicited during stimulation (figure adapted from Fernandez, *et al.*, 2013).
- (C) The association between upper-limb recovery and MEP responses in patients 6 months following a stroke is shown. Recovery was quantified as 'change in upper-extremity Fugl-Meyer score'. A relationship was found between available improvement and actual improvement in MEP-positive patients, whereas this was not the case for MEP-negative patients. MEP responses therefore provide an important indication of the recovery process in stroke patients (figure from Stinear, 2017, based on data from Byblow *et al.*, 2015).

1.6 The posterior parietal cortex (PPC)

The PPC resides posterior to the somatosensory cortex (figure 1.9A - figure adapted from the Florida Institute for Neurologic Rehabilitation, Inc: Atlas of Brain Injury and Anatomy: <u>http://www.finr.net/</u>), and throughout history it has evolved in terms of its structure and function. For example, studies in early placental mammals revealed that the PPC did not contain many multisensory segments, whereas studies in developed mammals have revealed that various PPC multi-sensory segments exist, which can be probed with TMS to elicit different behavioural responses, ranging from reaching to grasping (Kaas and Stepniewska, 2016). In human subjects compared to non-human primates (such as the macaque) the PPC differs in its morphology and functionality (Hill *et al.*, 2010; Kaas and Stepniewska, 2016). For example, in humans the region is extended (i.e. morphology) and it has also been suggested to have additional functional areas related to gesturing behaviours (Frey 2008; Konen et al., 2013; Kaas and Stepniewska, 2016).

The various motor related functions of the PPC include novel reaching, movement planning, motor intent, navigation and spatial awareness (Batista *et al.*, 1999; Connolly, Andersen and Goodale, 2003; Della-Maggiore *et al.*, 2004; Kaas and Stepniewska, 2016; Whitlock, 2017). The PPC has a vast range of connections with other structures (such as the SMA, M1 and the pre-frontal cortex) and this has resulted in its multiple motor and cognitive functions (Kass and Stepniewska 2016; Whitlock, 2017). Kaas and Stepniewska (2016) suggested that the neural communication between the PPC and other motor related regions enables successful motor responses. This has been further supported in clinical studies among patients with PPC impairments. For example, patients with optic apraxia (caused by PPC lesions) have disrupted movement control during reaching and grasping (Andersen *et al.*, 2014). This has been argued to be due to the disruption of communication between the PPC and other motor related domains associated with movement control, such as the M1, PMC and the frontal cortex (Andersen *et al.*, 2014). Figure 1.9B (Kaas and Stepniewska 2016) illustrates the neural communication between the PPC and other motor output.

Further evidence for the role of the PPC in motor behaviour comes from neuroimaging studies. For example, fMRI research has revealed significantly increased neural activity in the PPC during different types of motor tasks, including motor execution (typing

task) (Gordon et al., 1998), motor planning (virtual driving task among taxi drivers) (Maguire, Woollett and Spiers, 2006) and spatial navigation and memory (navigation memory task in patients with PPC lesions) (Ciaramelli *et al.*, 2010; Whitlock, 2017). Studies have also revealed the role of the PPC in motor reaching, for example evidence has illustrated that a posterior parietal reach (PPR) region exists in humans which is activated during pointing tasks (Batista *et al.*, 1999; Connolly, Andersen and Goodale, 2003). Researchers have also found that the neurons located in the PPR region encode hand movement goals during a task, thus providing evidence for the functional role of the PPC with regards to motor intent (Batista *et al.*, 1999; Whitlock, 2017).

Although studies have demonstrated that the left PPC is involved in motor adaptation there have been some controversies regarding its role. For example, researchers have argued that it is unclear as to whether it is the left or the right PPC that plays a more important role in learning (Bedard and Sanes, 2014). Bedard and Sanes (2014) explored bilateral activation patterns of parietal structures when participants were performing reaching outward movements using an MRI compatible joystick. Their findings concluded that during visuo-motor adaptation learning, both the right inferior parietal lobe and the left superior parietal lobe had similar levels of activation, therefore making it hard to decifer specific roles for different parietal structures. Perfetti et al., (2011) on the other hand concluded that learning during a visuo-motor task was accompanied with greater changes in the right compared to the left PPC, as quantified with EEG methods. However, Della-Maggiore et al., (2004) found greater changes in the left PPC during TMS whilst participants were performing a motor adaptation task. Based on these mixed findings it can be argued that the functional differences between the right and left PPC remain unclear. In addition to this, previous findings have not fully investigated the role of a number of regions during the different phases of motor adaptation (Bedard and Sanes, 2014). In this thesis, both the left and right PPC will be targeted and the findings will aim to elucidate the functional differences between the two hemispheres during preparation for motor reaching.

1.6.1 TMS to the PPC

Targeting other cortical regions, such as the PPC with TMS, does not have the same experimental output as the M1 (i.e. an elicited MEP; Janssen, Oostendorp and Stegeman 2015). However, comparable columnar arrangements exist across the cerebral cortex (Von Bonin and Mehler, 1971; Janssen, Oostendorp and Stegeman, 2015) and similar

coil principles and models (to the M1) have been developed to determine the best orientation for stimulating other regions. Several imaging studies (e.g. PET) have been conducted to create new TMS protocols for stimulating areas such as the occipital lobe, pre-motor cortex, supplementary motor area, parietal cortices and inferior frontal gyrus (Janssen, Oostendorp and Stegeman, 2015; Cona, Marino and Semenza, 2017).

TMS protocols targeting the left and right PPC include single pulse, paired pulse and repetitive pulse paradigms (Della-Maggiore et al., 2004; Busan et al., 2009). Studies have demonstrated that the left PPC plays an important role in motor adaptation. For example, Della-Maggiore et al., (2004) TMS delivered to the left PPC following movement onset resulted in disrupted reaching trajectories, however this was not the case when TMS was delivered to a control region such as the occipital lobe. Repetitive TMS protocols targeting the left PPC was found to have an impact on motor synchronisation patterns in the right hand and also delayed responses during a finger tapping task, however this was not the case with right PPC rTMS (Krause et al., 2012). Based on these studies it can be argued the left PPC has important networking pathways with the M1, and TMS to the PPC could disrupt neural connections, resulting in delayed motor activity (Krause et al., 2012). Similarly Vesia et al., (2007) explored the role of the left and right PPC, however they employed a reaching and pointing task. They found that left PPC TMS disrupted the end-point of reaching whereas right PPC TMS caused a shift in movement trajectories and affected subject's gaze during the task. The left and right PPC therefore play separate roles in motor activity.

Paired pulse TMS studies have shown that TMS to the right PPC had impacted reaching only when targets were presented on the left-hand side rather than the right-hand side (Koch *et al.*, 2008b). Most studies, have explored the effect of TMS on the PPC either 1) following the presentation of an auditory cue (Koch *et al.*, 2008b) or 2) when movement onset had occurred (Della-Maggiore *et al.*, 2004). Therefore, it is unknown whether TMS delivered a range of different time points during the preparation phase of motor adaptation results in the same findings. This will be addressed in this thesis when targeting regions such as the PPC. This paragraph aimed to briefly outline details regarding the action of TMS on the PPC during motor reaching. Sections 6.1 and 10.1 further describe the behavioural responses that have been observed when the PPC (left

and right) has been targeted with TMS protocols during both normal and novel reaching.



В.



Figure 1.9: PPC location and a schematic presentation of neural pathways between motor regions and their contribution to motor behaviours

- (A) The parietal lobe is shown in green, as is the PPC which is presented by the red dotted line (figure adapted from the Florida Institute for Neurologic Rehabilitation, Inc: Atlas of Brain Injury and Anatomy: <u>http://www.finr.net/</u>)
- (B) The interaction between the M1, PMC and rostral region of the PPC (rPPC) is shown with regards to three types of domains: reaching, grasping and defence. The different domains of the rPPC become activated via commands that are received from different regions. For example, the reaching domain is activated due to 1) cerebellar and thalamus excitatory inputs (+) to the M1 and 2) basal ganglia excitatory inputs to the PMC. Overall, these connections facilitate specific rPPC domain activation (figure from Kaas and Stepniewska, 2016).

40

1.7 The supplementary motor area (SMA)

The SMA is located in the frontal cortex and lies between the prefrontal cortex and M1 cortices (Nachev et al., 2007) (figure 1.10A - figure adapted from the Florida Institute for Neurologic Rehabilitation, Inc: Atlas of Brain Injury and Anatomy: <u>http://www.finr.net/</u>). The functions of the SMA include; regulating posture, facilitating complex/demanding motor tasks, and also planning and executing movements (Goldberg, 1985; Tanji, 1994; Akkal, Dum and Strick, 2007; Hiroshima *et al.*, 2014). Although previously thought of as one cortical structure, studies have shown that the SMA in fact consists of a different sub-regions (Akkal, Dum and Strick, 2007; Tanji, 1994). For example, the rostral part of the SMA is known as the pre-SMA, and the caudal part is known as the SMA proper, an additional part of the SMA forms what is known as the supplementary eye-field (SEF) (Akkal, Dum and Strick, 2007; Tanji, 1994).

Evidence for the SMA sub-regions has been illustrated in neuroimaging studies which have confirmed distinct functions of the pre-SMA and SMA-proper (Tanji, 1994; Akkal, Dum and Strick, 2007). For instance, in a motor sequence learning task (button pressing) using fMRI methods, Hikosaka et al., (1996) concluded that the pre-SMA had significantly greater neural activity only during the learning procedure but not when movements were performed. However, the SMA-proper was significantly active during movement performance but not during learning. Similarly, PET scanning techniques have revealed less activity in the pre-SMA compared to the SMA proper in tasks that involved movement control (Picard and Strick, 1996; Akkal, Dum and Strick, 2007). The pre-SMA has been associated with motor tasks such as, motor sequencing and retrieving motor memories, whereas the SMA-proper has been associated with movement execution (Halsband et al., 1993; He, Dum and Strick, 1995; Hikosaka et al., 1996; Alario et al., 2006). The SEF SMA sub-region plays a role in producing eye movements (Tehovnik et al., 2000; Fujii, Mushiake and Tanji, 2002; Nachev et al., 2007) and evidence for both its existence and function comes from patient models (Nachev et al., 2007; Husain et al., 2003). For example, Husain et al., (2003) reported a case study of a patient with a lesion to the SEF, who had impaired eye control movements.

The sub-regions communicate with each other and other areas of the motor circuit to facilitate motor output (Tanji, 1994). For example, the SMA-proper has direct

projections with the spinal cord (Tanji, 1994; Wang et al., 2001; Akkal, Dum and Strick, 2007). Furthermore, the pre-SMA is not densely connected with the M1, but rather with the prefrontal regions and the cerebellum which it receives input from (Wiesendanger and Wiesendanger, 1985; Tanji, 1994; Wang et al., 2001; Akkal, Dum and Strick, 2007). Furthermore, Akkal, Dum and Strick, (2007) revealed that the SMAproper and pre-SMA inputs arise from separate regions in the globus pallidus which has resulted in the distinct motor functions of these two sub-regions. Despite studies demonstrating differences between SMA sub-regions during motor tasks (Picard and Strick, 1996; Akkal, Dum and Strick, 2007) there have been controversies surrounding pre-SMA and SMA proper functioning. For instance, researchers have found that the pre-SMA and SMA-proper are similarly activated in different types of motor states (Wang et al., 2010). With regards to movement control, Chen, Scangos and Stuphorn (2010) reported that the signals originating from the pre-SMA could have an impact on SMA proper activity (Wardak, 2011). Similarly, Vergani (2015) reported that SMA motor activity is influenced by the anatomical connectivity between the pre-SMA and SMA proper (Lima, Krishnan and Scott, 2016). Based on such findings it could be argued that the connectivity between the pre-SMA and SMA proper has made it difficult to determine the precise functional role of SMA subregions (Wardak, 2011; Lima, Krishnan and Scott, 2016). This thesis will both address and consider these findings by using specific coil orientations to target specific sub-regions of the SMA, and this is further outlined below in section 1.7.1.

1.7.1 TMS to the SMA

Different coil positions have been used to target different regions of the SMA. For example, a coil handle positioned to the right has previously been used to target the pre-SMA (Cona, Marino and Semenza, 2017), whereas a coil position which induces a posterior-anterior current has been used to target the SMA-proper (Terao *et al.*, 2001). Repetitive TMS paradigms have mainly been implemented to explore SMA activity during motor related tasks. For instance, Kim *et al.*, (2014) applied rTMS to the M1 and the SMA during a motor sequence task and their results showed that rTMS to the SMA compared to the M1 caused a significant increase in movement time. The impact of SMA function on other regions has also been demonstrated. For example, Oliveri *et al.*, (2003) found an increase in M1 excitability (quantified by MEPs) following pre-SMA stimulation (Chouinard and Paus, 2010). Matsunaga *et al.*, (2005) similarly reported

heightened MEP responses after the SMA had been stimulated in an rTMS paradigm (Makoshi, Kroliczak and Van Donkelaar, 2011).

Differences have been found when comparing the results of pre-SMA stimulation to SMA-proper stimulation (Chouinard and Paus, 2010). While pre-SMA stimulation has been found to cause a greater disruption to tasks that have cognitive requirements ranging from switching to inhibition (Rushworth et al., 2002), SMA-proper stimulation has resulted in a disruption to 1) motor sequencing, 2) complex motor tasks and 3) bimanual tasks (Gerloff et al., 1997; Obhi et al., 2002; Steyvers et al., 2003) (Chouinard and Paus, 2010). Furthermore, Makoshi, Kroliczak and Van Donkelaar (2011) found that TMS to the SMA caused a disruption in movement onset responses in a loadholding task, whereas this was not the case when an area of the visual cortex was stimulated. This was also noted to be time-dependent (at approximately 400ms before motor activity). Makoshi, Kroliczak and Van Donkelaar (2011) therefore inferred that the SMA plays a key role in planning and predicting behaviours prior to the movement occurring. Studies exploring the role of the SMA during motor adaptation have primarily focused on non-human primate responses (Padoa-Schioppa, Li and Bizzi, 2004). In addition to this, SMA functioning has commonly been investigated using motor sequence paradigms compared to motor reaching paradigms (Gerloff et al., 1997; Kim et al., 2014). This thesis will address this by targeting the SMA using a novel reaching paradigm. Further details regarding the behavioural and neural responses that have been found with TMS delivered to the SMA during normal and novel reaching are described in section 7.1, as this section outlined experiment 4 whereby a virtual disruption approach was used to explore the role of the SMA in a motor control and motor reaching.



Figure 1.10: SMA location and coil position for targeting this region

A.

B.

- (A) The red dotted line represents the location of the SMA (figure adapted from the Florida Institute for Neurologic Rehabilitation, Inc: Atlas of Brain Injury and Anatomy: <u>http://www.finr.net/</u>).
- (B) Different coil positions have been used to target the sub-regions of the SMA. To target the pre-SMA, a coil position with the handle pointing to the right has been used (first image on panel B), however, to target the SMA-proper, a coil position inducing a posterior-anterior current has been used (second image on panel B) (positions based on Cona, Marino and Semenza, 2017; Terao *et al.*, 2001) (canonical brain figures adapted from an MRIcron template – Rorden and Brett, 2000).

1.8 The Pre-Motor cortex

The Pre-Motor cortex (PMC) is located between the dorsolateral prefrontal cortex and the M1 (Kantak *et al.*, 2012). The PMC has axonal signals that project to 1) the M1, 2) the corticobulbar pathway and 3) the corticospinal pathway, thus the region has a direct effect on motor-neurons in the brainstem and spinal cord (Purves *et al.*, 2001). The PMC is an important structure that forms part of the motor circuit and one of its key functions is to facilitate movement selection and preparation (Beck *et al.*, 2009; Chang *et al.*, 2010). Evidence for its role in movement selection comes from studies in nonhuman primates. For example, Kurata and Hoffman (1994) found that PMC lesions in monkeys resulted in significantly greater choice-selection errors during a reaction time task.

A key feature of the PMC is that it contains two structures, the dorsal PMC (dPMC) and the ventral PMC (vPMC) both of which have different functions. For example, the dPMC has been found to be involved in sequential motor learning, visually guided movements and also controlling kinematic responses (Johnson *et al.*, 1996; Davare *et al.*, 2015; Ohbayashi, Picard and Strick, 2016; Solopchuk, Alamia and Zénon, 2016). The functions of the ventral PMC range from speech control and production, to goaldirected behaviours (Binkofski and Buccino 2006; Meister *et al.*, 2007). For example, neuronal firing in the vPMC was found to be increased during goal-directed behaviours, including grasping and holding (Sakata *et al.*, 1995 Binkofski *et al.*, 2000; Binkofski and Buccino, 2006). Imaging studies have also revealed additional functions of the vPMC including its role in motor imagery and its importance in motor tasks with increased levels of difficulty. For instance, Winstein, Grafton and Pohl (1997) found that the vPMC forms part of an important cortical loop in the motor circuit. They reported increased regional cerebral blood flow (rCBF) in the vPMC which was associated with greater levels of precision during motor performance.

Functional asymmetries exist between PMC sub-regions (Schluter *et al.*, 1998; Schluter *et al.*, 2001; Beck *et al.*, 2009). For example, fMRI studies have revealed that the left dPMC has greater levels of neural activity when subjects performed motor tasks with both their right- and left-hand. However, the right dPMC was found to be more activated when subjects performed motor tasks with their left-hand (Schluter *et al.*, 2001). Although researchers have illustrated that PMC subregions contribute to motor-related activity, Freund and Hummelsheim (1985) reported that uncertainties still exist

regarding the specific functional role of the PMC and they studied this notion through lesion studies. For example, Kennard, Viets and Fulton (1934) reported that PMC damage can result in skilled movement deficits, forced motions during grasping tasks and muscle stiffness (Freund and Hummelsheim, 1985). On the other hand, Derousne (1973) reported that PMC lesions were associated with a lost of rhythmic motion and uncoordinated movement (Freund and Hummelsheim, 1985). Therefore controversies exist with regards to specific PMC functions because different behavioural observations have been reported in lesion studies (Freund and Hummelsheim, 1985). In this thesis, motor control and adaptation will be explored, therefore specific PMC functions only related to this type of motor activity will be investigated. Additionally, participants will have similar age ranges and no history of neurological conditions that could effect PMC functioning and cause varied results. This thesis will thus help answer questions relating to the specific role of the PMC during preparation for motor control and novel motor reaching.

1.8.1 TMS to the PMC

TMS studies targeting the PMC have explored differences between its subregions (e.g. vPMC vs. dPMC) during motor tasks. For instance, paired pulse TMS protocols have revealed that the vPMC influences M1 activity during grasp-to-hold tasks (Buch *et al.*, 2010). More specifically, Buch *et al.*, (2010) reported that the impact of the vPMC on the M1 depended on the type of activity that was being performed. For example, their physiological measurements of corticospinal excitability revealed that during movement planning and execution the vPMC had a facilitatory effect on M1 activity compared to tasks in which movements had to be updated in which the vPMC had an inhibitory effect on M1 activity. Buch *et al.*, (2010) also demonstrated behavioural effects of vPMC stimulation as their findings showed that subjects had a delay in adapting their grasping when their movements had to be updated (target switching from grasping a small cylinder to a large cylinder).

Furthermore, rTMS paradigms have been used in virtual lesion studies to provide a distinction between dPMC vs. vPMC functions. For example, Davare *et al.*, (2006) applied rTMS to the left and right dPMC and vPMC during a grasp-lifting task. They found that: 1) rTMS to the left vPMC interrupted sequential hand movement activity, 2) rTMS to both the left and right vPMC similarly resulted in disrupted finger positioning during grasping and, 3) rTMS to the left dPMC interrupted the phase between grasping

and lifting, and thus caused a delay in motor related activity. Davare *et al.*, (2006) therefore concluded that whilst the vPMC plays a role in finger positioning and grasping, the dPMC plays a role in managing the timing of sequential movements, e.g. from grasping to lifting, which is in line with studies in non-human primates (Murata *et al.*, 1997; Marconi *et al.*, 2001).

Furthermore, Schluter et al., (1998) explored functional hemispheric differences with regards to the role of the dPMC, and found that targeting the left dPMC with TMS protocols led to greater motor performance disruptions during bilateral tasks, whereas right dPMC TMS only affected motor tasks performed with the left-hand. As well as grasping tasks, PMC activity during reaching have also been investigated. For example, Ma et al., (2017) found that neural activity in the dPMC differed depending on the type of reaching being performed. They reported that whilst some neurons were only modulated by the location of the reaching target, other neurons in the dPMC were heightened and modulated during path switching of reaching trials. It should be noted that most studies exploring the role of the dPMC have implemented rTMS and pairedpulse TMS compared to SP-TMS paradigms and have mainly used grip/grasping and visuo-motor adaptation tasks as opposed to novel motor reaching tasks (Davare et al., 2006; Lee and Van Donkelaar, 2006; Buch et al., 2010). Chapter 9 and 11 therefore aim to provide novel findings regarding left and right dPMC functioning during preparation for normal and novel reaching with SP-TMS (virtual disruption approach). Sections 9.1 and 11.1 further describe the behavioural responses that have been observed with TMS protocols that have targeted the dPMC (left and right).



Figure 1.11: PMC location

The location of the PMC is demonstrated (purple) as are its two key structures: the dPMC (solid red line) and the vPMC (dotted red line) (figure adapted from the Florida Institute for Neurologic Rehabilitation, Inc: Atlas of Brain Injury and Anatomy: <u>http://www.finr.net/</u>).

1.9 Other regions of the motor network (the prefrontal cortex [dorsolateral] and the cerebellum) targeted with TMS

The prefrontal cortex has various neural connections with different regions of the motor network and therefore has a direct impact on motor behaviour (Goldman-Rakic, 1987). The functions of the prefrontal cortex range from executive functioning to organisation and planning (Tanji and Hoshi, 2001; Curtis and D'Esposito, 2003; Ball et al., 2011). An area of the prefrontal cortex involved in both movement control and motor output behaviour is the dorsolateral prefrontal cortex (DLPFC) (Hasan et al., 2012) (figure 1.12A - figure adapted from the Florida Institute for Neurologic Rehabilitation, Inc: Atlas of Brain Injury and Anatomy: http://www.finr.net). The DLPFC contains representations of motor sequences which is important in tasks where repetitive rhythmic motions are required (Shima et al., 2007). The DLPFC is also fundamental for motor learning due to its role in spatial attention, storing memories of sensory information and selecting appropriate motor responses (Badoud et al., 2017). Neuroimaging studies provide support for the role of the DLPFC in movement selection. For example, fMRI studies have found greater activity in the DLPFC when subjects were instructed to carry out an action selection task (Rowe et al., 2005). Similar findings have been illustrated with PET imaging techniques (Deiber et al.,1996). The DLPFC also plays a role in directing motor attention in a task which enables learning to take place (Shallice, 1982). This notion has been supported with repetitive TMS (rTMS) protocols. For example, Kantak et al., (2010) revealed that suppressing neuronal activity of the DLPFC with an rTMS paradigm during motor-skill learning had a detrimental effect on the motor learning process. Similarly, Robertson et al., (2001) found that rTMS delivered to the DLPFC disrupted spatial-sequential learning in a reaction time task.

The cerebellum is also a key structure in the motor network that is vital for motor control (Manto *et al.*, 2012) (see figure 1.12B - figure adapted from the Florida Institute for Neurologic Rehabilitation, Inc: Atlas of Brain Injury and Anatomy: <u>http://www.finr.net</u>). Research has found that this region regulates the timings of movements, assists in accurate motor output (particularly during reaching) and it is involved in motor learning and motor memory (Albus, 1971; Attwell, Cooke and Yeo *et al.*, 2002; D'Angelo and Casali, 2012; Koziol *et al.*, 2014). During motor adaptation (i.e. force-field [FF] learning), the cerebellum encodes the kinematic elements of movement

and stores them as motor memories (Overduin, Richardson and Bizzi, 2009). This can explain the overshooting phenomena that occurs in normal reaching trajectories (i.e. no FF) after motor adaptation has occurred (Hunter *et al.*, 2009; Hunter, Sacco and Turner, 2011; Pizzamiglio *et al.*, 2017b). The role of the cerebellum in motor functioning has been highlighted in patients with cerebellar damage, whereby poor performance has been reported (Landi, Baguear and Della-Maggiore, 2011). This is because the damage leads to patients no longer being able to store and recall the kinematic memories for motor adaptation (Criscimagna-Hemminger *et al.*, 2010; Landi, Baguear, and Della-Maggiore, 2011). Similarly, Bastian (2011) reported that impairments to the cerebellum resulted in movement control deficits (measured by trajectory reaching errors), as patients were not able to sustain accurate kinematics of movement control.

Studies using TMS have helped investigate the connectivity between different cortical regions, and how a region with deficits could impair motor function and physiological responses (Spampinato, Block and Celnik, 2017). For instance, in patients with cerebellar deterioration, MEP responses elicited during pre-movement were reduced when TMS was delivered to the M1 (Nomura, Takeshima, Nakashima, 2001). TMS studies during motor tasks have further elucidated the functional role of the cerebellum in the motor network. For example, rTMS of the medial cerebellum in a tapping task significantly affected the variability of tapping intervals during performance, however this was not the case during sham stimulation and also when targeting other regions, such as the M1 (Théoret, Haque and Pascual-Leone, 2001).

In summary, a variety of regions are involved and contribute to both upper limb reaching and adaptation. These regions range from the M1 to the PPC (Day, Rothwell and Marsden, 1983; Meyer and Voss, 2000) and other areas include the pre-motor cortex, as well as specific structures within it, such as the dPMC (Kantak *et al.*, 2012; Hardwick *et al.*, 2015). The role of the SMA should also be considered when noting its role in the planning of motor movements (Padoa-Schioppa, Li and Bizzi, 2004). Most TMS studies targeting these regions have implemented different TMS protocols, such as rTMS (Overduin, Richardson, and Bizzi, 2009; Ma, *et al.*, 2017). Therefore, the studies conducted within this thesis employed a standardised SP-TMS protocol to examine the role of several cortical regions during motor adaptation (MA). Additionally, very few studies have explored the effect of stimulation given at different preparatory time points on novel motor behaviour (i.e. during motor adaptation). Experiments in this thesis aim

to provide novel behavioural biomarkers of region-specific mechanisms operating during motor adaptation.



В.

A.



Figure 1.12: DLPFC and cerebellum location

The location of: (A) the DLPFC (green) and (B) the cerebellum (blue) is shown (figure adapted from the Florida Institute for Neurologic Rehabilitation, Inc: Atlas of Brain Injury and Anatomy: <u>http://www.finr.net/</u>).

Chapter 2

2 Purpose and aims of this thesis

2.1 Purpose and impact

This thesis aims to explore cortical region-specific mechanisms mediating functional reaching tasks using TMS. This knowledge could help bridge the gap between neural/cellular function and motor behaviour. Through exploring detailed limb kinematics, our understanding of how the brain interacts with muscles in a reaching task could be enhanced. Furthermore, by investigating how TMS induced disruption of cortical regions impacts on reaching behaviour we would gain an insight into how different cortical regions play a role in enabling us to learn new skills (i.e. notion of neuroplasticity) and how a change in neural activity can affect the ability to tackle new environmental demands.

The general questions that arise include:

- What neural cortical mechanisms are involved in both upper limb reaching and motor adaptation?
- Does delivering TMS at different times during the preparation of a reach alter MEP responses in different muscles?
- How does SP-TMS delivered at different times to different brain regions affect kinematics of reaching (i.e. behavioural responses) (in both normal reaching and motor adaptation)?
- How do MEP responses change in a novel motor reaching task (motor adaptation task vs. simple motor reaching task)?

The specific aims of the project are:

- To explore basic neural mechanisms of right arm reaching with single pulse TMS delivered to both the contralateral M1 and ipsilateral M1 (Experiment 1).
- To examine neural mechanisms (experiment 1 and 2) and behavioural responses (i.e. kinematics) (experiment 1 -8) during right arm reaching in a robot-mediated force-field with single pulse TMS delivered to the:

1) Left primary motor cortex (experiment 2)

- 2) Left posterior parietal cortex (experiment 3)
- 3) Supplementary motor area (experiment 4)

- 4) Right primary motor cortex (experiment 5)
- 5) Left dorsal premotor cortex (experiment 6)
- 6) Right posterior parietal cortex (experiment 7)
- 7) Right dorsal premotor cortex (experiment 8)
- To compare region-specific roles in motor control and motor adaptation

2.2 Specific hypotheses

Experiment 1 – Exploration of 1) contralateral and 2) ipsilateral neural mechanisms of right arm reaching using single pulse TMS:

Hypothesis I: TMS pulses delivered in the preparation phase of a reach will have a significant disruptive impact on the kinematic behaviour of the subsequent reach. This is interesting because if true, the impact of disruption of M1 neurotransmission to the CST (i.e. by the TMS pulse) can be directly linked to disruption of motor behaviour. In other words, TMS is used as a tool for a virtual disruption in the M1/CST pathway.

<u>Experiment 2</u> – Exploration of neural mechanisms of right arm reaching in a force field using single pulse TMS applied to the left M1:

Hypothesis I: There will be an increase in the MEP measured in the biceps brachii (BB) and flexor carpi radialis (FCR) (flexor) muscles because the force field is a clockwise velocity-dependent field, but not in the triceps brachii and extensor carpi radialis (extensor) muscles (see Pizzamiglio et al., [2017b] for comparisons of muscle responses to clockwise and counter clockwise force fields).

Hypothesis II. TMS will delay the movement onset by inhibiting a group of neurons in the brain (as previously demonstrated by Day, Rothwell and Marsden, 1983). The virtual disruption approach using TMS experimentally here can be compared to neurocomputational models where similar "data-based virtual lesions" are introduced into models of stroke (Small, Buccino and Solodkin, 2013).

<u>Experiment 3</u> – Exploring the impact of SP-TMS to the left posterior parietal cortex (PPC) during right arm reaching in a motor adaptation protocol:

Hypothesis I. The disruption of PPC function using TMS will impair novel motor performance (as demonstrated by Della-Maggiore *et al.*, 2004). This is because
the PPC feeds into other motor related cortical structures, signalling commands for successful motor output.

Hypothesis II. Different TMS timings will affect kinematic behaviour of the subsequent reach. These effects are not yet known and this is the first attempt to describe such behavioural data during motor adaptation.

<u>Experiment 4</u> – Exploring the impact of SP-TMS to the supplementary motor area (SMA) during right arm reaching in a novel motor learning protocol:

Hypothesis I. TMS applied to the SMA would have a disruptive effect on novel motor performance. This is because the SMA is known to be involved in fast-learning which is required for motor adaptation to take place (King *et al.*, 2013).

Hypothesis II. Different TMS timings will affect the kinematic behaviour of the subsequent reach, as the SMA is involved in motor planning and preparation (Padoa-Schioppa, Li and Bizzi, 2004).

<u>Experiment 5</u> – The impact of SP-TMS to the right primary motor cortex (M1) during right arm reaching in a novel motor learning protocol

Hypothesis I. The ipsilateral motor cortex will be modulated during movement preparation, because it has been shown that the ipsilateral M1 can too undergo task-related modulations of activity (Van den Berg, Swinnen and Wenderoth, 2011).

Hypothesis II. TMS will delay the movement onset by inhibiting a group of neurons in the brain (Day, Rothwell and Marsden, 1983).

<u>Experiment 6</u> – Exploring the impact of SP-TMS to the left dorsal pre-motor cortex (dPMC) during right arm reaching in a novel motor learning protocol:

Hypothesis I: TMS applied to the left dPMC during force-field reaching will have an effect on behavioural responses, demonstrating that this region is also actively involved in movement planning and preparation.

Hypothesis II: In simple reaching different TMS timings will have an effect on kinematic behaviours. This is because TMS can pre-activate the stimulated region before its typical activation begins, which as a result may speed up the onset of movements (Silvanto and Muggleton, 2008). It is unknown whether this is the case for motor adaptation, this will be therefore be explored.

<u>Experiment 7</u> – Exploring the impact of SP-TMS to the right posterior parietal cortex (PPC) during right arm reaching in a novel motor learning protocol:

Hypothesis I: TMS applied to the right PPC during force-field reaching will have an effect on behavioural responses, demonstrating that this region is actively involved in movement preparation.

Hypothesis II: Although TMS may have an impact on reaching trajectories, the impact of TMS may not be detrimental at all time points of when TMS is delivered. This is because studies have shown that the right PPC is involved only in planning left handed, and not right handed reaching movements (Schluter *et al.*, 2001; Oliveira *et al.*, 2010).

<u>Experiment 8</u> – Exploration of behavioural mechanisms of right arm reaching in a force field using single pulse TMS applied to the right dPMC:

Hypothesis I: TMS applied to the right dPMC during force-field reaching will have an effect on behavioural responses, because studies have shown that the region is densely connected to different structures in the motor circuitry (such as the prefrontal cortex and PPC) (Genon *et al.*, 2017), and is therefore involved in motor reaching and planning. This study is the first attempt to describe the effect of TMS delivered to the right dPMC (at different time points) during preparation for novel reaching.

<u>Experiment 9</u> – Comparing region-specific roles in the motor network during motor control and motor adaptation:

Hypothesis I: Statistical comparisons may reveal different levels of influence across different regions (quantified by summed error) thereby indicating the relative importance of specific regions in motor control and adaptation. This chapter builds a novel model for studying the motor network to demonstrate the different functional properties of cortical structures.

The general methodology implemented to study the aims and hypotheses' in this thesis are described in section 3.

Chapter 3

3 General methodology

This chapter describes the general procedures that were employed in this thesis and illustrates the apparatus used in each experiment. Although all studies employ similar apparatus (i.e. use of TMS, robotics and EMG), some experimental paradigms differ in terms of the task being carried out (e.g. force-field [FF] reaching vs. null FF reaching [FAM]) and the region of interest being stimulated with TMS. Each experimental chapter contains the methodological protocol that was specifically designed for the different tested hypotheses.

3.1 Ethics

All the procedures in this thesis were approved by the University Research Ethics Committee (UREC 1516_108, Appendix 1) and conducted according to the Declaration of Helsinki (World Medical Association, 2013). The approval of the research was issued to the researcher (Pegah Mohajer Shojaii), the principle investigator (Professor Duncan Turner) and the research site (University of East London - NeuroRehabilitation Unit). Before each experiment was conducted, participants were verbally instructed regarding the aims of the project.

In all experiments, the participants also received:

- A participant information sheet outlining the details of the experimental protocol (Appendix 2)
- A consent form to participate in a study involving the use of human participants (Appendix 2).
- A medical questionnaire to complete in order to ensure that TMS is a safe procedure for them to undergo (i.e. they do not have any metal or electrical implants) (Appendix 3).
- The researchers' contact details, including an e-mail address if they wish to obtain further details about the study, their performance, or their data/results (Appendix 1).

Participants were also made aware that they were free to withdraw from the experiment at any given time and were also welcome to discuss their thoughts and feelings with regards to their participation in the research.

3.2 Participants

Participant recruitment took place through online advertising (call for participants recruitment website – <u>https://www.callforparticipants.com/</u>), through emails and leaflets and 12-15 healthy young adults were recruited per experiment. The sample size selected was based on, 1) established methods used in the NeuroRehabilitation Unit, 2) similar studies that have been published (Hunter *et al.*, 2009; Hunter, Sacco and Turner, 2011), 3) taking into account the variation of M1 excitability in healthy young adults, and 4) non-learning which can occur in motor adaptation paradigms (Pizzamiglio *et al.*, 2017b). For example, we anticipated that one or two subjects per hypothesis will:

- a) Not perform reaching correctly (i.e. their reaching movements would either be too fast or too slow)
- b) They would not learn very well (i.e. more than 2 standard deviations away from the average learning rate)
- c) They may not exhibit significant MEPs in the muscles of interest (biceps brachii).

Furthermore, only right handed participants were recruited and only the dominant hand was investigated. This is due to different brain responses in the dominant and non-dominant brain hemispheres (Haaland and Harington, 1996). The inclusion and exclusion criteria for participant recruitment are further illustrated in table 3.1.

Throughout this thesis TMS was used to target seven regions in the motor network, and the participant demographics with regards to each experiment is outlined in table 3.2. It should be noted that once all of the experiments were complete, the next experimental approach that was taken aimed to explore the importance and contribution of different cortical regions in motor control and motor adaptation. In order to investigate this, all of the experimental data (chapters 4 - 11) were pooled together and synthesised. This then formed the outline for chapter 12 in this thesis.

Table 3.1: Inclusion and exclusion criteria for participation

| Category: | Inclusion criteria: | Exclusion criteria: | | |
|------------------------|---------------------------------|--|--|--|
| Participant health: | Healthy participants | History of neurological or psychiatric | | |
| | | conditions | | |
| | | Neuromuscular disease | | |
| Age range: | Young adults aged > 18 | Aged <18 and >40 | | |
| Handedness: | Right handed | Left handed | | |
| TMS safety guidelines: | No metal or electrical implants | Metal or electrical implants | | |

| | Experiment 1 (Chapter 4): | Experiment 2 (Chapter 5): | Experiment 3 (Chapter 6): | Experiment 4 (Chapter 7): | Experiment 5 (Chapter 8): | Experiment 6 (Chapter 9): | Experiment 7 (Chapter 10): | Experiment 8 (Chapter 11): | Chapter 12: | |
|--|--|--|--|--|--|--|---|---|---|--|
| Ethical approval/ participants health: | Following ethical approval from the University of East London Research Ethics Committee (Appendix 1), all of the participants for each study met the inclusion criteria and had no personal or family history of neurological/psychiatric conditions, no neuromuscular disease and no metal/electrical implants (based on Appendix 3) and gave their written consent to participate (Appendix 2). All aspects of each study were conducted in accordance with the Declaration of Helsinki (World Medical Association, 2013). | | | | | | | | | |
| Age range (mean, +/- SEM): | 21 - 32 (26, +/- 1) | 18 – 37 (26, +/- 1) | 20 – 32 (26, +/- 1) | 20 – 32 (26, +/- 1) | 18 – 32 (24, +/- 1) | 18 – 32 (24, +/- 1) | 18 – 32 (23, +/- 1) | 18 – 32 (23, +/- 1) | 18 – 37 (25, +/- 1) | |
| Gender: | 5 male; 8 female | Prior to data exclusion: 9 male: 7 female Following data exclusion: 8 male: 6 female | 5 male: 9 female | 5 male: 9 female | 3 male: 10 female | 3 male: 10 female | 3 male: 10 female | 3 male: 10 female | 30 male: 64 female | |
| Number of participants excluded and reasons for exclusion: | N/A (no participants were excluded) | Data from two subjects were excluded due to non-learning in the motor adaptation paradigm. | N/A (no participants were excluded) | N/A (no participants were excluded) | N/A (no participants were excluded) | N/A (no participants were excluded) | N/A (no participants were excluded) | N/A (no participants were excluded) | N/A (no participants were excluded) | |
| Number of participants (N): | 13 | 14 following data exclusion | 14 | 14 | 13 | 13 | 13 | 13 | 94 | |
| Mean stimulus intensity (i.e. at 110% RMT) | 56% | 56% | 57% | 57% | 56% | 57% | 55% | 56% | | |

3.3 Transcranial magnetic stimulation

3.3.1 Electromyography (EMG) recordings for MEPs:

Overall electrical signal and activity that is generated from muscles can be recorded either invasively or non-invasively using EMG (Mills, 2005; Chowdhury *et al.*, 2013). Invasive methods require the use of needle electrodes that are fit into the muscle(s) to capture the activity of muscle-fibres from a single unit (Chowdhury *et al.*, 2013). Noninvasive methods on the other hand, employ the use of surface electrodes that are positioned onto the muscle(s) of interest and the signal recorded is based on overall motor-unit activity (Chowdhury *et al.*, 2013). MEP responses in TMS research have commonly been recorded as a signal with surface EMG electrodes (Hunter, Sacco and Turner, 2011; Orban de Xivry *et al.*, 2013; Summers, Chen, Kimberley, 2017).

The use of surface EMG to record MEPs have helped demonstrate patho-physiological responses in clinical populations, which have been found to differ when compared to responses from healthy controls. For example, Brum, Cabib and Valls-Sole (2015) recorded MEPs from the first dorsal interosseous (FDI) muscle with EMG surface electrodes in stroke patients and they found a significant decrease in peak-to-peak MEP amplitude among patients compared to healthy subjects. Additionally, MEPs from the FDI muscle in patients with multiple sclerosis (MS) were found to have a longer duration compared to a healthy control group (Kukowski, 1993; Brum, Cabib and Valls-Sole, 2015). Therefore, surface EMG recordings have been used to characterise differences in MEP responses and this has helped provide further information regarding patho-physiological mechanisms in different clinical populations (Chowdhury *et al.*, 2013; Fernandez *et al.*, 2013).

In this thesis, surface EMG was used to record MEP signals (EMG, mV) using Signal Software Version 6 (Cambridge Electronic Design LTD, Cambridge UK). For experiments 1 and 2, MEPs were only recorded as a signal for the duration of the experiment when TMS was delivered to the contralateral M1 in order to explore corticospinal excitability (CSE) responses during right arm reaching. For experiments 3 (left PPC TMS), 4 (SMA TMS), 6 (left dPMC TMS), 7 (right PPC TMS) and 8 (right dPMC TMS), MEPs were not elicited during stimulation. For experiment 5, MEPs were not collected as kinematics of right arm reaching was of interest. However, one of the

steps of the TMS protocol was to identify the participants resting motor threshold (RMT) (left M1) to deliver TMS to the region of interest at 110% of their RMT. Therefore, MEP responses were only acquired and recorded during the assessment of RMT for those studies. With regards to experiment 5, the right M1 was targeted to identify resting motor threshold (RMT) by measuring MEPs from the left arm. The TMS protocol for this experimental set-up was different compared to the other experiments within this thesis (except for experiment 1 and experiment 2) whereby the left M1 was targeted and MEPs were elicited from the right arm to identify RMT. Targeting the right M1 in experiment 5 helped ensure that an appropriate stimulation intensity was used for each participant during TMS when considering that motor thresholds can fluctuate and differences in RMT have been identified between the left M1 and right M1 (Alm et al., 2013; Karabanov, Raffin and Siebner, 2015). Paired-pulse TMS studies that have targeted both the right and left M1 have also used similar protocols to separately identify hemispheric motor thresholds between the two regions prior to stimulation (Morishita *et al.*, 2014). Overall, the data for experiment 3 - 8 were based on kinematics only and not MEPs.

Where MEPs were obtained, the EMG signal was amplified (1000x; 1902 amplifier; Power 1401, CED LTD, UK), band-pass filtered (45 Hz high pass, 1kHz low pass) and a notch filter was applied (50Hz). The data were digitised at 5kHz (Micro 1401; Cambridge Electronic Design LTD, Cambridge UK). MEPs were always recorded with two disposable solid gel surface electrodes (Unimed Electrode Supplies LTD, Surrey, UK) positioned 1.5 centimetres apart (SENIAM [Surface Electromyography for the Non-invasive Assessment of Muscles] guidelines; Hermens *et al.*, 2000) along the muscle fibre direction of the right-sided biceps brachii (BB), triceps brachii (TB), extensor carpi radialis (ECR) and flexor carpi radialis (FCR) muscles. A ground electrode was also placed over the ulna styloid process bone (where the ulna joins the wrist) on the left arm.

These muscles have been studied in motor reaching paradigms, as they have been reported to be involved in horizontal planar reaching (Hunter, Sacco and Turner, 2011; Orban de Xivry *et al.*, 2013). For example, the BB is not only a rotator of the forearm muscles, but it is also an elbow flexor (which is essential for reaching outwards) (de Bruin *et al.*, 2011; Landin, Thompson and Jackson, 2017) and the TB is known to be involved in elbow extension and straightening the arm (O'Driscoll, 1992; Singh and

Pooley, 2002). The FCR plays a role in opposing a clockwise force (i.e. during motor adaptation) (Pizzamiglio *et al.*, 2017b) and is a wrist flexor muscle, while the ECR is known to be a wrist extension muscle.

These muscles are graphically illustrated in figure 3.1 (figure adapted from Learn Muscles: https://learnmuscles.com/wp-content/uploads/2016/12/Wrist-Sp-St-Blog-Post-Photo-1.jpg).



Figure 3.1: Muscles of interest chosen for electrode placement (outlined in red)

Muscles including those involved in extension (TB and ECR) and flexion (BB and FCR) are illustrated. The BB consists of a short head and long head within the shoulder region, which both connect at the elbow, the BB also connects to the ECR. The TB is comprised of different structures, including a long head and medial head which originate at the mid-section of the humerus (Watson and Wilson, 2007; Landin, Thompson and Jackson, 2017). These were the four muscles of interest in this thesis. Figure adapted from Learn Muscles: https://learnmuscles.com/wp-content/uploads/2016/12/Wrist-Sp-St-Blog-Post-Photo-1.jpg.



Figure 3.2: Visualisation of surface EMG recordings of MEPs

TMS applied to the left M1 during motor reaching (top panel) and data collected in Signal Version 6 and CED data acquisition software (Cambridge Electronic Design LTD, Cambridge UK) (second panel). Examples of MEP data from one specific TMS pulse time (250ms) are shown in the coloured traces. Each channel on the left-hand side of the bottom panel is associated with one muscle (blue for BB, black for TB, pink for ECR and green for FCR). The grey activity in the background represents overall EMG from each reaching trial and has been overlapped to illustrate the degree of muscle activity during a reaching trial. This raw data trace also illustrates the formal description of the preparation phase of reaching and the chosen timing at which TMS was delivered. Zero on X-axis represents the time (sec) of the visual cue to reach and the Y-axis represents elicited (by TMS) and ongoing muscle activity (mV) during the trial.

3.3.2 TMS Protocol:

In all experiments, TMS was delivered with a figure of eight coil connected to a Magstim Stimulator 200² (Magstim Company, Whitland, UK). The stimulator was triggered by a custom-built programme using Signal Version 6 and CED data acquisition software (Cambridge Electronic Design LTD, Cambridge, UK). Participants also wore a washable flat cap for the duration of the experiment, to mark the regions of interest that were stimulated, depending on the hypothesis being studied (left or right M1; left or right posterior parietal cortex, PPC; the supplementary motor area, SMA; left or right dorsolateral premotor cortex, PMC).

The TMS testing procedure always began with the coil being positioned above the scalp on the participant's left M1 to locate their motor 'hotspot' for the right-sided BB muscle. The area of the M1 stimulated for this was the upper arm representation which is approximately 4cm lateral to the top of head (vertex). To determine the optimal coil position for evoking MEPs in the right BB muscle, TMS was applied in 1cm steps around the upper limb region, with the coil handle positioned backwards and angled 45° away from the midsagittal line. The direction of the TMS induced current was posterioranterior (PA). This coil position, and current flow has consistently been identified as the optimal position when targeting the BB, as well as other muscles, such as the first dorsal interossei (FDI) (Abdalla, 2011; Mills, Boniface and Schubert, 1992).

Once the hotspot was located and coil position was determined, it was marked with a washable pen on the participant's cap. This was to ensure consistent coil position for the duration of the experiment. In all experiments, resting motor threshold (RMT) was defined as the minimum intensity that induced MEPs \geq to 50 µV peak to peak in the BB muscle (hotspot) in 5/10 trials (Hunter, Sacco and Turner, 2011; Rossi *et al.*, 2009).

It should be noted that in all experiments, single pulse TMS (SP-TMS) was applied at 110% of RMT in all TMS reaching blocks (FAM, FF and WO) at a range of specific times after the visual cue (10ms, 100ms, 130ms, 160ms, 190ms, 220ms, 250ms, 280ms and 310ms) during the preparation phase of a reach - this is at a time after the visual cue to reach and before the actual onset of the physical reach. It should also be noted that for each experiment, 20 single pulses at rest (with the right arm relaxed in the robotic arm) was obtained from each participant before they began the motor reaching tasks. For all experiments, the time points at which a TMS pulse was delivered during a trial of

reaching was randomly selected through a custom built program using Signal Version 6 and CED data acquisition software (Cambridge Electronic Design LTD, Cambridge, UK).

3.3.2.1 Cortical targeting with TMS:

The participants in this experiment did not have individual structural MRI scans, therefore an electroencephalography (EEG) cap was used to locate the different regions that were targeted with TMS. An EEG cap contains various electrodes which correspond to different cortical regions (Herwig, Satrapi and Schönfeldt-Lecuona, 2003). The use of EEG to locate cortical regions for brain stimulation with TMS methods have consistently been employed (Herwig, Satrapi and Schönfeldt-Lecuona, 2003; Schutter and van Honk, 2006; Beam et al., 2009).

For the experiments in this thesis that had no physiological experimental output (i.e. an MEP) (experiment 3, 4, 6, 7 and 8) a 10-20 systems EEG cap (64 channels) (ANT Neuro, Enschede, Netherlands) was fitted on top of a washable flat cap that participants wore. Following cap placement two measurements were taken with the Cz electrode (which overlies the vertex) as the common reference point to locate the mid-sagittal plane of the skull. The first measurement that was marked was 50% of the distance between the nasion and the inion (this corresponded to the Cz electrode). A second measurement was then taken to confirm the location of the Cz electrode and this was taken from the left ear, to the right area (two pre-auricular points - 50% of the distance between the two). These landmark measurements individualised the electrode positions for each participant. The electrode position of interest (depending on the experiment that was being conducted at the time) was then marked with a pen through the EEG cap to the flat cap that the participants wore. The EEG cap was then removed and the inner markings of the electrode position for stimulation remained on the flat cap that the participants wore. It should be noted that when EEG cap was removed, the flat cap was not adjusted or shifted as it was tightly fitted on the subjects head and this helped further ensure the accuracy of cortical targeting during stimulation.

3.3.2.2 Cortical targeting of regions without an experimental output and coil positions employed:

Regions of interest that had no measured output (i.e. an MEP; left PPC, SMA, left dPMC, right PPC and left dPMC) were located using an EEG cap (10-20 system; ANT

Neuro, Enschede, Netherlands) once measurements were taken with the Cz electrode as the reference point for the mid-sagittal plane of the skull (section 3.3.2.1). It is also important to consider the different types of coil positions and orientations that can be used to assure that the behavioural responses that occur is due to the stimulation of the targeted area, rather than its surrounding cortical regions. It has been argued that employing one stimulation intensity across participants could result in differences in task responses, this is because the effects of stimulation could either be too much, or too little for each participant (Bolognini and Ro, 2010). Researchers have therefore reported that when targeting regions with no experimental output TMS intensity should be set individually for each participant through functional measures, such as resting motor threshold for an MEP obtained in left M1 (Bolognini and Ro, 2010). The following paragraphs below outline: 1) the electrode position selected for the different regions of interest and 2) the optimal TMS coil position that was used across stimulated sites.

3.3.2.2.1 Left PPC location and coil position:

For experiment 3, the P3 electrode location was used for left PPC stimulation. It has been reported that P3 corresponds to the left PPC with very little variation across participants (< 2cm) (Herwig, Satrapi and Schönfeldt-Lecuona, 2003). Various TMS studies that have employed EEG methods for location purposes have also used the P3 electrode for left PPC stimulation (Pourtois *et al.*, 2001; Salatino *et al.*, 2014; Parks *et al.*, 2015). Additionally, the coil was positioned at an angle that was tangential and perpendicular to the midline, with the handle pointing sideways. Therefore, the current flow was in a lateral-medial direction. The coil position and orientation used for left PPC stimulation was based on the protocol of Della-Maggiore *et al.*, (2004).

3.3.2.2.2 SMA location and coil position:

With regards to experiment 4, it has been reported that the SMA is located 3cm anterior to the vertex and 0.5cm to the left, which corresponds to the FCz EEG electrode (Cunnington *et al.*, 1996; Oliveri *et al.*, 2003). This area was therefore marked for stimulation and the coil was positioned tangentially to the skull with the handle pointing to the right (based on Cona et al., 2017 protocol for SMA stimulation).

3.3.2.2.3 Left dPMC location and coil position:

The left dPMC was targeted in experiment 6 and was located as 2cm anterior and 1cm medial to the left M1 motor hotspot with the coil angled at 45° with respect to the

interhemispheric fissure (handle backwards) as this has commonly been used in TMS paradigms (Fink et al., 1997; Münchau *et al.*, 2002; Lee and Van Donkelaar, 2006; Zanon et al., 2013; Lega et al., 2016).

3.3.2.2.4 Right PPC location and coil position:

For experiment 7 in this thesis, the P4 electrode was marked for stimulation because it has been argued to correspond to the right PPC (Herwig, Satrapi and Schönfeldt-Lecuona, 2003; Prime, Versia and Crawford, 2008; Koch *et al.*, 2009; Vernet, Yang and Kapoula, 2011). Additionally, the coil handle was positioned backwards (angled 45° to the midline) to induce a posterior-anterior current (Prime, Versia and Crawford, 2008; Koch *et al.*, 2014).

3.3.2.2.5 Right dPMC location and coil position:

The final experiment in this thesis was based on targeting the right dPMC, this region was located as 2 - 2.5cm anterior and 1cm medial to the motor hotspot and the coil was angled 45° from the midline (handle backwards and downwards) (Cincotta *et al.*, 2004; Bestmann et al., 2005; Murase *et al.*, 2005; Ruitenberg *et al.*, 2014).

Figure 3.3 illustrates specific EEG locations for the stimulated regions and the different coil positions that were used (and electrical current that was induced) for each stimulated site (canonical brain figures adapted from an MRIcron template – Rorden and Brett, 2000).

3.3.2.3 Ensuring accurate targeting:

Researchers have reported the importance of accurate coil positioning during stimulation and different methods have been employed to ensure this, including TMS coil holders and neuronavigation (Chronicle, Pearson and Matthews, 2005; de Goede, Braack and van Putten, 2018). With regards to coil holders, Chronicle, Pearson and Matthews (2005) explored the precision, stability and durability of this method during M1 stimulation and their findings illustrated accuracy across different testing sessions. Automated robotic coil holders have also been used during stimulation and they have been argued to provide increased stability which has been reported to be particularly useful for longer experimental protocols and rTMS paradigms (Goetz *et al.*, 2019). Coil apparatus for positioning purposes have also been useful during multimodal MRI-TMS imaging studies and have provided clinicians with a compatible way to accurately explore human behaviour (Navarro de Lara *et al.*, 2015). On the other hand, there are

limitations to consider. For example, Goetz *et al.*, (2019) reported the absence of manual adjustments, as well as the lack of coil pressure feedback during stimulation. Despite these shortfalls, coil holders have been reported to be both an effective and reliable method for improving TMS coil accuracy (Sparing *et al.*, 2008; Rodseth, WashaBaugh and Krishnan 2017),

Neuronavigation methods have also been implemented in TMS protocols for: 1) marking anatomical regions and, 2) to ensure accurate coil positioning during stimulation. Neuronavigation functions via mapping out various physical landmarks on the subject, such as their cranium, ears and nasion. Following this, the system registers the TMS coil for the navigation of cortical structures (Sparing, Hesse and Fink, 2010). Neuronavigation methods take coil orientation in both space and time into account, which helps minimise any structural inaccuracies that may occur during stimulation (Sparing, Hesse and Fink, 2010). However, de Geode *et al.*, (2018) argued that research is yet to confirm the significant impact of neuronavigation with regards to coil orientation for SP and PP TMS protocols.

For each experiment in this thesis, the coil was positioned and held manually on the region of interest on the participants scalp. This free-hand method does have limitations that should be considered, particularly when taking into account that increased accuracy has been reported with the use of coil positioning equipment, and that manual coil holding has been associated with operator fatigue (Goetz et al., 2019). For location marking, an EEG 10/20 system method was used rather than neuronavigation and this is because the participants in each study did not have a structural MRI scan that could have been imported into the neuronavigation system. Furthermore, the use of neuronavigation for accurate coil positioning would have increased the duration of the reaching protocol (from 2 hours 30 minutes, to three hours) (this was tested in a pilot study) which could have been arduous for the participant during motor reaching. Despite coil position being determined manually, it should be considered that participant's head movements were minimised because the reaching protocol only required arm motions, additionally seatbelt straps were attached to the participants (section 3.4.1) to avoid trunk movements that may have occurred during reaching. Furthermore, there was manual control of the TMS coil, therefore, compensations for any head motions that did take place were taken into account. In addition to this, continuous attention was given to the coil position during the experimental protocol to

ensure precise coil placement and orientation, which has been argued to be important for TMS protocols that use manual methods for stimulation (Goetz *et al.*, 2019). Nonetheless, the use of manual methods still remains a limitation that should be considered, and if this experiment was to be replicated it would be important to implement techniques that aid coil accuracy, including neuronavigation or automated coil holders (Sparing, Hesse and Fink, 2010; Goetz *et al.*, 2019).



Figure 3.1: EEG locations for stimulated regions and optimal coil positions for the different locations targeted with TMS

- (A) The different locations of the stimulated areas with reference to a 64-channel EEG cap is demonstrated (ANT Neuro, Entschede, Netherlands) (figure adapted from <u>www.ant-neuro.com</u>). The outlined figure below figure 3.3A illustrates a canonical model of all of the regions targeted with TMS (canonical brain figure adapted from an MRIcron template Rorden and Brett, 2000).
- (B) The coil positions for the various cortical regions targeted with TMS are illustrated. This includes an induced posterior-anterior current flow for the M1 and dPMC (both left and right hemispheres). For the left PPC, but not for right PPC stimulation, a lateral-medial current flow was induced. For SMA stimulation, the coil was positioned tangentially to the skull (handle to the right) (canonical brain figures adapted from an MRIcron template Rorden and Brett, 2000).

3.4 Robotics

In all experiments, an interactive robot was used for motor reaching (MIT - Manus, Interactive Motion Technologies, Cambridge, MA, USA). The robot functioned via Linux software and also contained encoders which recorded different angular positions of the robotic manipulandum. There are different modes in which the robot could be used, this includes:

1. Assistive mode - this has been employed is rehabilitation programmes whereby the robot assists movements for motor output and different degrees of assistance can be provided (Patton *et al.*, 2006).

2. Resistive mode - this has been used in motor adaptation paradigms, where subjects are required to counter a force-produced in order to perform a reach with an ideal trajectory (Pizzamiglio *et al.*, 2017a; Pizzamiglio *et al.*, 2017b).

3. Non-assistive mode - In this mode there is no assistance or resistance elicited by the robot during reaching.

The use of robotics in assistive and resistive forms has commonly been employed in motor reaching and adaptation in both healthy and clinical populations (Patton *et al.*, 2006; Hunter, Sacco and Turner, 2011; Orban de Xivry *et al.*, 2013). Clinically, Patton *et al.*, (2006) noted that a resistive, rather than assistive mode led to enhanced therapeutic outcomes among stroke patients. They reported that in contrast to assistive reaching, reaching in a resistive mode, resulted in larger errors by which the patients were able to make significant improvements on. This demonstrates the importance of error-induced reaching through FF paradigms as a key model for motor recovery (Patton *et al.*, 2006).

3.4.1 Reaching task: motor reaching and motor adaptation:

In this thesis the reaching task that subjects performed differed between experimental protocols:

1) Experiment 1 (chapter 3):

This motor task was performed in the non-assistive mode of robot operation as unperturbed (no force field present) reaching was explored. Therefore no robotic assistance or resistance was elicited during reaching. Participants were instructed to make outward arm reaching movements towards a north-west target (135°) from a central starting position (15cm reach) within a target time of 1.0 - 1.2 seconds following the onset of a visual cue. Visual feedback for the timing of the reaches was shown on the screen (Section 3.4.2 describes further details regarding feedback). When participants reached towards the target and held their arm position, the robotic arm repositioned the participants arm into the central starting point for the next reaching trial. The reaching protocol for this experiment contained two blocks of unperturbed reaching with no TMS (N = 24 trials per block), followed by 3 blocks of unperturbed reaching with TMS trials to the contralateral M1 (N = 48 x 3 trials total). Participants then had a break (5 minutes). The same protocol was then followed, however the ipsilateral M1 was stimulated. In all stimulation trials, SP-TMS was delivered at nine different time points (10 - 310ms) (randomly) during the preparation phase of motor reaching, at 110% resting motor threshold.

A graphical illustration of the paradigm for experiment 1 is demonstrated in figure 3.4.

2) Experiments 2-8 (chapters 4 -11):

Motor adaptation was introduced into the reaching protocols for experiments 2-8 and the robot was used in a resistive mode. In these experiments familiarisation (FAM), force field (FF) and washout (WO) reaching conditions were explored. The robot was set to a non-assistive mode (i.e. unperturbed reaching) for FAM and WO reaching, and a resistive mode for FF reaching. During resistive FF reaching, the robot administered a velocity dependent clockwise force field (25 N ms⁻¹). In all reaching trials, participants were instructed to make outward arm reaching movements (away from the body) towards a north-west target (135°), from a central starting position (15cm reach) following the onset of a visual cue.

The experimental paradigm that was implemented for experiment 2 - 8 is shown in figure 3.5.



Figure 3.4: Graphical illustration of the paradigm implemented for experiment 1

The protocol for experiment 1 included two blocks of unperturbed reaching with no TMS, this was for the participants to familiarise themselves with the reaching task. TMS was then introduced (three blocks of unperturbed reaching) and single-pulses were delivered on the participant's left M1at a range of different times (10-310ms) to capture changes in M1 cortical excitability during the preparation of a reach. Following this, participants took a 5 minute break. The same protocol was undertaken, however with TMS delivered to the right M1 (canonical brain figures adapted from an MRIcron template – Rorden and Brett, 2000).



Figure 3.5: Graphical illustration of the reaching paradigm implemented for experiments 2 - 8:

The paradigm for experiments 2 – 8 included single-pulse TMS delivered to specific regions of the motor network. The area that was targeted with TMS depended on the experiment that was being conducted at the time (e.g. experiment 2, left m1 TMS; experiment 3, left PPC TMS; experiment 4, SMA TMS; experiment 5, right M1 TMS; experiment 6; left dPMC TMS; experiment 7; right PPC TMS and experiment 8, right dPMC TMS). TMS was delivered at a range of specific times to explore cortical excitability (specifically for experiment 2) and behavioural factors (for all experiments) during movement preparation. The motor paradigm consisted of different phases of reaching including, FAM reaching - firstly without TMS [2 blocks], and then with TMS delivery [2 blocks]. Following FAM blocks of reaching, motor adaptation was introduced, with FF blocks of reaching, firstly without TMS [4 blocks], and then with TMS [4 blocks]. The final 4 blocks at the end of the protocol acted as WO blocks, with 2 blocks of without TMS and a further 2 blocks with TMS (Canonical brain figures adapted from an MRIcron template – Rorden and Brett, 2000).

3.4.2 Reaching task set up and instructions:

Although experimental protocols differed for location of TMS (as outlined in section 3.3.2) the experimental set up was the same and instructions were only different when explaining the motor adaptation protocol. In all experiments, participants were seated in a customised chair in front of the robotic visual screen with their right arm rested in the robotic arm for support against gravity during reaching. The robotic arm contained an end-effector handle that the participants were asked to hold during reaching. Shoulder straps connected to the customised chair were attached to the participant to restrict trunk movement, particularly during motor adaptation (a common factor implemented in robotic motor adaptation paradigms - Hunter, Sacco and Turner, 2011; Pizzamiglio *et al.*, 2017b).

A number of measurements were also made to ensure that the arm was semi-pronated at a 70° shoulder extension and with 120° elbow flexion (Hunter, Sacco and Turner, 2011). Additional measurements included making sure that the participants shoulder and end-effector handle were at an equal height. A vertical screen was attached to the robot and positioned at the participant's eye-level. This presented both: a) visual cues for the reach, and b) online feedback of their arm position during each reach (Pizzamiglio *et al.*, 2017b). In all motor tasks, participants made outward arm reaching movements from the central starting point towards a north-west direction target (135°) within a 1.0 - 1.2 second time frame, following the onset of a visual cue (target was highlighted in an orange colour). After each single reaching trial, the robotic arm was automatically re-positioned into the central starting target, for the next reaching trial to take place.

A graphical illustration of the lab set up and robot reaching task is shown in figure 3.6A and 3.6B.



B)



Figure 3.6: Experimental set up and reaching task

The experimental set up is demonstrated with the main equipment outlined in yellow (EMG channel electrode adaptor, CED data acquisition system, the TMS machine and the robotic system [screen and robotic arm]. The different connections between systems are shown in the red dotted line, including the links between the TMS machine, robotic computer and CED system, and the EMG adaptor box and the CED system. (B) The motor reaching task has a number of continuous components; B1. The reach begins at the central starting point; B2. An orange visual cue for a reach appears in the 135° NW target; B3. The participant reaches towards the NW target, and then remains in this target until the robotic arm automatically places their arm in the central starting position for the next reaching trial (B4-B5).

These were the oral instructions given to the participants before the start of the experiment:

- 1. I would like you to make linear "straight-line" reaching movements from the central starting position, to the north-west target (135°).
- 2. You will only have to make the reach when the target lights up in an orange colour.
- You have a time frame of 1.0 -1.2 seconds to reach towards the target and will receive feedback for each reach you make from the screen, i.e. if you are too fast (> 1.2 seconds) or too slow (< 1.0 seconds). If you are on time (1.0 1.2 seconds), the feedback will be shown as "Good".
- 4. Once you have reached towards the north-west target, hold the position, until the robotic arm guides you back to the central starting position for the next reach to begin.
- 5. You may have short breaks after each reaching block if you wish to.
- 6. Are these instructions clear and are there any questions that you would like to ask before we begin.

When motor adaptation was introduced into the reaching paradigm, the additional instruction that the participant received was:

 The robot will now administer a clock-wise force-field during each reaching trial. All other aspects of the reaching task will remain the same, as well as the target that you will be reaching towards (135° NW).

The reaching protocol implemented for the experiments included FAM, FF and WO blocks. When motor adaptation was not explored (experiment 1), only FAM and WO reaching was performed. However, when motor adaptation was introduced into the reaching protocol (experiments 2 - 8) a velocity dependent FF (25 N sm⁻¹) was applied during reaching. The aim of the FAM blocks was for the participants to become familiarised with the task, while the WO blocks intended to "wash-out" the effects of the FF. Each experimental chapter in this thesis describes the reaching blocks (FAM, FF and WO), and number of trials in each block for the different experiments that were undertaken.

3.5 Synchronisation

In all experiments, kinematics (from robotics) and physiological signals were synchronised via a BNC cable connected from the robot to the CED system. Before each reach (at the time of visual cue to reach at 0 seconds), the robot sent a TTL pulse to the CED system which triggered the start of a reaching trial in the Signal Software (Version 6, CED LTD, UK) for data acquisition.

3.6 Data acquisition and analysis

The data that was acquired and analysed in this thesis focused on the effects of TMS pulse administration timings (10ms, 100ms, 130ms, 160ms, 190ms, 220ms, 250ms, 280ms and 310ms) on kinematics of reaching and MEPS where expected (i.e. stimulating the left M1). Therefore the main outcome measures reported in this thesis included kinematics and MEPs.

3.6.1 Kinematics

Kinematics are different types of motion measurements that give an insight into the performance functions of limbs in different conditions (An and Chao, 1984). Kinematics of the upper limb emphasises the joint angles and position of the arm in relation to reaching in a specific direction (Soechting and Ross, 1984; Soechting, Lacquaniti and Terzuolo, 1986; Borghese, Bianchi and Lacquaniti, 1996). In this thesis, we used an end-effector robot which does not measure individual joint angles. Kinematic measures of the upper limb were investigated in this thesis to explore whether changes occurred in reaching movements during motor adaptation and also subsequent to TMS delivery.

Kinematic data was acquired by 16-bit encoders located in the robot armature (sampled at 200hz) which recorded the position of reaches made. Various kinematic parameters were measured during each reaching trial, including movement onset, movement offset, movement duration, maximum velocity, maximum force and summed error.

Kinematic parameters were all analysed offline in MatLab 2017b (The MathWorks Inc, Natick MA, USA), and were quantified as:

1. Movement onset (ms): This represents the starting time of movement and was quantified as a speed profile exceeding 0.03 ms^{-1}

2. Movement offset (ms): This represents the end movement time point and was quantified as a speed profile that was lower than 0.03 ms^{-1}

3. Movement duration (ms): This reflected the time-span of the movement (movement offset minus movement onset)

4. Maximum velocity $(m.s^{-1})$: The maximum velocity during each reach

5. Maximum force (N): The maximum amount of force produced in each reaching trial

6. Summed error: The cumulative perpendicular distance between the participant's reaching trajectory and the ideal reaching trajectory (Hunter, Sacco, Nitsche and Turner, 2009; Pizzamiglio *et al.*, 2017a; Pizzamiglio *et al.*, 2017b) (figure 3.7).

Summed error has been used as a measure of error in a number of motor reaching protocols (Hunter *et al.*, 2009; Pizzamiglio *et al.*, 2017a; Pizzamiglio *et al.*, 2017b). In the reaching protocol there was an ideal linear reaching trajectory from the central starting point towards the north-west 135° target (figure 3.4). Summed error captured changes in reaching throughout the movement duration, i.e. from the start of the reach (onset) to the end of the reach (offset) (Pizzamiglio *et al.*, 2017b). This measure was of particular interest in the final chapter of this thesis as the impact of TMS on different cortical regions (left and right M1, left and right PPC, the SMA, and left and right dPMC) during novel reaching was compared.

It should be noted that prior to statistical analysis, online data cleaning procedures took place automatically during reaching (via the robotic encoders). This was based on the kinematic criteria for movement onset and movement offset. For example, for single reaching trials, if movement onset values fell below 0.03 ms⁻¹ and movement offset values exceeded 0.03 ms⁻¹ this was noted as a missed trial. Additionally, if participants failed to reach the target this was also calculated as a missed trial. Following automated data cleaning, single trial-by-trial kinematic measures were collected and averaged for each participant in each TMS reaching block. This data was used in statistical analysis (see section 3.7) to explore whether: 1) TMS time pulses affected kinematic measures, and 2) whether the effects of TMS were condition specific.



Figure 3.7: Calculation for summed error

The reach produced by one participant is shown (solid red line), as is the ideal linear reaching trajectory (solid blue line). The deviation between the two (i.e. ideal reach [blue] vs. participant's reach [red]) was calculated by summing the distance at each time point from movement onset to movement offset (Hunter et al., 2009; Pizzamiglio et al., 2017a; Pizzamiglio et al., 2017b).

3.6.2 MEPs

MEPs were obtained as a measure of corticospinal excitability (CSE) (Summers, Chen, and Kimberley, 2017) in experiment 1 and 2 (chapter 3 and 4) where TMS was delivered to the left M1. In these experiments, the MEPs evoked in muscles (BB, TB, ECR and FCR) during each reaching trial were recorded with EMG electrodes. It should be noted that quality control took place before the reaching protocol began to avoid the risk of artefacts during data collection. For example, prior to electrode placement, the muscles of interest were cleaned with alcohol wipes. This procedure has commonly been employed in TMS paradigms to reduce impendences that may arise during EMG/MEP recordings (Jubeau *et al.*, 2014; Cantone et al., 2019).

Once MEPs were collected They were then quantified by calculating the peak-to-peak amplitude as this has been reported as the most common measure of CSE (Talelli, Greenwood and Rothwell 2006; Taube *et al.*, 2006; Summers, Chen, Kimberley, 2017).

A custom built sampling configuration toolbox in Signal Version 6 and CED data acquisition software (Cambridge Electronic Design LTD, Cambridge, UK) was used to deliver TMS randomly at nine different time points (10ms, 100ms, 130ms, 160ms, 190ms, 220ms, 250ms, 280ms, and 310ms) during the preparation phase of reaching.

In the experiments were MEPs were elicited, a number of steps were taken to quantify the peak-to-peak amplitude. Firstly MEPs were assessed to ensure that they fell within the +15ms to +55ms time frame. The time window chosen for peak-to-peak analysis was +15ms to +55ms after the TMS pulse as this was when MEPs typically occurred (see figure 3.8). This time window has previously been used when measuring BB MEP amplitudes in motor reaching protocols (Harris-Love *et al.*, 2011). If MEPs did not fall within this range, they were discarded from the analysis. Therefore, there was quality control against experimenter bias because the same protocol was used to discard data for participants. Following data cleaning procedures, an average waveform was firstly created for each time point (i.e. >5 separate MEPs from each time point were averaged to form one MEP waveform). The peak-to-peak MEP amplitude was then calculated from the resulting average waveform. All MEP analysis was performed in Signal Software version 6 (CED LTD, Cambridge, UK). The steps taken are visually presented in figure 3.8.

Individual subject differences in cortical physiology and structure can cause MEP amplitudes to vary (Kiers et al., 1993; Burke et al., 1995; Thickbroom, Byrnes and

Mastaglia, 1999; Darling, Wolf and Butler, 2013; Okamoto *et al.*, 2015). Therefore, MEPs were normalised to adjust for the possible variability in peak-to-peak responses across participants. The MEPs of each TMS pulse time were normalised to the 10ms TMS pulse time. This was performed by dividing each averaged MEP at time x, by the 10ms MEP average. The 10ms pulse time was chosen as a baseline value for comparison, because it represents an active internal control for illustrating changes in MEP responses (from a fully relaxed state with arms resting on table during hotspot evaluation to a quiet resting state with the arm in the robot trough) and occurs at a time before any preparatory brain activity has been recorded (Klein-Flügge *et al.*, 2013). MEPs ratios were represented as percentages rather than raw values.

In experiments where cortical regions did not have a measured output (left and right PPC, left and right dPMC and the SMA) TMS intensity was always set individually for each participant through functional measures - i.e. left M1 TMS (hotspot for the rightsided BB muscle – at 110% RMT). Considering that all of the participants were right handed and performed a right hand reaching task, the left M1 rather than the right M1 was chosen for BB "hotspotting" because of the neural correlates of handedness (i.e. the contralateral (left) hemisphere controls the right hand) (Gut et al., 2007). Left M1 TMS as a functional measure ensured that stimulation intensity was individually set for each participant to avoid possible under- or over- stimulation of the targeted cortical regions. Similar functional measures have been used in TMS experiments (Della-Maggiore et al., 2004; Bolognini 2010). and Ro,



Figure 3.8: Protocol for MEP analysis

The MEP evoked in the BB for each TMS pulse for example delivered at 250ms after the visual cue to reach for one participant.

An average MEP waveform based on data from the five individual MEPs at 250ms is shown in the last window (black outline). In the red outlined box, the horizontal cursors display the peak-to-peak amplitude of the MEP trace, and the vertical cursors show the time that was chosen to quantify the peak-to-peak amplitude (0.265ms and 0.305ms - i.e. +15 to +55 ms after the TMS pulse).

3.7 Statistical analysis

Statistical analysis was performed in the Statistical Package for the Social Sciences (SPSS) 23 (IBM). Block by block statistics were gathered for all measures (kinematics and MEPs). Generally, the paradigm of all experiments carried out was designed to compare the results of whole reach kinematics (experiments 1 - 8) and MEPs (in experiments were MEPs were collected; experiments 1 and 2) at different TMS pulse times (time: 10ms, 100ms, 130ms, 160ms, 190ms, 220ms, 250ms, 280ms and 310ms). In each experimental chapter, a within-subjects analysis design was undertaken as all participants were exposed to the same conditions in reaching (i.e. same TMS pulse times, and same reaching conditions). Therefore, the analysis for the kinematic data, and MEP data was performed using well established tests and this is further described below.

3.7.1 Kinematic data (experiments 1-8):

Kinematic data were collected in all of the experiments carried out in this thesis. A number of steps were taken for the statistical testing of kinematic data. For example, in the first step of statistical testing, a two-way repeated measures analysis of variance (ANOVA) (repeated measures; RMANOVA) was performed using two factors:

1. Factor 1: TMS pulse time (10ms, 100ms, 130ms, 160ms, 190ms, 220ms, 250ms, 280ms and 310ms)

2. Factor 2: Condition (FAM, FF and WO)

Kinematic measures of movement onset, movement offset, movement duration, maximum velocity, maximum force and summed error were tested for statistical significance, with TIME and CONDITION as the main within subjects factors. The significance level for RMANOVA testing was set at p < 0.05. Interaction effects were also noted in the outputs (TIME*CONDITION). For RMANOVA testing, sphericity assumptions were tested for with Maulchy's Test of Sphericity. If sphericity was violated (i.e. p < 0.05), degrees of freedom were adjusted for using the Greenhouse Geisser correction, as this has been reported to reduce the risk of type 1 error rates by producing a more effective *F*-value (Winer, Brown and Michels 1991 cited in, Smith, 2017). Once sphericity assumptions were tested for and subsequent adjustments were made to the degrees of freedom, where statistical significance was found in the

RMANOVA (i.e. p < 0.05), post-hoc testing performed in order to confirm where the specific differences in results occurred. Post-hoc testing was conducted using paired Student's t-test and corrections for multiple comparisons for post-hoc testing were performed with the Bonferroni method, whereby the p value was set to 0.05/number of comparisons performed. For example, where TIME was found to be a significant factor in kinematic variables following RMANOVA testing, the 10ms pulse was taken as a baseline value for comparison in post-hoc statistical testing. Therefore, the significance value for was set as p < 0.006, as:

$$\frac{0.05}{8} = 0.006$$

For the factor of CONDITION where the RMANOVA revealed significance, 3 paired comparisons were made between the different conditions (i.e. FAM vs. FF, FAM vs. WO and FF vs. WO), with the p value set as < 0.016 as:

$$\frac{0.05}{3} = 0.016$$

The Bonferroni method used helps control for the family-wise error rate by avoiding false positives occurring in the results (Armstrong, 2014) and it has been implemented in previous motor adaptation protocols (Pizzamiglio *et al.*, 2017a; Pizzamiglio *et al.*, 2017b).

3.7.2 MEP data (experiment 1 and 2)

There were three factors to consider when exploring MEP data; TIME, MUSCLE and CONDITION. Therefore, each TMS pulse time, and individual muscle response was tested for statistical significance with an RMANOVA with TIME (10ms, 100ms, 130ms, 160ms, 190ms, 220ms, 250ms, 280ms, and 310ms), MUSCLE (BB, TB, ECR and FCR) and CONDITION (FAM, FF and WO) as the within-subject factors. Statistical outputs included main effects of TIME and CONDITION for muscles as well as interaction effects.

As with kinematic data, the RMANOVA significance value for MEP data was set as p < 0.05 and sphericity assumptions were tested for using Maulchy's Test of Sphericity. Greenhouse Geisser correction was used if sphericity assumptions were violated to adjust for the degrees of freedom. Following this, where main effects revealed significance, post-hoc testing performed (paired Student's t-test, with Bonferroni methods). If RMANOVA testing for the factor of TIME was found have a significant

effect on MEP responses in muscles, the p-value for post-hoc testing was set as p < 0.006. In post-hoc testing for the factor of CONDITION, significance was set as p < 0.016, and for the factor of MUSCLE significance was set as p < 0.0125.

All data in the results section for kinematics are presented as mean values \pm the standard error mean (SEM).

3.7.3 Additional statistical testing:

TMS can have a disruptive effect on subsequent motor reaching, depending on the region that is stimulated. For example, stimulating the left PPC was found to result in greater disruption in novel learning compared to FAM reaching (Della-Maggiore et al., 2004). However, the effect of stimulating other regions have mainly been studied with repetitive TMS protocols and whilst observing non-human primates. Therefore, in the penultimate chapter of this thesis (chapter 12), there was an aim to explore the impact of TMS delivered to different cortical regions during 1) motor control and 2) motor adaptation. In order to investigate this, performance during TMS FAM, FF and WO reaching were compared between all of the cortical regions with a three-factorial mixed ANOVA. Therefore, analysis consisted of a between-subjects factor (cortical region stimulated - REGION) and within-subject factors (TIME and CONDITION). For each kinematic variable, statistical outputs included main effects for region, condition and time. Various interaction effects were also produced from the ANOVA including, **REGION*TIME**, **REGION*CONDITION**, CONDITION*TIME and TIME*REGION*CONDITION. Individual ANOVA tests were then carried out for the significant mixed ANOVA findings (p < 0.05). Where significance was revealed in the individual ANOVA tests (p < 0.05), post-hoc testing with Bonferroni methods to correct for multiple comparisons were performed (paired Student's t-test for CONDITION [FAM, FF and WO comparisons, p-value set at 0.016] and TIME [vs. T10, p value set at 0.006] comparisons, and Independent t-test for REGION comparisons [p value set at 0.002]).

Chapter 4

4 Experiment 1

The neural and behavioural mechanisms mediating right arm reaching probed with TMS delivered to the contralateral (left) and ipsilateral (right) primary motor cortex

4.1 Introduction

Motor behaviours fall into one of two categories; discrete or continuous. Reaching with the upper limb has been classed within the discrete category, because it is a motion that has a defined beginning and end point (Muratori *et al.*, 2013). The concept of 'fine' and 'gross' motor skills should also be considered when defining reaching. While fine motor skills refer to actions that use small muscles (i.e. hands), gross motor skills are actions that use larger muscles (i.e. upper arm and the trunk of the body; Muratori *et al.*, 2013). In this experiment, reaching involved both gross and fine muscles. This is because the hand was used to grasp the joystick on the robotic arm, and the upper limb muscles were used to carry out the reaching motion. The role of the corticospinal tract (CST) is vital during reaching. This has been shown in both human studies (such as stroke patients) and animal models, whereby a damaged CST resulted in impaired motor function and poor reaching accuracy (Martin and Ghez 1991; Maraka *et al.*, 2014).

The act of reaching begins with preparatory neural activity within pre-motor and motor cortices before actual movement of the arm (Jones, 2012). These regions help regulate timing and outputs of motor behaviour (Halsband *et al.*, 1993; Overduin, Richardson, and Bizzi, 2009; Chang *et al.*, 2015; Panouilleres *et al.*, 2015). Reaching also involves complex information flows between different brain regions. For example, the dorsolateral prefrontal cortex (DLPFC) transmits organisation and planning information to the pre-motor cortex (PMC) to transfer to the primary motor cortex (M1). The PMC also receives information from the posterior parietal cortices (PPC) for preparation and spatial navigation of movements (Kantak et al., 2012). Section 1.5 provides further specific details with regards to how motor output occurs.

Although unilateral movements rely on recruiting contralateral brain regions for motor output, lateralisation exists between the contralateral and ipsilateral M1, therefore the two hemispheres may not exhibit the same functions (Hayashi *et al.*, 2008; Barber *et al.*,

2011). For example, TMS and functional magnetic resonance imaging (fMRI) during thumb abduction and finger dexterity tasks illustrated that the contralateral (left) M1 in right handed participants showed a greater recruitment of activity in both right and left hand motor tasks compared to the ipsilateral (right) M1 which was only recruited during left hand motor tasks (Kim *et al.*, 1993; Ghacibeh *et al.*, 2007; Muellbacher *et al.*, 2000). Therefore, exploring the role of both the contralateral and ipsilateral M1 can be important in investigating how these two regions differ with regards to their contribution to reaching behaviours.

TMS protocols have used surface electromyography (EMG) electrodes on different upper-limb arm muscles to capture physiological responses (e.g. motor evoked potentials [MEPs]) during the preparation and execution phases of reaching (Groppa *et al.*, 2012; Hunter, Sacco and Turner, 2011; Orban de Xivry *et al.*, 2013). Single-pulse TMS (SP-TMS) delivered to the M1 in reaching paradigms has shown dynamic changes in MEP amplitude. For example, during the preparation phase of right arm reaching towards the body (270° target on a visually presented dart-board), Hunter, Sacco and Turner (2011) found that the biceps brachii (BB) exhibited larger MEP responses than the triceps brachii (TB). However reaching away from the body (135° target) resulted in no differences in MEPs measured in the two muscles. Hunter, Sacco and Turner (2011) also demonstrated time-specific differences with regards to physiological responses between the BB and TB. For example, when TMS was delivered closer to movement onset (during 135° reaching) MEP amplitudes were significantly increased at 190ms and 220ms in the BB as opposed to the TB.

These physiological changes are important to consider because they show that there are complex mechanisms that underlie the time- and direction-tuning of an MEP. However, there is little literature on the time-dependent changes in MEPs comparing the contralateral and ipsilateral M1 and capturing the whole preparation phase of reaching. This new evidence might be important in revealing whether hemispheric-specific cortical excitability changes enable effective reaching.

As well as physiological changes being illustrated in both reaching preparation and reaching execution tasks, researchers have also explored how TMS affects behaviour. Behavioural changes as a consequence of TMS include errors made during reaching (e.g. trajectory errors) and altered maximum velocity and reaction time. With regards to errors made during reaching, Hunter, Sacco and Turner (2011) found that TMS to the
left M1 during reaching with the right hand towards different targets (135° and 270°) did not lead to any significant changes in trajectory errors made. However, studies have shown that TMS application to the contralateral M1 during movement preparation for unperturbed reaching delayed reaction time (Day, Rothwell and Marsden, 1983). Meyer and Voss (2000) also found similar results and reported a delay of up to 40ms during contralateral M1 stimulation during right-limb finger movement. Whether reaching errors and delays are affected by TMS delivered to the ipsilateral M1 during upper-limb robotic reaching are yet to be confirmed. Identifying differences between the two hemispheres is important in revealing whether there are specific reaching processes (determined by kinematics) that the contralateral M1 contributes to during right arm reaching, that the ipsilateral M1 does not, and vice versa. For example, the left hemisphere has been reported to be more specialised in motor sequence learning compared to the right hemisphere (Mutha, Haaland and Sainburg, 2012). This notion has support from studies on stroke patients with left-hemispheric lesions who performed poorly in sequential hand movement tasks (involving the ipsi-lesional arm and contralesional arm) compared to stroke patients with right-hemispheric lesions (Kimura, 1977). Motor planning, action and attention in finger and hand selection are also specific to functions of the left hemisphere (Schluter et al., 2001; Mutha, Haaland and Sainburg, 2012; Oliveira et al., 2010) and therefore the role of the contralateral hemisphere has been noted to have a greater prominence in motor tasks than the ipsilateral hemisphere.

This experiment set out to investigate whether TMS delivered at different time points, to either the contralateral or ipsilateral M1 during the preparation of a reach has an effect on distinct reaching kinematic parameters (movement onset, movement offset, maximum velocity, summed error and movement duration), and whether changes are common to stimulation of both hemispheres or selective to one hemisphere only. This is important in showing how reaching behaviours are modulated between hemispheres.

4.2 Methodology

Table 3.2 outlines participant's specific details for this experiment (e.g. N, age and gender). The ways in which the left and right M1 was located for this study is also described in section 3.3.2 and demonstrated in figure 3.3. The experimental paradigm

for this experiment is shown in figure 3.4. Section 3.41 and 3.4.2 outlines specific details regarding the instructions given to the participants in this experiment.

4.3 Data acquisition

4.3.1 MEPS recorded with EMG:

It should be noted that in this experiment MEPs were only recorded when TMS was delivered to the contralateral M1 to explore corticospinal excitability (CSE) responses during right arm reaching. The muscles that were explored and the filtering protocols used for signal acquisition are identified in section 3.3.1. Furthermore, section 3.6.2 describes the steps taken for MEP analysis (such as MEP normalisation) and figure 3.8 demonstrates the protocol for MEP quantification (i.e. calculating the peak to peak amplitude).

4.3.2 Kinematics recorded with the robot:

The kinematic variables that were recorded and the ways in which they were quantified and analysed (Matlab 2017b - The MathWorks Inc, Natick MA, USA) are explained in section 3.6.1.

4.4 Statistical analysis

Generally, the paradigm of this experiment was designed to compare the results of MEPs and kinematics during different TMS pulse times (TIME factor) and muscles (MUSCLE Factor for MEP data) and to compare contra-lateral and ipsi-lateral hemisphere stimulation (HEMISPHERE factor for kinematic data only).

4.4.1 MEPs:

Peak-to-peak MEP amplitudes for each TMS pulse time (TIME factor) and individual muscles (MUSCLE factor; BB, TB, ECR and FCR) were tested for significance with a two-way RMANOVA with TIME and MUSCLE as within-subjects factors in SPSS 23 (IBM). The ways in which sphericity assumptions were tested for are outlined in section 3.7.2. Post-hoc testing was performed (where significance was found (RMANOVA, p < 0.05) with a paired Student's t-test for TIME (p < 0.006) and MUSCLE (p < 0.0125; both corrected for multiple comparisons using Bonferroni methods).

4.4.2 Kinematics:

Kinematic responses for each TMS pulse time was calculated trial by trial and averaged across participants for each hemisphere. Overall, the aim of the analysis was to identify significant differences for TIME and HEMISPHERE factors between the contralateral and ipsilateral M1 for kinematic variables. Section 3.6.1 describes the ways in which kinematic parameters were explored and analysed in SPSS 23 (IBM) (including how sphericity assumptions were tested and how post-hoc testing was carried out).

4.5 Results

4.5.1 MEP results: Contralateral M1 stimulation only:

The results of average peak-to-peak MEP amplitudes during reaching (values as ratio of TMS 10ms) are shown in table 4.1. TIME factor had a significant main effect on peak-to-peak MEP amplitude (RMANOVA p < 0.05). However, there was no significant effect of MUSCLE factor and no MUSCLE*TIME interaction was found (RMANOVA p > 0.05) (table 4.1).

4.5.2 Kinematics: contralateral and ipsilateral stimulation:

The results of the two-way RMANOVA revealed a significant difference for TIME for movement onset, offset and maximum velocity (all F > 4.558, all p < 0.05) (table 4.2), but not for maximum force, movement duration and summed error (all F < 0.593, all p > 0.05). Kinematic conditions were not significantly different between the contralateral and ipsilateral M1 (HEMISPHERE factor; all F < 0.022, all p > 0.05) (therefore post hoc testing was not performed for this factor). A significant interaction was only found for movement onset (F = 3.61, p < 0.05) (see table 4.2).

Post hoc testing for time with regards to movement onset revealed a significant increase in all pulse times during contralateral and ipsilateral M1 stimulation when compared to 10ms (all p < 0.006). This was also the case for movement offset (all p < 0.006) for contralateral M1 stimulation. During ipsilateral M1 stimulation however, TMS delivered at 100ms was not significantly different from 10ms (p > 0.006). Maximum velocity was significantly increased when TMS was delivered at 280ms and 310ms when compared to 10ms during contralateral reaching (p < 0.006), whereas maximum velocity responses did not differ between 10ms and all other time points of TMS delivery during ipsilateral stimulation (p > 0.006). Table 4.3 and figure 4.1 further demonstrates post-hoc testing results.

| | TMS Pulse Time | | | | | | | | | |
|-------------------------------|----------------|-----------------|-------------|-------------|-----------------|-------------|-------------|-------------|-------------|--|
| | 10ms | 100ms | 130ms | 160ms | 190ms | 220ms | 250ms | 280ms | 310ms | |
| | Mean [SE] | Mean [SE] | Mean [SE] | Mean [SE] | Mean [SE] | Mean [SE] | Mean [SE] | Mean [SE] | Mean [SE] | |
| Biceps brachii (BB) | 1.00 [0.00] | 1.05 [0.13] | 1.32 [0.17] | 1.20 [0.08] | 1.29 [0.13] | 1.31 [0.10] | 1.42 [0.17] | 1.31 [0.17] | 1.40 [0.18] | |
| Triceps brachii (TB) | 1.00 [0.00] | 1.17 [0.17] | 1.32 [0.25] | 1.20 [0.12] | 1.22 [0.11] | 1.33 [0.14] | 1.30 [0.11] | 1.25 [0.09] | 1.30 [0.11] | |
| Extensor carpi radialis (ECR) | 1.00 [0.00] | 0.89 [0.06] | 1.13 [0.06] | 1.06 [0.06] | 1.01 [0.05] | 1.03 [0.07] | 1.02 [0.09] | 0.96 [0.05] | 0.99 [0.05] | |
| Flexor carpi radialis (FCR) | 1.00 [0.00] | 0.91[0.08] | 0.99 [0.07] | 0.93 [0.06] | 1.02 [0.07] | 1.05 [0.06] | 1.07 [0.08] | 1.02 [0.07] | 1.12[0.09] | |
| | | | А | NOVA TESTIN | G | | | | | |
| | Т | time (10-310ms) | 1 | Musc | le (BB, TB, ECR | , FCR) | | Time*Muscle | | |
| Statistical output | Df (Errors) | F | Sig | Df (Errors) | F | Sig | Df (Errors) | F | Sig | |
| Results: | 3.7 (44.2) | 3.631 | 0.014 | 1.6 (18.1) | 2.737 | 0.103 | 24 *288) | 0.899 | 0.604 | |

Table 4.1: Average peak-to-peak MEP amplitudes during FAM reaching at different time points, and results of RMANOVA testing:

| | | Two-way ANOVATIME factor: 10ms - 310msHEMISPHERE factor (Contralateral and Ipsilateral M1)Interaction effects $(Error)$ FSig. $df(Error)$ FSig. $df(Error)$ FSig.9[34.5]76.65< 0.0011 [12]0.5470.4743.3 [39.4]3.6080.0198[96]23.27< 0.0011 [12]0.4610.5108 [96]1.9280.064 | | | | | | | | |
|---------------------------------------|-------------------|---|---------|-------------------|---------------|-------|---------------------|-------|-------|--|
| | | TIME factor: | | HEMISPH | ERE factor | | Interaction effects | | | |
| | | 10ms - 310ms | | (Contralateral an | d Ipsilateral | M1) | TIME * HEMISPHERE | | | |
| | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | |
| Movement Onset (ms) | 2.9[34.5] | 76.65 | < 0.001 | 1 [12] | 0.547 | 0.474 | 3.3 [39.4] | 3.608 | 0.019 | |
| Movement Offset (ms) | 8[96] | 23.27 | < 0.001 | 1 [12] | 0.461 | 0.510 | 8 [96] | 1.928 | 0.064 | |
| Maximum Velocity (m.s ⁻¹) | 8[96] | 4.558 | < 0.001 | 1 [12] | 0.022 | 0.883 | 8 [96] | 1.554 | 0.149 | |
| Duration (ms) | 8[96] | 1.935 | 0.063 | 1 [12] | 2.231 | 0.154 | 8 [96] | 1.346 | 0.230 | |
| Summed Error (distance: cm) | 8[96] | 1.457 | 0.183 | 1 [12] | 0.976 | 0.343 | 8 [96] | 0.734 | 0.661 | |
| Maximum Force (N) | 8[96] | 0.593 | 0.781 | 1 [12] | 0.208 | 0.656 | 8 [96] | 0.874 | 0.541 | |

Table 4.2. Two-way RMANOVA results for kinematic variables during contralateral and ipsilateral M1 stimulation

| | TIME | TIME | | | | | | | | | |
|---------------------------------------|------------------|------------------|-------------|-------------|-------------|-------------|--------------|--------------|--------------|--|--|
| | 10ms | 100ms | 130ms | 160ms | 190ms | 220ms | 250ms | 280ms | 310ms | | |
| | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | | |
| | Contralateral M1 | Contralateral M1 | | | | | | | | | |
| Movement Onset (ms) | 298 [12] | 351 [10] * | 386 [12]* | 395 [11]* | 403 [9]* | 428 [12]* | 440 [12]* | 433 [12]* | 494 [18]* | | |
| Movement Offset (ms) | 1167 [18] | 1242 [16]* | 1244 [24]* | 1265 [22]* | 1258 [21]* | 1282 [21]* | 1282 [22]* | 1303 [23]* | 1323 [26]* | | |
| Maximum Velocity (m.s ⁻¹) | 0.27 [0.01] | 0.27 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.28 [0.01]* | 0.29 [0.01]* | 0.29 [0.01]* | | |
| Duration (ms) | 868 [17] | 827 [62] | 856 [23] | 823 [65] | 856 [23] | 854 [19] | 841 [19] | 842 [19] | 843 [21] | | |
| Summed Error (distance: cm) | 2.51 [0.23] | 2.50 [0.24] | 2.31[0.18] | 2.81[0.29] | 2.67 [0.22] | 2.67 [0.23] | 2.65 [0.20] | 2.96 [0.36] | 2.82 [0.27] | | |
| Force (N) | 4.5 [0.10] | 4.6 [0.20] | 4.6 [0.20] | 4.5 [0.10] | 4.6 [0.10] | 4.5 [0.20] | 4.6 [0.20] | 4.6 [0.20] | 4.6 [0.20] | | |
| | Ipsilateral M1 | | | | | | | | | | |
| Movement Onset (ms) | 325 [13] | 367 [10] * | 401 [12] * | 414 [12] * | 430 [12] * | 423 [9] * | 437 [13] * | 469 [18] * | 466 [16] * | | |
| Movement Offset (ms) | 1195 [20] | 1224 [20] | 1244 [20] * | 1270 [24] * | 1294 [21] * | 1267 [19] * | 1273 [22] * | 1321 [21] * | 1310 [17] * | | |
| Maximum Velocity (m.s ⁻¹) | 0.27 [0.01] | 0.27[0.01] | 0.27 [0.01] | 0.27 [0.01] | 0.27 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.27 [0.01] | 0.28 [0.01] | | |
| Duration (ms) | 870 [17] | 854 [13] | 844 [26] | 855 [27] | 864 [21] | 844 [76] | 835 [25] | 854 [21] | 844 [15] | | |
| Summed Error (distance: cm) | 2.64 [0.24] | 2.55 [0.18] | 2.34 [0.15] | 2.46 [0.19] | 2.51 [0.17] | 2.60 [0.22] | 2.40 [0.19] | 2.46 [0.16] | 2.39 [0.12] | | |
| Force (N) | 4.4 [0.21] | 4.5 [0.20] | 4.5 [0.20] | 4.5 [0.23] | 4.5 [0.25] | 4.5 [0.21] | 4.5 [0.20] | 4.5 [0.19] | 4.5 [0.20] | | |

Table 4.3. Post-hoc testing following significant RMANOVA findings:

Significant findings (Student's t-tests) with Bonferroni correction for multiple comparisons are marked with an asterisk (*)

Symbols represent significance following post hoc testing: * = significant difference vs 10ms for TIME factor

Figure 4.1[A-C]: Graphical presentation of the post-hoc testing results (based on table 4.3) for the significant kinematic variables:



4.6 Discussion

4.6.1 MEP responses:

ANOVA results revealed a significant main effect for the TIME factor for MEP amplitude (p < 0.05) demonstrating that as TMS pulses became closer to the onset of movement, MEP amplitudes of all muscles increased, although this was a small effect size as no further significance was illustrated in post-hoc testing (all p > 0.006). Additionally, no significant effect of MUSCLE factor or MUSCLE*TIME interaction was found. These results are similar to findings illustrated by Hunter, Sacco and Turner (2011) as they reported that TMS delivered to the contralateral M1 when reaching towards the body (270° target) induced MEP changes between the BB and the TB, with the BB exhibiting larger MEP responses, however this was not case when reaching away from the body (135° target as in this study). Direction MEP changes were also noted by Kantak et al., (2013) in upper-limb arm robotic reaching demonstrating an increase in TB MEP amplitude when reaching away from the body. It could be argued that the reason as to why MEP responses were not significantly altered in this experiment was due to the nature of the reaching protocol (i.e. 135° unperturbed reaching). Implementing different motor reaching protocols with different directions can demonstrate the complex mechanisms that underlie the time- and direction-tuning of an MEP. This thesis does not address directional tuning however. Furthermore, the impact of TMS on corticospinal projections during unperturbed reaching relies on the extent of training that has been undertaken (Kantak et al., 2013). For example, TB MEP responses during elbow extension reaching significantly increased following repeated reaching trials (160 reaching trials x 3 blocks of reaching) (Kantak et al., 2013). With regards to MEP findings in experiment 1, it could be suggested that MEP amplitudes between muscles may have significantly differed if subjects were exposed to a greater number of reaching trials. This is because practice-related activity can induce neuroplasticity, which can in turn effect muscle representations in the M1 (even in unperturbed reaching) (Kantak et al., 2013).

4.6.2 Kinematics:

Overall, experiment 1 illustrated the behavioural effects of TMS on reaching kinematics. For example, it was found that both contralateral and ipsilateral M1 TMS (right arm reaching) prolonged the movement onset time (also commonly known as

reaction time; this addresses experiment 1, hypothesis 2) and movement offset. More specifically for both hemispheres, the later the TMS pulse was delivered during movement onset, the greater the delay in movement onset, which was found in robust multiple comparisons testing (p < 0.006 - table 4.3).

The shift in movement onset reported in this experiment is in line with results from previous studies (Day, Rothwell and Marsden, 1983). Meyer and Voss (2000) noted that TMS can cause delays of up to 40ms in responses. However, it should be taken into account that previous experiments (e.g. Day, Rothwell and Marsden, 1983) reporting similar delays in movement onset have not explored upper limb reaching, but rather hand/finger pointing. Therefore experiment 1 demonstrated that similar delays can also be found during upper limb reaching (table 4.3).

A mechanistic effect of TMS on the M1 is that it excites pyramidal neurons transsynaptically (via depolarised interneurons) (Rothwell *et al.*, 1987; Terao and Ugawa, 2002; Hallett, 2007; Farzan *et al.*, 2016; Goss, Hoffman and Clark, 2012). The delay in movement onset found in this experiment can be explained mechanistically by the inhibitory effects of TMS. More specifically, TMS inhibits the 'release channel' of neurons that are responsible for movement preparation, making them unresponsive for a brief period, thus causing delays in signals that facilitate movement onset. The movement command can still be released for motor output because 'pre-movement facilitation' in the M1 was initiated (i.e. motor planning) (Day, Rothwell and Marsden, 1983; Ziemann *et al.*, 1997). Therefore, TMS can cause GABA_b-like activity effects (a neurotransmitter that influences inhibition processes) (Werhahn et al., 1999; Auriat *et al.*, 2015).

An interesting finding from the RMANOVA was that maximum velocity during reaching was significantly increased during the latter time points of contralateral M1 stimulation, which was not the case for ipsilateral M1 stimulation. The state dependency theory, specifically dispersed excitation to other connected brain regions can explain this finding. This theory suggests that TMS may not only directly impact synaptic activity in the targeted region, but it could also indirectly influence cortical activity by affecting its connected networks (Siebner *et al.*, 2009). Numerous regions make up the motor network including, the dorsolateral prefrontal cortex (DLPFC), the premotor cortices (PMC), the supplementary motor area (SMA), and the posterior parietal cortex (PPC) (Kantak *et al.*, 2012). Studies exploring the functional role of the left M1 have

shown that during reaching tasks it encodes kinematic parameters, including velocity, and the speed of movement via neural communication with connecting regions, such as the SMA (Hore and Flament, 1988; Aizawa and Tanji, 1994; Tanji and Mushiake, 1996; Moran and Schwartz, 1999; Tankus et al., 2009; Teka et al., 2017). Based on this theory, and findings from functional motor connectivity imaging studies (i.e. contralateral vs. ipsilateral M1 – see Kim *et al.*, 1993; Guye *et al.*, 2003) it can be argued that TMS caused a significant change in velocity measures only in the contralateral M1, because it had a greater 'knock-on' effect on its connecting nodes, which have been found to be more extensively connected in the left hemisphere as opposed to the right hemisphere (Kim *et al.*, 1993; Guye *et al.*, 2003; Siebner *et al.*, 2009).

Hemispheric asymmetries can also explain why TMS had a greater effect on kinematic variables in one hemisphere compared to another hemisphere (e.g. maximum velocity and movement offset in the contralateral M1 compared to the ipsilateral M1), which has been noted in fMRI and structural voxel based morphometry (VBM) studies. For example, in a resting fMRI study where connectivity patterns were correlated with motor performance, Barber et al., (2011) found a link between enhanced motor performance and increased left-hemispheric M1 connectivity, compared to righthemispheric M1 activity in right handed participants. VBM imaging methods use volumetric measures to explore anatomical differences between structures in the left and right hemisphere (Ashburner and Friston, 2000; Watkins et al., 2001; Büchel et al., 2004). Diffusion tensor imaging (DTI) VBM studies in right-handed subjects revealed that the contralateral M1 had a greater volume, and greater degrees of dendrites and axons than the ipsilateral M1 (Amunts et al., 1996). Such hemispheric asymmetries help explain the concept of lateralisation which was found in this study when considering that movement preparation was effected differently between the ipsilateral and contralateral M1. Although the left M1 has been found to be more prominently active than the right M1 in right handed subjects during arm-reaching and pointing tasks (Kim et al., 1993; Muellbacher et al., 2000; Barber et al., 2011) it should be taken into account that in this study using TMS to disrupt ipsilateral M1 functioning also resulted in delays in reaction time, which is in line with previous research that has found the ipsilateral M1 to undergo task related changes to the moving body side (Van den Berg, Swinnen and Wenderoth, 2011). This indicates that the ipsilateral cortex is also

involved and interacts with movement preparation processes, even during unilateral reaching. Summed error values were not significantly affected with TMS to either the contralateral or ipsilateral M1. This result can be due to unperturbed reaching that was performed, as similar studies have also not demonstrated any changes in this variable during unperturbed reaching (Hunter, Sacco and Turner, 2011).

4.7 Chapter conclusions:

In this study, TMS stimulation was found to delay movement onset and movement offset responses during both contralateral and ipsilateral M1 stimulation which can be due to the inhibitory effects of SP-TMS. Significant changes were not observed in MEP responses between muscles which could have been be due to the nature of the task (e.g. direction of reaching), and the number of reaching trials that participants were exposed to, when considering that continuous repeated practice can influence MEP responses through inducing plasticity. The differences between the two hemispheres with regards to the degree of delay as well as other kinematic parameters (e.g. movement velocity) can be attributed to 1) hemispheric asymmetries noted in VBM studies, 2) lateralisation of motor function between the left and right M1, and 3) differences in functional connectivity which have been studied with resting state fMRI studies. Overall this chapter highlighted reaching processes that the contralateral and ipsilateral M1 contribute to (planning and preparation of reaching) and the kinematic behaviours that may be more distinct for one hemisphere only (e.g. velocity; left M1). In the upcoming chapters, motor adaptation will be studied in order to investigate whether such hemispheric differences are also evident with TMS during a force-field reaching paradigm.

Chapter 5

5 Experiment 2

The neural and behavioural mechanisms mediating right arm reaching in a forcefield studied with single-pulse TMS to the left primary motor cortex (M1)

In the previous chapter, neural mechanisms and kinematic behaviour of unperturbed reaching were explored with contralateral and ipsilateral M1 TMS delivery. The purpose of this experiment was to introduce motor adaptation within the protocol. Therefore, this experiment aimed to investigate whether motor adaptation (i.e. force-field [FF] reaching) was accompanied by changes in corticospinal excitability (CSE) and in subsequent kinematic responses studied with TMS delivered to the left M1.

5.1 Introduction

Motor adaptation places emphasis on the brain's flexibility in meeting the demands of a changing environment compared to motor skill learning (Bastian, 2008). Motor adaptation is vital in the field of rehabilitation because repeated adaptation can lead to individuals learning a new motor skill (Bastian, 2008; Hunter et al., 2009). For example, in the study of Reisman (2005) stroke patients who had an impaired leg, were asked to adapt the affected leg to walk faster than the unaffected leg on a split-belt treadmill. Following motor adaptation trials, the patients gradually learned to walk with both legs at the same speed. It can thus be argued that motor adaptation can facilitate the re-learning of motor skills and aid neurological recovery. The motor adaptation process can be explored in robotic training with the robot in 'resistive mode' (see Section 3.4) (Patton et al., 2006). In this mode, the robot administers a force that opposes movements made by the patient. Forces can be applied in different directions and magnitude and aim to strengthen weakened muscles (Poli et al., 2013; Eiammanussakul and Sangveraphunsiri, 2018). Studies have shown that resistive robotic cycling training in stroke patients resulted in improved walking distance performance and an increase in comfortable speed - as quantified by walking assessments (both practical sessions and mobility scales) (Kamps and Schüle 2005; Eiammanussakul and Sangveraphunsiri, 2018). Therefore, the 'resistive mode' in robotic motor adaptation protocols can be used as an effective tool in assisting the recovery process in clinical populations (i.e. stroke patients).

Changes in corticospinal excitability have previously been investigated with TMS in motor adaptation paradigms. For example, an increase in BB compared to TB peak-topeak MEP amplitude was demonstrated in the preparatory phase of reaching during motor adaptation (Orban de Xivry *et al.*, 2013). More specifically, BB MEP increased when measured more closely to movement onset (Hunter, Sacco and Turner, 2011; Orban de Xivry *et al.*, 2013). However, it should be considered that TMS to the left M1 has not been delivered during the whole preparation phase prior to FF reaching onset. This could be interesting to explore, as it is currently not known whether different kinematic parameters have specific time-related effects with TMS.

Errors during reaching are greater during the initial stages of motor adaptation, however they gradually reduce trial by trial following additional blocks of force-field reaching (Hunter *et al.*, 2009; Hunter, Sacco and Turner 2011; Pizzamiglio *et al.*, 2017a; Pizzamiglio *et al.*, 2017b). Interestingly, studies with non-human primates have found that major changes within the M1 do not occur at the earlier stages of adaptation when errors are amplified, but rather at the latter stages when errors are reduced (Paz *et al.*, 2003). However, this has not been established with TMS protocols in man.

The motor adaptation protocol implemented in this experiment, contains 8 blocks of force-field reaching (i.e. approximately 200 reaches in a force-field) which should allow a fuller degree adaptation and consolidate a robust change in left M1 function. Thus, we tested the hypothesis that the left M1 is functionally altered by motor adaptation. Secondarily, as TMS appears to alter subsequent movement kinematics in normal reaching (see chapter 4), this experiment also addresses whether the impact of TMS disrupts subsequent reaching in a different fashion in motor adaptation. This would further suggest an alteration in left M1 function occurs during adaptation.

5.2 Methodology

Table 3.2 outlines participant demographics for this experiment, as well as the number of participants excluded and the reasons why. The TMS protocol for this experiment including how RMT was identified as well as TMS coil position for left M1 location is described in section 3.3.2. This protocol differs from experiment 1, as motor adaptation was explored. The experimental set up and the full details of the instructions given to the participants is outlined in Section 3.4.1 and 3.4.2. Furthermore, figure 3.5

graphically demonstrates the protocol with regards to the different blocks of reaching (FAM, FF and WO).

5.3 Data acquisition: MEPs and Kinematics

When TMS was delivered to the left M1, MEPs were recorded from four muscles (BB, TB, ECR and FCR) with two disposable surface electrodes. The specific guidelines (SENIAM) that were used for electrode positioning are outlined in Section 3.3.1. The ways in which MEP responses were recorded and quantified for this experiment is described section 3.6.2 and illustrated in figure 3.8. The kinematic variables recorded and their quantification and analysis process (Matlab 2017b - The MathWorks Inc, Natick MA, USA) is explained in section 3.6.1.

5.4 Statistical analysis: MEPs and Kinematics

Section 3.7.2 describes the ways in which statistical testing was performed for this experiment using a three-factorial RMANOVA (RMANOVA) with TIME (T10 to T310), CONDITION (FAM, FF, WO) and MUSCLE (BB, TB, ECR and FCR) as main factors in SPSS 23 (IBM). The statistical analysis that was undertaken for kinematic parameters is described in section 3.7.1. Section 3.7.2 also outlines the procedure for sphericity testing and how post hoc testing was performed (i.e. Student's t-tests and correcting for multiple comparisons using Bonferroni methods).

5.5 Results

5.5.1 MEPs:

The results of the RMANOVA (table 5.1) revealed a significant main effect for TIME (F: 9.86, p < 0.05) and CONDITION (F: 10.67, p < 0.05). However, MEP responses did not significantly differ between MUSCLES (F: 1.00, p > 0.05). A significant interaction was found for TIME*CONDITION (F: 6.00, p < 0.05), but not for TIME*MUSCLE, MUSCLE*CONDITION and TIME*MUSCLE*CONDITION (F < 2.22, p > 0.05).

Post hoc testing for TIME (table 5.1) showed no significant differences in MEP amplitude for FAM reaching between T10 and all other TMS pulse times in the BB, TB, ECR and FCR (all p > 0.006). However, there was a significant increase in MEP amplitude in the BB at 220ms and 250ms compared to T10 in FF reaching (p < 0.006). There were no significant differences in MEP amplitude during FF reaching between T10 and all other TMS pulse times in the TB, ECR and FCR (p > 0.006). No significant differences in MEP amplitude were found during WO reaching between T10 and all

other TMS pulse times in the BB, TB and ECR (all p > 0.006), but there was a significant increase in MEP amplitude in the FCR at 130ms compared to T10 (p < 0.006).

Post hoc testing for CONDITION (table 5.1) for the BB showed a significant increase in MEP amplitude in FF reaching compared to FAM reaching (at 220ms, 250ms and 310ms p < 0.016). MEP amplitude in FF reaching compared to WO reaching was also significantly increased (at 160ms, 220ms, 250ms and 310ms p < 0.016). No significant differences in MEP amplitude were found for WO vs. FAM reaching. TB MEP amplitude was significantly increased in FF reaching compared to FAM reaching at 220ms (p < 0.016). No significant differences were found for FF vs. WO reaching in the TB. However, there was a significant increase in MEP amplitude in the TB during WO reaching compared to FAM reaching (only at 310ms p < 0.016). With regards to the ECR and FCR, there were no significant differences in MEP amplitude during FF vs. FAM reaching, FF vs. WO reaching, and FAM vs. WO reaching (all p > 0.016).

5.5.2 Kinematics:

Patterns of reaching were as follows: during TMS FAM reaching, participant's successfully familiarised themselves with the task. Motor adaptation introduced with FF reaching (no TMS) initially caused deviations in reaching trajectories. However, following blocks of repeated trials, participant's learned to adapt to the FF and optimise their reaching – which is indicative of successful motor adaptation. TMS FF reaching however disrupted reaching trajectories (see ANOVA, table – 5.2). When unperturbed reaching was re-introduced into the protocol (WO blocks), 'overshoot errors' were visible whereby participants reached in the direction of the expected force (clockwise) which resulted in a deviation in the opposite direction (counter-clockwise). The errors however faded quickly and reaching became ideal (indicative of a successful deadaptation).

The results of the two-way repeated measures ANOVA (table 5.2) revealed a significant main effect of TIME for movement onset and offset (all F > 30.71, all p < 0.05), but not for summed error, maximum velocity, duration and maximum force (all F < 0.778 p > 0.05; table 4.2). A significant effect of CONDITION (FAM, FF and WO) was found for movement onset, maximum velocity, movement duration, summed error and maximum force (all F > 9.066, all p < 0.05) but not movement offset (F: 1.862, p > 0.05). With

regards to all kinematic variables explored, no significant interaction effects were found (all F < 1.084, < all p > 0.05).

Post hoc testing for TIME (table 5.3, figure 5.3[A-F]) revealed that during both FAM reaching and FF reaching, movement onset responses were significantly increased at all time points compared to T10 (p < 0.006). During WO reaching, movement onset was significantly increased at 190ms, 220ms, 250ms, 280ms and 310ms compared to T10 (p < 0.006). Movement offset was significantly increased at all time points compared to T10 (p < 0.006). Movement offset was significantly increased at all time points compared to T10 uring FAM reaching and FF reaching (p < 0.006), but not WO reaching (at all time points p > 0.006).

Post hoc testing for CONDITION (table 5.3, figure 5.3[A-F]) for movement onset showed no significant differences between FF and FAM reaching. However, movement onset in FF reaching was significantly increased compared to WO reaching (at 10ms, 190ms, 220ms, 250ms, 280ms and 310ms p < 0.016). No significant differences were revealed for movement onset during WO vs. FAM reaching (p > 0.016). Maximum velocity was significantly increased in FF reaching compared to FAM reaching (at 100ms, 130ms, 190ms, 220ms, 250ms, 280ms and 310ms p < 0.016). Maximum velocity was also significantly increased in FF reaching compared to WO reaching (at 250ms, p < 0.016). No significant difference in maximum velocity was found between WO vs. FAM reaching. Movement duration did not significantly differ between FF vs. FAM reaching. However, duration was significantly increased in FF reaching compared to WO reaching (at 100ms, 130ms, 160ms, 190ms, 220ms and 280ms, p < 0.016). Movement duration in FAM reaching was significantly increased compared to WO reaching (at 160ms and 280ms p < 0.016). Summed error was significantly increased in FF reaching compared to FAM reaching (at all time points, p < 0.016). Summed error was also significantly increased in FF reaching in contrast to WO reaching at all time points except for 100ms and 130ms (p > 0.016). There were no significant differences in summed error for FAM vs. WO reaching (all p > 0.016). Maximum force was significantly increased in FF reaching (p < 0.016) compared to both FAM and WO reaching which were not significantly different (p > 0.016).

All post-hoc testing results are shown in table 5.3 and figure 5.3[A-F].

| Three-facto | orial RMA | NOVA | | | | | | Post hoc t | esting | | | |
|-------------------|-----------|--------|---------|-----------|------------|--------------|-------------|------------|----------------|--------------|------------|----------------|
| , | TIME | | | TIME | | | | | | | | |
| <i>df</i> (Error) | F | Sig. | | 10ms | 100ms | 130ms | 160ms | 190ms | 220ms | 250ms | 280ms | 310ms |
| | | | | Mean | Mean | Mean | Mean | Mean | Mean | Mean | Mean | Mean |
| | | | | [SEM] | [SEM] | [SEM] | [SEM] | [SEM] | [SEM] | [SEM] | [SEM] | [SEM] |
| 3.0(38.2) | 9.86 | 0.001 | CONDITI | ON | | | BI | 3 | | | | |
| | | | FAM | 1.0 [0.0] | 1.03 [.05] | 1.09 [.07] | 1.07 [.07] | 1.02 [.05] | 1.03 [.07] | 1.09 [.08] | 1.15 [.11] | 1.09 [.12] |
| COl | NDITION | | FF | 1.0 [0.0] | 1.19 [.11] | 1.37 [.14] | 1.39 [.13]+ | 1.55 [.19] | 1.98 [.26] * 🔶 | 1.65 [.22] + | 1.89 [.36] | 2.49 [.42] * + |
| <i>df</i> (Error) | F | Sig. | WO | 1.0 [0.0] | 1.05 [.06] | 1.07 [.04] | 1.07 [.05] | 1.13 [.06] | 1.13 [.11] | 1.07 [.05] | 1.12 [.09] | 1.18 [.12] |
| 1.4(17.4) | 10.67 | 0.002 | | ТВ | | | | | | | | |
| M | USCLE | | FAM | 1.0 [0.0] | 1.02 [.05] | 1.00 [.06] | 1.01 [.03] | 1.05 [.04] | 1.06 [.06] | 1.07 [.08] | 1.07 [.08] | 1.09 [.08] |
| <i>df</i> (Error) | F | Sig. | FF | 1.0 [0.0] | 1.24 [.08] | 1.04 [.06] | 1.22 [.08] | 1.17 [.09] | 1.30 [.10] | 1.27 [.11] | 1.29 [.18] | 1.29 [.13] |
| 3(39) | 1.00 | 0.400 | WO | 1.0 [0.0] | 1.25 [.26] | 1.21 [.13] | 1.16 [.13] | 1.24 [.25] | 1.20 [.18] | 1.28 [.19] | 1.21 [.17] | 1.28 [.11] |
| TIME* | CONDITIC | DN | | | | | | ECR | | | | |
| <i>df</i> (Error) | F | Sig. | FAM | 1.0 [0.0] | 1.11 [.06] | 1.04 [.07] | 1.14 [.05] | 1.09 [.05] | 1.13 [.07] | 1.18 [.11] | 1.06 [.08] | 1.14 [.05] |
| 5.5 (70.9) | 6.00 | 0.001 | FF | 1.0 [0.0] | 1.04 [.05] | 1.08 [.04] | 1.13 [.07] | 1.28[.17] | 1.20 [.08] | 1.23 [.09] | 1.16 [.07] | 1.50 [.22] |
| TIME | *MUSCLE | Ξ | WO | 1.0 [0.0] | 1.11 [.06] | 1.06 [.07] | 1.09 [.06] | 1.08 [.05] | 1.06 [.05] | 1.13 [.05] | 1.18 [.08] | 1.14 [.11] |
| <i>df</i> (Error) | F | Sig. | | | | | | FCR | | | | |
| 9.0(116.9) | 1.20 | 0.305 | FAM | 1.0 [0.0] | 0.99 [.06] | 0.99 [.07] | 1.03 [.05] | 0.97 [.05] | 1.00 [.07] | 1.13 [.11] | 1.06 [.08] | 1.97 [.05] |
| MUSCLE | *CONDIT | ION | FF | 1.0 [0.0] | 1.09 [.05] | 1.24 [.10] | 1.24 [.09] | 1.41 [.24] | 1.27 [.14] | 1.57 [.38] | 1.70 [.37] | 1.98 [.52] |
| <i>df</i> (Error) | F | Sig. | WO | 1.0 [0.0] | 1.14 [.08] | 1.15 [.04] * | 1.13 [.06] | 1.08 [.05] | 1.06 [.04] | 1.08 [.04] | 1.06 [.04] | 1.07 [.05] |
| 2.9(36.7) | 2.22 | 0.106 | | | | | | | | | | |
| TIME*MUSC | CLE*CON | DITION | | | | | | | | | | |
| <i>df</i> (Error) | F | Sig. | | | | | | | | | | |
| 7.7(99.4) | 1.52 | 0.165 | | | | | | | | | | |

Table 5.1. Results of three-factorial RMANOVA for peak-to-peak MEP responses, followed by post hoc testing:

| | TIME: | | | (| CONDITION | | TIME * CONDITION | | | |
|---------------------------------------|-------------------|-------|---------|-------------------|-----------|---------|-------------------|-------|-------|--|
| | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | |
| Movement Onset (ms) | 3.0 [38.4] | 30.71 | < 0.001 | 2 [26] | 10.25 | 0.002 | 16[208] | 0.496 | 0.948 | |
| Movement Offset (ms) | 8[2] | 22.50 | < 0.001 | 2[26] | 1.862 | 0.175 | 16[208] | 1.084 | 0.372 | |
| Maximum Velocity (m.s ⁻¹) | 3.4[44.1] | 0.932 | 0.442 | 1.4 [18.2] | 9.066 | 0.004 | 16[208] | 0.837 | 0.643 | |
| Duration (ms) | 8 [104] | 0.778 | 0.569 | 2[26] | 13.14 | 0.001 | 16[208] | 0.973 | 0.487 | |
| Summed Error (distance: cm) | 3.9 [49.5] | 2.295 | 0.075 | 1.4 [18.1] | 31.37 | < 0.001 | 16[208] | 1.358 | 0.165 | |
| Maximum Force (N) | 8 [104] | 1.443 | 0.187 | 1.3 [15.7] | 313.06 | < 0.001 | 16[208] | 0.871 | 0.603 | |

Table 5.2. Results of the two-way RMANOVA for kinematics:

Table 5.3. Results of the two-way RMANOVA for kinematics:

The table shows kinematic responses during FAM, FF and WO blocks of reaching when TMS was applied to the left M1 at different time points during movement preparation. Values represent means and standard errors.

| TIME | TMS: 10ms | TMS: 100ms | TMS: 130ms | TMS: 160ms | TMS: 190ms | TMS: 220ms | TMS: 250ms | TMS: 280ms | TMS: 310ms |
|-----------|---------------|--------------|-------------|---------------|--------------------|------------------|---------------|---------------|---------------|
| | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] |
| CONDITION | | | | Mov | vement Onset (ms): | : | | | |
| FAM | 337 [15] | 384 [15]* | 407 [17]* | 408[14]* | 416 [11]* | 417 [13]* | 436 [12]* | 447 [14]* | 448 [13]* |
| FF | 310 [15]+ | 365 [16]* | 385 [17]* | 393 [21]* | 395 [17]*+ | 392 [16]*+ | 404 [16]*+ | 425 [21]* | 420 [16]*+ |
| WO | 365 [20] | 397 [17] | 411 [15] | 419 [16] | 440 [18]* | 437 [18]* | 451 [16]* | 457 [17]* | 468 [13]* |
| | | | | Mov | vement Offset (ms) | : | | | • |
| FAM | 1185 [16] | 1241 [15]* | 1278 [17]* | 1271 [14]* | 1277 [13]* | 1263 [22]* | 1300 [21]* | 1301 [19]* | 1303 [18]* |
| FF | 1166 [20] | 1247 [14]* | 1254 [16]* | 1278[16]* | 1291 [15]* | 1283 [12]* | 1268 [20]* | 1295 [21]* | 1284 [16]* |
| WO | 1185 [17] | 1220 [17] | 1239 [14] | 1233 [12] | 1257 [17] | 1236 [15] | 1276 [9] | 1266 [20] | 1293 [17] |
| | | | | Maxin | num Velocity (m.s | ⁻¹): | | | |
| FAM | 0.28 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.28 [0.01] |
| FF | 0.32 [0.02] | 0.31 [0.01] | 0.31 [0.01] | 0.31 [0.01] | 0.31 [0.02] | 0.31 [0.01] | 0.32 [0.01]■✦ | 0.31 [0.01] | 0.32 [0.01] |
| WO | 0.29 [0.01] | 0.29 [0.01] | 0.29 [0.01] | 0.29 [0.01] | 0.29 [0.01] | 0.30 [0.01] | 0.29 [0.01] | 0.30 [0.01] | 0.30 [0.01] |
| | | | | Move | ment Duration (ms | s): | | | |
| FAM | 848 [23] | 858 [22] | 871 [27] | 863 [20] 🛦 | 861 [16] | 847 [26] | 862 [22] | 854 [18] 🔺 | 855 [22] |
| FF | 862 [20] | 882 [15]+ | 869 [22]+ | 885 [23]+ | 895 [18]+ | 894 [17]+ | 864 [21] | 870 [23]+ | 865 [14] |
| WO | 816 [17] | 818 [16] | 823 [17] | 814 [18] | 819 [22] | 796 [17] | 824 [17] | 803 [15] | 821 [15] |
| | | | | Summe | d Error (distance: | cm) | | | · |
| FAM | 2.48[0.17] | 2.46[0.21] | 2.51[0.21] | 2.78[0.19] | 2.61[0.15] | 2.50[0.18] | 2.72[0.21] | 2.63[0.21] | 2.75[0.26] |
| FF | 4.17 [0.29]■+ | 4.94 [0.71]■ | 3.98 [0.29] | 4.25 [0.21]■+ | 4.69 [0.34]■+ | 5.20 [0.47]■✦ | 5.23 [0.43] | 4.59 [0.25]■+ | 5.35 [0.41]■+ |
| WO | 3.12 [0.32] | 3.07 [0.41] | 3.07 [0.37] | 3.28 [0.26] | 3.05 [0.35] | 3.10 [0.49] | 3.05 [0.33] | 3.02 [0.35] | 3.81 [0.56] |
| | | | | | Force (N): | • | | | |
| FAM | 4.6[0.3] | 4.7[0.3] | 4.7[0.3] | 4.7[0.3] | 4.8[0.3] | 4.6[0.3] | 4.7[0.3] | 4.7[0.3] | 4.7[0.3] |
| FF | 10.6[0.3] | 10.5[0.3] | 10.5[0.3]■+ | 10.4[0.3]■✦ | 10.6[0.3] | 10.3[0.3] | 10.5[0.3]■✦ | 10.5[0.3]■✦ | 10.8 [0.3] |
| WO | 4.8[0.2] | 4.9[0.2] | 4.9[0.2] | 4.9[0.2] | 4.9[0.2] | 4.9[0.2] | 4.9[0.2] | 4.9[0.2] | 5.0[0.2] |

Symbols represent significance following post hoc testing:

* = significant difference vs T10 for time in FAM blocks, * = significant difference vs T10 for time for FF blocks, * = significant difference vs T10 for time for WO blocks,

 \blacksquare = significant difference between FAM and FF, \blacktriangle = significant difference between FAM and WO, \bigstar = significant difference between FF and WO



Figure 5.1[A-F]: Graphical presentation of the post-hoc testing results (based on table 5.3) for the significant kinematic variables:



Symbols represent significance following post hoc testing:

* = significant difference vs T10 for time in FAM blocks, * = significant difference vs T10 for time for FF blocks, * = significant difference vs T10 for time for WO blocks, = significant difference between FAM and FF, \blacktriangle = significant difference between FAM and WO, + = significant difference between FF and WO

5.6 Discussion

The purpose of this chapter was to explore the effects of TMS delivered to the left M1 during reaching preparation in motor adaptation. Previous experiments investigating the role of the left M1 regarding motor function have implemented different TMS protocols and delivered TMS at different time points to target the M1 (Hunter, Sacco and Turner, 2011; Orban de Xivry *et al.*, 2013). Studies have also been conducted in non-human primates to explore neuronal changes in the left M1 during the early stages of motor adaptation (Paz *et al.*, 2003). This experiment aimed to provide novel findings regarding left M1 function in human subjects with single pulse TMS in a robotic-mediated upper-limb reaching FF paradigm.

5.6.1 MEP responses:

This study demonstrated significantly larger MEP responses in the BB muscle during FF reaching compared to both FAM and WO reaching, whereas MEP amplitude in the TB was significantly larger when comparing FF reaching to FAM reaching only. No other muscles (ECR and FCR) were significantly affected by CONDITION. The results from this study showed a significant increase in BB MEP amplitude as TMS pulses became closer to movement onset, and this trend for the increase in CSE during the latter stages of movement preparation is in line with previous studies (Leocani *et al.*, 2000; Orban de Xivry *et al.*, 2013).

Furthermore, the changes in peak-to-peak MEP responses that were greater in the BB muscle (compared to the TB muscle – whereby only FF reaching responses at 220ms differed from FAM reaching; see table 5.1) can be explained by the distribution of corticospinal projections (Richardson *et al.*, 2006). For instance, studies in both human subjects and non-human primates have shown that significantly greater numbers of corticomotor neuronal cells project to the BB motor neurons compared to the TB motor neurons (Palmer and Ashby 1992; Richardson *et al.*, 2006). Further evidence for this notion comes from patients with M1 lesions resulting in arm paresis, whereby weakness in the limbs have been found to be more prominent in elbow flexors as opposed to elbow extensors (Andrews and Bohannon, 2000; Richardson *et al.*, 2006). Therefore MEP findings in this study provided evidence for the notion that adaptive changes are

affected differently in muscles, resulting in their varying CSE, possibly driven by different neural pathways (Hunter, Sacco and Turner, 2011).

5.6.2 Kinematics:

The results of this experiment demonstrated that TMS to the left M1 had a behavioural effect by prolonging the movement onset time. This addresses hypothesis two for this experiment. The findings showed that the later the TMS pulse was delivered during movement preparation, the greater the delay in movement onset. TMS to the left M1 during movement preparation had both time and condition effects, which was not the case for the other kinematic variables explored (movement offset, duration, maximum velocity, summed error and maximum force). These results with regards to the delay in movement onset have been similarly highlighted by Day, Rothwell and Marsden (1983) and Meyer and Voss (2000). Day, Rothwell and Marsden (1983) proposed that TMS stimulation leads to neuronal inhibition, which results in neurons becoming unresponsive for a brief moment, therefore delaying the signals that facilitate movement onset. Kimura, Haggard and Gomi (2006) similarly reported that the effect of TMS during perturbed reaching significantly postponed movement onset. The cause of the delay following stimulation that were found can thus be attributed to the inhibitory effects of SP-TMS (see Section 4.6.2).

The delay that occurred in movement onset and offset can also be explained by the 'waiting period' phenomena as described by Hasegawa *et al.*, (2017). Prior to motor output there is a waiting period (i.e. the wait when preparing for action). Movement preparation has been argued to be a result of cortical changes that occur during the waiting period (Wise, 1985; Hasegawa *et al.*, 2017). Using calcium imaging methods to explore motor cortical neuronal activity in mice during preparation for reaching, Hasegawa *et al.*, (2017) found that motor preparation was accompanied by a selective inhibition of neural networks mediated by interneurons (Pfeffer *et al.*, 2013) which resulted in delays in movement responses due to the waiting period. This finding was therefore important because it highlighted the link between motor behaviours and the M1 circuitry, specifically how the neural networks can influence reaching preparation.

An additional behavioural finding from this study was the significant increase in maximum velocity during FF TMS reaching compared to TMS FAM reaching. Motor adaptation paradigms without TMS have also demonstrated that FF reaching increases

the speed of movements during a reach (Pizzamiglio *et al.*, 2017a). Researchers have argued that when there is a sudden change in the environment (e.g. a perturbation), velocity shifts occur (i.e. larger peaks in movement) to allow more time for participants to reach a target (Izawa *et al.*, 2008).

Furthermore, in this study TMS interfered with novel reaching, as summed error was significantly larger during FF reaching compared to both FAM and WO reaching. This result could be due to TMS disrupting neural communication processes between the M1 and other motor regions. For example, the cerebellum plays a key role in transmitting signals to the M1 to enable accurate motor reaching (i.e. straight and smooth trajectories) (Bastian, 2011). However it has been argued that following neuroplasticity, cells in the M1 become altered and motor responses in reaching occur as a result of input from other cortical structures, such as the pre-motor cortex and parietal cortex (rather than the cerebellum), which can cause errors in reaching (Tseng *et al.*, 2007; Orban de Xivry *et al.*, 2013). Errors in FF reaching could have also been a result of the recruitment of other brain regions (e.g. pre-motor cortex, posterior parietal regions, and the cerebellum) (Lee, *et al.*, 2003). For example, stimulation could have effected related motor cortical structures via trans-synaptic transmission (Chouinard *et al.*, 2003; Richardson *et al.*, 2006).

Overall, although various behaviours (neural and kinematics) were investigated in this experiment there are a number of factors to consider. For example, only individuals aged between 18 - 37 were tested, therefore these findings may not be generalisable to other age groups (e.g. older adults). Furthermore, motor adaptation learning without TMS was confined to four blocks only. It could be argued that these four blocks of reaching may not have been enough to fully explore the learning process and adding additional blocks may have warranted different results with TMS to the left M1. However, this would have been too demanding and fatiguing for the participant, particularly with TMS stimulation delivered after FF blocks of reaching. Therefore, there was a trade-off between participants completing the full motor adaptation and avoiding fatigue.

5.7 Chapter conclusions:

The purpose of this chapter was to illustrate the neural and behavioural effects of TMS to the left M1 during right arm novel reaching. Behavioural changes that were observed

such as delays in movement onset and offset can be attributed to the inhibitory effects of SP-TMS to the motor cortex. Additionally the effects of TMS on trajectory errors (summed error) were significantly larger during FF reaching as opposed to FAM and WO reaching, showing the specific disruptive effect of stimulation during novel TMS reaching only. This can be a result of various factors, including 1) neuroplasticity affecting communication between different cortical motor structures, and 2) the possible trans-synaptic effect of TMS to related cortical regions. Furthermore, MEP findings were also in line with previous studies, showing an increase in BB amplitude as opposed to TB amplitude when reaching away from the body (135° target) during FF reaching. MEPs were also found to be time- and condition-tuned, with significantly larger responses when TMS pulses became closer to movement onset, and mainly in FF blocks of reaching (see table 5.1). The differences in CSE muscles responses (i.e. flexors vs. extensors) in adaptive reaching in this study can be attributed to a possible difference in neural pathways between muscles, a notion that has been further supported by patients with M1 lesions. The next experimental chapter in this thesis explores whether TMS to the left PPC results in similar behavioural findings, especially when considering that the left PPC plays a vital role along with the M1 in novel reaching.

Chapter 6

6 Experiment 3

The impact of SP-TMS to the left posterior parietal cortex (PPC) during right arm reaching in a motor adaptation protocol - A virtual disruption study

6.1 Introduction

TMS has been used as a virtual disruption tool to gain an insight into the functional role of different brain regions (Della-Maggiore *et al.*, 2004; Sliwinska, Vitello and Devlin, 2014). The current that is induced in the brain when TMS is delivered to a specific site can have an impact on local brain activity and functional connectivity with other more remote regions such that disruptions in behavioural responses can result (Bolognini and Ro, 2010). Therefore, TMS can explore whether a cortical region is necessary for a particular function and this makes it different from other imaging modalities, such as functional magnetic resonance imaging (fMRI), which is more correlation-oriented, rather than casual-oriented in studying brain-behaviour functions (Bolognini and Ro 2010; Sliwinska, Vitello and Devlin, 2014).

Virtual disruption effects can be explored when TMS is delivered before and during a behavioural task. The changes that can occur in behavioural responses range from changes in the behavioural reaction time in response selection (this can be either increased or decreased) to enhanced errors rates in performance (Della-Maggiore *et al.*, 2004; Paus 2005; Sack, 2006; Sliwinska, Vitello and Devlin, 2014). For example, repetitive TMS (rTMS) to the left inferior frontal cortex has been found to induce errors in speech production (Pascual-Leone, Gates and Dhuna, 1991; Sliwinska, Vitello and Devlin, 2014), whereas single-pulse to the visual cortex has been reported to disturb processes of visual perception (Corthout *et al.*, 1999; Bolognini and Ro 2010).

Throughout this thesis, a robot-mediated force-field learning paradigm was used to explore novel motor learning. Although various regions in the neural motor circuit are involved in motor skill learning and aid the motor adaptation process, the left posterior parietal cortex (PPC) has been reported to be a key region that facilitates novel motor learning. The left PPC contributes to movement preparation and motor planning, as well as navigation and spatial awareness (Kaas and Stepniewska, 2016; Whitlock, 2017). For example, in a functional magnetic resonance imaging study (fMRI) study in which taxi

drivers carried out a virtual driving task by imagining driving passengers through a route, findings showed an increase in cortical activity within the parietal regions, particularly during the planning stage of the virtual journey (Maguire, Woollett and Spiers, 2006). The left PPC also has specific functions related to motor reaching, with evidence pointing to a posterior parietal reach region (PPR) in both macaques and human subjects (Connolly, Andersen and Goodale, 2003; Kaas and Stepniewska, 2016). In non-human primates, reaching and pointing tasks showed maximum neuronal firing in the PPR region, and in human imaging studies an area, homologous to where it was found in monkeys was activated (Batista *et al.*, 1999; Connolly, Andersen and Goodale, 2003; Whitlock, 2017).

The PPC is also involved in motor sequence learning (Jenkins et al., 1994; Catalan et al., 1998). Positron emission tomography (PET) studies exploring regional cerebral blood flow (rCBF) during sequential finger and thumb tapping have demonstrated that rCBF in parietal regions is specific to the length and complexity of the motor sequence; the longer and more complex the sequence, the greater the rCBF (Jenkins et al., 1994; Catalan et al., 1998). This activation has been explained by the link between the prefrontal cortex and the PPC which enables sustained attention during complex sequencing task (Friston et al., 1991; Catalan et al., 1998). As well as sequential motor learning, the function of the left PPC in novel motor learning was demonstrated by Della-Maggiore et al., (2004). In a robotic reaching task, they delivered single-pulse TMS to both the left PPC and an area of the occipital lobe (as a control) following movement onset and explored differences in reaching trajectories between the two regions during FAM and FF reaching. They found that FAM reaching with TMS did not affect kinematics of reaching and similar reaching performance was noted between the two regions when stimulated. During FF TMS reaching however, although performance was similar between the two regions in the initial stages of reaching, this was not the case in the final stages of FF reaching. Reaching deviated from the ideal path only with left PPC stimulation and not with occipital stimulation, demonstrating the importance of the left PPC in signalling the correct motor responses for regulating reaching trajectories during novel motor learning (Della-Maggiore et al., 2004).

The left PPC was targeted in this experiment (with TMS) due to its functions in motor reaching and purported importance in novel motor learning. This experiment provides a novel insight into kinematic parameters related to reaching that have not previously been explored with left PPC stimulation such as movement onset, movement offset and maximum velocity. However, in line with other experiments (i.e. Della-Maggiore *et al.*, 2004) errors in reaching will also be explored (summed error).

TMS has been noted to disrupt behaviours when delivered at specific times. For example, single-pulse TMS to the visual cortex between 80ms to 140ms impaired visual perception processes (Amassian *et al.*, 1989; Bolognini and Ro, 2010). In this experiment TMS was administered at a range of different time points during reaching preparation. This is because it is currently unknown whether TMS during motor preparation impacts behavioural responses, as time-related disruptions of the left PPC have only been studied following movement onset (Della-Maggiore *et al.*, 2004). This experiment therefore set out to provide a novel insight into left PPC function with regards to its role in motor planning during a novel form of motor adaptation.

6.2 Methodology

Table 3.2 illustrates participant demographics for this experiment. Section 3.3.2.2 highlights how regions with no visible experimental output (i.e. an MEP) were identified (with an EEG cap) and how TMS intensity was set using functional measures. Section 3.3.2.2.1 specifically describes the location that was used for left PPC stimulation (P3 electrode) and the coil position employed. This is also graphically illustrated in figure 3.3. Furthermore, the reaching task used for motor adaptation is described in section 3.4.1 and 3.4.2, and is shown in figure 3.5.

6.3 Data acquisition: MEPs and Kinematics

MEP responses were only obtained and recorded when locating the motor hotspot for the BB. This was carried out in order to identify the participant's resting motor threshold (RMT) (left M1) to deliver TMS to the left PPC at 110% of their RMT. MEPs were not elicited during left PPC stimulation. The main data from this experiment were kinematics; section 3.6.1 describes how kinematic parameters were all analysed and quantified offline in MatLab 2017b (The MathWorks Inc, Natick MA, USA).

6.4 Statistical analysis: Kinematics

The analysis performed for the kinematic variables, including the statisitical tests carried out (RMANOVA), how sphericity assumptions were met and the steps taken for post-hoc testing (Student's t-test) for the main factors of TIME and CONDITION, and for TIME*CONDITION interactions is described in section 3.7.1.

6.5 Results

6.5.1 Kinematics - parameters and trajectories of reaching:

When TMS was introduced in FF reaching, trajectory deviations were visible (even during the final block of reaching), which was not the case for FAM and WO reaching (see figure 6.1). This demonstrated that TMS had a specific disruptive effect only during novel motor learning.

The RMANOVA (table 6.1) revealed a significant main effect for TIME on movement onset and movement offset (p < 0.05), but not for maximum velocity, movement duration, summed error and maximum force (p > 0.05). There was also a significant main effect of CONDITION for movement onset, maximum velocity, movement duration, summed error and maximum force (p < 0.05) but not for movement offset (p > 0.05). There were no significant interaction effects found for any of the kinematic variables (p > 0.05) (see table 6.1).

Post hoc testing for TIME (table 6.2, figure 6.2[A-F]) demonstrated that during FAM reaching with TMS, movement onset was significantly increased compared to T10 at all time points apart from T100. Movement offset was significantly increased from T10 during 190ms, 220ms, 250ms, 280ms and 310ms. During FF reaching with TMS, movement onset was increased at all time points compared to T10 (all p < 0.006). Movement offset was significantly increased at all time points apart from 100ms and 250ms compared to the T10. During WO reaching, movement onset was significantly increased at 190ms, 220ms, 280ms and 310ms (p < 0.006), whilst movement offset was only significantly increased at 280ms compared to T10.

Post hoc testing for CONDITION (table 6.2, figure [6.2A-F]) comparing FF vs. FAM found no significant differences between responses for movement onset (p > 0.016). Maximum velocity was greater in FF reaching compared to FAM reaching (all time points; p < 0.016). Movement duration was greater in FF reaching although only at 130ms (p < 0.016). Summed error and maximum force were both increased in FF reaching compared to FAM reaching (at all time points all p < 0.016). Generally, kinematics values returned to or towards FAM values in WO reaching.

Post hoc testing results are shown in table 6.2 and figure 6.2[A-F].



Figure 6.1: Reaching block trajectories with TMS delivered to the left PPC in a single-participant

TMS to the left PPC during FAM reaching only resulted in a slight deviation from the ideal reaching trajectory in the first few trials (red coloured traces - A). FF reaching trajectories are shown in figures 6.2B and 6.2C. Figure 6.2B illustrates reaching trajectories in the final block of FF reaching without TMS, and figure 6.2C shows reaching trajectories in the final block of FF reaching with TMS. It can be seen that in the final block of FF reaching with no TMS delivered, the participant was able to adapt to the force-field because reaching became closer to the ideal, however, when TMS was introduced into the FF reaching, this resulted in major deviations from the ideal reaching trajectory (C). During WO, reaching trajectories returned FAM values with without TMS to or (D).

| | | TIME: | | (| CONDITION | | TIME * CONDITION INTERACTION | | | |
|---------------------------------------|-------------------|-------|---------|-------------------|-----------|---------|------------------------------|-------|-------|--|
| | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | |
| Movement Onset (ms) | 3.0 [38.1] | 17.15 | < 0.001 | 2 [26] | 9.11 | < 0.001 | 16[208] | 0.568 | 0.906 | |
| Movement Offset (ms) | 8[104] | 8.511 | < 0.001 | 2[26] | 0.176 | 0.840 | 16[208] | 1.285 | 0.209 | |
| Maximum Velocity (m.s ⁻¹) | 8[104] | 0.257 | 0.916 | 2 [26] | 10.15 | 0.001 | 16[208] | 0.935 | 0.530 | |
| Duration (ms) | 4.6 [58.7] | 0.538 | 0.729 | 2[26] | 5.159 | 0.019 | 16[208] | 1.233 | 0.245 | |
| Summed Error (distance: cm) | 3.3 [42.5] | 1.374 | 0.263 | 1.2 [14.8] | 13.34 | < 0.001 | 16[208] | 1.363 | 0.162 | |
| Maximum Force (N) | 3.1 [40.2] | 1.362 | 0.268 | 1.2 [14.4] | 243.01 | < 0.001 | 16[208] | 1.119 | 0.339 | |

Table 6.1. Results of the two-way RMANOVA:

Table 6.2. Post-hoc testing results

The table shows kinematic responses during FAM, FF and WO blocks of reaching when TMS was applied to the left PPC at different time points. Values represent means and standard error means (SEM).

| TIME | TMS: 10ms | TMS: 100ms | TMS: 130ms | TMS: 160ms | TMS: 190ms | TMS: 220ms | TMS: 250ms | TMS: 280ms | TMS: 310ms |
|-----------|------------------------|----------------------|-----------------------|-----------------------|---------------------------------|------------------------|------------------------|------------------------|-----------------------|
| | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] |
| CONDITION | | | | Mov | vement Onset (ms): | | | | |
| FAM | 362 [20] | 396 [20] | 412 [21]* | 362 [17]* | 427 [23]*▲ | 428 [22]* | 428 [21]* | 434 [26]* | 438 [21]* |
| FF | 350 [20]+ | 383 [18]* | 395 [20]* | 350 [23]* | 393 [17]*+ | 402 [23]*+ | 406 [20]* | 408 [22]*+ | 403 [21]*+ |
| WO | 384 [22] | 418 [24] | 425 [29] | 384 [23] | 437 [20]* | 453 [28]* | 443 [26] | 470 [32]* | 454 [39]* |
| | | | | Mov | vement Offset (ms): | | | | |
| FAM | 1204 [21] | 1223 [26] | 1226 [16] | 1245 [24] | 1274 [21]* | 1259 [26]* | 1270 [28]* | 1272 [31]* | 1276 [29]* |
| FF | 1220 [26] | 1244 [30] | 1270 [27] | 1281 [35]* | 1261 [30]* | 1264 [32]* | 1263 [31] | 1265 [30]* | 1263 [28]* |
| WO | 1210 [27] | 1227 [27] | 1252 [33] | 1237 [20] | 1253 [22] | 1271 [34] | 1258 [34] | 1305 [39]* | 1275 [38] |
| | | | | Maxin | num Velocity (m.s ⁻¹ | ¹): | | | |
| FAM | 0.28 [0.01] | 0.28 [0.01] | 0.29 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.28 [0.01] |
| FF | 0.32 [0.02] ★ ■ | 0.31 [0.02] | 0.31 [0.02] | 0.31 [0.02] | 0.32 [0.02] + ■ | 0.31 [0.02] + ■ | 0.32 [0.02] + ■ | 0.31 [0.01] + ■ | 0.33 [0.02] |
| WO | 0.28 [0.01] | 0.29 [0.01] | 0.29 [0.01] | 0.29 [0.01] | 0.29 [0.01] | 0.28 [0.01] | 0.28 [0.01] | 0.29 [0.01] | 0.29 [0.01] |
| | | | | Move | ement Duration (ms) |): | | | |
| FAM | 842 [19] | 827 [21] | 814 [18] | 836 [24] | 849 [22] | 832 [17] | 843 [21] | 838 [19] | 839 [19] |
| FF | 870 [18] 🕇 | 861 [21]+ | 876 [22] | 884 [25] + | 868 [24] | 862 [24] | 857 [27] | 857 [18] | 864 [20] |
| WO | 826 [23] | 809 [15] | 827 [17] | 806 [16] | 816 [18] | 817 [20] | 815 [19] | 835 [19] | 821 [21] |
| | | | • | Summe | d Error (distance: cr | m): | • | | |
| FAM | 2.83[0.18] | 2.71 [0.21] | 2.49[0.21] | 2.61[0.22] | 2.77[0.22] | 2.92[0.20] | 2.98[0.31] | 2.62[0.22] | 2.67[0.15] |
| FF | 4.41[0.46] ★ ■ | 4.40[0.42] ★■ | 4.71[0.50] ↓ ■ | 4.89[0.56] ↓ ■ | 4.92[0.69] ↓ ■ | 4.39[0.51] ↓ ■ | 5.23[0.86] ↓ ■ | 4.27[0.42] ↓ ■ | 4.21[0.38] ↓ ■ |
| WO | 2.29[0.11] | 2.43[0.21] | 2.72[0.27] | 2.59[0.13] | 2.36[0.20] | 2.65[0.26] | 2.53[0.19] | 2.67[0.16] | 2.48[0.26] |
| | | | • | • | Force (N): | | • | • | • |
| FAM | 4.6[0.1] | 4.7[0.2] | 4.7[0.1] | 4.7[0.2] | 4.7[0.1] | 4.7 [0.2] | 4.7[0.1] | 4.7[0.1] | 4.6[0.2] |
| FF | 10.0[0.4] + ■ | 10.5[0.4] + ■ | 10.4[0.4] ↓ ■ | 10.2[0.4] ↓ ■ | 10.5[0.4] + ■ | 10.3[0.3] ↓ ■ | 10.5[0.4] + ■ | 10.4[0.4] + ■ | 10.3[0.4] ★ ■ |
| WO | 4.6[0.1] | 4.6[0.1] | 4.8[0.1] | 4.6[0.1] | 4.6[0.13] | 4.6[0.1] | 4.6[0.2] | 4.6[0.1] | 4.6[0.1] |

Symbols represent significance following post hoc testing:

* = significant difference vs T10 for time in familiarisation blocks, * = significant difference vs T10 for time for FF blocks, * = significant difference vs T10 for time for WO blocks,
■ = significant difference between FAM and FF, += significant difference between FF and WO, ▲ = significant difference between FAM and WO



Figure 6.2[A-F]: Graphical presentation of the post-hoc testing results (based on table 6.2) for the significant kinematic variables:



Symbols represent significance following post hoc testing:

* = significant difference vs T10 for time in familiarisation blocks, * = significant difference vs T10 for time for FF blocks, * = significant difference vs T10 for time for WO blocks,
■ = significant difference between FAM and FF, += significant difference between FF and WO, ▲ = significant difference between FAM and WO

6.6 Discussion

The results of this experiment demonstrated the effects of single-pulse TMS to the left PPC during a novel motor task. Previous research of left PPC stimulation have not examined whether TMS at various of different time points during the preparation of a reach affects specific kinematic variables, such as movement onset and movement offset, but have rather placed emphasis on trajectory errors of reaching that occur after movement onset (Della-Maggiore *et al.*, 2004). This experiment therefore provided a novel insight into how TMS delivered to the left PPC during motor preparation can have an impact on different kinematic variables.

6.6.1 Kinematics - The shift in movement onset:

Stimulation the left PPC with TMS during motor preparation for reaching delayed the movement onset time, in FAM and FF reaching (this addresses hypothesis 2 - experiment 3). Similar delays also occurred in movement offset.

This delay in movement onset and offset can be explained in terms of the statedependency TMS theory (Silvanto and Muggleton, 2008). This theory proposes that the impact of TMS depends on the degree of excitability that the stimulated region exhibits. In this study, the subjects were already involved in the task, i.e. preparing themselves for action (in this case reaching), the region was therefore pre-activated. It has been suggested that the effects of stimulation on an area that has been pre-activated can result in neuronal noise, which can interrupt the region's functioning, therefore resulting in delayed responses (Miniussi, Ruzzoli and Walsh, 2010; Busan et al., 2012). Although previous studies have used other quantification techniques to determine when motor preparation occurred (e.g. participant's mean reaction time) for TMS to be delivered, the results regarding movement onset are similar. For example, it has been found that stimulating mid-point through the preparation phase leads to a delay in movement onset, which is not the case when stimulating prior to movement preparation (Busan et al., 2009; Busan et al., 2012). In this experiment, TMS was administered during reaching preparation, and not prior preparation (i.e. before visual cue), therefore the delayed responses that occurred regarding reaction times are in line with previous studies.

6.6.2 Kinematics - Impact of TMS on trajectories of reaching (summed error) and reaching velocity:

Another key finding in this experiment was that disrupting the left PPC with TMS during preparation for novel reaching impaired reaching performance as revealed by greater trajectory errors (this addresses hypothesis 1 - experiment 3). Left PPC stimulation significantly impaired reaching only during motor adaptation (FF reaching). The fact that TMS did not cause disruption during FAM and WO reaching suggests that the left PPC has specific functions related to novel motor learning per se. These findings are in line with what has previously been reported by Della-Maggiore *et al.*, (2004) who suggested that during FF reaching, different brain regions, including the PPC feed into other cortical structures, signalling commands for successful motor output. Disrupting PPC function with TMS may lead to error signal processing in visuo-motor transformations, thus causing trajectory errors in FF reaching (Chouinard *et al.*, 2003; Della-Maggiore, Malfait and Ostry, 2004). Experiment 3 therefore provided further evidence for the importance of the PPC in adjusting arm position to meet the requirements of a novel task.

The role of the left PPC in error signal processing has been further identified by researchers (Oliveira *et al.*, 2010), who have argued that the specific network connecting the left PPC with other cortical structures enables the formation of an internal model of error-driven responses for motor output (Oliveira *et al.*, 2010; Smith and Shadmehr, 2005). These regions therefore store movement dynamic information (Kawato 1999; ; Della-Maggiore *et al.*, 2004; Malfait and Ostry, 2004; Oliveira *et al.*, 2010;). In this experiment, TMS to the left PPC over repeated trials led to errors in responses, which based on the error-signal processing theory, could have been due to a disruption of movement dynamics (i.e. deviated reaching responses) being stored as an internal model for novel reaching.

It can be argued TMS to the left PPC creates a virtual disruption model of optic ataxia. Optic ataxia is a neurological condition that causes impairments in visually guided behaviour, and in many patient cases, lesions to the PPC contributes to the condition (Cavina-Pratesi, Connolly and Milner, 2013; Andersen *et al.*, 2014). Using similar reaching paradigms, increased trajectory errors have also been found among patients with optic ataxia (Pisella *et al.*, 2000) who were not able produce an ideal reach to a target of interest during novel reaching.
Furthermore, TMS stimulation to the left PPC increased maximum velocity responses during FF reaching compared to FAM and WO reaching. It could be argued that participant's increase their speed because they require more time to reach towards the target (due to the robotic perturbation). Specific structures in the motor network system connected to the left PPC have also been found to facilitate this action. For example, the cerebellum and the premotor cortices have been suggested to provide the appropriate strategies to respond to movements, through eliciting fast responses or slow responses depending on the task (Desmurget and Grafton, 2000). Based on this notion, various networks in the M1 enable individuals to adjust their speed of movements which can help explain why velocity responses differ between reaching blocks.

6.7 Chapter conclusions:

The purpose of this chapter was to explore the effects of left PPC stimulation during novel motor learning. TMS only had a significant disruptive impact on summed error during novel reaching, which is in line with conclusions drawn from previous research and can be explained by disrupted error-signal processing. Considering that TMS did not disrupt reaching processes in null-field reaching (FAM and WO) implies that the left PPC has a specific function related to novel motor learning. Overall, this experiment demonstrated that TMS to the left PPC caused a virtual disruption in learning. The mechanism may be via disruption of local processing or disrupted pathways involving the PPC and more remote nodes of the motor network circuit. Investigating whether novel motor learning is also disrupted during stimulation of other motor regions could help in developing a model of cortical regions that are specific only for novel motor reaching.

Chapter 7

7 Experiment 4

Exploring the impact of SP-TMS to the supplementary motor area (SMA) during right arm reaching in a novel motor learning protocol

7.1 Introduction

The supplementary motor area (SMA) is located in the frontal lobe of the brain and is situated near motor regions, such as the pre-motor cortex (PMC) and M1 (Nachev *et al.*, 2007). It consists of two regions; the pre-SMA and the SMA proper, and their functions differ regarding neural input and output underpinning motor behaviours. For example, the pre-SMA receives input from the prefrontal cortex, whereas the SMA-proper projects the input to the M1 for motor output (Tanji, 1994).

The SMA is known to have multiple functions which facilitate motor learning. For example, it assists skilled motor performance, plans movements for execution, helps retrieve motor memories and is also involved in sequential motor learning (i.e. learning patterns of movements; Tanji, 1994; Borich *et al.*, 2015). Early imaging studies, including positron emission tomography (PET) scans have illustrated the role of the SMA in motor behaviours. For example, Roland *et al.*, (1980) reported an increase in SMA neural activity when subjects imagined producing finger sequencing movements. Functional MRI neuro-feedback studies have also demonstrated the active role of the SMA in motor imagery. For instance, Mehler *et al.*, (2019) found increased blood oxygenation level dependent (BOLD) responses in the SMA compared to the M1 (which was de-activated) during a hand motor imagery task.

Other PET imaging studies have explored SMA patterns of cortical activity during motor skill learning and have found increased levels of practise induced activity in the SMA, but not in other cortical regions, such as the cerebellum and right PMC (van Mier, Perlmutter and Petersen, 2004). Functional magnetic resonance imaging (fMRI) studies have compared differences in neural activity between the SMA and the pre-SMA with results illustrating that fast motor learning is accompanied with greater SMA activity but decreased pre-SMA activity (Grafton, Hazeltine and Ivry, 1995; Sakai *et al.*, 1999; Floyer-Lea and Matthews, 2005).

Although neuroimaging studies have enabled researchers to explore the role of the SMA in motor skill learning, the specific functions of the SMA in relation to novel motor learning (i.e. motor adaptation) have primarily been investigated in non-human primates. Animal studies have provided an insight into how neuronal activity is changed in the SMA during motor adaptation compared to null-field reaching. For example, in a motor adaptation paradigm, Padoa-Schioppa, Li and Bizzi (2004) found that SMA neurons exhibited plasticity following a novel motor task and cells became differently tuned. Therefore it can be argued that induced plasticity in the SMA facilitates novel motor learning (Padoa-Schioppa, Li and Bizzi 2004).

Studies applying TMS to disrupt the functioning of the SMA have mainly been investigated during motor sequencing tasks and motor tasks performed with both hands (bimanual). For example, Gerloff et al., (1997) used repetitive TMS (rTMS) to target the SMA during complex and simple finger motor sequencing tasks. They found that rTMS only reduced motor accuracy in the complex sequencing task. With regards to bimanual motor paradigms, rTMS delivered to the SMA had a detrimental impact in performance during anti-phase (asymmetrical movements) bimanual hand movements compared to in-phase (synchronised movements) hand movements (Serrien et al., 2002; Steyvers et al., 2003). Therefore, the disruption of the SMA with TMS depends on the complexity of the task (Gerloff et al., 1997; Hallett, 2007; Kim and Shin, 2014). Paired pulse TMS (PP-TMS) protocols targeting the SMA, specifically the pre-SMA have shown its involvement in higher cognitive functions (Nachev, Kennard and Husain, 2008). For instance, in a motor response selection paradigm, the pre-SMA was reported to influence selection responses and only effected the excitability of the M1 when motor selection responses were switched in trials (Mars et al., 2009; Chouinard and Paus, 2010).

Although studies have been conducted with TMS to demonstrate the functional role of the SMA for motor behaviours, these experiments have mostly been conducted with:

1. rTMS and paired-pulse TMS protocols as opposed to single-pulse TMS protocols.

2. Motor sequencing and bimanual tasks compared to motor adaptation tasks.

3. Non-human primates (in studies concerning motor adaptation) in contrast to human subjects.

Therefore, it is currently unknown whether single-pulse TMS to the SMA during motor preparation disrupts novel motor reaching (i.e. motor adaptation protocol) in human subjects and it is for this reason that this experiment was conducted. The experiment specifically investigated whether trajectory errors or other kinematic measures of reaching (such as movement onset, movement offset, maximum velocity, and movement duration) were affected by applying TMS to the SMA.

7.2 Methodology

Table 3.2 demonstrates the participant demographics for SMA stimulation. TMS intensity was set using functional measures for each participant (section 3.2.2.2). The location of the SMA and coil orientation that was used for stimulation is described in section 3.3.2.2 and graphically demonstrated in figure 3.3 (the orientation and position used was based on Cona, Marino and Semenza, 2017 protocol). Furthermore, the reaching task for this experiment is outlined in section 3.4.1 and figure 3.5.

7.3 Data acquisition: MEPs and Kinematics

MEPs were only acquired during assessment of RMT for each participant at the start of the experiment (see section 3.3.1). Therefore, MEPs were not elicited or collected during SMA stimulation. Kinematic data were collected in this experiment and were analysed and quantified offline in MatLab 2017b (The MathWorks Inc, Natick MA, USA) (section 3.6.1).

7.4 Statistical analysis: Kinematics

A two-way RMANOVA was performed to explore whether different kinematic variables in different reaching blocks (FAM, FF and WO) were affected by TMS to the SMA during reaching preparation. Main effects for TIME, CONDITION and interactions (TIME*CONDITION) were obtained. Section 3.7.1 outlines how sphericity assumptions were tested and the steps taken for post-hoc testing with regards statistically significant findings.

7.5 Results

7.5.1 Kinematics - Repeated measures ANOVA and post-hoc testing:

The RMANOVA (table 7.1) revealed a significant effect of TIME for movement onset and movement offset (F > 8.102, p < 0.05), but not for maximum velocity, movement duration, summed error and maximum force (all F < 0.351, all p > 0.05). The RMANOVA showed that TMS to the SMA had a significant effect on CONDITION, for kinematic variables including movement onset, maximum velocity, summed error and maximum force (all F > 3.989, all p < 0.019), but not movement offset and duration (all F < 0.16, all p > 0.05). No interaction effects were found with SMA stimulation (all F > 0.378, all p > 0.05).

Post hoc testing for TIME (table 7.2, figure 7.1[A-E]) demonstrated a significant increase in all time points except 280ms in FAM reaching for movement onset (p < 0.006). For FF reaching, movement onset was significantly increased at all time points compared to T10 (p < 0.006). In WO reaching, movement onset only at 220ms and 280ms were significantly increased compared to T10 (p < 0.006). Movement offset significantly increased only at 130ms compared to T10 during FAM reaching (p < 0.006). During FF reaching, all time points for movement offset were significantly increased compared to T10 (p < 0.006). In WO reaching, no significant differences were found at any time points compared to T10 (p < 0.006).

Post hoc testing for movement onset for CONDITION (table 7.2, figure 7.1[A-E]) for movement onset demonstrated a significant reduction in the FF condition compared to FAM reaching (only at 190ms; p < 0.016). A significant increase in movement onset was found in WO reaching compared to FF reaching (only 220ms; p < 0.016). There were no significant differences in WO vs. FAM reaching (all p > 0.016). Maximum velocity was significantly increased in FF reaching compared to FAM reaching (only 280ms; p < 0.016). There was a significant decrease in maximum velocity in WO compared to FF reaching (only 280ms and 10ms; p < 0.016). There were no significant differences in WO vs. FAM reaching (all p > 0.016). Summed error (SE) was significantly increased in FF compared to FAM reaching (100ms and 130ms; p <0.016). SE was significantly increased in FF reaching compared to WO reaching (100ms, 130ms, 160ms, p < 0.016). There were no significant differences in WO vs. FAM SE responses (p > 0.016). Maximum force was significantly increased in FF compared to FAM and WO reaching as expected (all time points; p < 0.016). There were no significant differences in WO vs. FAM reaching.

| | TIME | | | (| CONDITION | | TIME*CONDITION INTERACTION | | | |
|---------------------------------------|-------------------|-------|----------------|-------------------|-----------|---------|----------------------------|-------|-------|--|
| | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | |
| Movement Onset (ms) | 8[104] | 13.26 | < 0.001 | 2 [26] | 3.989 | 0.031 | 16[208] | 0.722 | 0.770 | |
| Movement Offset (ms) | 8[104] | 8.102 | < 0.001 | 2[26] | 0.157 | 0.856 | 16[208] | 0.378 | 0.903 | |
| Maximum Velocity (m.s ⁻¹) | 8[104] | 0.351 | 0.857 | 2 [26] | 4.079 | 0.033 | 16[208] | 0.636 | 0.852 | |
| Duration (ms) | 8 [104] | 1.127 | 0.352 | 2[26] | 1.953 | 0.162 | 16[208] | 0.494 | 0.948 | |
| Summed Error (distance: cm) | 8 [104] | 0.837 | 0.572 | 1.3[16.0] | 6.597 | 0.016 | 16[208] | 1.387 | 0.150 | |
| Maximum Force (N) | 8 [104] | 0.80 | 0.608 | 1.1 [13.4] | 223.72 | < 0.001 | 16[208] | 1.163 | 0.301 | |

Table 7.1. Results of the two-way RMANOVA:

Table 7.2. Results following post-hoc testing

The kinematic data shown in this table is based on FAM, FF and WO blocks of reaching when TMS was delivered to the SMA. Means, \pm the standard error mean (SEM) are shown.

| TIME | TMS: 10ms | TMS: 100ms | TMS: 130ms | TMS: 160ms | TMS: 190ms | TMS: 220ms | TMS: 250ms | TMS: 280ms | TMS: 310ms | | |
|-----------|-----------------------|---------------|---------------|--------------|--------------------------------|-------------|-------------|---------------|--------------|--|--|
| | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | | |
| CONDITION | | | | Mov | ement Onset (ms): | | | · | | | |
| FAM | 371 [19] | 412 [18] * | 425 [18] * | 423 [18] * | 429 [17] * | 429 [19] * | 432 [17] * | 433 [18] | 442 [24] * | | |
| FF | 361 [22] | 391 [19] * | 403 [18] * | 398 [16] * | 409 [19] *■ | 412 [16] *+ | 427 [24] * | 420 [22]* | 411 [20] * | | |
| WO | 382 [25] | 409 [24] | 421 [22] | 423 [23] | 430 [23] | 440 [20] * | 427 [21] | 436 [24] * | 420 [21] | | |
| | Movement Offset (ms): | | | | | | | | | | |
| FAM | 1197 [33] | 1223 [35] | 1250 [35] * | 1246 [30] | 1249 [33] | 1255 [26] | 1254 [40] | 1246 [31] | 1264 [41] | | |
| FF | 1187 [26] | 1233 [28] * | 1265 [31] * | 1251 [27] * | 1259 [28] * | 1262 [22] * | 1273 [29] * | 1262 [29] * | 1257 [28] * | | |
| WO | 1203 [27] | 1230 [24] | 1251 [26] | 1244 [24] | 1232 [23] | 1261 [27] | 1249 [24] | 1262 [32] | 1247 [16] | | |
| | | | | Maxim | um Velocity (m.s ⁻¹ |): | | | | | |
| FAM | 0.29[0.02] | 0.29[0.02] | 0.29[0.02] | 0.29[0.02] | 0.29[0.02] | 0.29[0.02] | 0.29[0.02] | 0.29[0.01] | 0.29[0.02] | | |
| FF | 0.31[0.01] | 0.31[0.02] + | 0.31[0.01] | 0.31[0.01] | 0.31[0.01] | 0.31[0.01] | 0.30[0.01] | 0.31[0.01] ■+ | 0.31[0.02] | | |
| WO | 0.29[0.01] | 0.29[0.01] | 0.28[0.01] | 0.28[0.01] | 0.29[0.01] | 0.29[0.01] | 0.29[0.01] | 0.29[0.01] | 0.29[0.01] | | |
| | | | | Mover | ment Duration (ms) |): | | | | | |
| FAM | 826 [31] | 813 [31] | 825 [30] | 823 [28] | 820 [29] | 823 [28] | 822 [34] | 813 [25] | 822 [32] | | |
| FF | 826 [24] | 842 [22] | 862 [21] | 853 [21] | 850 [20] | 851 [17] | 847 [25] | 843 [19] | 848 [21] | | |
| WO | 821 [22] | 820 [20] | 830 [25] | 824 [25] | 803 [21] | 820 [24] | 822 [20] | 825 [28] | 827 [21] | | |
| | | | | Summed | l Error (distance: c | m): | | • | | | |
| FAM | 2.53[0.20] | 2.31 [0.19] | 2.38[0.19] | 2.56[0.27] | 2.64[0.22] | 2.48[0.24] | 2.58[0.27] | 2.44[0.23] | 2.71[0.24] | | |
| FF | 3.26[0.25] | 3.33[0.21] ■+ | 3.61[0.24] ■+ | 3.36[0.20] + | 3.30[0.22] | 3.35[0.26] | 3.17[0.20] | 3.13[0.17] | 3.26[0.27] | | |
| WO | 2.69[0.21] | 2.56[0.21] | 2.43[0.23] | 2.56[0.17] | 2.57[0.16] | 2.92[0.27] | 2.67[0.21] | 2.68[0.19] | 2.90[0.27] | | |
| | | | | | Force (N): | | | | | | |
| FAM | 4.6 [0.2] | 4.5[0.2] | 4.5 [0.2] | 4.5[0.2] | 4.6 [0.2] | 4.6[0.2] | 4.5[0.1] | 4.6 [0.2] | 4.6 [0.2] | | |
| FF | 10.0[0.4] | 9.9[0.4] ■✦ | 9.8 [0.3] ■+ | 9.8[0.4] ■+ | 9.9 [0.4] ■+ | 9.9[0.3] ■+ | 9.8[0.4] ■+ | 10.0[0.4] | 9.9 [0.4] ■+ | | |
| WO | 4.5[0.1] | 4.5[0.1] | 4.5 [0.1] | 4.4[0.1] | 4.5 [0.2] | 4.4[0.1] | 4.5[0.1] | 4.5 [0.1] | 4.5 [0.1] | | |

Symbols represent significance following post hoc testing:

* = significant difference vs T10 for time in familiarisation blocks, * = significant difference vs T10 for time for FF blocks, * = significant difference vs T10 for time for WO blocks,
■ = significant difference between FAM and FF, += significant difference between FF and WO, ▲ = significant difference between FAM and WO



Figure 7.1 [A-E]. Graphical presentation of the post-hoc testing results (based on table 7.2) for the significant kinematic variables:



Symbols represent significance following post hoc testing:

* = significant difference vs T10 for time in familiarisation blocks, * = significant difference vs T10 for time for FF blocks, * = significant difference vs T10 for time for WO blocks,
■ = significant difference between FAM and FF, += significant difference between FF and WO, ▲ = significant difference between FAM and WO

7.6 Discussion

This main purpose of this chapter was to investigate whether TMS delivered to the SMA during preparation for novel motor reaching impairs performance and whether TMS delivered at different time points had an impact on different kinematic variables. This experiment provided a novel approach in exploring behavioural mechanisms operating in the SMA during preparation for novel reaching. It employed a single-pulse TMS approach and used human subjects in a robot-mediated FF paradigm, as opposed to previous studies that have used rTMS protocols, motor sequencing tasks and focused mainly on non-human primate responses (Padoa-Schioppa, Li and Bizzi 2004; Serrien *et al.*, 2002; Steyvers *et al.*, 2003).

7.6.1 Kinematics

7.6.1.2 Shift in movement onset and offset:

TMS delivered to the SMA had time specific effects for movement onset and movement offset. As with previous experiments conducted (left M1 and left PPC stimulation), in this study a significant delay in movement onset was noted (this addresses hypothesis 2, experiment 4), when TMS was applied during FAM and FF reaching, whereas in WO reaching significant delays were minimal. Similar shifts in responses were found for movement offset in FAM and FF reaching, but not during WO reaching. This study therefore provided a novel insight into time-specific differences that can occur during SMA stimulation. The shift that occurred in movement onset can be explained by the 'functional overlap' in motor regions (Alexander and Crutcher, 1990; Scott and Kalaska, 1997). Interactions exist between motor networks during planning and reaching execution (He, Dum and Strick, 1995; Padoa-Schioppa, Li and Bizzi, 2004). For example, corticospinal projections from the SMA feed to the premotor cortices and the M1, which project onto the spinal cord for movement execution (Padoa-Schioppa, Li and Bizzi, 2004). This neural communication enables the movement dynamic processing which can influence kinematic behaviours (He, Dum and Strick, 1995).

7.6.1.3 Impact of SMA stimulation on reaching trajectories (SE):

RMANOVA testing revealed a significant effect of condition on SE induced by TMS with significantly greater trajectory errors in FF reaching compared to FAM and WO reaching. Therefore it can be argued that TMS to the SMA disrupted novel reaching performance. However, robust multiple comparisons testing (table 7.2) showed that SE

errors induced by TMS in FF reaching as opposed to other conditions (FAM and WO) were not as prominent as the deviations noted in TMS FF reaching in other experiments (e.g. left M1 and left PPC).

The SMA consists of two sub-regions; the pre-SMA and the SMA-proper and imaging studies (namely fMRI) have aimed to provide functional distinctions between these two sub-structures. For example increased neural activations have been revealed in the pre-SMA in complex motor tasks, whereas the SMA-proper has been found to be activated more so during motor planning and execution (Bates and Goldman-Rakic, 1993; Picard and Strick 2003; Shen and Alexander, 1997; Moran and Schwartz, 1999; Padoa-Schioppa, Li and Bizzi, 2004). In this study it could be argued that the pre-SMA was more likely stimulated because SMA-proper stimulation may have resulted in greater disruptions in reaching considering that it is more involved in motor preparation than the pre-SMA. Furthermore, stimulation of the SMA-proper can cause an indirect activation of the motor cortex and elicit MEPs which is not the case for pre-SMA stimulation (Narayana *et al.*, 2012; Cona, Marino and Semenza, 2017). In this study no MEPs were elicited (as inspected with EMG signals recorded throughout the experiment) and this provides further evidence supporting the notion that the pre-SMA was stimulated here.

However, there are challenges when trying to distinctively define activation patterns between sub-regions of the SMA because both the pre-SMA and SMA-proper are similarly activated in various types of motor states (Wang *et al.*, 2010; Courson, Macoir and Tremblay, 2017). This is supported by fMRI studies that have illustrated similar neural activation patterns in SMA sub-regions during motor learning (Hardwick *et al.*, 2013). Different co-ordinate systems including Montreal Neurological Institute (MNI) templates have been used to locate SMA sub-regions (Chau and McIntosh, 2005; Bracht *et al.*, 2012; Chung et al., 2005). In this experiment, using individual subject structural-MRI scans to locate different SMA sub-regions (with an MNI co-ordinate system) could have been a solution in determining the region of the SMA that was stimulated. Nonetheless, based on previous SMA TMS protocols (Cona, Marino and Semenza, 2017), it could be speculated that the pre-SMA was stimulated in this experiment because no visible physiological responses were noted in EMG signals which has been found with SMA-proper stimulation. Pre-SMA stimulation can also explain why trajectory errors did not occur at all time points of SMA stimulation in FAM and FF

reaching blocks (only significant for 100ms, 130ms and 160ms), in other words motor planning has not been *as* associated with the pre-SMA, whereas it has been with the SMA-proper.

7.7 Chapter conclusions:

The purpose of this chapter was to provide a novel insight into how delivering SP-TMS at different times to the SMA during the preparation of a reaching affected kinematic measures of reaching. TMS was found to delay movement onset and offset, particularly during preparation for FF reaching, which can be explained in terms of the functional overlap that has been found between different motor regions. Time specific differences of SMA TMS were also found in this study which can be due to the transient nature of TMS. TMS to the SMA during novel reaching did not have as much of a disruptive impact on reaching trajectories when compared to other regions explored in this thesis (such as the left M1 and left PPC) which could have been be due to the pre-SMA being stimulated as opposed to the SMA-proper. This finding could be further determined by employing structural MRI scans for subjects and locating sub-regions of the SMA for stimulation which have previously been conducted using different MRI co-ordinate systems. Exploring whether TMS delivered to other regions in the motor circuitry result in similar outcomes can be important in developing a cortical network model of brain structures that are specific for novel reaching.

Chapter 8

8 Experiment 5

The impact of SP-TMS to the right primary motor cortex (M1) during right arm reaching in a novel motor learning protocol

8.1 Introduction

To move voluntarily, a region of the brain called the primary motor cortex (M1) sends complex descending signals via the corticospinal tract to motor neurons that innervate skeletal muscles (Drew, Prentice and Schepens, 2004). The crossing of the corticospinal pathway has resulted the right side of our brain being responsible for moving the left side of our body and vice versa (Sun and Walsh, 2006). This has been demonstrated experimentally with TMS to the M1 of one hemisphere and by recording MEPs (Hallett, 2007).

Although contralateral motor activation patterns occur during unilateral movements, (shown both experimentally with TMS and in imaging studies) (Barber *et al.*, 2011; Hardwick *et al.*, 2013), researchers have shown lateralisation between the two hemispheres (Barber *et al.*, 2011). For example, findings from Kim *et al.*, (1993) concluded that although right handed and left handed finger movements resulted in similar left M1 cortical activation patterns, cortical activity in the right M1 was only noted during left-handed finger movement. Further hemispheric asymmetries were noted by Verstynen *et al.*, (2005) who reported greater cortical activity in the left hemisphere during left hand movements compared to the right hemisphere during right hand movements.

Motor network connectivity enables the transfer of information between different cortical areas and this facilitates synaptic plasticity during motor adaptation (Shannon *et al.*, 2016). Resting-MRI studies have found an association between enhanced motor performance and increased left-hemispheric motor network connectivity, as opposed to right-hemispheric motor connectivity in right handed subjects (Barber *et al.*, 2012). Voxel based connectivity techniques have been used to further illustrate hemispheric connectivity differences. For example Buckner *et al.*, (2011) reported a greater number of voxels in the right-cerebellar hemisphere which were connected to the left M1, in contrast to the quantity of voxels that were found in the left cerebellar-hemisphere which were connected to the right M1 (Schlerf *et al.*, 2014). Generally, the right M1 has

been found to exhibit less motor network connectivity than the left M1 (Guye *et al.*, 2003). Considering that hemispheric connectivity differences have been established between the right and left M1, this experiment employed SP-TMS as a virtual disruption tool to test the hypothesis that that TMS to the right M1 may not have a detrimental impact on reaching trajectories, because less network communication processes may be interrupted.

Furthermore, this experiment employed a motor adaptation paradigm and a single-pulse TMS protocol to explore the functional role of right M1 in novel reaching. This differs from previous studies that have implemented visuo-motor adaptation tasks and paired-pulse TMS protocols (Schlerf *et al.*, 2014). In this theses, in experiment 2, TMS to the left M1 was found to have a disruptive impact on right arm reaching, however, whether this is the case for TMS delivered to the right M1 is yet to be established. Considering that asymmetries between the two hemispheres have been found, and the connectivity of right motor regions are not as prominent as the left M1, particularly during right hand movements (Cramer *et al.*, 1999; Verstynen *et al.*, 2005; Schlerf *et al.*, 2014), it can be argued that TMS to the right M1 may not have as much of a disruptive impact that was noted in the left M1 during motor adaptation. This study therefore set out to explore whether the asymmetries that have been noted in the literature, effect the process of motor adaptation, and whether TMS has a disruptive effect on the right M1 during ipsilateral reaching.

8.2 Methodology

Participant demographics for this experiment are shown in table 3.2. The ways in which RMT was identified for right M1 stimulation is described in section 3.3.1 of the general methodology chapter. The location for right M1 TMS and the coil orientation is highlighted in figure 3.3. Additionally, the reaching paradigm for this experiment is outlined in section 3.4.1 and figure 3.5.

8.3 Data acquisition: MEPs and Kinematics

No MEPs were measured from the right arm in this experiment and data acquisition was based on kinematics only. Kinematic data were analysed and quantified offline in MatLab 2017b (The MathWorks Inc, Natick MA, USA) (section 3.6.1).

8.4 Statistical analysis: Kinematics

A two-way RMANOVA for each kinematic variable was performed in SPSS 23 (IBM). The main factors (TIME and CONDITION) and interactions explored (TIME*CONDITION), as well as the ways in which sphericity assumptions were tested, and how post-hoc testing was were carried out are explained in section 3.7.1.

8.5 Results

8.5.1 Kinematics: Repeated measures ANOVA and post-hoc testing:

The RMANOVA revealed a significant effect for TIME on movement onset and offset (all F > 4.342, all p < 0.05) but not for maximum velocity, movement duration, summed error and maximum force (all F < 0.849, all p > 0.05) (table 8.1). There was also a significant main effect of CONDITION for movement duration, summed error and maximum force (all F > 5.635, all p < 0.05), but not for movement onset, movement offset and maximum velocity (all F < 0.802, p > 0.05) (table 8.1). No significant interaction effects were found for any of the kinematic variables (all F < 0.517, all p < 0.05, table 8.1).

Post-hoc testing for TIME (see table 8.2, figure 8.2[A-E]) for movement onset showed that during FAM reaching all time points, apart from 100ms were significantly increased compared to T10. During FF reaching all time points were significantly increased compared to T10 (all p < 0.006), but in WO reaching only responses during 280ms and 310ms were significantly increased from T10 (p < 0.006). For movement offset, all time points apart from 190ms and 220ms were significantly increased compared to T10 during FF reaching (p < 0.006). However, there were no significant differences from T10 in movement offset during FAM and WO reaching (all p > 0.006). Post-hoc testing for CONDITION (see table 8.2, figure 8.1[A-E]) showed that movement duration was significantly increased in FF reaching compared to FAM reaching (only at 130ms p < 0.016). Movement duration during FF reaching was significantly longer compared to WO reaching (at 130ms, 160ms, 250ms, 280ms and 310ms, p < 0.016). There were no significant differences regarding FAM vs. WO reaching (all p > 0.016). Summed error was significantly increased in FF vs. FAM reaching at all time points (p < 0.016). Trajectory errors were significantly increased (160ms, 190ms, 20ms, 280ms and 310ms) during FF vs. WO reaching. There were no significant differences in FAM vs. WO reaching. Maximum force was significantly

increased in FF reaching at all time points compared to both FAM and WO reaching as expected.

All post-hoc testing results are illustrated in table 8.2 and figure 8.1[A-E].

| | TIME | | | (| CONDITION | | TIME *CONDITION INTERACTION | | | |
|---------------------------------------|-------------------|-------|---------|-------------------|-----------|---------|-----------------------------|-------|-------|--|
| | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | |
| Movement Onset (ms) | 8[96] | 12.27 | < 0.001 | 2 [24] | 1.696 | 0.213 | 16[192] | 1.010 | 0.447 | |
| Movement Offset (ms) | 8[96] | 4.342 | < 0.001 | 2 [24] | 2.200 | 0.133 | 16[192] | 0.834 | 0.645 | |
| Maximum Velocity (m.s ⁻¹) | 8[96] | 1.526 | 0.158 | 2 [24] | 0.802 | 0.460 | 16[192] | 1.179 | 0.288 | |
| Duration (ms) | 8[96] | 0.873 | 0.542 | 2 [24] | 5.635 | 0.010 | 16[192] | 1.093 | 0.364 | |
| Summed Error (distance: cm) | 3.2 [38.1] | 0.849 | 0.481 | 1.2 [13.4] | 12.92 | 0.003 | 16[192] | 0.944 | 0.520 | |
| Maximum Force (N) | 8[96] | 0.972 | 0.463 | 1.1 [12.6] | 176.78 | < 0.001 | 16[192] | 0.517 | 0.936 | |

Table 8.1. Results of the two-way RMANOVA:

Table 8.2. Post-hoc testing results: The table below shows kinematic responses during FAM, FF and WO reaching when TMS was delivered to the right M1 at different time points during reaching preparation. Values represent means, \pm the standard error mean (SEM).

| TIME | TMS: 10ms | TMS: 100ms | TMS: 130ms | TMS: 160ms | TMS: 190ms | TMS: 220ms | TMS: 250ms | TMS: 280ms | TMS: 310ms | | | | |
|-----------|-----------------------|----------------------|--------------|----------------|----------------------|----------------------|---------------|----------------|----------------|--|--|--|--|
| | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | | | | |
| CONDITION | | Movement Onset (ms): | | | | | | | | | | | |
| FAM | 387 [19] | 413 [16] | 416 [19] * | 429 [14] * | 445 [23] * | 445 [22] * | 445 [22] * | 444 [23] * | 444 [23] * | | | | |
| FF | 381 [15] | 417[15] * | 430 [18] * | 424 [13] * | 423 [14] * | 430 [18] * | 431 [17] * | 437 [14] * | 439 [16] * | | | | |
| WO | 405 [20] | 423 [21] | 441 [16] | 445 [26] | 434 [21] | 443 [20] | 443 [19] | 464 [25] * | 464[24] * | | | | |
| | Movement Offset (ms): | | | | | | | | | | | | |
| FAM | 1215 [26] | 1223 [30] | 1228 [33] | 1249 [37] | 1239 [33] | 1254 [33] | 1251 [36] | 1243 [36] | 1250 [37] | | | | |
| FF | 1216 [17] | 1272 [22] * | 1278 [25] * | 1272 [22] * | 1263 [21] | 1269 [19] | 1274 [24] * | 1291 [24] * | 1296 [23] * | | | | |
| WO | 1225 [25] | 1223 [25] | 1228 [24] | 1245 [31] | 1238 [24] | 1246 [29] | 1233 [24] | 1249 [28] | 1263 [28] | | | | |
| | | | | Ν | Maximum Velocity (n | n.s ⁻¹): | | | | | | | |
| FAM | 0.29 [0.01] | 0.29 [0.02] | 0.29 [0.02] | 0.28 [0.02] | 0.30 [0.02] | 0.29 [0.02] | 0.29 [0.02] | 0.29 [0.02] | 0.29 [0.02] | | | | |
| FF | 0.30 [0.01] | 0.30 [0.01] | 0.30 [0.02] | 0.30 [0.01] | 0.30 [0.02] | 0.30 [0.01] | 0.30 [0.02] | 0.29 [0.01] | 0.30 [0.01] | | | | |
| WO | 0.28 [0.01] | 0.29 [0.01] | 0.29 [0.01] | 0.29 [0.01] | 0.29 [0.01] | 0.29 [0.01] | 0.30 [0.01] | 0.29 [0.01] | 0.29 [0.01] | | | | |
| | | | |] | Movement Duration (| (ms): | | | | | | | |
| FAM | 827 [27] | 810 [27] | 812 [33] | 820 [36] | 804 [31] | 809 [30] | 805 [31] | 799 [26] | 806 [26] | | | | |
| FF | 835 [21] | 855 [24] | 849 [24] ■✦ | 848 [21] + | 841 [20] | 840 [16] | 843 [22] 🕇 | 855 [23] + | 858 [16] 🕇 | | | | |
| WO | 820 [20] | 800 [19] | 787 [19] | 800 [21] | 803 [16] | 803 [22] | 791 [17] | 785 [19] | 800 [21] | | | | |
| | | | | Su | immed Error (distanc | e: cm): | | · | | | | | |
| FAM | 2.17 [0.19] | 2.35 [0.27] | 2.15 [0.21] | 1.91 [0.18] | 1.93[0.14] | 2.13[0.15] | 2.02 [0.16] | 1.90 [0.18] | 2.00 [0.16] | | | | |
| FF | 3.44 [0.44] 🔳 | 3.86 [0.59] | 3.44 [0.44] | 3.99 [0.69] ■+ | 3.06 [0.34] ■+ | 3.79 [0.47] ■+ | 3.81 [0.60] | 3.83 [0.63] ■+ | 3.59 [0.41] ■+ | | | | |
| WO | 2.24 [0.18] | 2.22 [0.17] | 2.21[0.12] | 1.99 [0.09] | 2.20 [0.14] | 2.08 [0.10] | 2.26[0.13] | 2.14[0.14] | 2.21[0.12] | | | | |
| | | | | | Force (N): | | | | | | | | |
| FAM | 4.7 [0.1] | 4.7 [0.2] | 4.8 [0.2] | 4.7 [0.2] | 4.8 [0.2] | 4.8 [0.2] | 4.8 [0.2] | 4.8 [0.2] | 4.8 [0.2] | | | | |
| FF | 9.9 [0.4] ■✦ | 9.8 [0.4] ■✦ | 9.9 [0.4] ■✦ | 9.8 [0.3] ■+ | 10.0 [0.4] ■+ | 9.9 [0.3] ■✦ | 10.0 [0.4] ■+ | 9.9 [0.4] ■✦ | 9.9 [0.3] ■+ | | | | |
| WO | 4.7 [0.1] | 4.8 [0.2] | 4.8 [0.2] | 4.8 [0.2] | 4.8 [0.2] | 4.8 [0.2] | 4.8 [0.2] | 4.7 [0.1] | 4.8 [0.2] | | | | |

Symbols represent significance following post hoc testing:

* = significant difference vs T10 for time in familiarisation blocks, * = significant difference vs T10 for time for FF blocks, * = significant difference vs T10 for time for WO blocks, = significant difference between FAM and FF, \neq = significant difference between FF and WO, \blacktriangle = significant difference between FAM and WO



Figure 8.1 [A-E]: Graphical presentation of the post-hoc testing results (based on table 8.2) for the significant kinematic variables:



Symbols represent significance following post hoc testing:

- * = significant difference vs T10 for time in familiarisation blocks, * = significant difference vs T10 for time for FF blocks, * = significant difference vs T10 for time for WO blocks,
 - \blacksquare = significant difference between FAM and FF, \blacklozenge = significant difference between FF and WO, \blacktriangle = significant difference between FAM and WO

8.6 Discussion

As the left M1 was stimulated in chapter 5 of this thesis, the main aim of this chapter was to explore the effects of TMS delivered to the right M1 during right arm reaching preparation. Previous experiments investigating left and right hemispheric differences have implemented visuo-adaptation reaching protocols as opposed to robotic adaptation protocols and have explored other kinematic variables such as reach amplitude and the travel distance time to reach a target (Schlerf *et al.*, 2014). This study therefore aimed to provide novel findings regarding right M1 function using a robot-mediated FF paradigm and a single-pulse TMS protocol.

8.6.1 Kinematics

8.6.1.2 The shift in movement onset and offset:

Results from this experiment revealed a significant delay in movement onset (FAM, FF and WO) and offset (FF only) responses (see table 8.2). This finding therefore suggests a functional role of the ipsilateral M1 during motor preparation for unilateral reaching. This result is in line with findings from non-human primates, where for example neuronal unit activity recordings via implanted electrodes in the motor cortex of Rhesus monkeys have illustrated correlations between neuronal activity in the ipsilateral M1, and movement preparation and motor output (Ganguly *et al.*, 2009).

The delay that occurred in movement onset and offset can be explained by the 'waiting period' phenomena which results in cortical changes (see section 5.6.2) as described by Hasegawa *et al.*, (2017). More specifically, motor preparation resulted in inhibition of neural networks which was accompanied by a selective inhibition of neural networks mediated by interneurons (Pfeffer *et al.*, 2013). This result demonstrated that there may be similar mechanisms occurring in the right and left M1 because the shifts that occurred with regards to movement onset in relation to the 'waiting period' theory was also noted during left M1 stimulation. This finding was important because it highlighted the link between motor behaviours and the M1 circuitry, in particular how the neural networks can have an influence reaching preparation.

8.6.1.3 Right M1 connectivity in facilitating novel reaching:

TMS delivered to the right M1 had a significantly disruptive effect on reaching trajectories during FF reaching, compared to FAM and WO reaching.

TMS virtual lesion studies have mainly reported that in right handed subjects, there are greater disruptions in motor output when TMS is delivered to the left M1 compared to when it is delivered to the right M1 (Van den Berg, Swinnen and Wenderoth, 2011). However, in this study we demonstrated that disruptions are also caused when TMS is administered to the right M1, thus illustrating that the ipsilateral M1 also undergoes modulations of activity during right arm reaching. This view has been supported with studies in stroke patients. For example, it has been reported that following lesions to one hemisphere, the contra-lesional hemisphere becomes important in aiding ipsilateral movements (Dancause et al., 2006; Hummel and Cohen, 2006; Ganguly et al., 2009). Findings from an fMRI study conducted by Grefkes et al., (2008) showed that when stroke patients made hand movements (the affected hand), the contra-lesional hemisphere (unaffected) influenced neural activity of the ipsi-lesional motor cortex. The left M1 is not the only region that facilitates neuronal activity of the right M1, and other areas such as the posterior parietal cortices also play a role. For example, using a twincoil TMS protocol, Koch et al., (2008b) reported that the right PPC influences the right M1 when participants performed a leftward reach. Similarly in this study, participants were instructed to reach towards a target that was presented on the left side of the screen (north-west - 135°). Based on these findings, it could be argued that TMS to the right M1 caused a disruption in neural network communication (i.e. between the PPC and right M1, or left M1 and right M1) which affected the final output of reaching (i.e. increase in summed error).

Brain machine/computer interfaces (BMI/BCI) are based on measuring activity (i.e. neural) and converting it to an artificial output to replace or restore behaviours, this can be done either invasively (i.e. via implants) or non-invasively (i.e. via neural signals) (Wolpaw *et al.*, 2002; Daly and Huggins, 2015). Therefore, signals can be used to control functions in patients (McFarland and Vaughan, 2016) and can provide the brain with channels that depend on cortical activity to perform actions (Ranjangam *et al.*, 2016). The use of BMIs has been successful in assisting cortical recovery in stroke patients (Carmena *et al.*, 2003; Daly *et al.*, 2009; Ang *et al.*, 2010; Daly and Huggins 2015). For example in motor learning following stroke, physical therapy training combined with EEG-BMI training vs. physical therapy training alone resulted in enhanced motor performance (quantified with Fugl-Meyer scores) (Ramos-Murguialday *et al.*, 2013). In this experiment, summed error results demonstrated the importance of

ipsilateral arm-reaching neural responses in novel motor learning. The data in this study could be important in brain machine interfaces and restoring "functional cortical reorganisation" in patients who have had a severe stoke (Carmena *et al.*, 2003). In other words, the right M1 can be used as a site for extracting signals to use for BMIs (as previously emphasised with the motor cortex – Schroeder and Chestek, 2016; Hatsopoulos and Suminksi, 2011) because signals from this region may be better suited for assisting novel reaching compared to signals from other regions that are damaged (Friehs *et al.*, 2004).

In this study, maximum force was greater in FF reaching as expected (table 8.2). However, maximum velocity was not modulated, which was not the case in experiment 2 with TMS to the left M1 (maximum velocity was significantly increased in FF reaching). This can be because of the lateralisation of function between the left and right M1, which has been illustrated in both human subjects, and non-human primates. For example, Ganguly *et al.*, (2009) carried out invasive motor unit recordings (electrode implantation) in macaque monkeys and found that the contralateral M1 is more directly associated with movement velocity and has a greater influence in coding velocity parameters during reaching compared to the ipsilateral M1.

No right or left upper limb MEPs were collected were collected in this experiment. It would be interesting in future studies to collect MEPs to provide a more detailed insight into the role of ipsilateral corticospinal projections to the right arm during motor adaptation. This might also have impact for future design of contralesional BMI design (see above). Furthermore, testing left handed participants during right M1 stimulation may be warranted because studies have reported the importance of hand dominance in effecting M1 hemispheric asymmetries during ipsilateral movement (Van den Berg, Swinnen and Wenderoth, 2011). Given the possible role of right M1 in right arm reaching, it would be interesting to study IHI between left-right and right-left M1s during preparation for reaching in motor adaptation as this "balance" of inhibition is thought to be important in recovery from stroke.

8.7 Chapter conclusions:

This chapter demonstrated that TMS delivered to the right M1 at different time points during reaching preparation had a significant impact on movement onset and movement offset, particularly during novel reaching. This finding can be attributed to the 'waiting

period' phenomena, which was similarly noted during left M1 stimulation. Based on this finding, it could be argued that the right and left M1 undergo similar mechanistic processes when TMS is delivered during the preparation of a novel reach, and this could have been a result of the contralateral M1 exerting an influence on the ipsilateral M1 (which has been shown in stroke patient studies). The data in this study demonstrated the possible role of the ipsilateral M1 in novel right-arm reaching, whether it be direct or indirect via the left M1 connections, which can be important finding in the field of BMIs in terms of assisting functional recovery in stroke patients. Overall this experiment highlighted the functional role of right M1 in motor preparation and the importance of motor connectivity in facilitating preparation for novel reaching. Due to the interesting findings that were illustrated between the right and left M1 (chapter 5), additional right and left hemispheric regions were targeted with TMS in the final chapters of this thesis (e.g. the right and left dPMC, and the right and left PPC) to explore hemispheric asymmetries and similarities between regions during novel reaching.

Chapter 9

9 Experiment 6

Exploring impact of SP-TMS to the left dorsal pre-motor cortex (dPMC) during right arm reaching in a novel motor learning protocol

9.1 Introduction

The premotor cortex (PMC) is situated between the dorsal prefrontal cortex and the primary motor cortex (Kantak *et al.*, 2012) and has a vast range of functions that contribute to motor performance including movement control, skilled performance and reaching (Halsband *et al.*, 1993; Cao *et al.*, 2013; Kantak et al., 2012). The PMC is key for sequential motor learning, as lesions to this region has led to poor performance in motor sequencing tasks (Tranel *et al.*, 2003; Gross and Grossman 2008; Ohbayashi, Picard and Strick, 2016; Solopchuk, Alamia and Zenon 2016).

The PMC has two sub-structures; the ventral PMC (vPMC) and the dorsal PMC (dPMC). As well as its role in language production and comprehension, the vPMC is also involved in spatial perception and motor imagery (Rizzolatti, Fogassi and Gallese, 2002; Binkofski and Buccino, 2006). For example, Binkofski *et al.*, (2000) conducted an fMRI study which illustrated neural activations of ventral regions of the PMC when participants were asked to imagine making finger movements. The dPMC on the other hand, plays a role in cued response selection and goal directed behaviour which may be important for motor adaptation (Beck et al., 2009; Yamagata *et al.*, 2012). For example, dPMC neuronal activity was tuned with target location and arm use during the preparation phase of a reaching action (Tanji and Hoshi, 2001).

Generally, movements are prepared via the use of cues and waiting for information before preparing an action (Dekleva, Kording and Miller, 2018). Visual cues are vital in aiding dPMC motor functions and this has been noted in studies of non-human primates whereby dPMC resection resulted in the inability to use cues to initiate or suppress movements (Petrides 1982; Halsband and Passingham, 1985; Chouinard, Leonard and Paus, 2005). With regards to motor output behaviour, the dPMC has also been noted to be involved in encoding kinematic parameters that help in the formation of motor memories (Overduin, Richardson, and Bizzi, 2009; Meehan *et al.*, 2011; Meehan *et al.*, 2013).

Considering that this thesis will focus on the dPMC, and target both the left and right dPMC, it is important to take into account whether differences have been demonstrated between the two during reaching and adaptation. In motor paradigms where there is a change in the motor task, action-reprogramming is necessary and the left dPMC in particular has been noted to be involved in this (Hartwigsen *et al.*, 2012). For example, experiments have shown that the left dPMC plays a role in immediately abandoning previously prepared motor action behaviours and replacing them with new ones, in order to accurately meet the demands of a task (Hartwigsen *et al.*, 2012). The left dPMC contributes to reaching behaviours because cells within its region transmit signals that are associated with the visual control of reaching (Cisek and Kalaska, 2002; Lee and Van Donkelaar, 2006), therefore using TMS to target the dPMC could be important in further demonstrating its functional role during motor reaching adaptation (Clower *et al.*, 1996; Lee and Van Donkelaar, 2006).

Thus far, different TMS protocols have been used to explore dPMC function. For example, paired pulse TMS to the left dPMC led to an increase in the decision making time of movement onsets (Mochizuki *et al.*, 2005). However, this was not the case when the right dPMC was stimulated (Mochizuki *et al.*, 2005). In a visual perturbation prism adaptation task, Lee and Van Donkelaar (2006) targeted the left dPMC and found that it did not significantly contribute in trial-to-trail learning. On the other hand, TMS to the left dPMC could disrupt adaptation performance depending on the time point at which it is delivered. For example, slower adaptation rates were found when TMS was administered at movement onset, compared to when it was delivered at movement offset (Lee and Van Donkelaar, 2006). An additional important factor to consider is that dPMC function could depend upon handedness. For example, neuron activation recordings from non-human primates in bimanual and unimanual tasks demonstrated greater right dPMC activation was found when actions and tasks were performed with two hands (Kermadi, Liu and Roullier, 2000).

Currently the impact of single pulse TMS to the left dPMC during right arm reaching in a motor adaptation task is unknown as most research has focused on:

 Studies in non-human primates rather than human subjects (Overduin, Richardson, and Bizzi, 2009). Visuo-motor adaptation paradigms compared to robot induced perturbations (Lee and Van Donkelaar, 2006).

Based on findings that have illustrated the role of the left dPMC in motor memory and response selection it can be argued that in this study TMS could disrupt the left dPMC functions in a motor adaptation task.

9.2 Methodology

The details for participant demographics regarding left dPMC stimulation are shown in table 3.2. Prior to stimulation during reaching, functional measures were implemented to identify RMT for each participant (section 3.2.2.2 details this). Section 3.3.2.2.3 and figure 3.3 outlines the location and coil orientation chosen for left dPMC stimulation (based on protocols from Fink et al., 1997, Münchau *et al.*, 2002, Lee and Van Donkelaar, 2006; Zanon et al., 2013 and Lega et al., 2016). The reaching task for this experiment is explained in section 3.4.1 and graphically shown in figure 3.5.

9.3 Data acquisition: MEPs and Kinematics

MEPs were only elicited when identifying the participants RMT (section 3.6.2). No MEPs were collected during left dPMC stimulation. Kinematics were of main interest in this study, and section 3.6.1 provides details with regards to how kinematic data were analysed and quantified offline in MatLab 2017b (The MathWorks Inc, Natick MA, USA).

9.4 Statistical analysis: Kinematics

A two-way RMANOVA was performed in SPSS 23 (IBM) for each kinematic variable. Factors such as TIME, CONDITION and TIME*CONDITION were investigated. Section 3.7.1 describes the ways in which sphericity assumptions were tested and how post-hoc testing was performed (paired Student's t-test with Bonferroni correction for multiple comparisons).

9.5 Results

9.5.1 Kinematics:

The RMANOVA revealed a significant main effect for TIME on movement onset and movement offset (p < 0.05), but not for maximum velocity, movement duration, summed error and maximum force (p > 0.05) (see table 9.1). There was also a significant main effect of CONDITION on summed error and maximum force (p < 0.05), but not for movement onset, movement offset, maximum velocity and movement

duration (p > 0.05) (table 9.1). A significant interaction was found for movement onset (p < 0.05), but this was not the case for movement offset, maximum velocity, summed error, movement duration and maximum force (p > 0.05) (table 9.1).

Post hoc testing for TIME (table 9.2, figure 9.1[A-D]) for movement onset showed that during FAM reaching responses were significantly increased at all time points (p < 0.006) apart from 160ms (p > 0.006) compared to T10. In FF reaching, responses at 100ms, 190ms, 220ms, 250ms, 280ms and 310ms were significantly increased compared to T10 (p < 0.006). Movement onset during WO reaching was significantly increased at 100ms, 160ms, 190ms, 220ms, 250ms, 250ms, 280 and 310ms compared to T10 (p < 0.006). Movement offset was significantly increased at all time points compared to T10 (p < 0.006). Movement offset was significantly increased at all time points compared to T10 during FAM reaching (p < 0.006). During FF reaching, only TMS delivered at 100ms, 190ms, 220ms and 310ms was significantly increased compared to T10. WO reaching responses were significantly increased at 160ms and 280ms in contrast to T10 (p < 0.006).

Post-hoc testing for CONDITION (table 9.2, figure 9.1[A-D]) regarding summed error revealed that errors in reaching were significantly increased in FF reaching vs. FAM reaching (at all time points p < 0.016). Summed error in FF reaching was significantly increased compared to WO reaching (at 100ms, 160ms, 190ms, 220ms, 280ms and 310ms p < 0.016). Summed error was significantly increased in WO vs. FAM reaching (only at 160ms p < 0.016). Maximum force was significantly increased in FF reaching as expected (p < 0.016), compared to FAM and WO reaching, which in turn were similar.

Table 9.2 and figure 9.1[A-D] illustrates the results from post-hoc testing.

| | TIME | | | С | ONDITION | | TIME*CONDITION INTERACTION | | | |
|---------------------------------------|-------------------|-------|---------|-------------------|----------|---------|----------------------------|-------|-------|--|
| | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | |
| Movement Onset (ms) | 8 [96] | 17.20 | < 0.001 | 1.4 [16.2] | 0.606 | 0.554 | 16 [192] | 2.032 | 0.013 | |
| Movement Offset (ms) | 8 [96] | 10.64 | < 0.001 | 2 [24] | 1.543 | 0.234 | 16 [192] | 1.675 | 0.054 | |
| Maximum Velocity (m.s ⁻¹) | 2.9 [34.7] | 0.247 | 0.856 | 2 [24] | 1.129 | 0.340 | 16 [192] | 0.358 | 0.990 | |
| Duration (ms) | 4.4 [52.2] | 1.27 | 0.293 | 2 [24] | 3.070 | 0.065 | 16 [192] | 0.887 | 0.585 | |
| Summed Error (distance: cm) | 8 [96] | 0.478 | 0.869 | 2 [24] | 24.80 | < 0.001 | 16 [192] | 0.936 | 0.529 | |
| Maximum Force (N) | 1.4 [15.8] | 0.124 | 0.769 | 1.2 [13.3] | 280.50 | < 0.001 | 16 [192] | 0.179 | 0.100 | |

Table 9.1. Results of the two-way RMANOVA:

Table 9.2. Post-hoc testing results:

The table shows kinematic responses during FAM, FF and WO blocks of reaching when TMS was applied to the left dPMC at different time points. Values represent means and standard error means.

| TIME | TMS: 10ms | TMS: 100ms | TMS: 130ms | TMS: 160ms | TMS: 190ms | TMS: 220ms | TMS: 250ms | TMS: 280ms | TMS: 310ms | | |
|-----------|-----------------------|-------------|--------------|-------------------------------------|--------------------------------|-----------------|--------------|--------------|--------------|--|--|
| | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | | |
| CONDITION | | | | Mov | vement Onset (ms): | | | | | | |
| FAM | 379 [17] | 419 [21] * | 418 [17] * | 431 [21] | 434 [20] * | 436 [20] * | 449 [24] * | 446 [23] * | 458 [25] * | | |
| FF | 383 [12] | 411 [12] * | 420 [12] | 410 [11] | 425 [12] * | 443 [16] * | 443 [15] * | 419 [14] | 436 [15] * | | |
| WO | 382 [15] | 410 [12] * | 413 [16] | 436 [19] * | 442 [16] * | 436 [13] * | 439 [16] * | 472 [17] * | 442 [17] * | | |
| | Movement Offset (ms): | | | | | | | | | | |
| FAM | 1188 [29] | 1236 [23] * | 1244 [33] * | 1255 [33] * | 1254 [29] * | 1254 [30] * | 1270 [29] * | 1272 [31] * | 1263 [37] * | | |
| FF | 1219 [18] | 1259 [19] * | 1276 [18] | 1250 [16] | 1282 [23] * | 1294 [21] * | 1275 [24] | 1248 [19] | 1289 [21] * | | |
| WO | 1188 [24] | 1223 [19] | 1222 [25] | 1249 [29] * | 1262 [27] | 1235 [22] | 1254 [32] | 1273 [26] * | 1241 [17] | | |
| | | | | Maxin | num Velocity (m.s ⁻ | ¹): | | | | | |
| FAM | 0.29 [0.02] | 0.29 [0.02] | 0.29 [0.02] | 0.29 [0.02] | 0.29 [0.02] | 0.29 [0.02] | 0.29 [0.02] | 0.29 [0.01] | 0.29 [0.02] | | |
| FF | 0.30 [0.01] | 0.30 [0.01] | 0.30 [0.01] | 0.30 [0.01] | 0.30 [0.02] | 0.30 [0.01] | 0.30 [0.01] | 0.30 [0.01] | 0.30 [0.01] | | |
| WO | 0.29 [0.01] | 0.29 [0.01] | 0.30 [0.02] | 0.29 [0.01] | 0.29 [0.01] | 0.29 [0.01] | 0.29 [0.02] | 0.30 [0.01] | 0.29 [0.01] | | |
| | | • | | Move | ement Duration (ms |): | | | | | |
| FAM | 809 [28] | 818 [26] | 826 [32] | 824 [28] | 816 [29] | 818 [29] | 820 [27] | 825 [26] | 804 [29] | | |
| FF | 835 [21] | 848 [22] | 856 [18] | 840 [15] | 858 [25] | 851 [15] | 833 [20] | 829 [21] | 853 [20] | | |
| WO | 805 [25] | 813 [21] | 809 [28] | 813 [27] | 819 [31] | 799 [27] | 815 [29] | 800 [24] | 803 [23] | | |
| | | • | | Summe | d Error (distance: c | em): | | | | | |
| FAM | 1.99 [0.09] | 1.86 [0.12] | 1.91 [0.16] | 1 92 [0 10] | 2.13 [0.12] | 2.06 [0.16] | 1.82 [0.09] | 2.01 [0.12] | 2.05 [0.16] | | |
| FF | 3.04 [0.26] | 3.55 [0.39] | 3.41 [0.33] | $3.41 [0.35] \blacksquare \bigstar$ | 3.40 [0.26] | 3.38 [0.30] | 3.13 [0.25] | 3.24 [0.36] | 3.23 [0.31] | | |
| WO | 2.25 [0.29] | 2.25 [0.24] | 2.29 [0.16] | 2.50 [0.22] | 2.22 [0.29] | 2.13 [0.17] | 2.59 [0.32] | 2.15 [0.21] | 2.19 [0.23] | | |
| | | • | | | Force (N): | | | | | | |
| FAM | 4.7 [0.2] | 4.7 [0.2] | 4.7 [0.2] | 4.7 [0.2] | 4.7 [0.2] | 4.7 [0.2] | 4.8 [0.2] | 4.8 [0.2] | 4.7 [0.2] | | |
| FF | 9.9 [0.4] ■+ | 10.0 [0.4] | 9.9 [0.3] ■+ | 10.0 [0.4] | 10.0 [0.4] | 9.9 [0.4] ■+ | 9.9 [0.3] ■+ | 9.9 [0.4] ■+ | 9.8 [0.3] ■+ | | |
| WO | 4.7 [0.1] | 4.7 [0.1] | 4.8 [0.1] | 4.7 [0.1] | 4.7 [0.1] | 4.7 [0.1] | 4.7 [0.1] | 4.7 [0.1] | 4.7 [0.1] | | |



Figure 9.1 [A-D]. Graphical presentation of the post-hoc testing results (based on table 9.2) for the significant kinematic variables:

Symbols represent significance following post hoc testing:

- * = significant difference vs T10 for time in familiarisation blocks, * = significant difference vs T10 for time for FF blocks, * = significant difference vs T10 for time for WO blocks, \blacksquare = significant difference between FAM and FF, + = significant difference between FF and WO, \blacktriangle = significant difference between FAM and WO

9.6 Discussion:

The purpose of this chapter was to explore the effects of TMS delivered to the left dPMC during reaching preparation. Previous experiments investigating the role of the dPMC regarding motor function have implemented visuo-adaptation pointing paradigms (Lee and Van Donkelaar, 2006) and have also explored non-human primate neural responses of reaching compared to human subjects (Kermadi, Liu and Roullier 2000; Overduin, Richardson and Bizzi, 2009). This experiment therefore aimed to provide novel findings regarding left dPMC function in human subjects with single pulse TMS in a robotic-mediated upper-limb reaching FF paradigm.

9.6.1 Kinematics

9.6.1 The role of the dPMC in reaching preparation:

The findings from this experiment revealed that TMS caused a significant shift in movement onset and offset responses (FAM, FF and WO) at different time points (see table 9.2). Similarly, repetitive TMS studies have illustrated delayed responses during finger action-selection tasks when the left dPMC was targeted (Hartwigsen *et al.*, 2012). This particular result is important because it illustrates the role of the left dPMC in reaching preparation. For example, studies have demonstrated heightened neural responses in the dPMC during motor preparation when selecting actions for execution (Boussaoud and Wise, 1993; Hoshi and Tanji, 2006; Hartwigsen *et al.*, 2012).

The nature of the reaching task in this experiment can be a reason as to why preparatory action was affected during left dPMC stimulation. This is because activity in dPMC neurons have been noted when reaching tasks were based on target-location and arm-use (Hoshi and Tanji, 2000; Hoshi and Tanji, 2006) (which were two factors in this experiment; i.e. reaching towards a 135° target and using the upper-limb right arm to do so). Studies that have explored functional differences between the dPMC and vPMC have also noted greater dPMC neuronal activity during movement execution towards targets compared to greater vPMC neuronal activity during visual processing of information when reaching towards targets (Boussaoud and Wise, 1993; Hoshi and Tanji, 2002).

The neural functions of the dPMC facilitates its role in reaching preparation. For example, neurons in the dPMC exhibit sustained activity during what is known as a 'motor-set period' (Hoshi and Tanji, 2002). The motor-set period is a phase in which

subjects prepare themselves for a movement-trigger in order to begin arm-reaching in time (Hoshi and Tanji, 2002). This sustained neuronal activity enables the dPMC to respond to upcoming movements (Johnson et al., 1996; Hoshi and Tanji 2002; Messier and Kalaska, 2000). In this experiment it could be argued that TMS disrupted the motor-set period which resulted in delayed movement responses (onset and offset) during reaching preparation. The findings obtained from this experiment are important because they describe behavioural left dPMC functions related to reaching preparation which have not previously explored with these protocols.

9.6.2 Disrupted novel reaching with dPMC stimulation:

A key finding from this experiment was that disrupting the left dPMC with SP-TMS during preparation for novel reaching, impaired performance which was revealed by increased summed error trajectory responses. Left dPMC TMS during FF reaching compared to FAM and WO reaching significantly induced trajectory errors in reaching (see table 9.2). Considering that TMS did not cause major disruptions during FAM and WO reaching suggests that the left dPMC has a specific function in novel reaching.

Similar trajectory errors have been noted in prism-adaptation finger-movement paradigms whereby TMS delivered to the left dPMC slowed down adaptation rates and subjects were not able to correct movement trajectories to facilitate successful motor adaptation (Lee and Van Donkelaar, 2006). It has been suggested that corrections in reaching and adaptation depend upon 1) knowing that the error has occurred and 2) the extent of visual feedback is available in a task (Lee and Van Donkelaar, 2006). By remapping the visual representation of the arm, errors induced by perturbations can be minimised (Lee and Van Donkelaar, 2006). The dPMC is involved in reaching because its neurons have been found to transmit signals that are linked to the visual control of movements to the M1 (Cisek and Kalaska 2002; Lee and Van Donkelaar, 2006). The role of other regions that facilitate left dPMC motor functions are also important to take into account. For example, MRI studies have demonstrated a stronger functional link between the left dPMC and the submarginal gyrus following rTMS which was associated with decreases in error rates during a visuo-spatial task (Ward et al., 2010; Hartwigsen et al., 2012). Based on these findings it could be argued that errors in trajectories occurred because TMS disrupted and impaired neural communication between the left dPMC and other cortical regions (e.g. the visual cortex and

submarginal gyrus) that contribute to successful reaching and motor learning Hartwigsen *et al.*, 2012).

For motor output to occur, planning and preparation takes place. However, when there is a change in the environment (e.g. a perturbation) re-programming is essential in order to replace old plans with new plans, and this is known as action re-programming (Chambers *et al.*, 2007; Neubert et al., 2010; Hartwigsen *et al.*, 2012). Although the parietal cortex plays a role in action re-programming (Chambers *et al.*, 2007) other cortical regions are also involved, particularly the left dPMC. The left dPMC has been reported to carry out prompt response mapping in order to provide the appropriate motor output for a task (Christensen *et al.*, 2007; Hartwigsen *et al.*, 2012). Studies have also supported this notion, for example rTMS delivered to the left dPMC resulted in a significant delays in action re-programming (Hartwigsen *et al.*, 2012). When considering this, it could be suggested that in this experiment TMS caused disruptions in novel reaching because it impaired left dPMC action re-programming functions (i.e. the left dPMC was not able to map out new planning actions during perturbed reaching). This study was therefore important in illustrating the ways in which the left dPMC contributes to different motor function processing, in this case – action re-programming.

9.7 Chapter conclusions:

This chapter demonstrated that TMS delivered to the left dPMC at different time points during reaching preparation had a significant impact on movement onset and movement offset during FAM, FF and WO reaching. This finding can be attributed to the functional role of the left dPMC in the planning and preparation of reaching, as well as the type of motor task that was undertaken in this experiment. TMS disrupted the motor-set period whereby the left dPMC could not sustain its motor activity in order to begin arm reaching promptly – thus causing a delay in responses. Interestingly, FF reaching resulted in larger trajectory errors as opposed to FAM and WO reaching, and this could be due to TMS impairing neural communication between the left dPMC and key cortical areas that it is highly connected with that facilitate successful motor adaptation, such as the visual cortex and the submarginal gyrus. As opposed to the other experiments in this thesis whereby movement duration and maximum velocity were affected with TMS, this was not the case for the left dPMC. Therefore, this experiment was able to highlight specific neural processes of the left dPMC using behavioural

output measures. Whether the right dPMC also exhibits similar behavioural outputs can be important to explore as it could help provide explanations into possible asymmetries between the two hemispheres (this region was therefore targeted in experiment 8). Whether similar responses are demonstrated with right PPC stimulation would also be interesting to explore, and this was therefore investigated in the next chapter.

Chapter 10

10 Experiment 7

Exploring the impact of SP-TMS to the right posterior parietal cortex (PPC) during right arm reaching in a novel motor learning protocol

10.1 Introduction

In chapter 6 of this thesis, the left PPC was targeted and findings from the experiment revealed that left PPC function was significantly disrupted with TMS during novel force-field reaching, compared to FAM and WO reaching (see Section 6.6.1). The purpose of this chapter was to explore the functional role of the right PPC and whether the same behavioural findings could be established, or whether the left and right PPC have distinct functions related to novel reaching.

As opposed to the left PPC which has been linked to functions including movement preparation and planning, as well as navigation (Kaas and Stepniewska, 2016; Whitlock, 2017), the right PPC has been associated with functions such as maintaining alertness and spatial attention during a task (Posner and Peterson, 1990; Malhotra, Coulthard and Husain, 2009). Support for the role of the right PPC in spatial-attention comes from patients with right PPC lesions who exhibit spatial neglect (Vallar and Perani, 1986; Husain and Nachev, 2007). Patients with impaired right PPC function are often unaware of contra-lesional stimuli (i.e. objects in the left hemi-field) (Husain and Nachev, 2007) and this differs from patients with left PPC damage who have impairments in action control (i.e. ideomotor apraxia) (Husain and Nachev, 2007). This provides evidence that the two hemispheres do have separate behavioural roles because damage to the left- and right-PPC results in distinct functional impairments. Based on this it could be argued that TMS may have a different impact when it is delivered to the right PPC compared to what was found in the left PPC.

Asymmetries in motor behaviours do exist and can be due to left and right hemispheric differences between cortical regions, including the PPC (Schluter *et al.*, 2001). For example in a visuospatial motor paradigm which was based on cues specifying which finger patients should use to respond to trials in a task, Rushworth *et al.*, (1997) demonstrated that patients with left hemispheric parietal impairments as opposed to those with right hemispheric parietal impairments had slower responses. In this study, TMS will be applied during the preparation phase of a reach and neuroimaging studies
including positron emission tomography (PET) scans have provided an insight into the role of the right PPC with regards to movement preparation. For example, Coull and Nobre (1998) found less neural activity in the right PPC compared to the left intraparietal cortex when participants were required to prepare and execute finger movements during specific times in a task. Coull and Nobre (1998) did however find that the right PPC exhibited increased activity in spatial orientation tasks. Similarly, PET scans during finger-movement reaction time tasks have revealed greater activity in the left PPC during the preparation phase of movement, and this was found in fingermovements of both the left and right hand (Deiber et al., 1996). This shows that the left PPC as opposed to the right PPC is involved in the motor preparation tasks involving both hands (Krams et al., 1998; Rushworth, Krams and Passingham, 2001; Schluter et al., 2001). Furthermore, with regards to reaching tasks, less reaching-activity has been noted in the right PPC compared to the left PPC (Diedrichsen et al., 2006; Oliveira et al., 2010). Such findings therefore illustrate that the left hemisphere has a greater role and is more dominant than the right hemisphere during reaching tasks, specifically during motor preparation for movement execution (Schluter et al., 2001), therefore it could be argued that right PPC stimulation may not cause significant impairments when delivered during reaching preparation.

TMS studies targeting the right PPC have mainly explored its functions in saccadic visual memory tasks as well as visuo-spatial attention task. For example, SP-TMS protocols targeting the right PPC as opposed to the left PPC resulted in impaired visual-memory performance, thus demonstrating its role in visual spatial processing (Prime, Vesia and Crawford, 2008). TMS to the right PPC resulted in a 'virtual disruption' during a visuo-spatial attention task, more specifically TMS impaired neural activity in the task and caused delays in performance (Chambers *et al.*, 2004; Woo, Kim and Lee, 2009). With regards to reaching tasks, twin-coil TMS studies targeting the right PPC and the ipsilateral M1 have found TMS to the right PPC had a facilitatory effect on the ipsilateral M1 during movement planning, and these were also found to be time specific (Koch *et al.*, 2008b) therefore the relationship between different cortical regions is important in facilitating reaching preparation.

Although studies have previously explored the role of the left PPC (Della-Maggiore et al., 2004) and have highlighted its importance in the motor network in terms of facilitating motor preparation for novel reaching (including experiment 6 in this thesis),

the functional role of the right PPC with regards to this is unknown. This is because studies have mainly explored its role in spatial attention using rTMS protocols (as emphasised by Woo, Kim and Lee, 2009). This experiment is therefore the first experimental attempt to provide behavioural findings regarding novel reaching functions of the right PPC with SP-TMS using a robot-mediated upper-limb (right arm) reaching paradigm.

10.2 Methodology

Participant demographic details for this experiment are highlighted in table 3.2. Once RMT was identified for each participant (section 3.2.2.2 and 3.6.2 and details this), right PPC location and coil orientation for was chosen for stimulation, and this is outlined in section 3.3.2.2.4, and illustrated in figure 3.3. Figure 3.5 demonstrates the reaching paradigm for this experiment.

10.3 Data acquisition: MEPs and Kinematics

MEPs were only collected to identify participants RMT (section 3.6.2). Throughout the duration of the reaching experiment, no MEPs were elicited during stimulation. Only kinematic data were acquired and analysed. Section 3.6.1 outlines the ways in which kinematic data were analysed and quantified offline in MatLab 2017b (The MathWorks Inc, Natick MA, USA).

10.4 Statistical analysis: Kinematics

A two-way RMANOVA was performed for each kinematic parameter. Factors including TIME, CONDITION and TIME*CONDITION interactions were investigated. Sphericity assumptions were tested and post-hoc testing with paired Student's t-tests was carried out for the significant findings (see section 3.7.1 for further detail).

10.5 Results

10.5.1 Kinematics

Findings from the RMANOVA revealed a significant main effect for TIME on movement onset and movement offset (p < 0.05), but not for maximum velocity, summed error, movement duration and maximum force (p > 0.05) (see table 10.1). A significant main effect of CONDITION was found for movement offset, summed error, movement duration and maximum force (p < 0.05), but not for movement onset and maximum velocity (p > 0.05) (see table 10.1). No significant interaction effects were found for any of the kinematic parameters (p > 0.05).

Post hoc testing for TIME (table 10.2, figure 10.1[A-E]) for movement onset showed that during FAM reaching, all time points were significantly increased compared to T10 (p < 0.006). During FF reaching, only TMS delivered at 160ms, 250ms, 280ms and 310ms were significantly increased compared to T10 (p < 0.006). In WO reaching responses at 160ms, 250ms and 280ms were significantly increased compared to T10 (p < 0.006). Movement offset during FAM reaching was only significantly increased at 220ms and 280ms when compared to T10. During FF reaching, movement offset was only significantly increased at 190ms compared to T10. There were no significant changes in WO reaching for movement offset (p > 0.006).

Post hoc testing for CONDITION (table 10.2, figure 10.1[A-E]) showed that movement offset was significantly increased during FF reaching compared to FAM reaching (at 10ms, 100ms, 190ms and 310ms, all p < 0.016). Movement offset during FF reaching was significantly increased compared to WO reaching (at 160ms and 190ms p < 0.016). There were no significant differences regarding FAM vs. WO reaching (p > 0.016). Summed error was significantly increased in FF reaching compared to FAM reaching (but only at 130ms, 250ms and 280ms; p < 0.016). FF summed error was also significantly increased compared to WO summed error (at 100ms, 160ms and 280ms; p < 0.016). There were no significant differences in post hoc testing for FAM vs. WO reaching. Movement duration was significantly increased in FF reaching compared to FAM reaching (at 10ms, 100ms, 220ms and 280ms p < 0.016). Movement duration was also significantly increased in FF reaching compared to WO reaching (at 10ms, 100ms, 160ms, 220ms and 280ms p < 0.016). There were no significant differences in movement duration for FAM vs. WO reaching (p > 0.016). Maximum force was significantly increased in FF reaching (p < 0.016) compared to FAM and WO reaching, which did not significantly differ (p > 0.016).

Post hoc testing results are shown in table 10.2 and figure 10.1[A-E].

| | | TIME | | (| CONDITION | | TIME*CONDITION INTERACTION | | | |
|---------------------------------------|-------------------|-------|---------|-------------------|-----------|---------|----------------------------|-------|-------|--|
| | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | |
| Movement Onset (ms) | 4.0 [47.3] | 7.876 | < 0.001 | 2 [24] | 1.397 | 0.267 | 16 [192] | 0.467 | 0.960 | |
| Movement Offset (ms) | 3.7 [43.5] | 3.825 | 0.012 | 2 [24] | 8.27 | 0.003 | 16 [192] | 0.475 | 0.956 | |
| Maximum Velocity (m.s ⁻¹) | 2.1 [25.6] | 0.152 | 0.864 | 2 [24] | 0.471 | 0.630 | 16 [192] | 0.260 | 0.998 | |
| Duration (ms) | 3.0 [35.6] | 0.359 | 0.780 | 2 [24] | 11.11 | < 0.001 | 16 [192] | 0.404 | 0.980 | |
| Summed Error (distance: cm) | 8 [96] | 1.042 | 0.410 | 1.3 [14.5] | 7.557 | 0.012 | 16 [192] | 1.570 | 0.080 | |
| Maximum Force (N) | 1.7 [19.8] | 0.245 | 0.743 | 1.2 [13.6] | 213.87 | < 0.001 | 16 [192] | 0.190 | 0.100 | |

Table 10.1. Results of the two-way RMANOVA:

Table 10.2. Post-hoc testing results

| The table shows kinematic responses during FAM, FF and WO blocks of reaching when TMS was applied to the right PPC at different time points. Values |
|---|
| represent means and standard errors means. |

| TIME | TMS: 10ms | TMS: 100ms | TMS: 130ms | TMS: 160ms | TMS: 190ms | TMS: 220ms | TMS: 250ms | TMS: 280ms | TMS: 310ms |
|-----------|--|---------------|---------------|---------------|-----------------------|---------------|---------------|----------------|---------------|
| | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] |
| CONDITION | | | | | Movement Onset (| ms): | | | |
| FAM | 385 [16] | 409 [17] * | 415 [14] * | 429 [18] * | 420 [19] * | 432 [19] * | 435 [19] * | 440 [23] * | 439 [18] * |
| FF | 393 [21] | 418 [17] | 419 [16] | 440 [20] * | 439 [20] | 438 [21] | 426 [17] * | 434 [22] * | 438 [18] * |
| WO | 401 [21] | 427 [19] | 428 [19] | 434 [24] * | 441 [22] | 431 [23] | 445 [26] * | 454 [20] * | 450 [25] |
| | | | | | Movement Offset (| (ms): | | | |
| FAM | 1158 [30] | 1173 [27] | 1202 [29] | 1193 [27] | 1193 [28] | 1197 [29] * | 1197 [32] | 1201 [32] * | 1199 [34] |
| FF | 1209 [29] 🔳 | 1250 [25] 🔳 | 1238 [22] | 1267 [28] 🕇 | 1270 [28] *■ ✦ | 1248 [29] | 1239 [24] | 1262 [29] | 1245 [29] 🔳 |
| WO | 1160 [36] | 1175 [40] | 1185 [39] | 1190 [38] | 1196 [36] | 1193 [35] | 1204 [40] | 1220 [38] | 1211 [38] |
| | Maximum Velocity (m.s ⁻¹): | | | | | | | | |
| FAM | 0.31 [0.02] | 0.31 [0.02] | 0.31 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] |
| FF | 0.32 [0.02] | 0.31 [0.02] | 0.32 [0.02] | 0.30 [0.02] | 0.31 [0.02] | 0.31 [0.02] | 0.32 [0.02] | 0.31 [0.02] | 0.31 [0.02] |
| WO | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] |
| | | | | Ν | Movement Duration | (ms): | | | |
| FAM | 773 [27] | 764 [25] | 787 [29] | 764 [25] | 774 [29] | 765 [26] | 762 [28] | 762 [26] | 760 [28] |
| FF | 816 [23] ■✦ | 832 [24] ■✦ | 819 [24] | 827 [27] + | 831 [22] | 810 [26] 🖛 🕇 | 812 [21] | 829 [18] ■+ | 807 [26] |
| WO | 759 [29] | 749 [34] | 757 [34] | 755 [29] | 755 [32] | 762 [27] | 759 [35] | 766 [31] | 760 [27] |
| | | | | Su | mmed Error (distan | ce: cm): | | | |
| FAM | 2.36 [0.22] | 2.38 [0.19] | 2.17 [0.14] | 2.47 [0.18] | 2.30 [0.19] | 2.49 [0.23] | 2.13 [0.21] | 2.32 [0.22] | 2.19 [0.22] |
| FF | 3.05 [0.35] | 3.11 [0.32] + | 3.23 [0.38] | 3.00 [0.29] + | 2.84 [0.30] | 2.84 [0.34] | 3.01 [0.28] | 3.28 [0.34] ■+ | 2.99 [0.30] + |
| WO | 2.33 [0.22] | 2.11 [0.13] | 2.31 [0.15] | 2.16 [0.14] | 2.26 [0.13] | 2.19 [0.18] | 2.31 [0.15] | 2.31 [0.15] | 2.16 [0.17] |
| | | | | | Force (N): | | | | |
| FAM | 4.8 [0.1] | 4.8 [0.1] | 4.8 [0.1] | 4.8 [0.1] | 4.8 [0.1] | 4.8 [0.1] | 4.8 [0.1] | 4.8 [0.1] | 4.9 [0.1] |
| FF | 10.6 [0.5] •+ | 10.5 [0.5] ■+ | 10.7 [0.5] ■+ | 10.4 [0.5] ■+ | 10.6 [0.5] ■+ | 10.6 [0.5] ■+ | 10.7 [0.4] ■+ | 10.6 [0.4] ■+ | 10.6 [0.5] |
| WO | 5.0 [0.2] | 5.0 [0.2] | 5.0 [0.2] | 5.0 [0.2] | 4.9 [0.2] | 4.9 [0.2] | 5.0 [0.2] | 4.9 [0.2] | 4.8 [0.2] |

Symbols represent significance following post hoc testing:

* = significant difference vs T10 for time in familiarisation blocks, * = significant difference vs T10 for time for FF blocks, * = significant difference vs T10 for time for WO blocks,
■ = significant difference between FAM and FF, += significant difference between FF and WO, ▲ = significant difference between FAM and WO



Figure 10.1 [A-E]. Graphical presentation of the post-hoc testing results (based on table 10.2) for the significant kinematic variables:



Symbols represent significance following post hoc testing:

- * = significant difference vs T10 for time in familiarisation blocks, * = significant difference vs T10 for time for FF blocks, * = significant difference vs T10 for time for WO blocks,
 - = significant difference between FAM and FF, += significant difference between FF and WO, ▲ = significant difference between FAM and WO

10.6 Discussion

This results of this experiment demonstrated the effects of single-pulse TMS to the right PPC during novel reaching preparation. Although studies have revealed that the left PPC is involved in novel reaching (Della-Maggiore *et al.*, 2004) which was also illustrated in chapter 6 within this thesis, researchers have not targeted the right PPC with SP-TMS during reaching preparation. This study therefore provided a novel insight into the ways in which TMS can disrupt right PPC function in a motor adaptation paradigm via exploring different kinematic parameters as behavioural measures.

10.6.1 Kinematics

10.6.1.2 The shift in movement onset and movement offset:

The findings of this experiment illustrated that stimulation to the right PPC during motor preparation for reaching delayed the movement onset responses during FAM, FF and WO reaching and movement offset responses during FAM and FF reaching. The results regarding the shift were similar to what was found when the left PPC was stimulated with TMS (see chapter 6, table 6.2).

The delays that were noted in the preparation of reaching have similarly been demonstrated in rTMS protocols of visuo-spatial attention tasks (Xu *et al.*, 2016) as well as SP-TMS FAM reaching tasks (Busan *et al.*, 2009). For example, Busan *et al.*, (2009) reported that a specific network in the motor circuit including areas in the posterior parietal cortex enables the preparation of reaching. They specifically found that TMS delivered to regions of the right PPC during reaching preparation resulted in delayed responses (similar to findings in this experiment), however when TMS was delivered prior to the reaching preparation phase there was an effect of facilitation (i.e. faster movement onset responses). The results regarding the shift that was found can be explained by the TMS state-dependent theory which was proposed by Silvanto and Muggleton (2008) (Busan *et al.*, 2009). This theory poses the notion that because the PPC is already involved (i.e. pre-activated) in planning motor behaviours for execution (via corticospinal projections to areas of the premotor cortex and primary motor cortex), TMS delivered to its region can cause 'neural noise'. This then interferes with the region's functioning and as a result leads to slower responses in a task (Busan *et*

al., 2009; Miniussi, Ruzzoli and Walsh, 2010).

It can be argued that the behavioural responses of both hemispheres for movement onset are alike (as TMS similarly affected the right and left PPC; chapter 6, table 6.2), and this poses questions regarding possible asymmetries that have been noted between the two hemispheres which have been suggested to contribute to their distinct functions. This notion is supported by patient lesion studies. For example, although apraxia has been linked to being caused by lesions in the left PPC, as opposed to the right PPC (Vallar and Perani, 1986; Husain and Nachev, 2007), studies have demonstrated that some forms of apraxia are in fact associated with right posterior lesions (as reported by Hamser, 1998). Neural activation studies have also demonstrated that both hemispheres can be similarly activated during motor tasks (Calton, Dickinson and Snyder, 2002; Busan et al., 2012). On the other hand, it should be taken into account that in this experiment TMS did not have a significant main effect on the condition of reaching for movement onset, whereas TMS to the left PPC significantly affected movement onset responses between different reaching conditions (FAM, FF and WO; see chapter 6, table 6.2). This finding thus does in fact offer evidence for different functional roles between the two hemispheres which is in line with evidence proposed by researchers such as Husain and Nachev (2007) and Vallar and Perani (1986). In summary, TMS-induced shifts in timing is present in both PPCs, whereas simulating the left PPC appears to preferentially be involved in modulation of timing during novel reaching adaptation. These findings were important in showing the ways in which some behavioural functions between the two hemispheres are specialised and how some are not (e.g. TMS effects for time vs. condition). The differences and similarities regarding behavioural functions between the two hemispheres can be explained by the functional connections in the motor network (Gharbawie et al., 2010). For example, the two hemispheres may be connected to similar functional zones in the motor circuit which as a result facilitates similar behavioural responses, however some regions in one hemisphere may share greater connections with sub-structures in the motor network, and therefore cause differences in behavioural responses (Gharbawi et al., 2010).

10.6.1.3 Errors in reaching with right PPC TMS:

Another key finding from this experiment was that disrupting the right PPC with TMS during preparation for FF reaching compared to FAM and WO reaching impaired performance, therefore demonstrating that the right PPC also plays a functional role in novel reaching, which was similarly shown with left PPC stimulation (Della-Maggiore et al., 2004). Although post-hoc testing revealed that FF reaching trajectories were significantly impaired during right PPC stimulation, this was not illustrated at all time points of stimulation. However, during left PPC stimulation, FF reaching was significantly disrupted at all time of TMS delivery (10ms-310ms) (see chapter 6, table 6.2). Studies have shown the left hemisphere to be the dominant hemisphere in righthanded subjects (Busan et al., 2012; Vingerhoets et al., 2013). The left hemisphere as opposed to the right hemisphere has also been reported to play a role in facilitating motor actions for movement execution, particularly in visually-guided reaching (Goodale 1988; Busan et al., 2012). Studies have also illustrated that the right PPC is involved only in planning left hand reaches compared to the left PPC which is activated during both left and right hand reaching motions (Schluter et al., 2001; Oliveira et al., 2010). Therefore, based on these studies, the less robust reaching disruptions in the right PPC (based on post-hoc testing results in this chapter) can be attributed to its less important functional role in reaching preparation of the right arm.

It could be argued that the nature of the motor task in this experiment could have also contributed to the lack of reaching deviations that were found during right PPC stimulation. For example, the experimental task was associated with reaching towards one target (135°), however if the motor task was based on random reaching targets, right PPC neural activity may have been heightened due to the region's involvement in attention selection-response behaviours (Posner and Peterson, 1990; Malhotra, Coulthard and Husain, 2009), and as a result trajectories in reaching may have been severely impaired. Nonetheless in this experiment TMS did have significant detrimental impact (albeit a smaller impact than what was found in the left PPC) on right PPC function. Considering that the right PPC plays a key role in enabling individuals to uphold an alert state during a task (Malhotra, Coulthard and Husain, 2009), it could be suggested that TMS interfered with this particular specialised right PPC function, thus impacting reaching.

10.7 Chapter conclusions:

The purpose of this chapter was to explore the effects of right PPC stimulation during novel reaching and to investigate whether TMS administered at different times of motor preparation had an effect on motor output. TMS only had a significant timerelated impact on movement onset, and no significant effect of condition for movement onset was found - which was not the case for left PPC stimulation. This shows that the two hemispheres exhibit distinct functions, which is in line with most lesion studies that have illustrated different symptomologies associated with left and right hemispheric PPC lesions. Although findings illustrated that TMS significantly increased summed error during FF reaching compared to FAM and WO reaching, summed error responses were not as significantly impaired in this experiment compared to what was found during left PPC stimulation (chapter 6, table 6.2). This study was important in establishing distinct hemispheric PPC functions (left vs. right) related to the preparation for novel reaching. Whether other cortical regions in different hemispheres exhibit distinct differences were investigated, for example in the next experimental chapter in this thesis the right dPMC was targeted with TMS to investigate whether it plays a similar functional role when compared to the left dPMC (chapter 9) with regards to novel reaching.

Chapter 11

11 Experiment 8

Exploring the impact of SP-TMS to right dorsal pre-motor cortex (dPMC) during right arm reaching in a novel motor learning protocol

11.1 Introduction

In Chapter 9, the left dPMC was stimulated during preparation for reaching and findings revealed that trajectory errors were significantly impaired with TMS during FF reaching, but not during FAM or WO reaching at all time points in which TMS was delivered. Other kinematic measures however were not significantly affected with TMS. The aim of this study was to investigate whether the right dPMC also exhibited similar behavioural responses with TMS, or whether it has separate functions with regards to reaching preparation.

Premotor structures are important for upper limb movement particularly during complex tasks (Pollok et al., 2017). The left dPMC, but not the right dPMC has been reported to be involved in response selection and goal directed behaviour (Hoshi and Tanji, 2002; Beck et al., 2009; Yamagata et al., 2012). Studies have also demonstrated functional hemispheric differences between the right and left dPMC (Genon et al., 2017) particularly during motor sequence tasks. For example, when motor sequences performed with either the dominant or non-dominant hand, the left dPMC was reported to have a greater neural activation during the earlier learning phase of a motor sequence, whereas the right dPMC has been found to have greater activity in the advanced learning phase and when the sequences had greater degrees of complexity (Sadato et al., 1996; Grafton, Hazeltine and Ivry, 2002; Schubotz and von Cramon, 2003). The right dPMC has also been noted to have a greater involvement in spatial tasks compared to the left dPMC. For instance, functional MRI studies have highlighted increased neural activity in the right dPMC during spatial attention tasks (Gitelman et al., 1999; Schubotz and von Cramon, 2003). The role of the right dPMC in spatial tasks has also been demonstrated in the cognitive domain (Jonides et al., 1993; Schubotz and von Cramon, 2003; Genon et al., 2017). For example, positron emission tomography (PET) scans have revealed enhanced right hemispheric dPMC activity during spatial working memory tasks (Jonides et al., 1993). Further evidence for functional hemispheric differences between the right and left dPMC comes from patients with PMC lesions who exhibit different symptoms. For example, left dPMC lesions have been associated with dystonia (Ceballos-Baumann and Brooks, 1998; Beck et al., 2009) which is a condition of abnormal muscle activity that results in impaired muscle function, and its symptoms range from tremors to jerked muscle movement (Fahn 1984; Elble 2013; Jinnah and Factor 2015). On the other hand, right dPMC lesions have been associated with symptoms of ideomotor apraxia; which demonstrates the role of the right hemisphere in skilled motor functions (Schnider et al., 1997; Gross and Grossman, 2008; Wheaton et al., 2008). Previous studies that have implemented TMS protocols have mainly explored functional differences between the two regions with precision hand grasping and lifting tasks. For example, with an rTMS protocol, Davare et al., (2006) revealed a disruption in left dPMC function regarding a shift and delay in muscle response lifting times, whereas right dPMC rTMS did not cause such disruptions and did not significantly impair dominant hand movements during the task. It is important to explore whether reaching preparation during motor adaptation is affected when TMS is applied to the right dPMC, because most studies exploring reaching have mainly targeted the left dPMC rather than the right dPMC. For example, a visual perturbation prism adaptation task with SP-TMS to the left dPMC revealed slower rates of motor adaptation when TMS was delivered during movement onset in contrast to when it was delivered at movement offset (Lee and Van Donkelaar, 2006). However, the right dPMC has been found to be associated with motor planning, as demonstrated in visuomotor learning tasks. For example, Praeg et al., (2005) targeted the right dPMC and a control region with TMS in a visuo-motor task and their findings revealed that TMS to the right dPMC caused a disruption in responses during preparation for learning, which was not the case when the control region was stimulated.

On the other hand, the functional role of the right dPMC has not been explored with 1) a reaching motor adaptation paradigm and 2) TMS applied at different time points during the preparation of a reach. Additionally, studies with right dPMC stimulation have also not explored whether TMS causes a virtual disruption of trajectory errors in reaching and have mostly focused on reaction time responses (Praeg *et al.*, 2005). Considering that hemispheric differences have been found between the two regions, it could be argued that TMS to the right dPMC would result in different behavioural effects when compared to the findings that were illustrated for the left dPMC (chapter 8, experiment

6) which can be due to different subdivisions that have been noted within the left and right dPMC (Genon *et al.*, 2017). This study is the first attempt to explore this.

11.2 Methodology

Participant details (e.g. N, age, gender, handedness) for this experiment is shown in table 3.2. Following RMT identification with functional measures (see section 3.2.2.2 and 3.6.2), the right dPMC was stimulated. Right dPMC location and the coil orientation used for stimulation is described in section 3.3.2.2.5 and also shown graphically in figure 3.3. The reaching paradigm is demonstrated in figure 3.5.

11.3 Data acquisition: MEPs and Kinematics

MEPs were only collected to identify RMT (section 3.6.2) and no MEPs were elicited or collected during right dPMC stimulation. Kinematic data were acquired throughout the experiment and analysed and quantified offline in MatLab 2017b (The MathWorks Inc, Natick MA, USA) (see section 3.6.1).

11.4 Statistical analysis: Kinematics

A two-way RMANOVA was performed for the kinematic variables whereby TIME, CONDITION and TIME*CONDITION interactions were explored. Sphericity assumptions were tested and post-hoc analysis was performed for the significant findings (section 3.7.1 describes this in further detail).

11.5 Results

11.5.1 Kinematics:

The RMANOVA revealed a significant main effect for TIME on movement onset and movement offset (p < 0.05) but not for maximum velocity, movement duration, summed error and maximum force (p > 0.05) (see table 11.1). A significant main effect of CONDITION was found for movement offset, summed error, movement duration and maximum force (p < 0.05) but not for movement onset and maximum velocity and (table 10.1). No significant interactions were found (p > 0.05) (table 11.1).

Post-hoc testing for TIME (table 11.2, figure 11.1[A-E]) revealed that movement onset significantly increased compared to T10 in both FAM reaching and FF reaching conditions (p < 0.006) but not during the WO condition. Movement offset was significantly increased at 250ms, 280ms and 310ms compared to T10 in the FAM condition (p < 0.006). Movement offset significantly increased at 130ms, 160ms,

220ms, 250ms and 310ms compared to T10 during the FF condition. Movement offset was not significantly changed in the WO condition compared to T10 (all p > 0.006).

Post hoc testing for CONDITION (table 11.2, figure 11.1[A-E]) with regards to movement offset showed that FF reaching was significantly increased compared to FAM reaching (at 130ms and 220ms p < 0.016). Movement offset in FF reaching was significantly increased compared to WO (at 100ms, 130ms, 160ms, 220ms, 280ms and 310ms p < 0.016). No significant differences were found for FAM vs. WO (p > 0.016). Summed error was significantly increased in FF reaching compared to FAM reaching only at 190ms (p < 0.016). No differences in summed error were revealed for WO vs. FF reaching, and WO vs. FAM reaching (all p < 0.016). Movement duration significantly increased in FF reaching compared to both FAM reaching (at 160ms and 220ms p < 0.016) and WO reaching (at 100ms, 100ms, 130ms, 160ms, 190ms, 220ms, 280ms and 310ms p < 0.016). There were no significant differences in movement duration when comparing FAM vs. WO responses (all p > 0.016). Maximum force was significantly increased in FF reaching (p < 0.016) compared to FAM and WO reaching which were not significantly different (p > 0.016).

Post-hoc testing results are further demonstrated in table 11.2 and figure 11.1[A-E].

| | | TIME | | | CONDITION | | | TIME*CONDITION INTERACTION | | | |
|---------------------------------------|-------------------|-------|---------|-------------------|-----------|---------|-------------------|----------------------------|-------|--|--|
| | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | <i>df</i> (Error) | F | Sig. | | |
| Movement Onset (ms) | 8 [96] | 12.22 | < 0.001 | 2 [24] | 0.602 | 0.556 | 16 [192] | 1.206 | 0.266 | | |
| Movement Offset (ms) | 8 [96] | 4.531 | < 0.001 | 2 [24] | 10.37 | 0.001 | 16 [192] | 1.411 | 0.140 | | |
| Maximum Velocity (m.s ⁻¹) | 8 [96] | 0.695 | 0.695 | 1.4 [16.1] | 0.140 | 0.785 | 16 [192] | 1.631 | 0.064 | | |
| Duration (ms) | 8 [96] | 0.598 | 0.777 | 2 [24] | 11.01 | < 0.001 | 16 [192] | 1.196 | 0.274 | | |
| Summed Error (distance: cm) | 2.6 [30.7] | 1.11 | 0.355 | 2 [24] | 3.554 | 0.044 | 16 [192] | 0.838 | 0.642 | | |
| Maximum Force (N) | 8 [96] | 0.822 | 0.585 | 1.1 [12.5] | 135.60 | < 0.001 | 16 [192] | 0.535 | 0.926 | | |

Table 11.1. Results of the two-way RMANOVA:

| Table 11.2. Post-hoc testing results: The table shows kinematic responses during FAM, FF and WO blocks of reaching when TMS was applied to the right |
|--|
| dPMC at different time points. Values represent means and standard errors of the mean. |

| TIME | TMS: 10ms | TMS: 100ms | TMS: 130ms | TMS: 160ms | TMS: 190ms | TMS: 220ms | TMS: 250ms | TMS: 280ms | TMS: 310ms |
|-----------|--|---------------|-----------------------|---------------|---------------------|---------------|---------------|---------------|---------------|
| | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] | Mean [SEM] |
| CONDITION | | | | Mo | vement Onset (ms) |): | | | |
| FAM | 389 [17] | 419 [19] * | 418 [15] * | 431 [18] * | 434 [22] * | 437 [14] * | 457 [18] * | 458 [26] * | 461 [24] * |
| FF | 384 [18] | 420 [14] * | 432 [16] * | 418 [13] * | 424 [12] * | 440 [16] * | 442 [16] * | 434 [16] * | 438 [19] * |
| WO | 396 [24] | 418 [18] | 423 [19] | 434 [19] | 442 [20] * | 443 [19] | 445 [23] | 440 [24] | 444 [21] |
| | | | | Мо | vement Offset (ms) |): | | | |
| FAM | 1164 [32] | 1194 [29] | 1185 [28] | 1204 [28] | 1198 [30] | 1204 [31] | 1227 [32] * | 1228 [33] * | 1242 [35] * |
| FF | 1198 [38] | 1242 [38]+ | 1266 [37] *■ ✦ | 1248 [35] *+ | 1250 [39] | 1264 [33] *■+ | 1261 [46] * | 1257 [38]+ | 1257 [41] *+ |
| WO | 1184 [32] | 1174 [32] | 1194 [34] | 1201 [37] | 1200 [32] | 1197 [36] | 1216 [35] | 1194 [41] | 1202 [29] |
| | Maximum Velocity (m.s ⁻¹): | | | | | | | | |
| FAM | 0.31 [0.02] | 0.31 [0.01] | 0.32 [0.02] | 0.31 [0.02] | 0.31 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.31 [0.02] | 0.31 [0.02] |
| FF | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.31 [0.02] | 0.32 [0.02] | 0.31 [0.02] | 0.31 [0.02] | 0.31 [0.02] | 0.32 [0.02] |
| WO | 0.31 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] | 0.32 [0.02] |
| | | | | Mov | ement Duration (m | s): | | | |
| FAM | 775 [29] | 775 [25] | 768 [27] | 773 [24] | 765 [24] | 767 [28] | 770 [29] | 771 [25] | 784 [24] |
| FF | 814 [31] | 822 [32] + | 834 [33] 🗲 | 829 [31] ■✦ | 826 [36] 🕇 | 825 [27] ■+ | 819 [36] | 823 [32] + | 818 [31] 🕇 |
| WO | 788 [22] | 756 [26] | 771 [28] | 766 [29] | 758 [25] | 754 [28] | 771 [26] | 754 [31] | 758 [22] |
| | | • | | Summe | ed Error (distance: | cm): | | | |
| FAM | 2.31 [0.30] | 2.14 [0.33] | 2.25 [0.32] | 2.23 [0.27] | 2.30 [0.39] | 2.12 [0.28] | 2.21 [0.31] | 2.26 [0.30] | 2.38 [0.34] |
| FF | 2.70 [0.21] | 2.73 [0.15] | 3.44 [0.61] | 2.95 [0.21] | 2.81 [0.17] ■ | 2.88 [0.20] | 2.90 [0.23] | 2.75 [0.16] | 2.89 [0.20] |
| WO | 2.35 [0.31] | 2.31 [0.27] | 2.35 [0.29] | 2.04 [0.21] | 2.33 [0.28] | 2.33 [0.28] | 2.30 [0.26] | 2.12 [0.21] | 2.32 [0.27] |
| | | • | | | Force (N): | | | | |
| FAM | 4.9 [0.2] | 4.9 [0.1] | 4.9 [0.1] | 4.9 [0.1] | 4.8 [0.1] | 4.8 [0.1] | 4.9 [0.1] | 4.8 [0.1] | 4.9 [0.1] |
| FF | 10.6 [0.5] ■+ | 10.6 [0.5] ■+ | 10.5 [0.5] ■+ | 10.4 [0.5] ■+ | 10.5 [0.5] ■+ | 10.4 [0.5] ■+ | 10.4 [0.5] ■+ | 10.5 [0.5] ■+ | 10.6 [0.6] ■+ |
| WO | 4.9 [0.2] | 4.9 [0.2] | 4.9 [0.2] | 4.9 [0.2] | 4.8 [0.2] | 4.9 [0.2] | 4.8 [0.2] | 4.8 [0.2] | 4.9 [0.2] |

Symbols represent significance following post hoc testing:

* = significant difference vs T10 for time in familiarisation blocks, * = significant difference vs T10 for time for FF blocks, * = significant difference vs T10 for time for WO blocks,
■ = significant difference between FAM and FF, += significant difference between FF and WO, ▲ = significant difference between FAM and WO



Figure 11.1 [A-E]: Graphical presentation of the post-hoc testing results (based on table 11.2) for the significant kinematic variables:

180



Symbols represent significance following post hoc testing:

* = significant difference vs T10 for time in familiarisation blocks, * = significant difference vs T10 for time for FF blocks, * = significant difference vs T10 for time for WO blocks,
■ = significant difference between FAM and FF, += significant difference between FF and WO, ▲ = significant difference between FAM and WO

11.6 Discussion

The findings from this experiment demonstrated the ways in which SP-TMS to the right dPMC had an impact of right-arm reaching preparation behaviour. Although studies have shown that motor planning is affected when the right dPMC is targeted (Praeg, *et al.*, 2005), other factors such as performance (e.g. summed error) during novel reaching preparation have not been extensively studied. This study therefore aimed to provide a novel insight into whether a virtual disruption (via TMS) to the right dPMC can have an impact on its function and impair kinematic behaviour during reaching preparation.

11.6.1 Kinematics

11.6.1.2 The role of the right dPMC regarding motor preparation:

The findings from this experiment demonstrated that TMS to the right dPMC significantly shifted movement onset responses in FAM and FF reaching. Shifts in responses were also noted in WO reaching but not to the extent as FAM and FF reaching (table 11.2). Movement offset was also found to be significantly affected with TMS as delays in responses were found during FAM and FF reaching, but not during WO reaching (table 10.2).

These findings are in line with results that have been obtained with SP-TMS protocols during visuo-spatial motor learning tasks. For example, Praeg *et al.*, (2005) found that TMS to the right dPMC significantly impacted and delayed reaction times during motor learning which was not the case when a control region was stimulated. Although most studies have illustrated that the left dPMC plays an important role in motor preparation (Rushworth et al., 2003; Davare et al., 2006), this experiment revealed that the right dPMC also plays a role in motor preparation because TMS delayed reaching responses. Therefore it could be argued that preparation behaviours for reaching may not be restricted to only one hemisphere. A possible reason as to why TMS delayed responses can be attributed to distinct right dPMC sub-regions that have been reported to have specific functions and connections with other cortical structures that facilitate motor planning and preparation (Genon *et al.*, 2017). For example, using structural-, resting-and functional-connectivity measures, Genon *et al.*, (2017) found 5 right dPMC sub-regions that were connected to different cortical structures, including:

- 1) A rostral region connected to the prefrontal cortex
- 2) A central region connected to parietal structures
- 3) A caudal region linked to the motor circuitry
- 4) A dorsal region associated with cognitive and motor structures
- 5) A ventral region that was linked to visuo-motor structures

Genon *et al.*, (2017) argued that different right dPMC connections assist different functional behaviours. For example, rostral connections with the prefrontal cortex aid cognitive functions such as memory (Jonides *et al.*, 1993; Genon *et al.*, 2017). On the other hand, dorsal and caudal right dPMC sub-regions and connections (with motor structures) have been suggested to facilitate motor planning and preparation (Genon *et al.*, 2017). This is in line with fMRI studies that have found greater neural activity within this particular sub-region in motor planning and execution tasks (Picard and Strick, 2001; Genon *et al.*, 2017). Studies in non-human primates have also supported this notion, with greater cellular activity noted in the caudal right dPMC sub-region during motor preparation (Boussaoud, 2001; Genon *et al.*, 2017). The findings obtained for movement onset and offset were important in demonstrating the functional role of the right dPMC in movement preparation which have not previously been studied with motor adaptation protocols.

11.6.1.3 The impact of right dPMC TMS on reaching trajectories:

Although RMANOVA statistical testing revealed that TMS to the right dPMC had a significant effect on reaching trajectories (table 11.1), post-hoc testing did not reveal major significant differences between FF and FAM reaching conditions. There were also no significant differences in summed error regarding 1) FAM vs. WO reaching, and 2) FF vs. WO reaching (table 11.2).

This finding is in line with previous studies that have used TMS as a virtual disruption tool to explore right dPMC functioning. For instance, studies ranging from finger movement tasks to hand grip and lifting tasks have found that TMS to the right dPMC did not have a significant detrimental impact on motor performance (Schluter *et al.*, 1998; Schluter *et al.*, 2001; Davare *et al.*, 2006). The reason as to why summed error was not significantly affected with right dPMC stimulation during preparation for novel reaching can be explained in terms of hemispheric functions relating to handedness (Schluter *et al.*, 1998; Beck et al., 2009). For instance, Schluter *et al.*, (1998)

implemented a TMS protocol during a hand motor task and their results showed that when the right dPMC was stimulated only left hand functions were disrupted, whereas left dPMC stimulation caused a disruption in both the left and right hand performance during the task. Functional MRI studies have also demonstrated similar results in a finger movement tasks of both hands, whereby neural activity in the right dPMC was heightened during left handed finger movement as opposed to left dPMC activity which was increased when finger movements were performed with either hand (Schluter *et al.*, 2001).

Neuroimaging cerebral blood flow experiments have found enhanced right dPMC activity during complex sequential finger tasks (Sadato *et al.*,1996) and have therefore provided evidence for the role of the right dPMC in maintaining selective attention in complex tasks (Sarter, Givens and Bruno, 2001; Schubotz and von Cramon, 2003). It could be argued that if the reaching task in this experiment was based on reaching towards different targets, right dPMC activity would have been heightened (because participants may have had to sustain greater attention during reaching) and as a result TMS may have caused greater disruptions in reaching. Based on this assumption, the nature of the reaching task may have contributed to the lower impact on trajectory errors that were found in this study.

11.7 Conclusion:

Previous studies have mainly explored right dPMC function with TMS during sequential finger movement tasks as well as hand grip and lifting tasks, however this experiment provided novel findings into how TMS to the right dPMC has an impact on reaching preparation which has not previously been explored. The study revealed that TMS to the right dPMC shifted both movement onset and offset responses, and this is in line with findings from visuo-adaptation tasks. The study therefore provided an insight into the functional role of the right dPMC with regards to reaching preparation, which is a function that has been argued to be facilitated via right dPMC sub-regional connections with other cortical structures. The findings for summed error in this experiment were different in comparison to what was revealed when the left dPMC was stimulated (chapter 9, table 9.2). Summed error findings thus provided evidence for dPMC hemispheric differences in preparation for novel reaching. Considering that various regions in this thesis were impaired differently with TMS stimulation during

novel reaching, the final chapter of this thesis incorporates a systematic comparative analysis to compare novel reaching performance between all of the regions that were stimulated (left and right M1, SMA, left and right PPC, and left and right dPMC) to explore how they statistically differ in terms of the disruption caused. The final experimental chapter therefore highlights a model of the motor network, illustrating the relative influence of several cortical regions involved in novel motor performance.

Chapter 12

The comparative importance of different cortical regions in motor control and motor adaptation

12.1 Introduction

The purpose of this chapter was to compare the relative importance of cortical motor regions that were involved in motor control and motor adaptation. The main disruptive impact of TMS on reaching preparation were quantified by measuring the summed error of reaching trajectories and timing of reaching; both of which have been shown to be disrupted by TMS in previous chapters. Summed error, in particular has been reported to represent successful motor adaptation; with less errors indicating enhanced performance (Pizzamiglio *et al.*, 2017a; Pizzamiglio *et al.*, 2017b).

By exploring the impact of TMS on kinematics during upper-limb reaching, the functional role of different cortical areas in the motor network can be quantified. This has been supported by previous studies that have also used TMS protocols to explore the functional purpose of a region. For example, Della-Maggiore *et al.*, (2004) found that TMS delivered to the left PPC during upper-limb novel reaching following movement onset impaired reaching trajectories. Other studies have illustrated that rTMS to the SMA had a negative effect on bimanual motor co-ordination performance, however this was not the case when the M1 was stimulated (Obhi *et al.*, 2002). TMS protocols targeting the cerebellum during a finger reaching-to-point task resulted in distance-to-point estimation errors, therefore demonstrating the role of the cerebellum in planning and preparing movements (Miall *et al.*, 2007).

Studies have reported increased neuronal firing in the dPMC during the encoding of kinematic parameters in a motor task (Alexander and Crutcher, 1990; Fu, Suarez and Ebner, 1993; Ward *et al.*, 2007). However, the type of kinematic parameter encoded by the dPMC is currently unclear. Therefore, investigating kinematic factors separately between regions can help reveal the structures in the motor network that may play a more distinctive role in encoding a specific type of kinematic parameter. Additionally, researchers have demonstrated an insight into the cortical networks that underpin motor control in grasping and reaching behaviours. As well as PPC, SMA and PMC projections to the M1, there are a large range of connections which feed into these

structures to facilitate motor activity. For example, the DLPFC has been reported to be involved in motor related behaviours, ranging from movement selection to performance attentiveness and researchers have noted that it transmits organisation and planning information to the PMC to transfer to the M1 (Ehrsson et al., 2000; Kantak et al., 2012; Badoud et al., 2017). Network projections between the DLPFC and the pre-SMA have also been illustrated (Wang et al., 2005; Silva Moreira, Marques and Magalhaes, 2016; Kim et al., 2018). Additionally, areas such as the visual cortex (V1), somatosensory cortex (S1), basal ganglia (BG) all interact with the M1. The projections from the M1 also feed into regions such as the red nucleus (RN), and the vestibular network (VN) which in turn interact with the cortical signals descending towards the spinal cord, enabling motor behaviours such as reaching to take place (Scott, 2004). The role of these regions in reaching and grasping has been supported by lesion studies. For example, Whishaw, Gorny and Sarna (1998) found that RN lesions in mice disrupted both limb aiming behaviours and paw rotation during a skilled reaching task, thus providing evidence for RN function with regards to limb motor control (Whishaw and Gorny, 1996). Similarly, with regards to Huntington's Diseases (HD) which is a condition that arises from basal ganglia impairments (Reiner, Dragatsis and Dietrich, 2011), researchers have found that reaching performance during both normal and novel (force-field) reaching was accompanied by muscle jerks and error-feedback processing in HD patients compared to healthy controls (Smith, Brandt and Shadmehr, 2000). Figure 1.4B (adapted from Briggs et al., 2018) also provides a graphical demonstration of the various projections from different regions that facilitate motor execution for goal directed behaviour.

Furthermore, the fact that the same upper-limb arm reaching paradigm and TMS protocol (SP-TMS delivered at 110% RMT) was used across participants, direct hemispheric asymmetries between regions could be explored in this chapter. Hemispheric asymmetries have been previously reported by researchers, for example when participants performed hand movements Verstynen *et al.*, (2005) noted greater neural activity in the left M1 during right hand movements compared to the right M1. Other studies have similarly found that right handed movement result in much greater neural activity in the left hemisphere compared to the right hemisphere (Cramer *et al.*, 1999; Schlerf *et al.*, 2014). Lesion studies also provide further evidence for hemispheric asymmetries between structures of the left and right hemisphere. For instance, damage

to the right dPMC has been associated with symptoms of ideomotor apraxia whereas damage to the left dPMC has been linked to dystonia (Schnider et al., 1997; Ceballos-Baumann and Brooks, 1998; Gross and Grossman, 2008; Wheaton et al., 2008; Beck et al., 2009). Asymmetries have been reported between different regions of the left and right hemisphere, but this has not been explored during a TMS robotic reaching paradigm, which is a novel element in this thesis.

In order to explore the comparative impact of TMS during upper-limb reaching, a three factorial mixed ANOVA was performed. A novel canonical model was then built based on the differential effects of TMS across regions. The overall purpose of this chapter was to demonstrate the relative importance of the role played by each region in motor control and motor adaptation.

12.2 Methodology

All of the experimental data from chapters 4 - 11 were pooled together for this chapter, and this is further explained in section 3.2. Participant demographics for this chapter are outlined in table 3.2. As previously mentioned throughout the chapters, TMS intensity was set for each region individually for the participant's via functional measures (i.e. left M1, right sided BB MEP – and set at 110% RMT). Figure 3.3 demonstrates the coil orientation and location used for the stimulation of the different regions (canonical brain figures adapted from an MRIcron template – Rorden and Brett, 2000). TMS was always delivered at 110% of RMT. The reaching task that was explored in this chapter was based on the motor adaptation protocol outlined in experimental chapters 5 - 11 and is shown in figure 3.5.

12.3 Data acquisition: Kinematics

The only type of data analysed in this chapter were kinematic parameters which were quantified in Matlab 2017b (The MathWorks Inc, Natick MA, USA) (see section 3.6.1).

12.4 Statistical analysis: Kinematics

Section 3.7.3 describes the statistical analysis that was undertaken for this chapter. Following the three-factorial mixed ANOVA (SPSS 23, IBM) that was undertaken for each kinematic variable, the following statistical tests were performed:

 A one-way between-subjects ANOVA was performed for significant REGION kinematic variables (p < 0.05) (collapsed for TIME factor [by averaging TMS pulse times across participants]). Post hoc testing included Independent t-tests for: Left M1 vs. left PPC, SMA, right M1, left dPMC, right dPMC and right PPC; Left PPC vs. SMA, right M1, left dPMC, right PPC and right dPMC; SMA vs. left PPC, right M1, left dPMC, right PPC and right dPMC; Right M1 vs. left PPC, left dPMC, right PPC and right dPMC; Right PPC vs. left dPMC, right dPMC; Left dPMC vs. right dPMC; (significance set at p < 0.002).

- 2) A one-way repeated measures ANOVA was performed for significant CONDITION kinematic variables, with maximum velocity, movement duration, summed error and maximum force collapsed for TIME factor by averaging TMS pulse times across participants. Post hoc testing included paired Student's t-test: FF vs. FAM, FF vs. WO and FAM vs. WO (significance set at p < 0.016).</p>
- A one-way repeated measures ANOVA was performed for significant TIME kinematic variables, followed by paired Student's t-test (vs. T10) (significance set at p < 0.006).
- Significant interactions found in the mixed ANOVA are graphically presented in the results section.

Overall, data are presented as mean values, \pm the standard error of the mean (SEM) (REGION, CONDITION and TIME).

12.5 Results

12.5.1 Kinematics

12.5.1.1 Mixed ANOVA findings:

The results of the three-factorial mixed ANOVA (table 12.1) revealed a significant effect for REGION on summed error (p < 0.05) but not movement onset, movement offset, maximum velocity, movement duration and maximum force (p > 0.05). There was a significant effect for CONDITION on movement onset, movement offset, maximum velocity, movement duration, summed error and maximum force (p < 0.05). A significant effect was found for TIME on movement onset and movement offset (p < 0.05) but not maximum velocity, movement duration, summed error and maximum force (p > 0.05). A significant interaction for REGION*CONDITION was found for movement onset, maximum velocity and summed error (p < 0.05) but not movement duration and maximum force (p > 0.05). The findings also revealed a significant REGION*TIME interaction for movement onset and movement offset (p < 0.05) but not maximum velocity, movement duration, summed error and maximum force (p > 0.05) but not movement duration and maximum force (p > 0.05). The findings also revealed a significant REGION*TIME interaction for movement onset and movement offset (p < 0.05) but not maximum velocity, movement duration, summed error and maximum force (p > 0.05).

force. No significant interactions for CONDITION*TIME and REGION*CONDITION*TIME were found (for all kinematic variables p > 0.05).

12.5.1.2 One way ANOVA and post-hoc testing findings for REGION:

The one-way between-subjects ANOVA (table 12.2A) revealed a significant effect for summed error for FF, FAM and WO reaching between regions stimulated (all p < 0.05) (TMS blocks).

Post-hoc testing (table 12.2B and figure 12.2[C-E]) for summed error (REGION factor) during FAM reaching was significantly greater for left M1 stimulation compared to right M1, left dPMC, right PPC and right dPMC stimulation (p < 0.002) but not left PPC and SMA stimulation (p > 0.002). With regards to FF reaching, summed error was significantly greater during left M1 stimulation compared to SMA, right M1, left dPMC, right PPC and right dPMC stimulation (p < 0.002) but not when compared to left PPC stimulation (p > 0.002). Summed error during WO reaching was significantly greater with left M1 stimulation when compared to all other regions (p < 0.002). With regards to FAM reaching summed errors, there was no significant difference between left PPC stimulation compared to SMA stimulation (p > 0.002), however summed error was significantly greater during left PPC stimulation compared to right M1, left dPMC, right PPC and right dPMC stimulation (all p < 0.002). Summed error during FF reaching was significantly greater with left PPC stimulation compared to SMA, right M1, left dPMC, right PPC and right dPMC stimulation. WO reaching summed errors with left PPC stimulation was significantly greater when compared to right M1, right PPC and right dPMC stimulation (p < 0.002) but not when compared to SMA and left dPMC stimulation responses (p > 0.002). FAM summed error responses with SMA stimulation was significantly greater compared to right M1, left dPMC and right dPMC responses (p < 0.002) but not compared to the right PPC (p > 0.002).

Summed error during FF reaching with SMA stimulation was significantly greater compared right PPC and right dPMC stimulation (p < 0.002). However, there were no significant differences in FF summed error when comparing the SMA to the right M1 and left dPMC (p > 0.002). With regards to WO responses, summed error was significantly greater during SMA stimulation compared to right M1, left dPMC, right PPC and right dPMC stimulation. Summed error FAM responses during right M1 stimulation was significant greater compared to right PPC stimulation (p < 0.002), but

not when compared to the left dPMC and right dPMC (p > 0.002). FF summed error responses were only significantly greater with right M1 stimulation when compared to right PPC and right dPMC stimulation (p < 0.002), but not when compared to the left dPMC. WO responses were not significantly different between the right M1 and the left dPMC, right PPC and right dPMC (p > 0.002). Summed error during FAM and FF reaching with left dPMC stimulation was significantly greater compared to right PPC and right dPMC stimulation (p < 0.002), but this was not the case with WO reaching (p > 0.002). No significant differences were found in FAM, FF and WO reaching when comparing right PPC and right dPMC summed error responses (p > 0.002).

12.5.1.3 One way ANOVA and post-hoc testing findings for CONDITION:

Although the overall N (94) was larger in this analysis, the results obtained for CONDITION corroborate the results that were found in the previous experimental chapters. See section 5.5.2 (left M1), section 6.6.1 (left PPC), section 7.5.1 (SMA), section 8.5.1 (right M1), section 9.5.1 (left dPMC), 10.5.1 (right PPC) and section 11.5.1 (right dPMC) for results. A summary of the significant findings are illustrated in table 12.3A.

12.5.1.4 One way ANOVA and post-hoc testing findings for TIME:

The results obtained for TIME reflect what was found in the previous experimental chapters (5 - 11). See section 5.5.2 (left M1), section 6.6.1 (left PPC), section 7.5.1 (SMA), section 8.5.1 (right M1), section 9.5.1 (left dPMC), 10.5.1 (right PPC) and section 11.5.1 (right dPMC) for results. The summative results for TIME are illustrated in table 12.3B.

12.5.1.5 Significant interaction effects:

A significant REGION*CONDITION interaction was found for movement onset, maximum velocity and summed error (p < 0.05) but not movement offset, movement duration and maximum force (p > 0.05). Therefore, significant parameters were not collapsed for time, and these results are graphically demonstrated in figures 12.4[A-C]. A significant REGION*TIME interaction was found for movement onset and movement offset (p < 0.05) but not for maximum velocity, movement duration, summed error and maximum force (p > 0.05). These were not collapsed for time because of the significant TIME factor within the interaction.

REGION*TIME interactions are demonstrated in figures 12.5[A-B].

| | Moven | nent onset | (ms) | Moveme | ent offset | (ms) | Maximum velocity (m.s ⁻¹) | | Movemen | nt durati | on (ms) | Sun | nmed err | or | Maxii | num force | e (N) | |
|-------|----------|------------|---------|----------|------------|---------|---------------------------------------|-------|---------|-----------|---------|---------|----------|-------|---------|-----------|--------|---------|
| | Df | F | Sig. | Df | F | Sig. | Df | F | Sig. | Df | F | Sig. | Df | F | Sig. | Df | F | Sig. |
| | (Error) | | | (Error) | | | (Error) | | | (Error) | | | (Error) | | | (Error) | | |
| R | 6 | 0.290 | 0.940 | 6 | 0.632 | 0.704 | 6 | 0.591 | 0.736 | 6 | 1.506 | 0.186 | 6 | 5.883 | < 0.001 | 6 | 0.983 | 0.442 |
| | [87] | | | [87] | | | [87] | | | [87] | | | [87] | | | [87] | | |
| С | 1.9 | 18.81 | < 0.001 | 2 | 13.80 | < 0.001 | 2 | 13.43 | < 0.001 | 2 | 42.30 | < 0.001 | 1.5 | 78.85 | < 0.001 | 1.2 | 1491.6 | < 0.001 |
| | [163.0] | | | [12] | | | [12] | | | [12] | | | [124.0] | | | [96.4] | | |
| Т | 4.8 | 104.61 | < 0.001 | 6.5 | 52.91 | < 0.001 | 5.3 | 0.689 | 0.638 | 6.7 | 1.169 | 0.320 | 6.0 | 1.126 | 0.346 | 4.9 | 1.107 | 0.355 |
| | [416.7] | | | [558.2] | | | [453.2] | | | [578.4] | | | [517.6] | | | [420.2] | | |
| R*C | 11.3 | 1.865 | 0.046 | 11.5 | 1.457 | 0.149 | 11.3 | 2.225 | 0.015 | 11.8 | 0.888 | 0.558 | 8.6 | 2.685 | 0.008 | 6.7 | 0.499 | 0.825 |
| | [163.0] | | | [165.6] | | | [163.2] | | | [171.0] | | | [124.0] | | | [96.4] | | |
| R*T | 28.8 | 1.613 | 0.025 | 38.5 | 1.456 | 0.040 | 31.3 | 0.475 | 0.994 | 38.9 | 0.650 | 0.954 | 35.7 | 1.430 | 0.054 | 29.0 | 0.524 | 0.982 |
| | [416.7] | | | [558.2] | | | [453.2] | | | [578.4] | | | [517.6] | | | [420.11] | | |
| C*T | 11.6 | 1.579 | 0.095 | 12.2 | 1.632 | 0.076 | 6.7 | 0.847 | 0.544 | 11.8 | 1.285 | 0.223 | 9.1 | 0.980 | 0.455 | 5.3 | 0.820 | 0.541 |
| | [1002.0] | | | [1057.6] | | | [579.3] | | | [1020.3] | | | [784.7] | | | [457.6] | | |
| R*C*T | 69.2 | 0.756 | 0.930 | 73.0 | 0.842 | 0.823 | 49.0 | 0.563 | 0.987 | 70.4 | 0.738 | 0.947 | 52.2 | 1.211 | 0.147 | 31.6 | 0.417 | 0.998 |
| | [1002.0] | | | [1057.6] | | | [579.3] | | | [1020.3] | | | [784.7] | | | [457.6] | | |

| Table 12.1. Resul | ts of the three-fact | torial mixed ANOV | A for TMS bloc | ks of reaching: |
|-------------------|----------------------|-------------------|----------------|-----------------|
| | | | | <u> </u> |

Table 12.2A. REGION specific summed error values (mean $[\pm SEM]$) for the different reaching conditions, followed by a one-way between subjects ANOVA for summed error (collapsed for TIME):

| | Summed error - TMS | | | | | | |
|------------|--------------------|-------------|-------------|--|--|--|--|
| REGION | FAM | FF | WO | | | | |
| Left M1 | 2.60 [0.20] | 4.71 [0.38] | 3.17 [0.38] | | | | |
| Left PPC | 2.73 [0.21] | 4.60 [0.53] | 2.53 [0.20] | | | | |
| Left dPMC | 1.97 [0.12] | 3.31 [0.30] | 2.29 [0.23] | | | | |
| SMA | 2.52 [0.23] | 3.31 [0.22] | 2.66 [0.21] | | | | |
| Right M1 | 2.06 [0.17] | 3.64 [0.49] | 2.17 [0.13] | | | | |
| Right PPC | 2.31 [0.18] | 3.04 [0.30] | 2.24 [0.13] | | | | |
| Right dPMC | 2.24 [0.30] | 2.89 [0.17] | 2.27 [0.24] | | | | |

| | One-way ANOVA (between subjects design) | | | | | | | |
|-----|---|------|---------|--|--|--|--|--|
| | TMS (collapsed for TIME) | | | | | | | |
| | Df [Error] | F | Sig. | | | | | |
| FAM | 6 [87] | 2.25 | 0.046 | | | | | |
| FF | 6 [87] | 5.50 | < 0.001 | | | | | |
| WO | 6 [87] | 3.04 | 0.009 | | | | | |

| REGI | ON | | | | |
|---------------|------------|---------|----------------|---------|--|
| | | | Summed error - | - TMS | |
| Comparisons | | FAM | FF | WO | |
| - | | (Sig.) | (Sig.) | (Sig.) | |
| Left M1 Vs. | Left PPC | 0.071 | 0.609 | 0.071 | |
| | SMA | 0.150 | < 0.001 | 0.150 | |
| | Right M1 | < 0.001 | < 0.001 | < 0.001 | |
| | Left dPMC | < 0.001 | < 0.001 | < 0.001 | |
| | Right PPC | < 0.001 | < 0.001 | < 0.001 | |
| | Right dPMC | < 0.001 | < 0.001 | < 0.001 | |
| Left PPC Vs. | SMA | 0.005 | < 0.001 | 0.005 | |
| | Right M1 | < 0.001 | < 0.001 | < 0.001 | |
| | Left dPMC | < 0.001 | < 0.001 | < 0.001 | |
| | Right PPC | < 0.001 | < 0.001 | < 0.001 | |
| | Right dPMC | < 0.001 | < 0.001 | < 0.001 | |
| Left dPMC Vs. | Right PPC | < 0.001 | 0.002 | < 0.001 | |
| | Right dPMC | < 0.001 | < 0.001 | < 0.001 | |
| SMA Vs. | Right M1 | < 0.001 | 0.009 | < 0.001 | |
| | Left dPMC | < 0.001 | 0.975 | < 0.001 | |
| | Right PPC | 0.004 | 0.001 | 0.004 | |
| | Right dPMC | < 0.001 | < 0.001 | < 0.001 | |
| Right M1 Vs. | Left dPMC | 0.173 | 0.008 | 0.173 | |
| | Right PPC | 0.002 | < 0.001 | 0.002 | |
| | Right dPMC | 0.005 | < 0.001 | 0.005 | |
| Right PPC Vs. | Right dPMC | 0.211 | 0.132 | 0.211 | |

Table 12.3B: Post hoc testing results for summed error TMS (collapsed for TIME) (REGION main factor):





Symbols represent:

Significant difference between left M1 and other cortical regions stimulated = * Significant difference between left PPC and other cortical regions stimulated = * Significant difference between left dPMC and other cortical regions stimulated = * Significant difference between SMA and other cortical regions stimulated = * Table 12.3A. One-way ANOVA findings for CONDITION main factor in TMS blocks:

| | | CONDITION | | | | | | | | | | |
|------------|---------------------|----------------------|---------------------------------------|------------------------|-----------------------------|-------------------|--|--|--|--|--|--|
| | Movement onset (ms) | Movement offset (ms) | Maximum velocity (m.s ⁻¹) | Movement duration (ms) | Summed error (distance: cm) | Maximum force (N) | | | | | | |
| Left M1 | \checkmark | ~ | ✓ | \checkmark | ✓ | ~ | | | | | | |
| Left PPC | ✓ | * | ✓ | \checkmark | ✓ | ~ | | | | | | |
| Left dPMC | * | ~ | ✓ | \checkmark | ✓ | \checkmark | | | | | | |
| SMA | \checkmark | * | ✓ | \checkmark | ✓ | \checkmark | | | | | | |
| Right M1 | \checkmark | ✓ | ✓ | \checkmark | ✓ | ~ | | | | | | |
| Right PPC | \checkmark | ~ | × | \checkmark | ✓ | ~ | | | | | | |
| Right dPMC | × | ~ | × | \checkmark | ✓ | ~ | | | | | | |

Table 12.2B. One-way ANOVA findings for TIME main factor in TMS blocks:

| | TIME | |
|------------|---------------------|----------------------|
| | Movement onset (ms) | Movement offset (ms) |
| Left M1 | \checkmark | √ |
| Left PPC | 1 | √ |
| Left dPMC | √ | ✓ |
| SMA | √ | ✓ |
| Right M1 | √ | ✓ |
| Right PPC | ~ | ✓ |
| Right dPMC | ✓ | ✓ |

Symbols represent:

 \checkmark = Significant ANOVA results (p < 0.05)

***** = Non-significant ANOVA results (p > 0.05)

196



Regions colour coded as:

Left M1, left PPC, left dPMC, SMA, right M1, right PPC and right dPMC

Figure 12.5 [A-B]. Graphical illustration of REGION*TIME interaction effects in TMS blocks:



Regions colour coded as:

Left M1, left PPC, left dPMC, SMA, right M1, right PPC and right dPMC

198
12.6 Discussion:

Previous studies exploring the impact of TMS on preparation for upper-limb reaching have not targeted a variety of regions using robotic methods but have rather targeted one or two regions using different types of reaching paradigms (such as pointing-to-reach tasks) and TMS protocols (rTMS and twin-coil methods). Throughout the motor adaptation paradigms in this thesis (chapters 5 - 11) the same reaching task and TMS protocol (SP TMS, 110% RMT, and TMS delivered randomly at different time points of reaching preparation) was undertaken. The use of consistent paradigms enabled direct kinematic comparisons between different cortical regions in this chapter. Overall, this final chapter provided a novel insight into the different levels of influence across regions of the motor network and demonstrated that TMS-based disruption to different cortical structures had a varied impact on motor control (FAM) and motor adaptation (FF) as measured by summed error and some interactions for timing of reaching.

12.6.1 Kinematics:

12.6.1.2 Region specific relative influence defined by summed error

The main finding from this experimental chapter was that TMS had a region-specific impact on motor control and motor adaptation because summed error was significantly different between regions across the different reaching conditions (FAM, FF and WO). Post hoc testing revealed the relative importance of a regions role in preparation for reaching, for example, TMS to the left M1 and the left PPC resulted in greater summed error responses during FF reaching when compared to all other regions. Based on this finding, it could therefore be inferred that based on this comparative chapter that the left M1 and left PPC play a greater role in the preparation for motor adaptation during upper-limb reaching compared to other regions in the motor network, such as the SMA, left and right dPMC and right PPC. These findings are in line with studies that have similarly demonstrated the important role of left M1 and left PPC in reaching behaviours. For example, Cavina-Pratesi, Connolly and Milner (2013) found that patients with optic ataxia due to left PPC lesions had arm reaching deficits, but not reaching-to-grasp deficits. TMS studies have also shown that virtual disruptions to the

left PPC compared to a control area (e.g. the occipital lobe) following movement onset resulted in impaired reaching trajectories (Della-Maggiore, *et al.*, 2004). This experiment provided a novel finding because it demonstrated that the left PPC is not only involved in reaching execution (as previously demonstrated by Della-Maggiore, *et al.*, 2004) but also reaching preparation and TMS to the left PPC during reaching preparation can result in reaching trajectories that resemble a model of optic ataxia. Left M1 stimulation also had a significant detrimental impact on reaching during motor adaptation. As well as suggesting an obvious direct role of left M1 in right arm reaching, this could have been due to TMS affecting other cortical structures (such as the cerebellum) (via trans-synaptic transmission) that facilitate M1 functioning for motor output (Chouinard *et al.*, 2003; Richardson *et al.*, 2006). Direct TMS disruption of cerebellar regions and their possible roles in motor adaptation would support this avenue of future work and the use of double-coil TMS induced disruption of cerebello-M1 connection function during motor adaptation.

Although TMS to the left M1 and left PPC resulted in a greater disruption to reaching preparation compared to other regions (e.g. the SMA), we should be cautious and not de-emphasise the role of the SMA particularly when experimental findings from previous chapters in this thesis (chapter 7) demonstrated its given role in movement preparation. This is supported by studies that have undertaken different motor protocols. For instance, in an fMRI motor imagery and execution task, Kasess et al., (2008) found that the SMA caused a suppression in movement planning and preparation which resulted in inhibited M1 activity (Kim et al., 2018). Studies in PD patients have also provided evidence for the importance of the SMA in motor preparation and execution (Chung et al., 2018). The SMA is strongly connected with the basal ganglia and has been found to be disrupted in PD patients (Jenkins et al., 1992; Haslinger et al., 2001; Chung et al., 2018). For instance, imaging studies have revealed reduced cerebral blood flow in the SMA of PD patients whilst they were performing a motor task with a joystick (Playford et al., 1992; Chung et al., 2018). Patients with PD also have increased neuronal desynchronisation during movement preparation, before movement onset (Rowland et al., 2015; Chung et al., 2018). Based on this notion, it can be argued that TMS effected reaching preparation because it disrupted communication between cortical networks that are connected with the SMA including the basal ganglia. Furthermore, the role of the SMA and left dPMC should not be disregarded because this comparative chapter revealed that TMS to these two regions during motor adaptation resulted in significantly greater summed error responses compared to other regions, such as the right PPC during preparation for FF reaching. Researchers have argued that that the SMA and PMC play an important role in facilitating the 'Bereitschaft (readiness) potential' in a motor task during movement preparation (Deecke and Kornhuber, 1987; Di Russo et al., 2005; Nguyen, Breakspear and Cunnington, 2014; Verleger et al., 2016). For example, by using a dipole EEG analysis method in a finger flexion task, Di Russo et al., (2005) revealed that the SMA and PMC produce these potentials prior to movement onset. Lesion studies have also shown that patients with SMA lesions have impaired Bereitschaft potential features (Deecke et al., 1978; Di Russo et al., 2005) thus providing further evidence for the role of the SMA in "premovement potential" generation. Therefore, it can be suggested that in this study TMS disrupted the Bereitschaft potential process, thereby causing increased SMA and left dPMC-based summed error responses. However, it has been argued that the impact of the SMA in facilitating the Bereitschaft potential depends on the type of motor task being performed (Praamstra et al., 1996). For example, when subjects performed fixed vs. freely selected movements with a joystick, Praamstra et al., (1996) reported that the movement potentials eliciting the Bereitschaft potential in the SMA were less affected when subjects performed fixed movements compared to freely selected movements. Furthermore, the complexity of a motor task has also been found to influence the Bereitschaft potential (Shibasaki and Hallett, 2006), a notion supported by Simonetta et al., (1991) who concluded that sequential movements compared to simple movements led to both larger and earlier Bereitschaft potential onset (Shibasaki and Hallett, 2006). Similar findings were illustrated by Benecke et al., (1985). Shibasaki and Hallett (2006) also noted that the speed of a movement, specifically faster motions were associated with a later Bereitschaft potential onset. Based on such studies, it could be argued that the reason as to why TMS did not severely affect SMA functioning was due to the nature of the task, i.e. simple, fixed reaching movements took place compared to complex sequential movements. Furthermore, studies have also shown that the Bereitschaft potential can arise from other regions depending on the task performed. For example, Wheaton, Shibasaki and Hallett (2005) explored how movement related cortical potentials are affected during everyday motor tasks in right handed participants.

Their findings revealed that the Bereitschaft potential developed from the left parietal cortex during movement planning and preparation (Shibasaki and Hallett, 2006). Considering this, it could be suggested that in this thesis TMS had a greater impact on the left PPC because the Bereitschaft potential firstly developed from the parietal cortex and then the medial-frontal areas (Wheaton, Shibasaki and Hallett, 2005; Shibasaki and Hallett, 2006). Therefore, a number of factors ranging from task complexity to physiological features (e.g. the Bereitschaft potential) can help explain why some regions were more severely impacted with TMS compared to others.

Furthermore, resting-state fMRI studies have demonstrated the importance of different connecting pathways in the motor circuit that facilitate motor function. For instance, studies in stroke patients with upper-limb impairment have shown that increased functional connectivity between the affected M1 (i.e. ipsilesional hemisphere) and areas such as the SMA and frontal lobe resulted in enhanced motor recovery (Park *et al.*, 2011; Auriat *et al.*, 2015). Other important connecting pathways include the ipsilesional M1 and the contralesional dPMC (Bestmann et al., 2010; Auriat *et al.*, 2015). For instance, a TMS-fMRI (dual-coil) study conducted by Bestmann *et al.*, (2010) in stroke patients revealed that a decrease in inhibition from the affected dPMC to the unaffected dPMC was associated with greater motor deficits. Based on these studies, it could be argued that TMS to one specific region had a distinct effect on the networks associated with that region, which provides an explanation for the varied region-specific impact of TMS.

The fact that TMS had differential effects in motor control and motor adaptation provides evidence for the task specific properties of different nodes in the motor network (Buchel, Coull and Friston, 1999; Wei *et al.*, 2018) which has been supported by neuroimaging studies. For example, in an fMRI motor task where participants were instructed to perform: 1) right ankle movement, 2) agonist muscle movement and 3) movement with pressure-stimulation, Wei *et al.*, (2018) found that agonist muscle movement compared to both right ankle movement and pressure-stimulation resulted in the greatest and most dispersed activity in in areas ranging from the thalamus to the left M1. Agonist muscle activity also takes place in motor adaptation protocols (Pizzamiglio *et al.*, 2017b). It can therefore be argued that in this experiment TMS may have disrupted the spread of cortical activity related to agonist muscle movement, thereby resulting in increased trajectory reaching errors.

12.6.1.3 Evidence for hemispheric asymmetries during motor adaptation

Post hoc testing results for summed error (table 12.2B) provides evidence for hemispheric asymmetries because some cortical structures between the left and right hemisphere responded differently during motor adaptation. This was particularly evident for the left PPC vs. the right PPC. These results are in line with lesion-studies that have also demonstrated asymmetries between regions. For example, researchers have reported that damage to the right PPC can result in spatial neglect whereas damage to the left PPC can result in right-handed defects during reaching (Perenin and Vighetto, 1988; Husain and Nachev, 2007; Koch *et al.*, 2011; Andersen et al., 2014). Such studies have therefore highlighted separate functional roles of the two regions, with the right PPC involved in visual/spatial attention and the left PPC involved in visually guided reaching (Della-Maggiore *et al.*, 2004; Koch *et al.*, 2011). The results for summed error support literature findings regarding the greater functional role of the left PPC (compared to the right PPC) in 1) visually guided-reaching and 2) novel reaching behaviours.

12.6.1.4 Regions and conditions impacted for maximum velocity (REGION*CONDITION)

As well as the region-specific impact on summed error, a significant REGION*CONDITION interaction was found for maximum velocity (table 12.1 and figure 12.4[A-C]. The findings revealed that regions such as the right dPMC and right PPC were highly impacted with TMS for maximum velocity during motor control [FAM and WO reaching] but not so much during motor adaptation [FF reaching]. This was not the case for other regions such as the left PPC, left M1, SMA, right M1 and left dPMC. Studies in non-human primates have revealed that the dPMC plays a critical role in encoding kinematic parameters of a movement (Alexander and Crutcher, 1990; Fu, Suarez and Ebner, 1993; Ward *et al.*, 2007), based on this it can be suggested that when TMS is delivered to the dPMC, it will have an impact on parameters such as velocity. However, it is important to note that TMS to the left dPMC did not impact on motor control, but rather during motor adaptation and this can be due to the functional asymmetries that have been reported between the left and right dPMC. For example, greater neural activity has been noted in the left dPMC compared to the right dPMC during the learning phase of a motor task (Hardwick *et al.*, 2013). Furthermore, it has

been argued that the left dPMC in contrast to the right dPMC facilitates movement selection, this has been supported by both left dPMC virtual disruption studies and clinical population studies as subjects were found to perform poorly during response selection tasks (Halsband *et al.*, 1993; Mochizuki *et al.*, 2005; Kantak *et al.*, 2012; Hardwick *et al.*, 2013). Considering these studies, it could be argued that TMS will have a greater impact in regions that play a more functional role in the learning of a new motor task, which is in line with what was found in this experiment (e.g. left dPMC was impacted more with TMS during MA compared to FAM and WO reaching).

12.6.1.5 Limitations to consider

This chapter revealed a region-specific influence for summed error particularly for the left M1 and left PPC compared to other regions that were targeted. However, this can be considered as a limitation of the experimental approach that was undertaken to disrupt normal processing during the preparation phase of reaching. For example, it could be argued that if partipants had performed complex tasks, including sequential movements towards different reaching targets compared to fixed movements towards one target, TMS to areas such as the SMA and dPMC may have had a more detrimental impact on summed error responses during reaching preparation (based on findings from Simonetta et al., 1991 and Praamstra et al., 1996). This notion has also been supported by Shibasaki et al., (1993). In a finger movement protocol using PET imaging methods, they revealed that task complexity increased cerebral blood flow in the SMA and M1 because the demanding nature of the task affected the components associated with movement-related cortical potentials (Shibasaki and Hallett, 2006). Therefore, if this study were to be replicated, altering the nature of the reaching task would be an important factor to take into account, i.e. making the task more demanding for the subject. However, there is a trade off between making the task more difficult and avoiding fatigue. Nonetheless, shortening the reaching paradigm could provide a solution in ensuring subject comfort during a complex and fatiguing motor task. Nonetheless, it important to not under emphasise the role of other regions that were targeted with TMS because 1) previous chapters revealed that TMS did have an impact on other structures targeted during preparation for novel reaching and, 2) table 12.2A revealed larger summed error responses during FF vs. FAM reaching for all regions, thus demonstating each regions' important functional role in movement preparation.

Furthermore, using a combination of methods could have led to more in depth findings regarding the neurophysiological impact of TMS during reaching preparation. For instance, if TMS-EEG or TMS-fMRI was employed, observing cortical potentials (with EEG) or cerebral blood flow (with fMRI) could have helped establish the network effects of TMS across different structures, particularly when considering that TMS does not only have a direct affect on the area that is being stimulated, but it also has an impact on regions connected with the stimulated area (Ruff, Driver and Bestmann, 2009). If this study was repeated it would be useful to employ additional imaging methodologies in conjuction with TMS. Mixed methodologies would enable different statistical analysis to be carried out. For example, neural activation patterns could be correlated with different performance measures (e.g. summed error) during preparation for novel reaching (Ruff, Driver and Bestmann, 2009), therefore both causal and correlational inferences could be drawn with regards to the results obtained (Ruff, Driver and Bestmann, 2009).

Moreover, it is important to consider the sample population that was studied in this thesis; individuals aged 18 - 40. A future consideration would be to develop the study by recruiting other groups of participants, including middle aged and older aged adults (e.g 55+ years). This would be interesting to explore because studies employing TMS protocols have demonstrated that neuroplasiticity in the M1 is reduced in older adults (Fathi et al., 2010; Porto et al., 2015). This notion is supported by Fathi et al., (2010) as their findings concluded reduced MEP amplitude responses in older aged subjects compared to both younger and middle-aged subjects. They argued that the reduced plasticity in older adults could be attributed to impaired cortical networks between the basal-ganglia and motor circuits (e.g. the M1, SMA and sensorimotor regions). In addition to recruiting older adults, recuiting patients with neurological impairments could also help expand the findings of this study. For example, the comparative chapter in thesis revealed the importance of the left PPC, and chapter 6 also demonstrated that a patient-like model of optic ataxia was created with the virtual lesion TMS approach (as the results were similar to clinical findings from Pisella et al., 2000). Therefore, directly studying motor adaptation in patients (such as those with optic ataxia) could help: 1) provide a comparison group for analysis purposes (i.e. healthy group vs. patient group), 2) explore the strength of TMS as a virtual lesion tool, and 3) further investigate the hypothesis regarding the role of the left PPC for the preparation of novel reaching.

Overall, if the studies in this thesis were to be replicated a number of factors would be considered, ranging from implementing a more complex motor reaching paradigm to employing other imaging methods in conjuction with TMS. In addition to this, it would also be interesting to explore other sample-populations, such as older-aged adults or clinical patients with neurological impairments. Despite these limitations, novel findings were illustrated, and the relevance of these findings with regards to future directions (e.g. in the field of neurorehabilitation and neurocomputional models) are discussed in chapter 13.

12.7 Conclusion:

This experiment provided novel findings by demonstrating the comparative impact of TMS to various nodes of the motor network. The main findings revealed a statistical difference in summed error between different cortical regions. More specifically, areas such as the left M1 and left PPC were found to play a significant role in motor adaptation compared to other regions such as the SMA, right M1, left dPMC, right PPC and right dPMC. Based on these findings it can be argued that damage to areas including the left M1 and left PPC may result in greater impairments during upper-limb reaching during motor adaptation, which is line with patient studies (e.g. stroke and traumatic brain injury patients). Furthermore, the fact that TMS had a varied impact on different regions provides evidence for the notion that different areas of the motor network play a different functional role in motor control and adaptation. This was illustrated by the interactions that were found, whereby the right dPMC and right PPC were highly impacted for maximum velocity during motor control and motor adaptation whereas other regions were only highly impacted during motor adaptation. Functional asymmetries that have been found between regions of the two hemispheres explains why TMS affected some regions more than others. Overall, this final chapter begins to build a unique and novel model for studying the motor network during a range of tasks including reaching and as a function of healthy ageing or disease whereby the regions may operate in compensatory roles resulting in a different pattern or signature of regionspecific importance.

Chapter 13

Final conclusion and Future directions

This thesis outlined a novel approach in studying the network nodes in the motor circuit during an upper-limb robot-mediated reaching tasks. The approach taken to explore behavioural responses using TMS protocols in this thesis was novel because it was the first attempt to describe the different behavioural effects (kinematic responses) of SP-TMS delivered to a number of cortical structures at nine different time points during the preparation phase of a reach with a robot-mediated force-field paradigm. The overall results that were obtained in this thesis were also novel as the findings identified regions of the motor circuit that play a more functional role in novel reaching. For instance, both individual chapters (chapter 5 and 6) and the final comparative chapter revealed that the left M1 and left PPC were similarly and most significantly disrupted with TMS during novel reaching.

These findings have a wide range of future implications, one of which includes its application in brain machine interfaces (BMI) in the field of neurorehabilitation. BMI's are based on obtaining brain signals, interpreting them into commands which are then transferred to devices to perform actions (Shih, Krusienski and Wolpaw, 2012). BMIs have therefore been used to control functions in patients with impairments (McFarland and Vaughan. 2016) by providing the brain with channels based on cortical activity (Ranjangam et al., 2016; Shih, Krusienski and Wolpaw, 2012). BMI applications have been used to help assist functional recovery among clinical populations, ranging from patients with stroke to those with multiple sclerosis (Shih, Krusienski and Wolpaw, 2012; Ramos-Murguialday et al., 2019). Ramos-Murguialday et al., (2019) reported that BMI training in stroke patients to assist upper-limb movement (i.e. the paretic arm) with a robotic device resulted in an increase in a clinical outcome measure, Fugl-Meyer score, therefore demonstrating its effectiveness in the field of neurorehabilitation and aiding functional recovery. The use of TMS with a virtual disruption approach has helped researchers identify regions that are important for particular functions (Bolognini and Ro, 2010; Carmena and Cohen, 2012) and due to this stimulation methods have been implemented in the BMI field (Reis et al., 2009; Carmena and Cohen, 2012). The final chapter of this study demonstrated the functional role of the left M1 compared to other regions of the motor network (e.g. right PPC and right dPMC) in novel motor behaviours. The role of the ipsilateral M1 should also not be ignored, as findings from this thesis demonstrated its importance in novel reaching. Therefore, the data from both the left M1 and right M1 can be used as a BMI signal source to aid functional recovery in stroke patients (Carmena, *et al.*, 2003; Schroeder and Chestek, 2016; Hatsopoulos and Suminksi, 2011).

BMI approaches are not only confined to assisting motor recovery but also cognitive recovery (Hauschild *et al.*, 2012) and this is important in terms of the left PPC findings in this thesis. For instance, studies in non-human primates have shown that extracted PPC signals increase performance rates, specifically reaching trajectories and this can be important in helping assist patients with left PPC lesions such as those with optic apraxia (Hauschild *et al.*, 2012). Overall this thesis has demonstrated regions that are involved in novel reaching that can be used as a site for signal extraction for BMI approaches (Ranjangam *et al.*, 2016; Andersen *et al.*, 2014).

The findings from this thesis also revealed that although the SMA, right M1, left dPMC and were not as disrupted as the left M1 and left PPC, their roles in preparation for novel reaching should not be de-emphasised, as the individual experimental chapters revealed their role in novel reaching too. For instance, with regards to the SMA, the findings from chapter 7 demonstrated that reaching trajectories were significantly impaired in FF reaching compared to FAM and WO reaching (albeit, not at all time time points). Additionally, the results from chapter 9 illustrated significant disruptions at all time points when TMS was delivered to the left dPMC. These findings are relevant within the neurorehabilitation field because they have helped provide further evidence regarding the importance of left dPMC- and SMA- M1 connectivity in aiding functional recovery among stroke patients. For example, studies have shown that rTMS delivered to the intact hemisphere of stroke patients enhanced neural communication between the affected SMA and M1, which led to improved motor performance when patients performed paretic hand movements (Grefkes et al., 2010; Liew et al., 2014). Therefore, the results from this thesis demonstrated the critical role of regions in the motor circuit (other than the M1 and left PPC) that facilitate motor preparation and execution.

Moreover, Huang and Krakauer (2009) reported that the use of robotic protocols in healthy participants have helped researchers gain an understanding into the mechanisms of motor control and motor learning. Based on this notion, it could be argued that this thesis demonstrated an insight into methodological factors that are important in the field of neurorehabilitation. For example, the findings highlighted that subjects were able to optimise their reaching following four blocks of no-TMS motor adaptation reaching trials (which is in line with motor adaptation research - Pizzamiglio et al, 2017a; Pizzamiglio et al., 2017b). Therefore, implementing and expanding such motor protocols in the field of neurorehabilitation can be considered, particularly when taking into account that researchers such as Patton et al., (2006) have demonstrated that resistive robotic tasks compared to assistive tasks resulted in better therapeutic outcomes in stroke patients. However, the principles of neuroplasticity during motor adaptation remain under-researched and some have argued that demanding and complex motor learning tasks facilitate neuroplasticity compared to repetitive motor adaptation tasks (Turner et al., 2013). Therefore, whether this particular model can be used for motor recovery among patients can be questioned. Nonetheless, further investigations into methodological factors in motor protocols (such as, the type of motor task undertaken, the duration of the task and number of sessions) can help build the foundations for new research within the field neurorehabilitation (Huang and Krakauer, 2009).

Furthermore, the main findings from the right M1 revealed that TMS had an disruptive impact on reaching trajectories particularly during force-field reaching compared to normal reaching (FAM and WO) (figure 12.2D and table 12.2A). This finding confirmed what previous researchers have noted regarding right M1 function during ipsilateral reaching (right arm) and that the right M1 also undergoes modulations of activity during right arm reaching (Van den Berg, Swinnen and Wenderoth, 2011). This result is not only useful for BMI neurorehabilitation applications (i.e. in stroke patients where the contra-lesional hemisphere becomes important in guiding ipsilateral functions - Dancause *et al.*, 2006; Ganguly *et al.*, 2009) but the finding can also be applied within neuro-computational models regarding data-based virtual lesions that have been incorporated into patient models (for example, stroke patients). Using TMS to create patient models via a virtual disruption can help researchers overcome many of the challenges faced when investigating the effects of lesions on functional connectivity in clinical populations (Sliwinska, Vitello and Devlin, 2014). For example, Sliwinska, Vitello and Devlin (2014) reported that the use of TMS: 1) has greater degrees of

accuracy and specific structures can be explored compared to patients lesions studies whereby the damage affects a wide range of structures, and 2) the same subjects can be used to explore the effects of TMS in different regions (i.e. greater degree of control vs. heterogeneous stroke lesion patients). Based on these notions, the data in this thesis can help gain a deeper understanding into how neural networks operate following disrupted functional activity.

It is important to note that findings from this thesis also revealed that TMS to some regions, such as the right PPC and the right dPMC was not so impactful. This could have been due to the methodology that was undertaken. For example, in this thesis only right handed subjects were recruited and studies have shown that TMS to the right dPMC effects left handed functions compared to right handed functions (Schluter et al., 1998). Similarly, the right PPC has been found to be involved in planning left handed reaching movements only compared to the left PPC which has been involved in planning both right and left handed movements (Schluter et al., 2001; Oliveira et al., 2010). In addition to this, imaging studies have revealed that 1) sequential motor tasks and 2) complex motor tasks affect the right dPMC and right PPC as both regions play a role in maintaining selective attention (Poster and Peterson, 1990; Sadato et al., 1996; Malhotra, Coultard and Husain, 2009) whereas a simple motor adaptation task was used in this thesis. Therefore, the future directions of this research would include providing a deeper insight on the function of the right dPMC and right PPC by refining methodological factors, including the motor task undertaken (i.e. using a complex task rather than simple task) and recruiting a different sample population (i.e. both right and left handed subjects). Such research findings can help demonstrate how handedness contributes to motor preparation and execution, and the differences with regards to how cortical regions interact in both left- and right-handed subjects during novel reaching.

Nonetheless, the studies in this thesis were novel as previous researchers that have explored the behavioural effects of TMS have mainly investigated behavioural responses following movement onset compared to exploring responses prior to movement onset (i.e. motor preparation) (*Della-Maggiore et al., 2004*). In addition to this, most researchers have employed; 1) paired-pulse TMS and rTMS paradigms to explore the functional role of cortical regions in motor tasks (Gerloff *et al., 1997;* Serrienet *et al., 2002;* Steyvers *et al., 2003;* Nachev, Kennard and Husain, 2008; Mars *et al., 2009;* Chouinard and Paus, 2010) and 2) different motor paradigms such as visuo-

motor adaptation tasks, finger sequencing tasks, 'point' reaching and 'grasp' reaching tasks (Gerloff et al., 1997; Culham, Cavina-Pratesi and Singhal, 2006; Lee and Van Donkelaar, 2006; Miall et al., 2007) compared to a novel SP-TMS, robotic induced arm reaching perturbation paradigm which was used in this thesis.

Overall, in this thesis a novel TMS mapping method was introduced to create virtual disruptions of different regions in the motor network in order to explore the functional role of different structures in upper-limb motor control and motor adaptation. Despite novel findings illustrating the pivotal role of regions such as the left and right M1, left PPC, SMA and left dPMC which can be implemented in the field of neurorehabilitation (e.g. BMIs) and also in neurocomputational patient models, there are future considerations to take into account. For example, other regions of the motor network can also be targeted such as the cerebellum and the DLPFC as these two structures have been argued to play a key role in encoding kinematic parameters and facilitating motor responses (Overduin, Richardson and Bizzi 2009; Badoud et al., 2017). In addition to this, it would be important to perform a "control" protocol of a NO-TMS paradigm in order to make direct comparisons of TMS virtual-disruption during reaching vs. no TMS during reaching over the course of the same number of trials. Moreover, developing a more demanding motor protocol could also help reveal the importance of regions such as the right PPC and right dPMC in motor control and motor adaptation, as the functioning of these two regions were not as effected with TMS compared to other regions that were targeted. Therefore: 1) stimulating further putative cortical regions in the motor circuit, 2) obtaining control findings with a NO-TMS protocol, and 3) modifying the adaptation protocol could help further refine our understanding of the regions involved in motor control and adaptation.

REFERENCES

- Aizawa, H. and Tanji J. (1994) 'Corticocortical and thalamocortical responses of neurons in the monkey primary motor cortex and their relation to a trained motor task', J Neurophysiol, 71(2), pp. 550 – 560. doi: 10.1152/jn.1994.71.2.550.
- Abdalla. G.M.G. (2011) 'The Optimal Position of the Coil Angle on Scurf by Using Transcranial Magnetic Stimulation on Biceps Branchii' World Journal of Sport Sciences, 5(1), pp. 23 – 26. Available at: https://idosi.org/wjss/5(1)11/5.pdf (Accessed: 14th March 2017).
- Ahonen, J.P., Jehkonen, M., Dastidar, P., Molnar, G. and Hakkinen, V. (1998) 'Cortical silent period evoked by transcranial magnetic stimulation in ischemic stroke', *Electroencephalogr Clin Neurophysiol*, 109(3), pp. 224 – 229. doi: 10.1016/s0924-980x(98)00014-9.
- Akbari Sari, A., Ravaghi, H., Mobinizadeh, M., and Sarvari, S. (2013) 'The Cost-Utility Analysis of PET-Scan in Diagnosis and Treatment of Non-Small Cell Lung Carcinoma in Iran', *Iranian journal of radiology: a quarterly journal published by the Iranian Radiological Society*, 10(2), pp. 61–67. doi:10.5812/iranjradiol.8559.
- Akkal, D., Dum, R.P. and Strick, P.L. (2007) 'Supplementary motor area and presupplementary motor area: targets of basal ganglia and cerebellar output', *J Neurosci*, 27(40), pp. 10659 – 10673. doi: doi.org/10.1523/JNEUROSCI.3134-07.2007.
- Alario, F.X, Chainay, H., Lehericy, S. and Cohen L. (2006) 'The role of the supplementary motor area (SMA) in word production', *Brain Res*, 1076(1), pp. 129 – 143. doi:10.1016/j.brainres.2005.11.104.
- Albus, J.S. (1971) 'A theory of cerebellar function', *Mathematical Biosciences*, 10(1-2), pp. 25 61. doi: 10.1016/0025-5564(71)90051-4.
- Aleman, A., Sommer, I.E. and Kahn, R.S. (2007) 'Efficacy of slow repetitive transcranial magnetic stimulation in the treatment of resistant auditory hallucinations in schizophrenia: a meta-analysis', *Journal of Clinical Psychiatry*, 68(3), pp. 416 – 421. Available at: https://www.ncbi.nlm.nih.gov/pubmed/17388712 (Accessed: 10th October 2018).
- Alexander, A. S. and Hasselmo, M. E. (2018) 'Shedding light on stellate cells', *eLife*, 7, e41041. doi:10.7554/eLife.41041.
- Alexander, G.E. and Crutcher, M.D. (1990). 'Functional architecture of basal ganglia circuits: neural substrates of parallel processing', *Trends Neurosci*, 13(7), pp. 266 – 271. doi: 10.1016/0166-2236(90)90107-L.
- Alm, P. A., Karlsson, R., Sundberg, M. and Axelson, H. W. (2013) 'Hemispheric lateralization of motor thresholds in relation to stuttering', *PloS one*, 8(10), e76824, pp. 1 – 5. doi:10.1371/journal.pone.0076824.

- Alonso, A. and Klink, R. (1992) 'Differential electroresponsiveness of stellate and pyramidallike cells of medial entorhinal cortex layer II', *Journal of Neurophysiology*, 70(1), pp. 128 – 143. Doi: 10.1152/jn.1993.70.1.128
- Amassian, V.E., Cracco, R.Q., Maccabee, P.J., Cracco, J.B., Rudell, A. Eberle, L. (1989) 'Suppression of visual perception by magnetic coil stimulation of human occipital cortex', Electroencephalogr Clin Neurophysiol, 74(6), pp. 458 – 462. doi: 10.1016/0168-5597(89)90036-1.
- Amunts, K., Schlaug, G., Schleicher, A., Steinmetz, H., Dabringhaus, A., Roland, P.E. and Zilles, K. (1996) 'Asymmetry in the human motor cortex and handedness', Neuroimage, 4(3 pt 1), pp. 216 – 222. doi: 10.1006/nimg.1996.0073.
- An, K.N. and Chao, E.Y. (1984) 'Kinematic analysis of human movement', Ann Biomed Eng, 12(6), pp. 585 – 597. Available at: https://www.ncbi.nlm.nih.gov/pubmed/6534225 (Accessed 7th September 2018).
- Andersen, R. A., Andersen, K. N., Hwang, E., and Hauschild, M. (2014) 'Optic ataxia: from Balint's syndrome to the parietal reach region', *Neuron*, 81(5), pp. 967–983. doi: 10.1016/j.neuron.2014.02.025.
- Andersen, R. A., Kellis, S., Klaes, C. and Aflalo, T. (2014). 'Toward more versatile and intuitive cortical brain-machine interfaces', *Current biology: CB*, 24(18), R885–R897, pp. 1 – 31. doi:10.1016/j.cub.2014.07.068.
- Andersen, R.A. and Buneo, C.A. (2002) 'Intentional maps in posterior parietal cortex', AnnualReviewofNeuroscience,25,pp.189–220.doi:10.1146/annurev.neuro.25.112701.142922.
- Andrews, A.W. and Bohannon, R.W. (2000) 'Distribution of muscle strength impairments following stroke', *Clin Rehabil*, 14(1), pp. 79 87. doi: 10.1191/026921500673950113.
- Ang, K.K., Guan, C., Chua, K.S., Ang, B.T., Kuah, C., Wang, C., Phua, K.S., Chin, Z.Y. and Zhang, H. (2010) 'Clinical study of neurorehabilitation in stroke using EEG-based motor imagery brain-computer interface with robotic feedback', *Conf Proc IEEE Eng Med Biol Soc*, 2010, pp. 5549 – 5552. doi: 10.1109/IEMBS.2010.5626782.
- ANT Neuro (1999 2019) Available at: <u>https://www.ant-neuro.com/</u> (EEG cap figure obrained from this web page).
- Armstrong, R.A. (2014) 'When to use the Bonferroni correction', *Ophthalmic Physiol*, 34(5), pp. 502 508. doi: 10.1111/opo.12131.
- Ashburner, J. and Friston, K.J. (2000) 'Voxel-based morphometry--the methods', *Neuroimage*, 11(6 pt 1), pp. 805 821. doi: 10.1006/nimg.2000.0582.
- Ashe, J. and Georgopoulos, A.P. (1994) 'Movement parameters and neural activity in motor cortex and area 5', *Cerebral Cortex*, 4(6), pp. 590–600. doi: 10.1093/cercor/4.6.590.
- Attwell, P.J.F., Cooke, S.F. and Yeo, C.H. (2002) 'Cerebellar Function n Consolidation of a Motor Motory', *Neuron*, 34(6), pp. 1011 1020. doi: 10.1016/S0896-6273(02)00719-5.

- Auriat, A. M., Neva, J. L., Peters, S., Ferris, J. K. and Boyd, L. A. (2015) 'A Review of Transcranial Magnetic Stimulation and Multimodal Neuroimaging to Characterize Post-Stroke Neuroplasticity', *Frontiers in neurology*, 6(article 226), pp. 1 – 20. doi: 10.3389/fneur.2015.00226.
- Badoud, S., Borgognon, S., Cottet, J., Chatagny, P., Moret, V., Fregosi, M., Kaeser, M., Fortis, E., Schmidlin, E., Bloch, J., Brunet, J.F. and Rouiller, E.M. (2017) 'Effects of dorsolateral prefrontal cortex lesion on motor habit and performance assessed with manual grasping and control of force in macaque monkeys', *Brain Struct Funct*, 222(3), pp. 1193 – 1206. doi: 10.1007/s00429-016-1268-z.
- Baizer, J.S., Kralj-Hans, I. and Glickstein, M. (1999) 'Cerebellar lesions and prism adaptation in macaque monkeys', *Journal of Neurophysiology*, 81(4), pp. 1960 – 1965. doi: 10.1152/jn.1999.81.4.1960.
- Ball, G., Stokes, P. R., Rhodes, R. A., Bose, S. K., Rezek, I., Wink, A.-M, Lord, L-D., Mehta, M.A., Grasby, P.M. and Turkheimer, F.E. (2011) 'Executive Functions and Prefrontal Cortex: A Matter of Persistence?' *Frontiers in Systems Neuroscience*, 5 (article 3), pp. 1 – 13. doi: 10.3389/fnsys.2011.00003.
- Bandettini, P. A. (2009) 'What's new in neuroimaging methods?', Annals of the New York Academy of Sciences, 1156, pp. 260–293. Available at: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC2716071/pdf/nihms116704.pdf (Accessed on 08 August 2018).
- Barber, A. D., Srinivasan, P., Joel, S. E., Caffo, B. S., Pekar, J. J. and Mostofsky, S. H. (2011) 'Motor "dexterity"?: Evidence that left hemisphere lateralization of motor circuit connectivity is associated with better motor performance in children', *Cerebral cortex* (New York, N.Y: 1991), 22(1), pp. 51 – 59. doi: 10.1093/cercor/bhr062.
- Bastian A. J. (2008) 'Understanding sensorimotor adaptation and learning for rehabilitation', *Current opinion in neurology*, 21(6), pp. 628 – 633. doi:10.1097/WCO.0b013e328315a293.
- Bastian A. J. (2011) 'Moving, sensing and learning with cerebellar damage', *Current opinion in neurobiology*, 21(4), pp. 596 601. doi:10.1016/j.conb.2011.06.007.
- Bateman, T.M. (2012) 'Advantages and disadvantages of PET and SPECT in a busy clinical practice', *J Nucl Cardiol*, 19(1), pp. S3 11. doi: 10.1007/s12350-011-9490-9.
- Bates, J.F. and Goldman-Rakic, P.S. (1993) 'Prefrontal connections of medial motor areas in the rhesus monkey', *J Comp Neurol*, 336(2), pp. 211 228. doi: 10.1002/cne.903360205.
- Batista, A. P., Buneo, C. A., Snyder, L. H., and Andersen, R. A. (1999) 'Reach plans in eyecentered coordinates', *Science*, 285, pp. 257 – 260. doi: 10.1126/science.285.5425.257.
- Beam, W., Borckardt, J. J., Reeves, S. T., and George, M. S. (2009), 'An efficient and accurate new method for locating the F3 position for prefrontal TMS applications', *Brain Stimulation*, 2(1), pp. 50 – 54. doi: 10.1016/j.brs.2008.09.006.

- Beck, S., Houdayer, E., Richardson, S. P. and Hallett, M. (2009) 'The role of inhibition from the left dorsal premotor cortex in right-sided focal hand dystonia', *Brain stimulation*, 2(4), pp. 208 – 214. doi:10.1016/j.brs.2009.03.004.
- Bedard, A.C., Nichols, S., Barbosa, J.A., Schachar, R., Logan, G.D. and Tannock, R. (2002) 'The development of selective inhibitory control across the life span', *Dev Neuropsychol*, 21(1), pp. 93 – 111. doi: 10.1207/S15326942DN2101_5.
- Bedard, P. and Sanes, J. N. (2014) 'Brain representations for acquiring and recalling visualmotor adaptations', *NeuroImage*, *101*, pp. 225–235. doi: 10.1016/j.neuroimage.2014.07.009.
- Beers, C.A. and Federico, P. (2012) 'Functional MRI applications in epilepsy surgery', *Can J Neurol Sci*, 39(3), pp. 271 – 285. Available at: https://www.ncbi.nlm.nih.gov/pubmed/22547506 (Accessed: 21st October 2018).
- Begliomini, C., De Sanctis, T., Marangon, M., Tarantino, V., Sartori, L., Miotto, D., Motta, R., Stramare, R. and Castiello, U. (2014) 'An investigation of the neural circuits underlying reaching and reach-to-grasp movements: from planning to execution', *Front Hum Neurosci*, 8(676), pp. 1–14. doi: 10.3389/fnhum.2014.00676.
- Bekkers, J.M. (2011) 'Pyramidal Neurons', *Curr Biol*, 21(24), R975. doi: 10.1016/j.cub.2011.10.037.
- Beloozerova, I.N., Sirota, M.G. and Swadlow, H.A. (2003) 'Activity of different classes of neurons of the motor cortex during locomotion', J Neurosci, 23(3), pp. 1087 – 1097. doi: 10.1523/JNEUROSCI.23-03-01087.2003.
- Benecke, R., Dick, J.P.R., Rothwell, J.C., Day, B.L. and Marsden, C.D. (1985) 'Increase of the Bereitschaftspotential in simultaneous and sequential movements', *Neurosci Lett*, 62(3), pp. 347 – 352. doi: 10.1016/0304-3940(85)90573-7.
- Berger A. (2002) 'Magnetic resonance imaging', *BMJ (Clinical research ed.)*, 324(7328), pp. 35. doi: 10.1136/bmj.324.7328.35
- Berger, A. (2003) 'How does it work? Positron emission tomography', *BMJ*, 326(7404), pp. 1449. doi. 10.1136/bmj.326.7404.1449.
- Bernabeu, M., Demirtas-Tatlidede, A., Opisso, E., Lopez, R., Tormos, J. M. and Pascual-Leone, A. (2009) 'Abnormal corticospinal excitability in traumatic diffuse axonal brain injury', *Journal of neurotrauma*, 26(12), pp. 2185 – 2193. doi:10.1089/neu.2008.0859.
- Berrington de González, A., Mahesh, M., Kim, K. P., Bhargavan, M., Lewis, R., Mettler, F., and Land, C. (2009) 'Projected cancer risks from computed tomographic scans performed in the United States in 2007', *Archives of internal medicine*, 169(22), pp. 2071–2077. doi:10.1001/archinternmed.2009.440.
- Bestmann, S., Baudewig, J., Siebner, H.R., Rothwell, J.C. and Frahm, J. (2005) 'BOLD MRI responses to repetitive TMS over human dorsal premotor cortex', *Neuroimage*, 28(1), pp. 22 29. doi: 10.1016/j.neuroimage.2005.05.027.

- Bestmann, S., Swayne, O., Blankenburg, F., Ruff, C. C., Haggard, P., Weiskopf, N., Josephs, O., Driver, J., Rothwell, J.C. and Ward, N. S. (2007) 'Dorsal premotor cortex exerts state-dependent causal influences on activity in contralateral primary motor and dorsal premotor cortex', *Cerebral cortex (New York, N.Y. : 1991)*, 18(6), pp. 1281 – 1291. doi:10.1093/cercor/bhm159
- Bestmann, S., Swayne, O., Blankenburg, F., Ruff, C. C., Teo, J., Weiskopf, N., Driver, J., Rothwell, J.C. and Ward, N. S. (2010) 'The role of contralesional dorsal premotor cortex after stroke as studied with concurrent TMS-fMRI', *The Journal of neuroscience: the official journal of the Society for Neuroscience*, 30(36), 11926– 11937. doi:10.1523/JNEUROSCI.5642-09.2010
- Betz, K. (1874) 'Anatomischer Nachweis zweier Gehirnzentra', Zentralbl. Med. Wiss. 12, pp. 578–580.
- Binkofski, F. and Buccino, G. (2006) 'The role of ventral premotor cortex in action execution and action understanding', *J Physiol Paris*, 99(4-6), pp. 396 – 405. doi: 10.1016/j.jphysparis.2006.03.005.
- Binkofski, F., Amunts, K., Stephan, K.M., Posse, S., Schormann, T., Freund, H.J., Zilles, K. and Seitz, R.J. (2000) 'Broca's region subserves imagery of motion: a combined cytoarchitectonic and fMRI study', *Hum Brain Mapp*, 11(4), pp. 273 – 285.
- Blumbergs, P. C., Jones, N. R. and North, J. B. (1989) 'Diffuse axonal injury in head trauma', Journal of neurology, neurosurgery, and psychiatry, 52(7), pp. 838 – 841. doi: 10.1136/jnnp.52.7.838.
- Bohn, S. S. (1937) 'The reactions of patients to encephalography. An analysis of one thousand consecutive cases', *Bulletin of the Neurological Institute of New York*, 6, pp. 540-568.
- Bolognini, N. and Ro, T. (2010) 'Transcranial magnetic stimulation: disrupting neural activity to alter and assess brain function', *Journal of Neuroscience: the official journal of the Society for Neuroscience*, 30(29), pp. 9647 – 9650. doi: 10.1523/JNEUROSCI.1990-10.2010.
- Borghese, N. A., Bianchi, L. and Lacquaniti, F. (1996) 'Kinematic determinants of human locomotion', *The Journal of physiology*, 494(Pt 3), pp. 863 879. doi: 10.1113/jphysiol.1996.sp021539.
- Borich, M.R., Brodie, S.M., Gray, W.A., Ionta, S. and Boyd, L.A. (2015) 'Understanding the role of the primary somatosensory cortex: Opportunities for rehabilitation', *Neuropsychologia*, 79(Pt B), pp. 246–255. doi: 10.1016/j.neuropsychologia.2015.07.007.
- Boussaoud, D. (2001) 'Attention versus intention in the primate premotor cortex', *Neuroimage*, 14(1 Pt 2), pp. S40 S45. doi: 10.1006/nimg.2001.0816.
- Boussaoud, D. and Wise, S.P. (1993) 'Primate frontal cortex: effects of stimulus and movement', *Exp Brain Res*, 95(1), pp. 28 40. doi: 10.1007/BF00229651.

- Bracht, T., Federspiel, A., Schnell, S., Horn, H., Höfle, O., Wiest, R., Dierks, T., Strik, W., Müller, TJ. and Walther, S. (2012). 'Cortico-cortical white matter motor pathway microstructure is related to psychomotor retardation in major depressive disorder'. *PloS* one, 7(12), e52238. doi: 10.1371/journal.pone.0052238
- Brenner, D. and Hall, E. (2007) 'Computed Tomography An Increasing Source of Radiation Exposure', New England Journal Of Medicine, 357(22), pp. 2277 - 2284. doi: 10.1056/nejmra072149.
- Briggs, F. (2010). 'Organizing principles of cortical layer 6', *Frontiers in neural circuits*, 4(3), pp. 1–8. doi:10.3389/neuro.04.003.2010.
- Briggs, R.G., Conner, A.K., Baker, C.M., Burks, J.D., Glenn, C.A., Sali, G., Battiste, J.D., O'Donoghue, D.L. and Sughrue, M.E. (2018) 'A Connectomic Atlas of the Human Cerebrum-Chapter 18: The Connectional Anatomy of Human Brain Networks', *Oper Neurosurg (Hagerstown)*, 15(1), S470- S480. doi: 10.1093/ons/opy272.
- Brück, W., Sommermeier, N., Bergmann, M., Zettl, U., Goebel, H.H., Kretzschmar, H.A. and Lassmann, H. (1996) 'Macrophages in multiple sclerosis', *Immunobiology*, 195(4-5), pp. 588 – 600. doi: 10.1016/S0171-2985(96)80024-6.
- Brum, M., Cabib, C. and Valls-Solé, J. (2016) 'Clinical Value of the Assessment of Changes in MEP Duration with Voluntary Contraction', *Frontiers in neuroscience*, 9, pp. 505. doi:10.3389/fnins.2015.00505.
- Buccino, G., Binkofski, F., Fink, G.F, Fadiga, L., Fogassi, L., Gallese, V., Seitz, R.J., Zilles, K., Rizzolatti, G. and Freund, H.J. (2001) 'Action observation activates premotor and parietal areas in a somatotopic manner: an fMRI study', *Euro J Neurosci*, 13(2), pp. 400–404. doi: 10.1111/j.1460-9568.2001.01385.x.
- Buch, E. R., Mars, R. B., Boorman, E. D. and Rushworth, M. F. (2010) 'A network centered on ventral premotor cortex exerts both facilitatory and inhibitory control over primary motor cortex during action reprogramming', *The Journal of neuroscience: the official journal of the Society for Neuroscience*, 30(4), pp. 1395 – 1401. doi: 10.1523/JNEUROSCI.4882-09.2010.
- Büchel, C., Coull, J.T. and Friston, K.J. (1999) 'The predictive value of changes in effective connectivity for human learning', Science, 283(5407), pp. 1538 – 1541. doi: 10.1126/science.283.5407.1538.
- Büchel, C., Raedler, T., Sommer, M., Sach, M., Weiller, C. and Koch, M.A. (2004) 'White matter asymmetry in the human brain: a diffusion tensor MRI study', *Cereb Cortex*, 14(9), pp. 945 – 951. doi: 10.1093/cercor/bhh055.
- Buchman, A. S. and Bennett, D. A. (2011) 'Loss of motor function in preclinical Alzheimer's disease', *Expert review of neurotherapeutics*, 11(5), pp. 665 – 676. doi:10.1586/ern.11.57.

- Buckner, R. L., Krienen, F. M., Castellanos, A., Diaz, J. C. and Yeo, B. T. (2011) 'The organization of the human cerebellum estimated by intrinsic functional connectivity'. *Journal of neurophysiology*, 106(5), pp. 2322 – 2345. doi:10.1152/jn.00339.2011.
- Burke, D., Hicks, R., Stephen, J., Woodforth, I. and Crawford, M. (1995) 'Trial-to-trial variability of corticospinal volleys in human subjects', *Electroencephalogr Clin Neurophysiol*, 97(5), pp. 231 – 237. doi: 10.1016/0924-980X(95)00005-4.
- Busan, P., Barbera, C., Semenic, M., Monti, F., Pizzolato, G., Pelamatti, G. and Battaglini, P. P. (2009) 'Effect of transcranial magnetic stimulation (TMS) on parietal and premotor cortex during planning of reaching movements', *PloS one*, 4(2), e4621, pp. 1 – 8. doi:10.1371/journal.pone.0004621.
- Busan, P., Zanon, M., Vinciati, F., Monti, F., Pizzolato, G. and Battaglini, P. P. (2012) 'Transcranial magnetic stimulation and preparation of visually-guided reaching movements', *Frontiers in neuroengineering*, 5(18), pp. 1–15. doi:10.3389/fneng.2012.00018.
- Buxton R. B. (2013) 'The physics of functional magnetic resonance imaging (fMRI)', *Reports* on progress in physics. Physical Society (Great Britain), 76(9), pp. 1 61. doi:10.1088/0034-4885/76/9/096601.
- Byblow, W.D., Stinear, C.M., Barber, P.A., Petoe, M.A. and Ackerley S.J. (2015) 'Proportional recovery after stroke depends on corticomotor integrity', *Ann Neurol*, 78(6), pp. 826 – 836. doi: 10.1002/ana.24472.
- Byrnes, M.L., Thickbroom, G.W., Phillips, B.A., Wilson, S.A. and Mastaglia F.L. (1999) 'Physiological studies of the corticomotor projection to the hand after subcortical stroke', *Clin Neurophysiol*, 110(3), pp. 487 - 498. doi: 10.1016/S1388-2457(98)00044-3.
- Cai, W., George, J. S., Chambers, C. D., Stokes, M. G., Verbruggen, F. and Aron, A. R. (2011) 'Stimulating deep cortical structures with the batwing coil: how to determine the intensity for transcranial magnetic stimulation using coil-cortex distance', *Journal of neuroscience methods*, 204(2), pp. 238–241. doi:10.1016/j.jneumeth.2011.11.020.
- Calabresi, P.A. (2004) 'Diagnosis and management of multiple sclerosis', *Am Fam Physician*, 70(10), pp. 1935 1944. Available at: https://www.aafp.org/afp/2004/1115/p1935.html (Accessed on: 19th September 2018).
- Call for Participants LTD (no date) *Call for Participants*. Available at: <u>https://www.callforparticipants.com/</u> (Participant recruitment website).
- Calton, J.L., Dickinson, A.R. and Snyder, L.H. (2002) 'Non-spatial, motor-specific activation in posterior parietal cortex', *Nat Neurosci*, 5(6), pp. 580 - 588. doi: 10.1038/nn0602-862.
- Cantone, M., Lanza, G., Vinciguerra, L., Puglisi, V., Ricceri, R., Fisicaro, F., Vagli, C., Bella, R., Ferri, R., Pennisi, G., Di Lazzaro, V. and Pennisi, M. (2019) 'Age, Height, and Sex on Motor Evoked Potentials: Translational Data From a Large Italian Cohort in a

Clinical Environment', Frontiers in human neuroscience, 13(185), pp. 1 – 20. doi: 10.3389/fnhum.2019.00185.

- Cao, Y., Hao, Y., Liao, Y., Xu, K., Wang, Y., Zhang, S., Chen, W. and Zheng, X. (2013) 'Information analysis on neural tuning in dorsal premotor cortex for reaching and grasping', *Computational and mathematical methods in medicine*, 2013(article ID 730374), pp. 1 – 9. doi:10.1155/2013/730374.
- Carmena, J. M. and Cohen, L. G. (2012) 'Brain-machine interfaces and transcranial stimulation: future implications for directing functional movement and improving function after spinal injury in humans', *Handbook of clinical neurology*, 109, pp. 435– 444. doi:10.1016/B978-0-444-52137-8.00027-9.
- Carmena, J. M., Lebedev, M. A., Crist, R. E., O'Doherty, J. E., Santucci, D. M., Dimitrov, D. F., Patil., P.G. Henriquez, C.S. and Nicolelis, M. A. (2003) 'Learning to control a brain-machine interface for reaching and grasping by primates', *PLoS biology*, 1(2), pp 193 208. doi:10.1371/journal.pbio.0000042.
- Carr, J.A and Shepherd, R.B. (2000) *Movement science: Foundations for physical therapy in rehabilitation*. U.S Gaithersburg: Aspen Publishers Inc.
- Castellanos, N.P, Leyva, I., Buldú, J.M., Bajo, R., Paúl, N., Cuesta, P., Ordóñez, V.E., Pascua, C.L., Boccaletti, S., Maestú, F. and del-Pozo, F. (2011) 'Principles of recovery from traumatic brain injury: reorganization of functional networks', *Neuroimage*, 55(3), pp. 1189 – 1199. doi: 10.1016/j.neuroimage.2010.12.046.
- Castro-Alamancos M. A. (2013) 'The motor cortex: a network tuned to 7-14 Hz', *Frontiers in neural circuits*, 7(21), pp. 1 7. doi:10.3389/fncir.2013.00021.
- Castro-Alamancos, M.A and Rigas, P. (2002) 'Synchronized oscillations caused by disinhibition in rodent neocortex are generated by recurrent synaptic activity mediated by AMPA receptors', *J Physiol (Lond)* 542(Pt 2), pp. 567 – 581. doi: 10.1113/jphysiol.2002.019059.
- Catalan, M.J. Honda, M., Weeks, R.A, Cohen, L.G. and Hallett, M. (1998) 'The functional neuroanatomy of simple and complex sequential finger movements: A PET study', *Brain*, 121(pt 2), pp. 253 – 264. doi: doi.org/10.1093/brain/121.2.253.
- Catani M. (2017) 'A little man of some importance', *Brain: a journal of neurology*, 140(11), pp. 3055 3061. doi:10.1093/brain/awx270.
- Cauller L. (1995) 'Layer I of primary sensory neocortex: where top-down converges upon bottom-up', *Behav Brain Res*, 71(1-2), pp. 163 – 170. Doi: 10.1016/0166-4328(95)00032-1.
- Cavina-Pratesi, C., Connolly, J. D. and Milner, A. D. (2013) 'Optic ataxia as a model to investigate the role of the posterior parietal cortex in visually guided action: evidence from studies of patient M.H', *Frontiers in human neuroscience*, 7(article 336), pp. 1 – 14. doi:10.3389/fnhum.2013.00336.

- Ceballos-Baumann, A.O. and Brooks, D.J. (1998) 'Activation positron emission tomography scanning in dystonia', *Adv Neurol*, 78, pp. 135 152. Available at: https://www.ncbi.nlm.nih.gov/pubmed/9750911 (Accessed 12th December 2018).
- Chae, J., Yang, G., Park, B.K. and Labatia, I. (2002) 'Muscle weakness and cocontraction in upper limb hemiparesis: relationship to motor impairment and physical disability', *Neurorehabil Neural Repair*, 16(3), pp. 241 248. doi: 10.1177/154596830201600303.
- Chambers, C.D, Bellgrove, M.A., Gould, I.C., English, T., Garavan, H., McNaught, E., Kamke, M. and Mattingley J.B. (2007) 'Dissociable mechanisms of cognitive control in prefrontal and premotor cortex', J Neurophysiol, 98(6), pp. 3638 – 3647. doi: 10.1152/jn.00685.2007.
- Chambers, C.D., Payne, J.M., Stokes, M.G. and Mattingley, J.B. (2004) 'Fast and slow parietal pathways mediate spatial attention', *Nat Neurosci*, 7(3), pp. 217- 218. doi: 10.1038/nn1203.
- Chang, W., O'Connell, N.E., Burns, E., Chipchase, L.S., Liston, M.B. and Schabrun, S.M. (2015) 'Organisation and function of the primary motor cortex in chronic pain: protocol for a systematic review and meta-analysis', *BMJ*, 5(11), e008540, pp. 1–4. doi: 10.1136/bmjopen-2015-008540.
- Chang, W.H., Tang, P.F., Wang, Y.H., Lin, K.H., Chiu, M.J. and Chen, S.H. (2010) 'Role of the premotor cortex in leg selection and anticipatory postural adjustments associated with a rapid stepping task in patients with stroke', *Gait Posture*, 32(4), pp. 487 493. doi: 10.1016/j.gaitpost.2010.07.007.
- Chau, W. and McIntosh, A.R. (2005) 'The Talairach coordinate of a point in the MNI space: how to interpret it', *Neuroimage*, 25(2), pp. 408 – 416. doi: 10.1016/j.neuroimage.2004.12.007.
- Chen, R., Jiao, Y. and Herskovits, E.H. (2011) 'Structural MRI in autism spectrum disorder', Pediatric Research, 69(5 Pt 2), pp. 63R – 68R. doi: 10.1203/PDR.0b013e318212c2b3.
- Chen, X., Scangos, K.W. and Stuphorn, V. (2010) 'Supplementary motor area exerts proactive and reactive control of arm movements', *J Neurosci*, 30(44), pp. 14657–14675. doi: 10.1523/JNEUROSCI.2669-10.2010.
- Chistyakov, A. V., Soustiel, J. F., Hafner, H., Trubnik, M., Levy, G. and Feinsod, M. (2001) 'Excitatory and inhibitory corticospinal responses to transcranial magnetic stimulation in patients with minor to moderate head injury', *Journal of neurology, neurosurgery, and psychiatry*, 70(5), pp. 580 – 587. doi:10.1136/jnnp.70.5.580.
- Chouinard, P. A. and Paus, T. (2010) 'What have We Learned from "Perturbing" the Human Cortical Motor System with Transcranial Magnetic Stimulation?' *Frontiers in human neuroscience*, 4(article 173), pp. 1 7. doi:10.3389/fnhum.2010.00173.
- Chouinard, P.A., Leonard, G. and Paus, T. (2005) 'Role of the primary motor and dorsal premotor cortices in the anticipation of forces during object lifting', *J Neurosci*, 25(9), pp. 2277 2284. doi: 10.1523/JNEUROSCI.4649-04.2005.

- Chouinard, P.A., Van Der Werf, Y.D., Leonard, G. and Paus, T. (2003). 'Modulating neural networks with transcranial magnetic stimulation applied over the dorsal premotor and primary motor cortices', *J Neurophysiol*, 90(2), pp. 1071 – 1083. doi: 10.1152/jn.01105.2002.
- Chowdhury, R. H., Reaz, M. B., Ali, M. A., Bakar, A. A., Chellappan, K. and Chang, T. G. (2013) 'Surface electromyography signal processing and classification techniques', *Sensors (Basel, Switzerland)*, 13(9), pp. 12431 – 12466. doi:10.3390/s130912431.
- Christensen, M.S., Lundbye-Jensen, J., Geertsen, S.S., Petersen, T.H., Paulson, O.B. and Nielsen, J.B. (2007) 'Premotor cortex modulates somatosensory cortex during voluntary movements without proprioceptive feedback', *Nat Neurosci*, 10(4), pp. 417 – 419. doi: 10.1038/nn1873.
- Chronicle, E.P., Pearson, A.J. and Matthews, C. (2005) 'Development and positioning reliability of a TMS coil holder for headache research', *Headache*, 45(1), pp. 37 41. doi: 10.1111/j.1526-4610.2005.05008.x.
- Chung, J. W., Burciu, R. G., Ofori, E., Coombes, S. A., Christou, E. A., Okun, M. S., Hess, C.W. and Vaillancourt, D. E. (2018) 'Beta-band oscillations in the supplementary motor cortex are modulated by levodopa and associated with functional activity in the basal ganglia', *NeuroImage Clinical*, 19, pp. 559 – 571. doi: 10.1016/j.nicl.2018.05.021.
- Chung, G.H., Han, Y.M., Jeong, S.H. and Jack, C.R Jr. (2005) 'Functional heterogeneity of the supplementary motor area', *AJNR Am J Neuroradiol*, 26(7), pp. 1819 – 1823. Available at: http://www.ajnr.org/content/ajnr/26/7/1819.full.pdf (Accessed 17th August 2018).
- Ciaramelli, E., Rosenbaum, R.S. Solcz, S., Levine, B. and Moscovitch, M. (2010) 'Mental space travel: damage to posterior parietal cortex prevents egocentric navigation and reexperiencing of remote spatial memories', *J Exp Psychol Learn Mem Cogn*, 36(3), pp. 619 – 634. doi: 10.1037/a0019181.
- Cincotta, M., Borgheresi, A., Balestrieri, F., Giovannelli, F., Rossi, S., Ragazzoni, A., Zaccara, G. and Ziemann, U. (2004) 'Involvement of the human dorsal premotor cortex in unimanual motor control: an interference approach using transcranial magnetic stimulation', *Neurosci Lett*, 367(2), pp. 189 193. doi: 10.1016/j.neulet.2004.06.003.
- Cirillo, J., Todd, G. and Semmler, J.G. (2011) 'Corticomotor excitability and plasticity following complex visuomotor training in young and old adults', *The European journal of neuroscience*, 34(11), pp. 1847–1856. doi: 10.1111/j.1460-9568.2011.07870.x.
- Cisek, P. and Kalaska, J.F. (2002) 'Simultaneous encoding of multiple potential reach directions in dorsal premotor cortex', *J Neurophysiol*, 87(2), pp. 1149 1154. doi: 10.1152/jn.00443.2001.
- Clower, D.M., Hoffman, J.M., Votaw, J.R., Faber, T.L., Woods, R.P. and Alexander, G.E. (1996) 'Role of posterior parietal cortex in the recalibration of visually guided reaching', *Nature*, 383(6601), pp. 618 621. doi: 10.1038/383618a0.

- Cona, G., Marino, G. and Semenza, C. (2017) 'TMS of supplementary motor area (SMA) facilitates mental rotation performance: Evidence for sequence processing in SMA', *Neuroimage*, 146, pp. 770-777. Doi: 10.1016/j.neuroimage.2016.10.032.
- Connolly, J.D., Andersen, R.A., Goodale, M.A. (2003) 'FMRI evidence for a 'parietal reach region' in the human brain', *Exp Brain Res*, 153(2), pp. 140-145. doi: 10.1007/s00221-003-1587-1.
- Corthout, E., Uttl, B., Ziemann, U., Cowey, A. and Hallett, M. (1999) 'Two periods of processing in the (circum)striate visual cortex as revealed by transcranial magnetic stimulation', *Neuropsychologia*, 37(2), pp. 137 – 145. doi: 10.1016/S0028-3932(98)00088-8.
- Coull, J.T. and Nobre, A.C. (1998) 'Where and when to pay attention: the neural systems for directing attention to spatial locations and to time intervals as revealed by both PET and fMRI', *J Neurosci*, 18(18), pp. 4626 – 7435. Available at: http://www.jneurosci.org/content/jneuro/18/18/7426.full.pdf (accessed: 4th September 2018).
- Courson, M., Macoir, J. and Tremblay, P. (2017) 'Role of medial premotor areas in action language processing in relation to motor skills', *Cortex*, 95, pp. 77 91. doi: 10.1016/j.cortex.2017.08.002.
- Cramer, S.C., Finklestein, S.P., Schaechter, J.D., Bush, G. and Rosen, B.R. (1999) 'Activation of distinct motor cortex regions during ipsilateral and contralateral finger movements', J Neurophysiol, 81(1), pp. 383 – 387. doi: 10.1152/jn.1999.81.1.383.
- Criscimagna-Hemminger, S. E., Bastian, A. J. and Shadmehr, R. (2010) 'Size of error affects cerebellar contributions to motor learning', *Journal of neurophysiology*, *103*(4), pp. 2275 2284. doi:10.1152/jn.00822.2009.
- Crossman, A.R. and Neary, D. (2014) *Neuroanatomy An illustrated colour text*. Churchill Livingstone. 5th edition.
- Culham, J.C., Cavina-Pratesi, C. and Singhal, A. (2006) 'The role of parietal cortex in visuomotor control: what have we learned from neuroimaging?', *Neuropsychologia*, 44(13), pp. 2668–2684. doi: 10.1016/j.neuropsychologia.2005.11.003.
- Cunningham, I.A. and Judy, P.F. (2000) 'Computed Tomography', in Bronzino, J.D (ed.) *The Biomedical Engineering Handbook*. Boca Raton: CRC Press LLC. (Chapter 62). Available at: https://www.k-space.org/ymk/CT_ch062.pdf (Accessed: 04th December 2018).
- Cunnington, R., Iansek, R., Thickbroom, G.W., Laing, B.A., Mastaglia, F.L, Bradshaw, J.L. and Phillips, J.G. (1996) 'Effects of magnetic stimulation over supplementary motor area on movement in Parkinson's disease', *Brain*, 119(Pt 3), pp. 815 – 822. doi: 10.1093/brain/119.3.815.
- Curtis, C.E. and D'Esposito, M. (2003) 'Persistent activity in the prefrontal cortex during working memory', *Trends Cogn Sci*, 7(9), pp. 415 423. doi: 10.1016/S1364-6613(03)00197-9.

- D'Angelo, E. and Casali, S. (2012) 'Seeking a unified framework for cerebellar function and dysfunction: from circuit operations to cognition', *Front Neural Circuits*, 6(116), pp. 1– 23. doi: 10.3389/fncir.2012.00116.
- D'Agata, F., Cicerale, A., Mingolla, A., Caroppo, P., Orsi, L., Mortara, P., Troni, W. and Pinessi, L. (2015). 'Double-Cone Coil TMS Stimulation of the Medial Cortex Inhibits Central Pain Habituation', *PLoS One*, 10(6), e0128765, pp. 1-11. doi: 10.1371/journal.pone.0128765.
- Dalamagkas, K., Tsintou, M., Rathi, Y., O'Donnell, L.J., Pasternak, O., Gong, X., Zhu, A., Savadjiev, P., Papadimitriou, G.M., Kubicki, M., Yeterian, E.H. and Makris, N. (2019) 'Individual variations of the human corticospinal tract and its hand-related motor fibers using diffusion MRI tractography', *Brain Imaging Behav*, pp. 1 – 40. doi: 10.1101/369199.
- Daly, J. J., and Huggins, J. E. (2015) 'Brain-computer interface: current and emerging rehabilitation applications', Archives of physical medicine and rehabilitation, 96(3 Suppl), S1–S7. doi:10.1016/j.apmr.2015.01.007.
- Daly, J.J., Cheng, R., Rogers, J., Litinas, K., Hrovat, K. and Dohring, M. (2009) 'Feasibility of a new application of noninvasive Brain Computer Interface (BCI): a case study of training for recovery of volitional motor control after stroke', *J Neurol Phys Ther*, 33(4), pp. 203 – 211. doi: 10.1097/NPT.0b013e3181c1fc0b.
- Dancause, N., Barbay, S., Frost, S.B., Zoubina, E.V., Plautz, E.J., Mahnken, J.D. and Nudo, R.J. (2006) 'Effects of small ischemic lesions in the primary motor cortex on neurophysiological organization in ventral premotor cortex', *J Neurophysiol*, 96(6), pp. 3506 – 3511. doi: 10.1152/jn.00792.2006.
- Dandy W.E. (1918) 'Ventriculography following the injection of air into the cerebral ventricles'. *Annals of surgery*, 68(1), pp. 5–11. [pp. 8 illus] Available at: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC1426769/ (Accessed on 08 August 2018).
- Darling, W. G., Pizzimenti, M. A. and Morecraft, R. J. (2011) 'Functional recovery following motor cortex lesions in non-human primates: experimental implications for human stroke patients', *Journal of integrative neuroscience*, 10(3), pp. 353 – 384. doi:10.1142/S0219635211002737.
- Darling, W. G., Wolf, S. L. and Butler, A. J. (2006) 'Variability of motor potentials evoked by transcranial magnetic stimulation depends on muscle activation', *Exp. Brain Res*, 174(2), pp. 376–385. doi: 10.1007/s00221-006-0468-9.
- Davare, M., Andres, M., Cosnard, G., Thonnard, J.L. and Olivier, E. (2006) 'Dissociating the role of ventral and dorsal premotor cortex in precision grasping', *J Neurosci*, 26(8), pp. 2260 – 2268. doi: 10.1523/JNEUROSCI.3386-05.2006.
- Davare, M., Zénon, A., Desmurget, M. and Olivier, E. (2015) 'Dissociable contribution of the parietal and frontal cortex to coding movement direction and amplitude', *Frontiers in human neuroscience*, 9(article 241), pp. 1–12. doi:10.3389/fnhum.2015.00241.

- Davidson, T. W., Bolic, M. and Tremblay, F. (2016) 'Predicting Modulation in Corticomotor Excitability and in Transcallosal Inhibition in Response to Anodal Transcranial Direct Current Stimulation', *Frontiers in human neuroscience*, 10, 49. doi: doi:10.3389/fnhum.2016.00049.
- Day, B. L., Rothwell, J. C. & Marsden, C. D. (1983) 'Interaction between the long-latency stretch reflex and voluntary electromyographic activity prior to a rapid voluntary motor reaction', *Brain Res*, 270(1), pp. 55 – 62. doi: 10.1016/0006-8993(83)90791-6.
- Day, B.L., Rothwell, J.C., Thompson, P.D., Maertens De Noordhout, A., Nakashima, K., Shannon, K. and Marsden, C.D. (1989) 'Delay in the execution of voluntary movement by electrical or magnetic brain stimulation in intact man: evidence for the storage of motor programs in the brain', *Brain*, 112(pt 3), pp. 649 – 663. doi: 10.1093/brain/112.3.649 pmid:2731025.
- de Bruin M., Veeger, H.E, Kreulen, M., Smeulders, M.J. and Bus, S.A. (2013) 'Biceps brachii can add to performance of tasks requiring supination in cerebral palsy patients', *J Electromyogr Kinesiol*, 23(2), pp. 516 – 522. doi: 10.1016/j.jelekin.2012.10.013.
- De Goede, A. A., Ter Braack, E. M. and van Putten, M. (2018) 'Accurate Coil Positioning is Important for Single and Paired Pulse TMS on the Subject Level', *Brain topography*, 31(6), pp. 917–930. doi: 10.1007/s10548-018-0655-6.
- De Leon, M.J., DeSanti, S., Zinkowski, R., Mehta, P.D., Pratico, D., Segal, S., Clark, C., Kerkman, D., DeBernardis, J., Li, J., Lair, L., Reisberg, B., Tsui, W. and Rusinek, H. (2004) 'MRI and CSF studies in the early diagnosis of Alzheimer's disease', *J Intern Med*, 256(3), pp. 205–223. doi: /10.1111/j.1365-2796.2004.01381.x.
- Decety, J., Kawashima, R., Gulyás, B. and Roland, P.E. (1992) 'Preparation for reaching: a PET study of the participating structures in the human brain', *Neuroreport*, 3(9), pp. 761-764. doi: 10.1097/00001756-199209000-00010
- Deecke, L. and Kornhuber, H.H. (1978) 'An electrical sign of participation of the mesial 'supplementary' motor cortex in human voluntary finger movement', *Brain Res*, 159(2), pp. 473 476. doi: 10.1016/0006-8993(78)90561-9.
- Deecke, L., Lang, W., Heller, H. J., Hufnagl, M. and Kornhuber, H. H. (1987) 'Bereitschaftspotential in patients with unilateral lesions of the supplementary motor area', *Journal of neurology, neurosurgery, and psychiatry*, 50(11), pp. 1430 – 1434. doi: 10.1136/jnnp.50.11.1430.
- DeFelipe, J. and Fariñas, I. (1992) 'The pyramidal neuron of the cerebral cortex: morphological and chemical characteristics of the synaptic inputs', *Prog Neurobiol*, 39(6), pp. 563 607. doi: 10.1016/0301-0082(92)90015-7.
- Deiber, M.P, Ibanez, V., Sadato, N. and Hallett, M. (1996) 'Cerebral structures participating in motor preparation in humans: a positron emission tomography study', *J Neurophysiol*, 75, pp. 233 – 247. doi: 10.1152/jn.1996.75.1.233.

- Dekleva, B. M., Kording, K. P. and Miller, L. E. (2018) 'Single reach plans in dorsal premotor cortex during a two-target task', *Nature communications*, 9(1), 3556, pp. 1 12. doi: 10.1038/s41467-018-05959-y.
- Della-Maggiore, V., Malfait, N., Ostry, D.J. and Paus, T. (2004) 'Stimulation of the posterior parietal cortex interferes with arm trajectory adjustments during the learning of new dynamics', J Neurosci, 24(44), pp. 9971 – 9976. doi: 10.1523/JNEUROSCI.2833-04.2004.
- Deng, Z. D., Lisanby, S. H., and Peterchev, A. V. (2013) 'Coil design considerations for deep transcranial magnetic stimulation', *Clinical neurophysiology: official journal of the International Federation of Clinical Neurophysiology*, 125(6), pp. 1202–1212. doi:10.1016/j.clinph.2013.11.038.
- Derousne, C. (1973) 'Le syndrome 'pre-moteur': a propos de 5 observations de tumeurs rolandoprerolandiques circonscrites', *Revue Neurologique*, 128, pp. 353 363.
- Desmurget, M. and Grafton, S. (2000) 'Forward modeling allows feedback control for fast reaching movements', Trends Cogn Sci, 4(11), pp. 423 431. doi: 10.1016/S1364-6613(00)01537-0.
- Devanand, D. P., Bansal, R., Liu, J., Hao, X., Pradhaban, G. and Peterson, B. S. (2012) 'MRI hippocampal and entorhinal cortex mapping in predicting conversion to Alzheimer's disease', *NeuroImage*, 60(3), pp. 1622–1629. doi:10.1016/j.neuroimage.2012.01.075.
- Di Lazzaro, V. and Ziemann, U. (2013) 'The contribution of transcranial magnetic stimulation in the functional evaluation of microcircuits in human motor cortex', *Frontiers in neural circuits*, 7(*article 18*), pp. 1–9. doi: 10.3389/fncir.2013.00018.
- Di Lazzaro, V., Oliviero, A., Profice, P., Saturno, E., Pilato, F., Insola, A., Mazzone, P., Tonali, P. and Rothwell, J.C. (1998) 'Comparison of descending volleys evoked by transcranial magnetic and electric stimulation in conscious humans', *Electroencephalogr Clin Neurophysiol*, 109(5), pp. 397 – 401. doi: 10.1016/S0924-980X(98)00038-1.
- Di Lazzaro, V., Pilato, F., Dileone, M., Ranieri, F., Ricci, V., Profice, P., Bria, P., Tonali, P.A. and Ziemann, U. (2006) 'GABA_A receptor subtype specific enhancement of inhibition in human motor cortex', *The Journal of Physiology*, 575(Pt 3), pp. 721–726. doi: 10.1113/jphysiol.2006.114694.
- Di Russo, F., Incoccia, C., Formisano, R., Sabatini, U., Zoccolotti, P. (2005) 'Abnormal motor preparation in severe traumatic brain injury with good recovery', J Neurotrauma, 22(2), pp. 297 312. doi: 10.1089/neu.2005.22.297.
- Diedrichsen, J., Grafton, S., Albert, N., Hazeltine, E. and Ivry, R.B. (2006) 'Goal-selection and movement-related conflict during bimanual reaching movements', *Cereb Cortex*, 16(12), pp. 1729 – 1738. doi: 10.1093/cercor/bhj108.
- Diehl, P., Kliesch, U., Dietz, V. and Curt, A. (2006) 'Impaired facilitation of motor evoked potentials in incomplete spinal cord injury', J Neurol, 253(1), pp. 51 – 57. doi: 10.1007/s00415-005-0921-x.

- Donoghue, J.P. and Wise, S.P. (1982) 'The motor cortex of the rat: cytoarchitecture and microstimulation mapping', *J Comp Neurol*, 212(1), pp. 76 88. doi: 10.1002/cne.902120106.
- Douglas, R.. and Martin, K. (1998) 'Neocortex', In G. M. Shepherd (ed.) *The synaptic* organization of the brain. New York, US: Oxford University Press, pp. 459 509.
- Douglas, R.J. and Martin, K.A. (2004) 'Neuronal circuits of the neocortex', Annu Rev Neurosci, 27, pp. 419 451. doi: 10.1146/annurev.neuro.27.070203.144152.
- Drew, T., Prentice, S. and Schepens, B. (2004) 'Cortical and brainstem control of locomotion', *Progress in brain research*, 143, pp. 251 261. doi: 10.1016/S0079-6123(03)43025-2.
- Dum, R. P., Levinthal, D. J. and Strick, P. L. (2016) 'Motor, cognitive, and affective areas of the cerebral cortex influence the adrenal medulla', *Proceedings of the National Academy of Sciences of the United States of America*, 113(35), pp. 9922–9927. doi:10.1073/pnas.1605044113.
- Dum R.P and Strick, P.L. (1991) 'The origin of corticospinal projections from the premotor areas in the frontal lobe', J Neurosci, 11(3), pp. 667 – 689. doi: 10.1523/JNEUROSCI.11-03-00667.1991.
- Ehrsson, H.H., Fagergren, A., Jonsson, T., Westling, G., Johansson, R.S. and Forssberg, H. (2000) 'Cortical activity in precision- versus power-grip tasks: an fMRI study', J Neurophysiol, 83 (1), pp. 528 – 536. doi: 10.1152/jn.2000.83.1.528.
- Eiammanussakul, T. and Sangveraphunsiri, V. (2018) 'A Lower Limb Rehabilitation Robot in Sitting Position with a Review of Training Activities', *Journal of Healthcare Engineering*, 2018, pp. 1 – 18. doi: 10.1155/2018/1927807.
- Eisenberg, M., Shmuelof, L., Vaadia, E. and Zohary, E. (2011) 'The representation of visual and motor aspects of reaching movements in the human motor cortex', *J Neurosci*, 31(34), pp. 12377–12384. doi: 10.1523/JNEUROSCI.0824-11.2011.
- Elble R. J. (2013) 'Defining dystonic tremor', *Current neuropharmacology*, *11*(1), pp. 48 52. doi:10.2174/157015913804999478.
- Eldaief, M. C., Press, D. Z. and Pascual-Leone, A. (2013) 'Transcranial magnetic stimulation in neurology: A review of established and prospective applications', *Neurology. Clinical practice*, 3(6), pp. 519–526. doi:10.1212/01.CPJ.0000436213.11132.8e.
- Escudero, J.V, Sancho, J., Bautista, D., Escudero, M. and López-Trigo, J. (1998) 'Prognostic value of motor evoked potential obtained by transcranial magnetic brain stimulation in motor function recovery in patients with acute ischemic stroke', *Stroke*, 29(9), pp. 1854 – 1859. doi: doi/full/10.1161/01.STR.29.9.1854.
- Fahn, S. (1984) 'The varied clinical expressions of dystonia', *Neurol Clin*, 2(3), pp. 541 554. Doi: 10.1016/S0733-8619(18)31090-9.
- Fame, R. M. MacDonald, J. L. and Macklis, J. D. (2010) 'Development, specification, and diversity of callosal projection neurons', *Trends in neurosciences*, 34(1), pp. 41–50. doi:10.1016/j.tins.2010.10.002.

- Farzan, F., Barr, M.S., Sun, Y., Fitzgerald, P.B. and Daskalakis, Z.J. (2012) 'Transcranial magnetic stimulation on the modulation of gamma oscillations in schizophrenia', *Annals of the New York Academy of Sciences*, 1265(1), pp. 25 – 35. doi: 10.1111/j.1749-6632.2012.06543.x.
- Farzan, F., Vernet, M., Shafi, M. M., Rotenberg, A., Daskalakis, Z. J. and Pascual-Leone, A. (2016) 'Characterizing and Modulating Brain Circuitry through Transcranial Magnetic Stimulation Combined with Electroencephalography', *Frontiers in neural circuits*, 10(73), pp. 1–24. doi:10.3389/fncir.2016.00073.
- Fathi, D., Ueki, Y., Mima, T., Koganemaru, S., Nagamine, T., Tawfik, A., Fukuyama, H. (2010) 'Effects of aging on the human motor cortical plasticity studied by paired associative stimulation', *Clin Neurophysiol*, 121(1), pp. 90 – 93. doi: 10.1016/j.clinph.2009.07.048.
- Feldmeyer, D., Roth, A. and Sakmann, B. (2005) 'Monosynaptic connections between pairs of spiny stellate cells in layer 4 and pyramidal cells in layer 5A indicate that lemniscal and paralemniscal afferent pathways converge in the infragranular somatosensory cortex', *The Journal of neuroscience: the official journal of the Society for Neuroscience*, 25(13), pp. 2423 – 2431. doi: 10.1523/JNEUROSCI.5227-04.2005.
- Ferbert, A., Priori, A., Rothwell, J. C., Day, B. L., Colebatch, J. G. and Marsden, C. D. (1992) 'Interhemispheric inhibition of the human motor cortex', *The Journal of Physiology*, 453, pp. 525–546. doi: 10.1113/jphysiol.1992.sp019243.
- Fernández, V., Valls-Sole, J., Relova, J.L., Raguer, N., Miralles, F., Dinca, L., Taramundi, S., Costa-Frossard, L., Ferrandiz, M., Ramió-Torrentà, L., Villoslada, P., Saiz, A., Calles, C., Antigüedad, A., Alvarez-Cermeño, J.C., Prieto, J.M., Izquierdo, G., Montalbán, X. and Fernández, O. (2013) 'Recommendations for the clinical use of motor evoked potentials in multiple sclerosis', Neurologia, 28(7), pp. 408 416. doi: 10.1016/j.nrl.2012.07.007.
- Filimon, F., Rieth, C.A., Sereno, M.I. and Cottrell, G.W. (2015) 'Observed, Executed, and Imagined Action Representations can be Decoded From Ventral and Dorsal Areas', *Cerebral Cortex*, 25(9), pp. 3144–3158. doi: 10.1093/cercor/bhu110.
- Fink, G.R., Frackowiak, R.S., Pietrzyk, U. and Passingham, R.E. (1997) 'Multiple nonprimary motor areas in the human cortex', J Neurophysiol, 77(4), pp. 2164 – 2174. doi: 10.1152/jn.1997.77.4.2164.
- Floeter, M.K. and Rothwell, J.C. (1999) 'Releasing the brakes before pressing the gas pedal', *Neurology*, 53(4), pp. 664–665. doi: 10.1212/WNL.53.4.664.
- Florida Institute for Neurologic Rehabilitation, Inc (2019) *Atlas of Brain Injury and Anatomy*. Available at: <u>http://www.finr.net/</u> (Accessed: January 2019).
- Floyer-Lea, A. and Matthews, P.M. (2005) 'Distinguishable brain activation networks for short- and long-term motor skill learning', *J Neurophysiol*, 94(1), pp. 512 – 518. doi: 10.1152/jn.00717.2004.

- Freund, H.J. and Hummelsheim, H. (1985) 'Lesions of premotor cortex in man', *Brain*, 108(pt 3), pp. 697 733. doi: 10.1093/brain/108.3.697.
- Frey S. H. (2008) 'Tool use, communicative gesture and cerebral asymmetries in the modern human brain', *Philosophical transactions of the Royal Society of London. Series B, Biological sciences*, 363(1499), pp. 1951 1957. doi:10.1098/rstb.2008.0008.
- Friehs, G.M., Zerris, V.A., Ojakangas, C.L., Fellows, M.R. and Donoghue, J.P. (2004) 'Brain– Machine and Brain–Computer Interfaces', *Stoke*, 33(11, Suppl. 1), pp. 2702 – 2705. doi: 10.1161/01.STR.0000143235.93497.
- Frisoni, G. B., Fox, N. C., Jack, C. R., Scheltens, P., and Thompson, P. M. (2010) 'The clinical use of structural MRI in Alzheimer disease', *Nature Reviews: Neurology*, 6(2), pp. 67– 77. doi: 10.1038/nrneurol.2009.215.
- Friston, K.J., Frith, C.D, Liddle, P.F. and Frackowiak, R.S. (1991) 'Comparing functional (PET) images: the assessment of significant change', *J Cereb Blood Flow Metab*, 11(4), pp. 690 – 699. doi: 10.1038/jcbfm.1991.122.
- Fu, Q.G., Flament, D., Coltz, J.D. and Ebner T.J. (1995) 'Temporal encoding of movement kinematics in the discharge of primate primary motor and premotor neurons', *Journal of Neurophysiology*, 72(2), pp. 836 – 854. doi: 10.1152/jn.1995.73.2.836
- Fu, Q.G., Suarez, J.I. and Ebner, T.J. (1993) 'Neuronal specification of direction and distance during reaching movements in the superior precentral premotor area and primary motor cortex of monkeys', J Neurophysiol, 70(5), pp. 2097 – 2116. doi: 10.1152/jn.1993.70.5.2097.
- Fujii, N., Mushiake, H. and Tanji, J. (2002) 'Distribution of eye- and arm-movement-related neuronal activity in the SEF and in the SMA and Pre-SMA of monkeys', J Neurophysiol, 87(4), pp. 2158 – 2166. doi: 10.1152/jn.00867.2001.
- Ganguly, K., Secundo, L., Ranade, G., Orsborn, A., Chang, E. F., Dimitrov, Wallis, J.D, Barbaro, N.M, Knight, R.T. and Carmena J.M. (2009) 'Cortical representation of ipsilateral arm movements in monkey and man', *The Journal of neuroscience: the* official journal of the Society for Neuroscience, 29(41), pp. 12948–12956. doi:10.1523/JNEUROSCI.2471-09.2009.
- García-Cabezas, M. Á. and Barbas, H. (2014) 'Area 4 has layer IV in adult primates', *The European journal of neuroscience*, 39(11), pp. 1824 1834. doi:10.1111/ejn.12585.
- Genon, S., Li, H., Fan, L., Müller, V. I., Cieslik, E. C., Hoffstaedter, F., Reid, A.T, Langner, R., Grefkes, C., Fox, P.T, Moebus, S., Caspers, S., Amunts, K., Jiang, T. and Eickhoff S.B. (2017) 'The Right Dorsal Premotor Mosaic: Organization, Functions, and Connectivity', *Cerebral cortex (New York, N.Y: 1991)*, 27(3), pp. 2095 2110. doi:10.1093/cercor/bhw065.
- George, M. S., Taylor, J. J. and Short, E. B. (2013) 'The expanding evidence base for rTMS treatment of depression', *Current opinion in psychiatry*, 26(1), pp. 13 18. doi:10.1097/YCO.0b013e32835ab46d.

- Gerloff, C., Corwell, B., Chen, R., Hallett, M. and Cohen, L.G. (1997) 'Stimulation over the human supplementary motor area interferes with the organization of future elements in complex motor sequences', *Brain*, 120(Pt 9), pp. 1587 – 1602. doi: 10.1093/brain/120.9.1587.
- Gersner, R., Oberman, L., Sanchez, M.J., Chiriboga, N., Kaye, H.L, Pascual-Leone, A., Libenson, M., Roth, Y., Zangen, A. and Rotenberg, A. (2016) 'H-coil repetitive transcranial magnetic stimulation for treatment of temporal lobe epilepsy: A case report', *Epilepsy Behav Case Rep*, 5, pp. 52–56. doi: 10.1016/j.ebcr.2016.03.001.
- Ghacibeh, G.A., Mirpuri, R., Drago, V., Jeong, Y., Heilman, K.M. and Triggs W.J. (2007) 'Ipsilateral motor activation during unimanual and bimanual motor tasks', *Clin Neurophysiol*, 118(2), pp. 325 – 332. doi: 10.1016/j.clinph.2006.10.003.
- Gharbawie, O. A., Stepniewska, I., Burish, M. J. and Kaas, J. H. (2010) 'Thalamocortical connections of functional zones in posterior parietal cortex and frontal cortex motor regions in New World monkeys', *Cerebral cortex (New York, N.Y: 1991)*, 20(10), pp. 2391 – 2410. doi:10.1093/cercor/bhp308.
- Ghasemi, N., Razavi, S. and Nikzad, E. (2016) 'Multiple Sclerosis: Pathogenesis, Symptoms, Diagnoses and Cell-Based Therapy', *Cell journal*, 19(1), pp. 1–10. doi:10.22074/cellj.2016.4867.
- Giguere, M. and Goldman-Rakic, P.S. (1988) 'Mediodorsal nucleus: areal, laminar, and tangential distribution of afferents and efferents in the frontal lobe of rhesus monkeys', *Journal of Comparative Neurology*, 277(2), pp. 195–213. doi: 10.1002/cne.902770204.
- Gilio, F., Iacovelli, E., Conte, A., Frasca, V., Gabriele, M., Giacomelli, E., Bettolo, C.M., Scaldaferri, N., Trebbastoni, A., Prencipe, M. and Inghilleri, M. (2008) 'Asymmetric responses to repetitive transcranial magnetic stimulation (rTMS) over the left and right primary motor cortex in a patient with lateralized progressive limb-kinetic apraxia', *Neuroscience Letters*, 437(2), pp. 125 – 129. doi: 10.1016/j.neulet.2008.03.072.
- Gitelman, D.R., Nobre, A.C., Parrish, T.B, LaBar, K.S., Kim, Y.H., Meyer, J.R. and Mesulam, M. (1999) 'A large-scale distributed network for covert spatial attention: further anatomical delineation based on stringent behavioural and cognitive controls', Brain, 122(Pt 6), pp. 1093 – 1106. doi: 10.1093/brain/122.6.1093.
- Glover G. H. (2011) 'Overview of functional magnetic resonance imaging', *Neurosurgery* clinics of North America, 22(2), pp. 133-vii. doi:10.1016/j.nec.2010.11.001.
- Goetz, S., Kozyrkov, I.C., Luber, B., Lisanby, S.H., Murphy, D.L., Grill, W.M. and Peterchev, A.V. (2019) 'Accuracy of robotic coil positioning during transcranial magnetic stimulation', *J Neural Eng*, pp. 1 – 7. doi: 10.1088/1741-2552/ab2953.
- Goldberg, G. (1985). 'Supplementary Motor Area Structure and Function: Review and Hypothesis', *Behavioral and Brain Sciences*, 8(4), pp. 567 588. doi: 10.1017/S0140525X00045167.

- Goldenberg M. M. (2012). 'Multiple sclerosis review' P & T: a peer-reviewed journal for formulary management, 37(3), pp. 175 – 184. Available at: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC3351877/pdf/ptj3703175.pdf (Accessed: 16th September 2018).
- Goldman-Rakic, P.S. (1987). 'Motor control function of the prefrontal cortex', *Ciba Found Symp*, 132, pp. 187-200. Available at: https://www.ncbi.nlm.nih.gov/pubmed/3322715 (Accessed: 15th September 2018).
- Goodale, M.A. (1988). 'Hemispheric differences in motor control', Behav Brain Res, 30(2), 203 214. doi: 10.1016/0166-4328(88)90149-0.
- Gordon, A., Lee, J.H., Flament, D., Ugurbil, K. and Ebner, T.J. (1998) 'Functional magnetic resonance imaging of motor, sensory, and posterior parietal cortical areas during performance of sequential typing movements', *Exp Brain Res*, 121(2), pp. 153 166. https://doi.org/10.1007/s002210050447.
- Gore J. C. (2003) 'Principles and practice of functional MRI of the human brain', *The Journal* of clinical investigation, 112(1), pp. 4–9. doi:10.1172/JCI19010.
- Goss, D. A., Hoffman, R. L. and Clark, B. C. (2012) 'Utilizing transcranial magnetic stimulation to study the human neuromuscular system', *Journal of visualized experiments: JoVE*, (59), E3387. doi:10.3791/3387.
- Grafton, S.T., Hazeltine, E. and Ivry, R. (1995) 'Functional mapping of sequence learning in normal humans', *J Cogn Neurosci*, 7(4), pp. 497 510. doi: 10.1162/jocn.1995.7.4.497.
- Grafton, S.T., Hazeltine, E. and Ivry, R.B. (2002) 'Motor sequence learning with the nondominant left hand. A PET functional imaging study', *Exp Brain Res*, 146(3), pp. 369-378. doi: 10.1007/s00221-002-1181-y.
- Grefkes, C., Nowak, D.A., Eickhoff, S.B., Dafotakis, M., Küst, J., Karbe, H. and Fink, G.R. (2008) 'Cortical connectivity after subcortical stroke assessed with functional magnetic resonance imaging', *Ann Neurol*, 63(2), pp. 236 – 246. doi: 10.1002/ana.21228.
- Griffeth L. K. (2005) 'Use of PET/CT scanning in cancer patients: technical and practical considerations', *Proceedings (Baylor University. Medical Center)*, 18(4), pp. 321-330. Available at: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC1255942/pdf/bumc0018-0321.pdf (Accessed: 10th October 2018).
- Groppa, S., Oliviero, A., Eisen, A., Quartarone, A., Cohen, L.G., Mall, V., Kaelin-Lang, A., Mima, T., Rossi, S., Thickbroom, G.W., Rossini, P.M., Ziemann, U., Valls-Solé, J. and Siebner, H.R. (2012) 'A practical guide to diagnostic transcranial magnetic stimulation: report of an IFCN committee', *Clin Neurophysiol*, 123(5), pp. 858 - 882. doi: 10.1016/j.clinph.2012.01.010.
- Gross, G.R. and Grossman, M. (2008) 'Update on Apraxia', *Current Neurology and Neuroscience Reports*, 8(6), pp. 490 – 496. doi:10.1007/s11910-008-0078-y.
- Gut, M., Urbanik, A., Forsberg, L., Binder, M., Rymarczyk, K., Sobiecka, B., Kozub, J. and Grabowska, A. (2007) 'Brain correlates of right-handedness', Acta Neurobiol Exp

(Wars), 67(1), pp. 43 – 51. Available at: https://www.ncbi.nlm.nih.gov/pubmed/17474320 (Accessed: 23rd October 2018).

- Guye, M., Parker, G.J., Symms, M., Boulby, P., Wheeler-Kingshott, C.A., Salek-Haddadi, A., Barker, G.J. and Duncan, J.S. (2003) 'Combined functional MRI and tractography to demonstrate the connectivity of the human primary motor cortex in vivo', *Neuroimage*, 19(4), pp. 1349 - 1360. doi: 10.1016/S1053-8119(03)00165-4.
- Haaland, K.Y. and Harrington, D.L. (1996) 'Hemispheric asymmetry of movement', *Curr Opin Neurobiol*, 6(6), pp. 796 800. doi: 10.1016/S0959-4388(96)80030-4.
- Hallett, M. (2000) 'Transcranial magnetic stimulation and the human brain', *Nature*, 13(406), pp. 147–150. doi: 10.1038/35018000.
- Hallett, M. (2007) 'Transcranial magnetic stimulation: a primer', *Neuron*, 55(2), 187-199. doi: 10.1016/j.neuron.2007.06.026.
- Halsband, U. and Passingham, R.E. (1985). 'Premotor cortex and the conditions for movement in monkeys (Macaca fascicularis)', *Behav Brain Res*, 18(3), pp. 269 – 277. doi: 10.1016/0166-4328(85)90035-X.
- Halsband, U., Ito, N., Tanji, J. and Freund, H.J. (1993) 'The role of premotor cortex and the supplementary motor area in the temporal control of movement in man', *Brain*, 116(Pt 1), pp. 242 - 266. doi: 10.1093/brain/116.1.243.
- Hamilton, L. D., Thomas, E., Almuklass, A. M. and Enoka, R. M. (2017) 'A framework for identifying the adaptations responsible for differences in pegboard times between middle-aged and older adults', *Experimental gerontology*, 97, pp. 9 – 16. doi:10.1016/j.exger.2017.07.003.
- Hamser, K. (1998) 'Intelligence and Aphasia', in Sarno, M.T. (ed.) *Acquired Aphasia 3rd Edition*, CA: US: Academic Press, pp. 341 373. doi: 10.1016/B978-0-12-619322-0.X5000-3.
- Han, X.M., Tian, H.J., Han, Z., Zhang, C., Liu, Y., Gu, J.B., Bakshi, R. and Cao, X. (2017) 'Correlation between white matter damage and gray matter lesions in multiple sclerosis patients', *Neural Regen Res*, 12(5), pp. 787 - 794. doi: 10.4103/1673-5374.206650.
- Hannah, R., Cavanagh, S. E., Tremblay, S., Simeoni, S. and Rothwell, J. C. (2018) 'Selective Suppression of Local Interneuron Circuits in Human Motor Cortex Contributes to Movement Preparation', *The Journal of neuroscience: the official journal of the Society for Neuroscience*, 38(5), pp. 1264 – 1276. doi:10.1523/JNEUROSCI.2869-17.2017.
- Hardwick, R. M., Lesage, E. and Miall, R. C. (2014) 'Cerebellar transcranial magnetic stimulation: the role of coil geometry and tissue depth', *Brain stimulation*, 7(5), pp. 643–649. doi:10.1016/j.brs.2014.04.009.
- Hardwick, R. M., Lesage, E., Eickhoff, C. R., Clos, M., Fox, P. and Eickhoff, S. B. (2015) 'Multimodal connectivity of motor learning-related dorsal premotor cortex', *NeuroImage*, 123, 114–128. doi:10.1016/j.neuroimage.2015.08.024.

- Hardwick, R. M., Rottschy, C., Miall, R. C. and Eickhoff, S. B. (2013) 'A quantitative metaanalysis and review of motor learning in the human brain', *NeuroImage*, 67, pp. 283 – 297. doi: 10.1016/j.neuroimage.2012.11.020.
- Harms, K. J., Rioult-Pedotti, M. S., Carter, D. R. and Dunaevsky, A. (2008) 'Transient spine expansion and learning-induced plasticity in layer 1 primary motor cortex', *The Journal* of neuroscience: the official journal of the Society for Neuroscience, 28(22), pp. 5686– 5690. doi:10.1523/JNEUROSCI.0584-08.2008.
- Harris, K. D. and Shepherd, G. M. (2015) 'The neocortical circuit: themes and variations', *Nature neuroscience*, 18(2), pp. 170 181. doi:10.1038/nn.3917.
- Harris-Love, M. L., Morton, S. M., Perez, M. A., & Cohen, L. G. (2011) 'Mechanisms of Short-Term Training-Induced Reaching Improvement in Severely Hemiparetic Stroke Patients: A TMS Study', *Neurorehabilitation and Neural Repair*, 25(5), pp. 398–411. doi: 10.1177/1545968310395600.
- Hartwigsen, G., Bestmann, S., Ward, N. S., Woerbel, S., Mastroeni, C., Granert, O. and Siebner, H. R. (2012) 'Left dorsal premotor cortex and supramarginal gyrus complement each other during rapid action reprogramming', *The Journal of neuroscience: the official journal of the Society for Neuroscience*, 32(46), 16162 – 16171a, pp. 1 – 21. doi:10.1523/JNEUROSCI.1010-12.2012.
- Hasan, A., Galea, J. M., Casula, E. P., Falkai, P., Bestmann, S. and Rothwell, J. C. (2012) 'Muscle and timing-specific functional connectivity between the dorsolateral prefrontal cortex and the primary motor cortex', *Journal of cognitive neuroscience*, 25(4), pp. 558 – 570. doi:10.1162/jocn_a_00338.
- Hasegawa, M., Majima, K., Itokazu, T., Maki, T., Albrecht, U.R., Castner, N., Izumo, M., Sohya, K., Sato, T.K., Kamitani, Y. and Sato, T.R. (2017) 'Selective Suppression of Local Circuits during Movement Preparation in the Mouse Motor Cortex', *Cell Rep*, 18(11), pp. 2676 – 2686. doi: 10.1016/j.celrep.2017.02.043.
- Haslinger, B., Erhard, P., Kämpfe, N., Boecker, H., Rummeny, E., Schwaiger, M., Conrad, B. and Ceballos-Baumann, A.O. (2001) 'Event-related functional magnetic resonance imaging in Parkinson's disease before and after levodopa', *Brain*, 124(Pt 3), pp. 558 – 570. doi: 10.1093/brain/124.3.558.
- Hatsopoulos, N. G. and Suminski, A. J. (2011) 'Sensing with the motor cortex', *Neuron*, 72(3), pp. 477–487. doi:10.1016/j.neuron.2011.10.020.
- Hauschild, M., Mulliken, G.H., Fineman, I., Loeb, G.E. and Andersen, R.A. (2012) 'Cognitive signals for brain-machine interfaces in posterior parietal cortex include continuous 3D trajectory commands', *PNAS*, 109(42), pp. 17075-17080. https://doi.org/10.1073/pnas.1215092109.
- Hayashi, M. J., Saito, D. N., Aramaki, Y., Asai, T., Fujibayashi, Y. and Sadato, N. (2008) 'Hemispheric asymmetry of frequency-dependent suppression in the ipsilateral primary motor cortex during finger movement: a functional magnetic resonance imaging study',

Cerebral cortex (New York, N.Y: 1991), 18(12), pp. 2932 – 2940. doi: 10.1093/cercor/bhn053.

- He, S-Q, Dum, R.P. and Strick, P.L (1995). 'Topographic organization of corticospinal projections from the frontal lobe: motor areas on the medial surface of the hemisphere', *J Neurosci*, 15(5 pt 1), pp. 3284 – 3306. doi: 10.1523/JNEUROSCI.15-05-03284.1995.
- Hermens, H. J., Freriks, B., Disselhorst-Klug, C., and Rau, G. (2000) 'Development of recommendations for SEMG sensors and sensor placement procedures', *J Electromyogr Kinesiol*, 10(5), pp. 361-374. doi: 10.1016/S1050-6411(00)00027-4.
- Herwig, U., Satrapi, P. and Schönfeldt-Lecuona, C. (2003) 'Using the international 10-20 EEG system for positioning of transcranial magnetic stimulation', *Brain Topogr*, 16(2), pp. 95 – 99. doi: 10.1023/B:BRAT.0000006333.93597.9d.
- Hikosaka, O., Sakai, K., Miyauchi, S., Takino, R., Sasaki, Y. and Pütz, B. (1996) 'Activation of human presupplementary motor area in learning of sequential procedures: a functional MRI study', *J Neurophysiol*, 76(1), pp. 617 621. doi: doi.org/10.1152/jn.1996.76.1.617.
- Hill, J., Inder, T., Neil, J., Dierker, D., Harwell, J. and Van Essen, D. (2010) 'Similar patterns of cortical expansion during human development and evolution', *Proceedings of the National Academy of Sciences of the United States of America*, 107(29), 13135–13140. doi:10.1073/pnas.1001229107.
- Hiroshima, S., Anei, R., Murakami, N. and Kamada, K. (2013) 'Functional localization of the supplementary motor area', *Neurologia medico-chirurgica*, 54(7), pp. 511 520. doi:10.2176/nmc.oa.2012-0321.
- Hoeffner, E.G., Mukherji, S.K., Srinivasan, A. And Quint, D.J (2012) ' Neuroradiology back to the future: brain imaging', AJNR Am J Neruoradial, 33(1), pp. 5-11. doi: 10.3174/ajnr.A2936.
- Hoffmann, E.J, Phelps, M.E, Mullani, N.A, Higgins, C.S, Ter-Pogossian, M.M. (1976) 'Design and performance characteristics of a whole-body positron transaxial tomograph', *J Nucl Med*, 17(6), pp. 493 – 502. Available at: (Accessed: 12th October 2018).
- Hoffmann, E.J., Phelps, M.E., Mullani, N.A., Higgins, C.S. and Ter-Pogossian, M.M. (1976) 'Design and performance characteristics of a whole body positron transaxial tomograph', *Journal of Nuclear Medicine*, 17(6), pp. 493 – 502. Available at: <u>http://jnm.snmjournals.org/content/17/6/493.long</u> (Accessed: 13 September 2018).
- Hore, J. and Flament, D. (1998) 'Changes in motor cortex neural discharge associated with the development of cerebellar limb ataxia', J Neurophysiol, 60(4), pp. 1285 – 1302. doi: 10.1152/jn.1988.60.4.1285.
- Hoshi, E. and Tanji, J. (2000) 'Integration of target and body-part information in the premotor cortex when planning action', *Nature*, 408(6811), pp. 466 470. doi: 10.1038/35044075.

- Hoshi, E. and Tanji, J. (2002). 'Contrasting neuronal activity in the dorsal and ventral premotor areas during preparation to reach', *J Neurophysiol*, 87(2), pp. 1123 1128. doi: 10.1152/jn.00496.2001.
- Hoshi, E. and Tanji, J. (2006) 'Differential involvement of neurons in the dorsal and ventral premotor cortex during processing of visual signals for action planning', J Neurophysiol, 95(6), pp. 3596 – 3616. doi: 10.1152/jn.01126.2005.
- Hounsfield, G.N. (1973) 'Computerized transverse axial scanning (tomography). 1. Description of system', *The British Journal of Radiology*, 46(552), pp. 1016 1022. doi: 10.1259/0007-1285-46-552-1016.
- Hounsfield, G.N. (1980) 'Computed medical imaging', *Medical Physics*, 7(4), pp. 283 290. doi: 10.1118/1.594709.
- Huang, V. S. and Krakauer, J. W. (2009) 'Robotic neurorehabilitation: a computational motor learning perspective', *Journal of neuroengineering and rehabilitation*, 6; 5, pp. 1 13. doi: 10.1186/1743-0003-6-5.
- Hubers, A., Orekhov, Y. and Ziemann, U. (2008) 'Interhemispheric motor inhibition: its role in controlling electromyographic mirror activity', *The European Journal of Neuroscience*, 28(2), pp. 364-371. doi: 10.1111/j.1460-9568.2008.06335.x.
- Hui, C., Taddi, P. and Patti, L. (2019, Jan). *Ischemic Stroke*. Treasure Island (FL): StatPearls Publishing. Available at: <u>https://www.ncbi.nlm.nih.gov/books/NBK499997/</u> (Accessed: February 2019).
- Hummel., F.C. and Cohen, L.G. (2006) 'Non-invasive brain stimulation: a new strategy to improve neurorehabilitation after stroke?', *Lancet Neurol*, 5(8), pp. 708 – 712. doi: 10.1016/S1474-4422(06)70525-7.
- Hunter, T., Sacco, P. and Turner, D. (2011) 'Changes in Excitability of the Motor Cortex Associated with Internal Model Formation during Intrinsic Visuomotor Learning in the Upper Arm', *Journal of Behavioral and Brain Science*, 1(3), pp. 140-152. doi: 10.4236/jbbs.2011.13019.
- Hunter, T., Sacco, P., Nitsche, M. A. and Turner, D. L. (2009) 'Modulation of internal model formation during force field-induced motor learning by anodal transcranial direct current stimulation of primary motor cortex', *The Journal of Physiology*, 587(Pt 12), pp. 2949 – 2961. doi: 10.1113/jphysiol.2009.169284.
- Husain, M. and Nachev, P. (2007) 'Space and the parietal cortex', *Trends in cognitive sciences*, 11(1), pp. 30 36. doi:10.1016/j.tics.2006.10.011.
- Husain, M., Parton, A., Hodgson, T.L., Mort, D. and Rees, G. (2003) 'Self-control during response conflict by human supplementary eye field', *Nat Neurosci*, 6(2), pp. 117 – 118. doi: 10.1038/nn1005.
- Ibanez, J., Hannah, R., Rocchi, L. and Rothwell, J.C. (2018) 'Premovement suppression of corticospinal excitability may be a necessary part of movement preparation', *bioRxiv*, pp. 1 – 28. doi: 10.1101/470153.
- Inghilleri, M., Berardelli, A., Marchetti, P. and Manfredi, M. (1996) 'Effects of diazepam, baclofen and thiopental on the silent period evoked by transcranial magnetic stimulation in humans', *Exp Brain Res*, 109(3), pp. 467–472. doi: 10.1007/bf00229631.
- Izawa, J., Rane, T., Donchin, O. and Shadmehr, R. (2008) 'Motor adaptation as a process of reoptimization', *The Journal of neuroscience: the official journal of the Society for Neuroscience*, 28(11), pp. 2883 – 2891. doi:10.1523/JNEUROSCI.5359-07.2008.
- Janssen, A. M., Oostendorp, T. F. and Stegeman, D. F. (2015) 'The coil orientation dependency of the electric field induced by TMS for M1 and other brain areas', *Journal of NeuroEngineering and Rehabilitation*, 12, pp. 47. doi: 10.1186/s12984-015-0036-2.
- Jenkins, I.H., Brooks, D.J., Nixon, P.D., Frackowiak, R.S. and Passingham, R.E. (1994) 'Motor sequence learning: a study with positron emission tomography', *J Neurosci*, 14(6), pp. 3775 – 3790. doi: 10.1523/JNEUROSCI.14-06-03775.1994.
- Jenkins, I.H., Fernandez, W., Playford, E.D., Lees, A.J., Frackowiak, R.S., Passingham, R.E. and Brooks, D.J. (1992) 'Impaired activation of the supplementary motor area in Parkinson's disease is reversed when akinesia is treated with apomorphine', *Ann Neurol*, 32(6), pp. 749 – 457. doi: 10.1002/ana.410320608.
- Jinnah, H. A. and Factor, S. A. (2015) 'Diagnosis and treatment of dystonia', *Neurologic clinics*, 33(1), pp. 77 100. doi:10.1016/j.ncl.2014.09.002.
- Johnson, P.B., Ferraina, S., Bianchi, L., Caminiti, R. (1996) 'Cortical networks for visual reaching: physiological and anatomical organization of frontal and parietal lobe arm regions', *Cereb Cortex*, 6(2), pp. 102 119. doi: 10.1093/cercor/6.2.102.
- Jones, R. (2012) 'Motor system: Getting ready to reach', *Nature Reviews Neuroscience*, 13(9), pp. 600. doi: 10.1038/nrn3321.
- Jonides, J., Smith, E.E., Koeppe, R.A., Awh, E., Minoshima, S. and Mintun, M.A. (1993) 'Spatial working memory in humans as revealed by PET', *Nature*, 363(6430) pp. 623 – 625. doi: 10.1038/363623a0.
- Jubeau, M., Rupp, T., Perrey, S., Temesi, J., Wuyam, B., Levy, P., Verges, S. and Millet, G.Y. (2014) 'Changes in voluntary activation assessed by transcranial magnetic stimulation during prolonged cycling exercise', *PloS one*, 9(2), e89157. Doi: 10.1371/journal.pone.0089157.
- Kaas, J. H. (2000) 'Why is brain size so important: design problems and solutions as neocortex gets bigger or smaller', *Brain and Mind*, 1(1), pp. 7 23. doi: 10.1023/A:1010028405318.
- Kaas, J. H., and Stepniewska, I. (2016) 'The evolution of posterior parietal cortex and parietalfrontal networks for specific actions in primates', *The Journal of Comparative Neurology*, 524(3), pp. 595–608. doi: 10.1002/cne.23838.
- Kale, N., Ozdemir, G., Coban, E. and Agaoglu, J. (2014). 'Evaluating TMS Motor Evoked Potentials in Early Stages of Multiple Sclerosis Aid in Predicting Disability (P4.191)',

Neurology, 82(10), P4. 191. Available at: https://n.neurology.org/content/82/10 Supplement/P4.191 (Accessed: 07 August 2018).

- Kamps, A. and Schüle, K. (2005) 'Cyclic movement training of the lower limb in stroke rehabilitation', Neurol Rehabil, 11(5), pp. S1 – S12. Available at: apxgroup.com.br/apxgroup2016/wp-content/uploads/2016/06/1_Cyclic-movementtraining-of-lower-limb-in-stroke-rehab.pdf (Accessed: 19th October 2018).
- Kaneko, T., Cho, R., Li, Y., Nomura, S. and Mizuno, N. (2000) 'Predominant information transfer from layer III pyramidal neurons to corticospinal neurons', *J Comp Neurol*, 423(1), pp. 52 – 65. doi: 10.1002/1096-9861(20000717)423:1<52::AID-CNE5>3.0.CO;2-F.
- Kantak, S. S., Jones-Lush, L. M., Narayanan, P., Judkins, T. N. and Wittenberg, G. F. (2013)
 'Rapid plasticity of motor corticospinal system with robotic reach training', *Neuroscience*, 247, pp. 55 – 64. doi:10.1016/j.neuroscience.2013.05.001.
- Kantak, S. S., Stinear, J. W., Buch, E. R. and Cohen, L. G. (2012). 'Rewiring the Brain: Potential Role of the Premotor Cortex in Motor Control, Learning, and Recovery of Function Following Brain Injury', *Neurorehabilitation and Neural Repair*, 26(3), pp. 282 – 292. doi: 10.1177/1545968311420845.
- Kantak, S.S, Sullivan, K.J., Fisher, B.E., Knowlton, B.J. and Winstein, C.J. (2010) 'Neural substrates of motor memory consolidation depend on practice structure', *Nat Neurosci*, 13(8), pp. 923 – 925. doi: 10.1038/nn.2596.
- Kantarci, K. and Jack C.R. (2003) 'Neuroimaging in Alzheimer disease: an evidence-based review', *Neuroimaging Clin N Am*, 13(2), pp. 197-209. doi: 10.1016/S1052-5149(03)00025-X.
- Kapogiannis, D. and Wassermann, E. M. (2008) 'Transcranial magnetic stimulation in Clinical Pharmacology', *Central nervous system agents in medicinal chemistry*, 8(4), pp. 234– 240. Available at: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC2613312/pdf/nihms53576.pdf (Accessed: 13th September 2018).
- Karabanov, A.N, Raffin, E. and Siebner, H.R. (2015) 'The Resting Motor Threshold--Restless or Resting? A Repeated Threshold Hunting Technique to Track Dynamic Changes in Resting Motor Threshold', *Brain Stimul*, 8(6), pp. 1191 – 1194. doi: 10.1016/j.brs.2015.07.001.
- Kasess, C.H., Windischberger, C., Cunnington, R., Lanzenberger, R., Pezawas, L. and Moser, E. (2008) 'The suppressive influence of SMA on M1 in motor imagery revealed by fMRI and dynamic causal modeling', *Neuroimage*, 40(2), pp. 828 – 837. doi: 10.1016/j.neuroimage.2007.11.040.
- Kawato, M. (1999) 'Internal models for motor control and trajectory planning', *Curr Opin Neurobiol*, 9(6), pp. 718 727. doi: 10.1016/S0959-4388(99)00028-8.

- Kennard, M.A., Viets, H.R. and Fulton, J.F. (1934) 'The syndrome of the premotor cortex in man: impairment of skilled movements, forced grasping, spasticity and vasomotor disturbance', *Brain*, 57, pp. 69 – 84.
- Kermadi, I., Liu, Y. and Rouiller, E.M. (2000). 'Do bimanual motor actions involve the dorsal premotor (PMd), cingulate (CMA) and posterior parietal (PPC) cortices? Comparison with primary and supplementary motor cortical areas', *Somatosens Mot Res*, 17(3), pp. 255 – 271. doi: 10.1080/08990220050117619.
- Kiers, L., Cros, D., Chiappa, K.H. and Fang, J. (1993) 'Variability of motor potentials evoked by transcranial magnetic stimulation', *Electroencephalogr Clin Neurophysiol*, 89(6), pp. 415 – 423. doi: 10.1016/0168-5597(93)90115-6.
- Kim G.W., Won, Y.H., Park, S.H., Seo, J.H. and Ko, M.H. (2015) 'Can motor evoked potentials be an objective parameter to assess extremity function at the acute or subacute stroke stage?', Ann Rehabil Med, pp. 253 – 261. doi: 10.5535/arm.2015.39.2.253.
- Kim, B. R., Moon, W. J., Kim, H., Jung, E. and Lee, J. (2016b) 'Transcranial Magnetic Stimulation and Diffusion Tensor Tractography for Evaluating Ambulation after Stroke', *Journal of stroke*, 18(2), pp. 220 – 226. doi:10.5853/jos.2015.01767.
- Kim, S.G., Ashe, J., Hendrich, K., Ellermann, J.M., Merkle, H., Uğurbil, K. and Georgopoulos A.P. (1993) 'Functional magnetic resonance imaging of motor cortex: hemispheric asymmetry and handedness', *Science*, 261(5121), pp. 615 – 617. doi: 10.1126/science.8342027.
- Kim, T., Oh, W. C., Choi, J. H. and Kwon, H. B. (2016a) 'Emergence of functional subnetworks in layer 2/3 cortex induced by sequential spikes in vivo', *Proceedings of* the National Academy of Sciences of the United States of America, 113(10), E1372– E1381. doi:10.1073/pnas.1513410113.
- Kim, Y. K., Park, E., Lee, A., Im, C. H. and Kim, Y. H. (2018) 'Changes in network connectivity during motor imagery and execution', *PloS one*, 13(1), e0190715. doi: 10.1371/journal.pone.0190715.
- Kim, Y. K. and Shin, S. H. (2014) 'Comparison of effects of transcranial magnetic stimulation on primary motor cortex and supplementary motor area in motor skill learning (randomized, cross over study)', *Frontiers in human neuroscience*, 8(article 937) pp. 1 – 7. doi:10.3389/fnhum.2014.00937.
- Kimura, D. (1977) 'Acquisition of a motor skill after left-hemisphere damage', *Brain*, 100(3), pp. 527 542. Doi: 10.1093/brain/100.3.527.
- Kimura, T., Haggard, P. and Gomi, H. (2006) 'Transcranial magnetic stimulation over sensorimotor cortex disrupts anticipatory reflex gain modulation for skilled action', J Neurosci, 26(36), pp. 9272 – 9281. doi: 10.1523/JNEUROSCI.3886-05.2006.
- King, B. R., Fogel, S. M., Albouy, G. and Doyon, J. (2013) 'Neural correlates of the agerelated changes in motor sequence learning and motor adaptation in older adults',

Frontiers in human neuroscience, 7(article 142), pp. 1 - 13. doi:10.3389/fnhum.2013.00142.

- Kitago, T. and Ratan, R. R. (2017) 'Rehabilitation following hemorrhagic stroke: building the case for stroke-subtype specific recovery therapies', *F1000Research*, 6(2044), pp. 1–8. doi:10.12688/f1000research.11913.1.
- Klein-Flügge, M. C., Nobbs, D., Pitcher, J. B. and Bestmann, S. (2013) 'Variability of human corticospinal excitability tracks the state of action preparation', *The Journal of neuroscience: the official journal of the Society for Neuroscience*, 33(13), pp. 5564 – 5572. doi:10.1523/JNEUROSCI.2448-12.2013.
- Klomjai, W., Katz, R. and Lackmy-Vallée, A. (2015) 'Basic principles of transcranial magnetic stimulation (TMS) and repetitive TMS (rTMS)', *Annuals of physical and rehabilitation medicine*, 54(4), pp. 208–213. doi: 10.1016/j.rehab.2015.05.005.
- Kobayashi, M. and Pascual-Leone, A. (2002) 'Transcranial magnetic stimulation in neurology', *The Lancet. Neurology*, 2(3), pp. 145–156. doi: 10.1016/S1474-4422(03)00321-1.
- Koch, G., Cercignani, M., Bonnì, S., Giacobbe, V., Bucchi, G., Versace, V., Caltagirone, C. and Bozzali, M. (2011) 'Asymmetry of parietal interhemispheric connections in humans', J Neurosci, 31(24), pp. 8967 – 8975. doi: 10.1523/JNEUROSCI.6567-10.2011.
- Koch, G., Fernandez Del Olmo, M., Cheeran, B., Schippling, S., Caltagirone, C., Driver, J. and Rothwell, J. C. (2008b) 'Functional interplay between posterior parietal and ipsilateral motor cortex revealed by twin-coil transcranial magnetic stimulation during reach planning toward contralateral space', *The Journal of neuroscience: the official journal* of the Society for Neuroscience, 28(23), pp. 5944 – 5953. doi:10.1523/JNEUROSCI.0957-08.2008
- Koch, G., Ribolsi, M., Mori, F., Sacchetti, L., Codecà, C., Rubino, I.A., Siracusano, A., Bernardi, G. and Centonze, D. (2008a) 'Connectivity between posterior parietal cortex and ipsilateral motor cortex is altered in schizophrenia', *Biological Psychiatry*, 64(9), pp. 815–819. doi: 10.1016/j.biopsych.2008.05.026.
- Koch, G., Ruge, D., Cheeran, B., Fernandez Del Olmo, M., Pecchioli, C., Marconi, B., Versace, V., Lo Gerfo, E., Torriero, S., Oliveri, M., Caltagirone, C. and Rothwell, J.C. (2009) 'TMS activation of interhemispheric pathways between the posterior parietal cortex and the contralateral motor cortex', *The Journal of physiology*, 587(Pt 17), pp. 4281 4292. doi:10.1113/jphysiol.2009.174086.
- Konen, C. S., Mruczek, R. E., Montoya, J. L. and Kastner, S. (2013) 'Functional organization of human posterior parietal cortex: grasping- and reaching-related activations relative to topographically organized cortex', *Journal of neurophysiology*, 109(12), pp. 2897 – 2908. doi:10.1152/jn.00657.2012.
- Kosaka, N., Tsuchida, T., Uematsu, H., Kimura, H., Okazawa, H. and Itoh, H. (2008) '18F-FDG PET of common enhancing malignant brain tumors', AJR Am J Roentgenol, 160(6), pp. W365 – W369. doi: doi: 10.2214/AJR.07.2660.

- Kotwaliwale, N., Singh, K., Kalne, A., Jha, S. N., Seth, N., and Kar, A. (2011) 'X-ray imaging methods for internal quality evaluation of agricultural produce', *Journal of food science* and technology, 51(1), pp. 1 – 15. doi:10.1007/s13197-011-0485-y.
- Koziol, L. F., Budding, D., Andreasen, N., D'Arrigo, S., Bulgheroni, S., Imamizu, H., Ito, M., Manto, M., Marvel, C., Parker, K., Pezzulo, G., Ramnani, N., Riva, D., Schmahmann, J., Vandervert, L. and Yamakazi, T. (2014) 'Consensus Paper: The Cerebellum's Role in Movement and Cognition', *Cerebellum (London, England)*, 13(1), pp. 151 – 177. doi: 10.1007/s12311-013-0511-x.
- Krakauer, J.W. and Mazzoni, P. (2011) 'Human sensorimotor learning: adaptation, skill, and beyond', *Current opinion in neurobiology*, 21(4), pp. 636 – 644. doi: 10.1016/j.conb.2011.06.012.
- Krams, M., Rushworth, M.F., Deiber, M.P., Frackowiak, R.S. and Passingham, R.E. (1998)
 'The preparation, execution and suppression of copied movements in the human brain', *Exp Brain Res*, 120(3), pp. 386 – 398. doi: 10.1007/s002210050412.
- Krause, V., Bashir, S., Pollok, B., Caipa, A., Schnitzler, A. and Pascual-Leone, A. (2012) '1 Hz rTMS of the left posterior parietal cortex (PPC) modifies sensorimotor timing', *Neuropsychologia*, 50(14), pp. 3729 3735. doi: 10.1016/j.neuropsychologia.2012.10.020.
- Krebs, I. H., Brashers-Krug, T., Rauch, S.L., Savage, C.R., Hogan, N., Rubin, R.H., Fischman, A.J. and Alpert, N.M. (1998) 'Robot-aided functional imaging: application to a motor learning study', *Human Brain Mapping*, 6(1), pp. 59-72. doi: 10.1002/(SICI)1097-0193(1998)6:1<59::AID-HBM5>3.0.CO;2-K.
- Kukowski, B. (1993) 'Duration, configuration and amplitude of the motor response evoked by magnetic brain stimulation in patients with multiple sclerosis', *Electromyography and Clinical Neurophysiology*, 33(5), pp. 295–297. Available at: https://europepmc.org/abstract/med/8404566 (Accessed: 13th October 2018).
- Kurata, K. and Hoffman, D.S. (1994) 'Differential effects of muscimol microinjection into dorsal and ventral aspects of the premotor cortex of monkeys', *J Neurophysiol*, 71(3), pp. 1151 – 1164. doi: 10.1152/jn.1994.71.3.1151.
- Landi, S. M., Baguear, F. and Della-Maggiore, V. (2011) 'One week of motor adaptation induces structural changes in primary motor cortex that predict long-term memory one year later', *The Journal of neuroscience: the official journal of the Society for Neuroscience*, 31(33), pp. 11808–11813. doi:10.1523/JNEUROSCI.2253-11.2011.
- Landin, D., Thompson, M. and Jackson, M. R. (2017) 'Actions of the Biceps Brachii at the Shoulder: A Review', *Journal of clinical medicine research*, 9(8), pp. 667–670. doi:10.14740/jocmr2901w.
- Lee, J.H. and Van Donkelaar, P. (2006) 'The human dorsal premotor cortex generates on-line error corrections during sensorimotor adaptation', *J Neurosci*, 26(12), pp. 3330 – 3334. doi: 10.1523/JNEUROSCI.3898-05.2006.

- Lee, L., Siebner, H.R., Rowe, J.B., Rizzo, V., Rothwell, J.C., Frackowiak. R.S. and Friston, K.J. (2003) 'Acute remapping within the motor system induced by low-frequency repetitive transcranial magnetic stimulation', *J Neurosci*, 23(12), pp. 5308 – 5318. doi: 10.1523/JNEUROSCI.23-12-05308.2003.
- Lefaucheur, J.P., André-Obadia, N., Antal, A., Ayache, S.S., Baeken, C., Benninger, D.H., Cantello, R.M., Cincotta, M., de Carvalho, M., De Ridder, D., Devanne, H., Di Lazzaro, V., Filipović, S.R., Hummel, F.C., Jääskeläinen, S.K., Kimiskidis, V.K., Koch, G., Langguth, B., Nyffeler, T., Oliviero, A., Padberg, F., Poulet, E., Rossi, S., Rossini, P.M., Rothwell, J.C., Schönfeldt-Lecuona, C., Siebner, H.R., Slotema, C.W., Stagg, C.J., Valls-Sole, J., Ziemann, U., Paulus, W. and Garcia-Larrea, L. (2014) 'Evidence-based guidelines on the therapeutic use of repetitive transcranial magnetic stimulation (rTMS)', *Clinical neurophysiology: official journal of the International federation of Clinical Neurophysiology*, 125(11), pp. 2150 2206. doi: 10.1016/j.clinph.2014.05.021.
- Lega, C., Stephan, M. A., Zatorre, R. J. and Penhune, V. (2016) 'Testing the Role of Dorsal Premotor Cortex in Auditory-Motor Association Learning Using Transcranical Magnetic Stimulation (TMS)', *PloS one*, 11(9), e0163380, pp. 1 – 16. doi:10.1371/journal.pone.0163380.
- Leocani, L., Cohen, L.G., Wassermann, E.M., Ikoma, K. and Hallett, M. (2000) 'Human corticospinal excitability evaluated with transcranial magnetic stimulation during different reaction time paradigms', *Brain*, 123(Pt 6), pp. 1161 – 1173. doi: 10.1093/brain/123.6.1161.
- *Learn Muscles* Available at: <u>https://learnmuscles.com/wp-content/uploads/2016/12/Wrist-Sp-St-Blog-Post-Photo-1.jpg</u> (Muscles of interest figure obtained from this webpage).
- Liew, S. L., Santarnecchi, E., Buch, E. R. and Cohen, L. G. (2014) 'Non-invasive brain stimulation in neurorehabilitation: local and distant effects for motor recovery', *Frontiers in human neuroscience*, 8(article 378), pp. 1 – 15. doi: 10.3389/fnhum.2014.00378.
- Lima, C. F., Krishnan, S. and Scott, S. K. (2016) 'Roles of Supplementary Motor Areas in Auditory Processing and Auditory Imagery', *Trends in neurosciences*, 39(8), pp. 527 – 542. doi: 10.1016/j.tins.2016.06.003
- Livingston, S.C., Goodkin, H.P. and Ingersoll, C.D. (2010) 'The influence of gender, hand dominance, and upper extremity length on motor evoked potentials', *J Clin Monit Comput*, 24(6), pp. 427 – 436. doi: 10.1007/s10877-010-9267-8.
- Lotze, M., Montoya, P., Erb, M., Hülsmann, E., Flor, H., Klose, U., Birbaumer, N. and Grodd, W. (1999) 'Activation of cortical and cerebellar motor areas during executed and imagined hand movements: an fMRI study', *J Cogn Neurosci*, 11(5), pp. 491 – 501. doi: 10.1162/089892999563553.
- Luft, A.R. and Buitrago, M.M. (2005) 'Stages of motor skill learning', *Molecular Neurobiology*, 32(3), pp. 205 216. doi: 10.1385/MN:32:3:205.

- Ma, X., Ma, C., Zhang., P., Kang., T. and He., J. (2017) 'Neurons in dorsal premotor cortex represent the switching of intended hand path in a delayed reaching task', *J Integr Neurosci*, 16(3), pp. 365 – 382. doi: 10.3233/JIN-170024.
- Machii, K., Cohen, D., Ramos-Estebanez, C. and Pascual-Leone, A. (2006) 'Safety of rTMS to non-motor cortical areas in healthy participants and patients', Clin Neurophysiol, 117(2), pp. 455 471. doi: 10.1016/j.clinph.2005.10.014.
- Maguire, E.A., Woollett, K. and Spiers, H.J. (2006) 'London taxi drivers and bus drivers: a structural MRI and neuropsychological analysis', *Hippocampus*, 16(12), pp. 1091 1101. doi: 10.1002/hipo.20233.
- Mahdavi, A., Azar, R., Shoar, M. H., Hooshmand, S., Mahdavi, A. and Kharrazi, H. H. (2015) 'Functional MRI in clinical practice: Assessment of language and motor for pre-surgical planning', *The neuroradiology journal*, 28(5), pp. 468–473. doi:10.1177/1971400915609343.
- Makoshi, Z., Kroliczak, G. and van Donkelaar, P. (2011) 'Human supplementary motor area contribution to predictive motor planning', *J Mot Behav*, 43(4), pp. 303 309. doi: 10.1080/00222895.2011.584085.
- Malfait, N. and Ostry, D.J. (2004) 'Is interlimb transfer of force-field adaptation a cognitive response to the sudden introduction of load?', *J Neurosci*, 24(37), pp. 8084 – 8089. doi: 10.1523/JNEUROSCI.1742-04.2004.
- Malhotra, P., Coulthard, E. J. and Husain, M. (2009) 'Role of right posterior parietal cortex in maintaining attention to spatial locations over time', *Brain: a journal of neurology*, 132(Pt 3), pp. 645 – 660. doi:10.1093/brain/awn350.
- Malone, L. A., Vasudevan, E. V. L. and Bastian, A. J. (2011) 'Motor adaptation training for faster re-learning', *The Journal of Neuroscience: The Official Journal of the Society for Neuroscience*, 31(42), pp. 15136 – 15143. doi: 10.1523/JNEUROSCI.1367-11.2011.
- Manto, M., Bower, J. M., Conforto, A. B., Delgado-García, J. M., da Guarda, S. N. F., Gerwig, M., Habas., C., Hagura, N., Ivry, R.B., Marien, P., Molinari, M., Naito, E., Nowak, D.A., Taib, N.O.B., Pelisson, D., Tesche, C.D., Tilikete, C. and Timmann, D. (2012) 'Consensus Paper: Roles of the Cerebellum in Motor Control—The Diversity of Ideas on Cerebellar Involvement in Movement', *Cerebellum*, 11(2), pp. 457 487. doi: 10.1007/s12311-011-0331-9.
- Maraka, S., Jiang, Q., Jafari-Khouzani, K., Li, L., Malik, S., Hamidian, H., Zhang, T., Lu, M., Soltanian-Zadeh, H., Chopp, M. and Mitsias, P.D. (2014) 'Degree of corticospinal tract damage correlates with motor function after stroke', *Annals of Clinical and Translational Neurology*, 1(11), pp. 891 – 899. doi: 10.1002/acn3.132.
- Marconi, B., Genovesio, A., Battaglia-Mayer, A., Ferraina, S., Squatrito, S., Molinari, M., Lacquaniti, F. and Caminiti, R. (2001) 'Eye-hand coordination during reaching. II. An analysis of the relationships between visuomanual signals in parietal cortex and parietofrontal association projections', *Cereb Cortex* 11(6), pp. 513 – 527. Doi: 10.1093/cercor/11.6.528.

- Marlinski, V., Nilaweera, W.U., Zelenin, P.V., Sirota, M.G. and Beloozerova I.N. (2012) 'Signals from the ventrolateral thalamus to the motor cortex during locomotion', J Neurophysiol, 107(1), pp. 455–472. doi: 10.1152/jn.01113.2010.
- Mars, R.B., Klein, M.C., Neubert, F.X, Olivier, E., Buch, E.R., Boorman, E.D. and Rushworth, M.F. (2009) 'Short-latency influence of medial frontal cortex on primary motor cortex during action selection under conflict', *J Neurosci*, 29(21), pp. 6926 – 6931. doi: 10.1523/JNEUROSCI.1396-09.2009.
- Martin, J.H. and Ghez, C. (1991) 'Impairments in reaching during reversible inactivation of the distal forelimb representation of the motor cortex in the cat', *Neuroscience Letters*, 133(1), pp. 61–64. doi: 10.1016/0304-3940(91)90057-Z.
- Martin, T.A., Keating, J.G., Goodkin, H.P., Bastian, A.J. and Thach, W.T. (1996). 'Throwing while looking through prisms. II. Specificity and storage of multiple gaze-throw calibrations', *Brain: A Journal of Neurology*, 119(pt 4), pp. 1199 – 1211. doi: 10.1093/brain/119.4.1199.
- Massie, C. L., Kantak, S. S., Narayanan, P. and Wittenberg, G. F. (2014) 'Timing of motor cortical stimulation during planar robotic training differentially impacts neuroplasticity in older adults', *Clinical neurophysiology: official journal of the International Federation of Clinical Neurophysiology*, 126(5), pp. 1024–1032. doi:10.1016/j.clinph.2014.06.053.
- Matsunaga, K., Maruyama, A., Fujiwara, T., Nakanishi, R., Tsuji, S. and Rothwell, J. C. (2005) 'Increased corticospinal excitability after 5Hz rTMS over the human supplementary motor area', Journal of Physiology, 562, pp. 295 – 306. doi: 10.1113/jphysiol.2004.070755.
- Mayo-Smith, W., Hara, A., Mahesh, M., Sahani, D., and Pavlicek, W. (2014) 'How I Do It: Managing Radiation Dose in CT', *Radiology*, 273(3), pp. 657-672. doi: 10.1148/radiol.14132328.
- McAllister, T. W. (2011) 'Neurobiological consequences of traumatic brain injury', *Dialogues in clinical neuroscience'*, 13(3), pp. 287 300. Available at: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC3182015/pdf/DialoguesClinNeurosci-13-287.pdf (Accessed 21st September 2018).
- McCollough, C. H., Guimarães, L., and Fletcher, J. G. (2009) 'In defense of body CT'. *AJR American journal of roentgenology*, 193(1), pp. 28–39. doi:10.2214/AJR.09.2754.
- McFarland, D.J. and Vaughan, T.M. (2016) 'BCI in practice', *Prog Brain Res*, 228, pp. 389-404. doi: 10.1016/bs.pbr.2016.06.005.
- Mcfarland, D.J. and Vaughan, T.M. (2016) 'Chapter 12 BCI in practice', in Coyle, D. (ed.) Brain-Computer Interfaces: Lab Experiments to Real-World Applications, Amsterdam, Netherlands: Elsevier, pp. 389 – 404.
- McFarland, N.R. and Haber S.N. (2002) 'Thalamic relay nuclei of the basal ganglia form both reciprocal and nonreciprocal cortical connections, linking multiple frontal cortical

areas', *J Neurosci*, 22(18), pp. 8117 – 8132. doi: https://doi.org/10.1523/JNEUROSCI.22-18-08117.2002.

- McGinley, M., Hoffman, R. L., Russ, D. W., Thomas, J. S. and Clark, B. C. (2010) 'Older adults exhibit more intracortical inhibition and less intracortical facilitation than young adults', *Experimental gerontology*, 45(9), pp. 671–678. doi:10.1016/j.exger.2010.04.005.
- Meehan, S. K., Randhawa, B., Wessel, B., and Boyd, L. A. (2011) 'Implicit sequence-specific motor learning after subcortical stroke is associated with increased prefrontal brain activations: an fMRI study', *Human brain mapping*, 32(2), pp. 290 – 303. doi:10.1002/hbm.21019.
- Meehan, S.K., Zabukovec, J.R., Dao, E., Cheung, K.L., Linsdell, M.A. and Boyd, L.A. (2013) 'One herz repetitive transcranial magnetic stimulation over dorsal premotor cortex enhances offline motor memory consolidation for sequence-specific implicit learning', Eur. J. Neurosci, 8, pp. 1 – 9. doi: 10.1111/ejn.12291.
- Mehler, D., Williams, A. N., Krause, F., Lührs, M., Wise, R. G., Turner, D. L., Linden, D.E.J. and Whittaker, J. R. (2019) 'The BOLD response in primary motor cortex and supplementary motor area during kinesthetic motor imagery based graded fMRI neurofeedback', *NeuroImage*, 184, pp. 36–44. doi:10.1016/j.neuroimage.2018.09.007.
- Meister, I. G., Wilson, S. M., Deblieck, C., Wu, A. D., and Iacoboni, M. (2007) 'The essential role of premotor cortex in speech perception', *Current biology : CB*, 17(19), 1692–1696. doi:10.1016/j.cub.2007.08.064.
- Messier, J. and Kalaska, J.F. (2000) 'Covariation of primate dorsal premotor cell activity with direction and amplitude during a memorized-delay reaching task', *J Neurophysiol*, 84(1), pp. 152 – 165. doi: 10.1152/jn.2000.84.1.152.
- Meyer, B.U. and Voss, M. (2000) 'Delay of the execution of rapid finger movement by magnetic stimulation of the ipsilateral hand-associated motor cortex', *Exp Brain Res*, 134(4), pp. 477 482. doi: /doi.org/10.1007/s002210000486.
- Meyer, G., Wahle, P., Castaneyra-Perdomo, A. and Ferres-Torres, R. (1992) 'Morphology of neurons in the white matter of the adult human neocortex', *Exp Brain Res*, 88(1), pp. 204 212. Available at: https://link.springer.com/article/10.1007/BF02259143 (Accessed: 19th September 2018).
- Miall, R. C., Christensen, L. O., Cain, O. and Stanley, J. (2007) 'Disruption of state estimation in the human lateral cerebellum', *PLoS biology*, 5(11), e316, pp. 2733 – 2744. doi:10.1371/journal.pbio.0050316.
- Mills K. R. (2005) 'The basics of electromyography', *Journal of neurology, neurosurgery, and psychiatry*, *76*(2), pp. ii32–ii35. doi:10.1136/jnnp.2005.069211.
- Mills, K.R., Boniface, S.J. and Schubert, M. (1992) 'Magnetic brain stimulation with a double coil: the importance of coil orientation', *Electroencephalogr Clin Neurophysiol*, 85(1), pp. 17 – 21. doi: 10.1016/0168-5597(92)90096-T.

- Miniussi, C., Ruzzoli, M. and Walsh, V. (2010) 'The mechanism of transcranial magnetic stimulation in cognition', *Cortex*, 46(1), pp. 128 – 130. doi: 10.1016/j.cortex.2009.03.004.
- Mishra, B. R., Sarkar, S., Praharaj, S. K., Mehta, V. S., Diwedi, S., and Nizamie, S. H. (2011) 'Repetitive transcranial magnetic stimulation in psychiatry', *Annals of Indian Academy* of Neurology, 14(4), pp. 245–251. doi: 10.4103/0972-2327.91935.
- Mitchell, G.A.G. and Patterson, E.L. (1954) *Basic Anatomy*. Edinburgh and London: Churchill Livingstone.
- Miyachi, S., Lu, X., Inoue, S., Iwasaki, T., Koike, S., Nambu, A. and Takada, M. (2005) 'Organization of multisynaptic inputs from prefrontal cortex to primary motor cortex as revealed by retrograde transneuronal transport of rabies virus', *J Neurosci*, 25(10), pp. 2547 - 2556.
- Mochizuki, H., Franca, M., Huang, Y.Z. and Rothwell, J.C. (2005) 'The role of dorsal premotor area in reaction task: comparing the "virtual lesion" effect of paired pulse or theta burst transcranial magnetic stimulation', *Exp Brain Res*, 167(3), pp. 414 421. doi: 10.1007/s00221-005-0047-5.
- Mochizuki, H., Huang, Y.-Z. and Rothwell, J. C. (2004) 'Interhemispheric interaction between human dorsal premotor and contralateral primary motor cortex', *The Journal of Physiology*, 561(Pt 1), pp. 331–338. doi: 10.1113/jphysiol.2004.072843.
- Moran, D.W. and Schwartz, A.B. (1999) 'Motor cortical representation of speed and direction during reaching', *Journal of Neurophysiology*, 82(5), pp. 2676–2692. doi: 10.1152/jn.1999.82.5.2676.
- Morin, A., Doyon, J., Dostie, V., Barakat, M., Tahar, A. H., Korman, M, Benali, H., Karni, A., Ungerleider, L.G. and Carrier, J. (2008) 'Motor Sequence Learning Increases Sleep Spindles and Fast Frequencies in Post-Training Sleep', *Sleep*, 31(8), pp. 1149 – 1156. doi: 10.5665/*sleep*/31.8.1149.
- Morishita, T., Kubota, S., Hirano, M. and Funase, K. (2014) 'Different modulation of short- and long-latency interhemispheric inhibition from active to resting primary motor cortex during a fine-motor manipulation task', *Physiological Reports*, 2(10), e12170, pp. 1– 14. doi: 10.14814/phy2.12170.
- Moseley, I. F., Loh, L., and du Boulay, G. H. (1977) 'Effects of general anaesthesia on size of cerebrospinal fluid spaces during and after pneumoencephalography', *Journal of neurology, neurosurgery, and psychiatry*, 40(11), pp. 1033–1048. Available at: https://jnnp.bmj.com/content/jnnp/40/11/1033.full.pdf (Accessed 09 September 2019).
- Muellbacher, W., Facchini, S., Boroojerdi, B. and Hallett, M. (2000) 'Changes in motor cortex excitability during ipsilateral hand muscle activation in humans', *Clin Neurophysiol*, 111(2), pp. 344 339. doi: 10.1016/S1388-2457(99)00243-6.
- Münchau, A., Bloem, B.R., Irlbacher, K., Trimble, M.R. and Rothwell, J.C. (2002) 'Functional connectivity of human premotor and motor cortex explored with repetitive transcranial

magnetic stimulation', *J Neurosci*, 22(2), pp. 554 – 561. doi: 10.1523/JNEUROSCI.22-02-00554.2002.

- Murase, N., Rothwell, J.C., Kaji, R., Urushihara, R., Nakamura, K., Murayama, N., Igasaki, T., Sakata-Igasaki, M., Mima, T., Ikeda, A. and Shibasaki, H. (2005) 'Subthreshold lowfrequency repetitive transcranial magnetic stimulation over the premotor cortex modulates writer's cramp', Brain, 128(Pt 1), pp. 104 – 115. doi: 10.1093/brain/awh315.
- Murata, A., Fadiga, L., Fogassi, L., Gallese, V., Raos, V. and Rizzolatti, G. (1997) 'Object representation in the ventral premotor cortex (area F5) of the monkey', *J Neurophysiol*, 78(4), pp. 2226 2230. doi: 10.1152/jn.1997.78.4.2226.
- Muratori, L. M., Lamberg, E. M., Quinn, L. and Duff, S. V. (2013) 'Applying principles of motor learning and control to upper extremity rehabilitation', *Journal of Hand Therapy: Official Journal of the American Society of Hand Therapists*, 26(2), pp. 94–103. doi: 10.1016/j.jht.2012.12.007.
- Murphy, A. C., Muldoon, S. F., Baker, D., Lastowka, A., Bennett, B., Yang, M. and Bassett, D. S. (2018) 'Structure, function, and control of the human musculoskeletal network', *PLoS biology*, 16(1), e2002811. doi:10.1371/journal.pbio.2002811.
- Musuka, T. D., Wilton, S. B., Traboulsi, M. and Hill, M. D. (2015) 'Diagnosis and management of acute ischemic stroke: speed is critical', *CMAJ: Canadian Medical Association journal = journal de l'Association medicale canadienne*, 187(12), pp. 887– 893. doi:10.1503/cmaj.140355.
- Mutha, P. K., Haaland, K. Y. and Sainburg, R. L. (2012) 'The effects of brain lateralization on motor control and adaptation', *Journal of motor behavior*, 44(6), pp. 455 469. doi:10.1080/00222895.2012.747482.
- Nachev, P., Kennard, C. and Husain, M. (2008). 'Functional role of the supplementary and presupplementary motor areas', Nat Rev Neurosci, 9(11), pp. 856 – 869. doi: 10.1038/nrn2478.
- Nachev, P., Wydell, H., O'Neill, K., Husain, M., and Kennard, C. (2007) 'The role of the presupplementary motor area in the control of action', *Neuroimage*, 36(3-3), pp. T155 – T163. doi: 10.1016/j.neuroimage.2007.03.034.
- Nadkarni, T.N., Andreoli, M.J., Nair, V.A., Yin, P., Young, B.M., Kundu, B., Pankratz, J., Radtke, A., Holdsworth, R., Kuo, J.S., Field, A.S., Baskaya, M.K., Moritz, C.H., Meyer and, M.E. and Prabhakaran V. (2014) 'Usage of fMRI for pre-surgical planning in brain tumor and vascular lesion patients: task and statistical threshold effects on language lateralization', *Neuroimage Clin*, 7, pp. 415-23. doi: 10.1016/j.nicl.2014.12.014.
- Naidech, A. M. (2011) 'Intracranial hemorrhage', *American journal of respiratory and critical care medicine*, 184(9), pp. 998 1006. doi:10.1164/rccm.201103-0475CI.
- Narayana, S., Laird, A. R., Tandon, N., Franklin, C., Lancaster, J. L. and Fox, P. T. (2012) 'Electrophysiological and functional connectivity of the human supplementary motor area', *NeuroImage*, 62(1), pp. 250 – 265. doi:10.1016/j.neuroimage.2012.04.060.

- National Health Service (2018) *PET Scan*. Available at: https://www.nhs.uk/conditions/pet-scan/ (Accessed: 2nd February 2019).
- Navarro de Lara, L. I., Windischberger, C., Kuehne, A., Woletz, M., Sieg, J., Bestmann, S., Weiskopf, N., Strasser, B. and Moser, E. (2015) 'A novel coil array for combined TMS/fMRI experiments at 3 T', *Magnetic resonance in medicine*, 74(5), pp. 1492 – 1501. doi: 10.1002/mrm.25535.
- Neubert, F. X., Mars, R. B., Buch, E. R., Olivier, E. and Rushworth, M. F. (2010) 'Cortical and subcortical interactions during action reprogramming and their related white matter pathways', *Proceedings of the National Academy of Sciences of the United States of America*, 107(30), pp. 13240 – 13245. doi:10.1073/pnas.1000674107.
- Newcombe, V.F.J., Outtrim, J.G, Chatfield, D.A., Manktelow, A., Hutchinson, P.J, Coles, J.P, Williams, G.B, Sahakian, B.J, Menon, D.K. (2011) 'Parcellating the neuroanatomical basis of impaired decision-making in traumatic brain injury', *Brain*, 134(3), pp. 759 – 768. doi: 10.1093/brain/awq388.
- Nezafat, R., Shadmehr, R. and Holcomb, H.H. (2001) 'Long-term adaptation to dynamics of reaching movements: a PET study.' *Exp Brain Res*, 140(1), pp. 66 76. doi: 10.1007/s002210100787.
- Nguyen, V.T., Breakspear, M. and Cunnington, R. (2014) 'Reciprocal interactions of the SMA and cingulate cortex sustain premovement activity for voluntary actions', *J Neurosci*, 34(49), pp. 16397 16407. doi: 10.1523/JNEUROSCI.2571-14.2014.
- Ni, Z., Gunraj, C., Nelson, A.J., Yeh, I.J., Castillo, G., Hogue, T. and Chen, R. (2009) 'Two phases of interhemispheric inhibition between motor related cortical areas and the primary motor cortex in the human', *Cerebral Cortex*, 19(7), pp. 1654-1665. doi: 10.1093/cercor/bhn201.
- Nikolova, M., Pondev, N., Christova, L., Wolf, W. and Kossey, A.R. (2006) 'Motor cortex excitability changes preceding voluntary muscle activity in simple reaction time task', *European Journal of Applied Physiology*, 98(2), pp. 212-219. doi: 10.1007/s00421-006-0265-y.
- Noda, Y., Barr, M. S., Zomorrodi, R., Cash, R. F. H., Farzan, F., Rajji, T. K., Chen, R., Daskalakis, Z.J. and Blumberger, D. M. (2017) 'Evaluation of short interval cortical inhibition and intracortical facilitation from the dorsolateral prefrontal cortex in patients with schizophrenia', *Scientific Reports*, 7(1) (article number 17106), pp. 1 - 12. doi: 10.1038/s41598-017-17052-3.
- Nomura, T., Takeshima, T. and Nakashima, K. (2001). 'Reduced pre-movement facilitation of motor evoked potentials in spinocerebellar degeneration', *J Neurol Sci*, 187(1-2), 41 -47. doi: 10.1016/S0022-510X(01)00522-6.
- Nudo, R.J. and Friel, K.M. (1999) 'Cortical plasticity after stroke: implications for rehabilitation', *Rev Neurol (Paris)*, 155(9), pp. 713 717. doi: RNE-08-1999-155-9-0000-0000-101019-ART70.

- O'Driscoll, S.W. (1992) 'Intramuscular triceps rupture', *Can J Surg*, 35(2), pp. 203 207. Available at: https://www.ncbi.nlm.nih.gov/pubmed/1562935 (Accessed 20th September 2018).
- Obhi, S.S., Haggard, P., Taylor, J. and Pascual-Leone, A. (2002) 'rTMS to the supplementary motor area disrupts bimanual coordination', Motor Control, 6(4), pp. 319 332. doi: 10.1123/mcj.6.4.319.
- Ohbayashi, M., Picard, N. and Strick, P.L. (2016) 'Inactivation of the Dorsal Premotor Area Disrupts Internally Generated, But Not Visually Guided, Sequential Movements', J Neurosci, 36(6), pp. 1971 – 1976. Doi: doi: 10.1523/JNEUROSCI.2356-15.2016.
- Okamoto, E., Ishikawa, E., Yamamoto, T., Matsuda, M., Nakai, K., Matsushita, A., Masuda, Y., Akutsu, H., Ayuzawa, S., Sakamaki, F., Matsumoto, Y. and Matsumura, A. (2015) 'Variability in amplitude and stimulation threshold values in motor evoked potential (MEP) monitoring during the resection of brain lesions', *Clin Neurophysiol*, 126(6), pp. 1271 – 1278. doi: 10.1016/j.clinph.2014.10.005.
- Oliveira, F. T., Diedrichsen, J., Verstynen, T., Duque, J. and Ivry, R. B. (2010) 'Transcranial magnetic stimulation of posterior parietal cortex affects decisions of hand choice', *Proceedings of the National Academy of Sciences of the United States of America*, 107(41), pp. 17751 – 17756. doi:10.1073/pnas.1006223107.
- Oliveri, M., Babiloni, C., Filippi, M.M., Caltagirone, C., Babiloni, F., Cicinelli, P., Traversa, R., Palmieri, M.G. and Rossini, P.M. (2003) 'Influence of the supplementary motor area on primary motor cortex excitability during movements triggered by neutral or emotionally unpleasant visual cues', *Exp Brain Res*, 149(2), pp. 214 – 221. doi: 10.1007/s00221-002-1346-8.
- Opie, G.M. and Semmler, J.G. (2014) 'Age-related differences in short- and long-interval intracortical inhibition in a human hand muscle', *Brain Stimul*, 7(5), pp. 665 672. doi: 10.1016/j.brs.2014.06.014.
- Orban de Xivry, J.J., Ahmadi-Pajouh, M. A., Harran, M. D., Salimpour, Y., and Shadmehr, R. (2013) 'Changes in corticospinal excitability during reach adaptation in force fields', *Journal of Neurophysiology*, 109(1), pp. 124–136. doi: 10.1152/jn.00785.2012.
- Overduin, S.A., Richardson, A.G. and Bizzi, E. (2009) 'Cortical processing during dynamic motor adaptation,' *Advances in experimental medicine and biology*, 629, pp. 423-438. doi: 10.1007/978-0-387-77064-2 22.
- Padberg, F., Zwanzger, P., Keck, M.E., Kathmann, N., Mikhaiel, P., Ella, R., Rupprecht, P., Thoma, H., Hampel, H., Toschi, N. and Möller, H.J. (2002) 'Repetitive transcranial magnetic stimulation (rTMS) in major depression: relation between efficacy and stimulation intensity', *Neuropsychopharmacology*, 27(4), pp. 638 – 645. doi: 10.1016/S0893-133X(02)00338-X.
- Padoa-Schioppa, C., Li, C.S. and Bizzi, E. (2004) 'Neuronal activity in the supplementary motor area of monkeys adapting to a new dynamic environment', *Journal of neurophysiology*, 91(9), pp. 449 – 473. Available at:

http://web.mit.edu/bcs/bizzilab/publications/padoa-schioppa2004.pdf (accessed 08 September 2018).

- Palmer, E. and Ashby, P. (1992) 'Corticospinal projections to upper limb motoneurones in humans', *The Journal of physiology*, 448, pp. 397 – 412. Available at: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC1176206/pdf/jphysiol00434-0399.pdf (Accessed: 17th October 2018).
- Panouilleres, M.T.N., Joundi, R.A., Brittain, J. and Jenkinson, N. (2015). 'Reversing motor adaptation deficits in the ageing brain using non-invasive stimulation', *The Journal of Physiology*, 593(pt 16), pp. 3645 – 3655. doi: 10.1113/JP270484.
- Papathanasiou, N.D, Gaze, M.N, Sullivan, K., Aldridge, M., Waddington, W., Almuhaideb, A., and Bomanji, J.B. (2011). '18F-FDG PET/CT and 123I-Metaiodobenzylguanidine Imaging in High-Risk Neuroblastoma: Diagnostic Comparison and Survival Analysis', *Journal Of Nuclear Medicine*, 52(4), pp. 519-525. doi: 10.2967/jnumed.110.083303.
- Parks, N. A., Mazzi, C., Tapia, E., Savazzi, S., Fabiani, M., Gratton, G. and Beck, D. M. (2015) 'The influence of posterior parietal cortex on extrastriate visual activity: A concurrent TMS and fast optical imaging study', *Neuropsychologia*, 78, pp. 153 – 158. doi:10.1016/j.neuropsychologia.2015.10.002.
- Pascual-Leone, A., Freitas, C., Oberman, L., Horvath, J.C., Halko, M., Eldaief, M., Bashir, S., Vernet, M., Shafi, M., Westover, B., Vahabzadeh-Hagh, A.M. and Rotenberg, A. (2011) 'Characterizing brain cortical plasticity and network dynamics across the age-span in health and disease with TMS-EEG and TMS-fMRI', *Brain Topogr*, 24(3-4), pp. 302 – 315. doi: 10.1007/s10548-011-0196-8.
- Pascual-Leone, A., Gates, J.R. and Dhuna, A. (1991) 'Induction of speech arrest and counting errors with rapid-rate transcranial magnetic stimulation', *Neurology*, 41(5), pp. 697 – 702. doi: 10.1212/WNL.41.5.697.
- Patton, J.L., Stoykov, M.E., Kovic, M., Mussa-Ivaldi, F.A. (2006) 'Evaluation of robotic training forces that either enhance or reduce error in chronic hemiparetic stroke survivors', *Exp Brain Res*, 168(3), pp. 368 – 383. doi: 10.1007/s00221-005-0097-8.
- Paus, T. (2005) 'Inferring causality in brain images: a perturbation approach', *Philos Trans R* Soc Lond B Biol Sci, 360(1457), pp. 1109 – 1114. doi: 10.1098/rstb.2005.1652.
- Pavlides, C., Miyashita, E. and Asanuma, H. (1993) 'Projection from the sensory to the motor cortex is important in learning motor skills in the monkey', *J Neurophysiol*, 70(2), pp. 733 – 741. doi: 10.1152/jn.1993.70.2.733.
- Paz, R., Boraud, T., Natan, C., Bergman, H. and Vaadia, E. (2003) 'Preparatory activity in motor cortex reflects learning of local visuomotor skills', *Nat Neurosci*, 6(8), pp. 882 – 890. doi: 10.1038/nn1097.
- Penfield, W. and Boldrey, E. (1937). 'Somatic motor and sensory representation in the cerebral cortex of man as studied by electrical stimulation', *Brain: A Journal of Neurology, 60*, pp. 389 – 443. Doi: 10.1093/brain/60.4.389 - figure

- Penfield, W. and Rasmussen, T. (1950) *The cerebral cortex of man; a clinical study of localization of function*. England Oxford: Macmillan.
- Pennisi, G., Rapisarda, G., Bella, R., Calabrese, V., Maertens De Noordhout, A. and Delwaide, P.J. (1999) 'Absence of response to early transcranial magnetic stimulation in ischemic stroke patients: prognostic value for hand motor recovery', *Stroke*, 30(12), pp. 2666 2670. Available at: https://www.ahajournals.org/doi/pdf/10.1161/01.str.30.12.2666 (Accessed: 13th October 2018).
- Perenin, M.T. and Vighetto, A. (1988) 'Optic ataxia: a specific disruption in visuomotor mechanisms. I. Different aspects of the deficit in reaching for objects', *Brain*, 111(Pt 3), pp. 643 – 674. Doi: 10.1093/brain/111.3.643.
- Perfetti, B., Moisello, C., Landsness, E. C., Kvint, S., Lanzafame, S., Onofrj, M., Tononi, G. and Ghilardi, M. F. (2011) 'Modulation of gamma and theta spectral amplitude and phase synchronization is associated with the development of visuo-motor learning', *The Journal of neuroscience: the official journal of the Society for Neuroscience*, 31(41), pp. 14810 14819. doi:10.1523/JNEUROSCI.1319-11.2011.
- Peters, R. (2006) 'Ageing and the brain', *Postgraduate medical journal*, 82(964), pp. 84 88. doi:10.1136/pgmj.2005.036665.
- Petersen, C.C. and Crochet, S. (2013) 'Synaptic computation and sensory processing in neocortical layer 2/3', *Neuron*, 78(1), pp. 28 48. doi: 10.1016/j.neuron.2013.03.020.
- Petrides, M. (1982) 'Motor conditional associative-learning after selective prefrontal lesions in the monkey', *Behav Brain Res*, 5(4), pp. 407 413. doi: 10.1016/0166-4328(82)90044-4.
- Petrides, M. (1982). 'Motor conditional associative-learning after selective prefrontal lesions in the monkey', *Behav Brain Res*, 5(4), pp. 407 413.
- Pfeffer, C. K., Xue, M., He, M., Huang, Z. J. and Scanziani, M. (2013). 'Inhibition of inhibition in visual cortex: the logic of connections between molecularly distinct interneurons', *Nature neuroscience*, 16(8), pp. 1068 – 1076. doi: 10.1038/nn.3446.
- Picard, N. and Strick, P.L. (1996) 'Motor areas of the medial wall: a review of their location and functional activation', *Cereb Cortex*, 6(3), pp. 342 353. Available at: https://www.ncbi.nlm.nih.gov/pubmed/8670662 (Accessed: 13th August 2018).
- Picard, N. and Strick, P.L. (2001) 'Imaging the premotor areas', *Curr Opin Neurobiol*, 11(6), pp. 663 672. doi: 10.1016/S0959-4388(01)00266-5.
- Picard, N. and Strick, P.L. (2003) 'Activation of the Supplementary Motor Area (SMA) during Performance of Visually Guided Movements', *Cerebral Cortex*, 13(9), 977-986. doi: 10.1093/cercor/13.9.977.
- Picht, T., Strack, V., Schulz, J., Zdunczyk, A., Frey, D., Schmidt, S. and Vajkoczy, P. (2012) 'Assessing the functional status of the motor system in brain tumor patients using transcranial magnetic stimulation'. *Acta Neurochir (Wien)*, 154(11), pp. 2075 – 2081. doi: 10.1007/s00701-012-1494-y.

- Pierri, J.N., Volk, C.L., Auh, S., Sampson, A. and Lewis D.A. (2001) 'Decreased somal size of deep layer 3 pyramidal neurons in the prefrontal cortex of subjects with schizophrenia', *Arch Gen Psychiatry*, 58(5), pp. 466 – 473. doi: 10.1001/archpsyc.58.5.466.
- Pisella, L., Grea, H., Tilikete, C., Vighetto, A., Desmurget, M., Rode, G., Boisson, D. and Rosetti, Y. (2000). 'An 'automatic pilot' for the hand in human posterior parietal cortex, toward reinterpreting optic ataxia', *Nat Neurosci*, 3(7), pp. 729 – 736. doi: 10.1038/76694.
- Pizzamiglio, S., De Lillo, M., Naeem, U., Abdalla, H. and Turner, D. L. (2017a) 'High-Frequency Intermuscular Coherence between Arm Muscles during Robot-Mediated Motor Adaptation', *Frontiers in physiology*, 7(668), pp. 1 – 14. doi:10.3389/fphys.2016.00668.
- Pizzamiglio, S., Desowska, A., Shojaii, P., Taga, M. and Turner, D.L. (2017b) 'Muscle cocontraction patterns in robot mediated force field learning to guide specific muscle group training', *NeuroRehabilitation*, 41(1), pp. 17-29. doi: 10.3233/NRE-171453.
- Playford, E.D., Jenkins, I.H., Passingham, R.E., Nutt, J., Frackowiak, R.S. and Brooks, D.J. (1992) 'Impaired mesial frontal and putamen activation in Parkinson's disease: a positron emission tomography study', *Ann Neurol*, 32(2), pp. 151 – 161. doi: 10.1002/ana.410320206.
- Poli, P., Morone, G., Rosati, G. and Masiero, S. (2013) 'Robotic technologies and rehabilitation: new tools for stroke patients' therapy', *BioMed research international*, 2013, 153872, pp. 1 – 8. doi:10.1155/2013/153872.
- Pollok, B., Overhagen, C. L., Keitel, A. and Krause, V. (2017) 'Transcranial direct current stimulation (tDCS) applied to the left dorsolateral premotor cortex (dPMC) interferes with rhythm reproduction', *Scientific reports*, 7(1), 11509, pp. 1 – 8. doi: 10.1038/s41598-017-11980-w.
- Porto, F. H., Fox, A. M., Tusch, E. S., Sorond, F., Mohammed, A. H. and Daffner, K. R. (2015) 'In vivo evidence for neuroplasticity in older adults', *Brain research bulletin*, 114, pp. 56 – 61. doi:10.1016/j.brainresbull.2015.03.004.
- Posner, M.I. and Petersen, S.E. (1990) 'The attention system of the human brain', *Annu Rev Neurosci*, 13, pp. 25 42. doi: 10.1146/annurev.ne.13.030190.000325.
- Poston, B., Kukke, S. N., Paine, R. W., Francis, S. and Hallett, M. (2012) 'Cortical silent period duration and its implications for surround inhibition of a hand muscle', *The European journal of neuroscience*, 36(7), pp. 2964 – 2971. doi:10.1111/j.1460-9568.2012.08212.x.
- Pourtois, G., Vandermeeren, Y., Olivier, E. and de Gelder, B. (2001) 'Event-related TMS over the right posterior parietal cortex induces ipsilateral visuo-spatial interference', *Neuroreport*, 12(11), pp. 2369 – 2374. Available at: https://pdfs.semanticscholar.org/93d8/da6884c6272a382f7db8521fb76c65f27c3a.pdf (Accessed: 16th August 2018).

- Povlishock, J.T. and Katz, D.I. (2005) 'Update of neuropathology and neurological recovery after traumatic brain injury', *J Head Trauma Rehabil*, 20(1), pp. 76 94. doi: 10.1097/00001199-200501000-00008.
- Praamstra, P., Stegeman, D.F., Horstink, M.W., Brunia, C.H. and Cools, A.R. (1996) 'Movement-related potentials preceding voluntary movement are modulated by the mode of movement selection', Exp Brain Res, 103(3), pp. 429 – 439. doi: 10.1007/bf00241502.
- Praeg, E., Herwig, U., Lutz, K. and Jancke, L. (2005) 'The role of the right dorsal premotor cortex in visuomotor learning: A transcranial magnetic stimulation study', *NeuroReport: For Rapid Communication of Neuroscience Research*, 16(15), pp. 1715 – 1718. doi: 10.1097/01.wnr.0000183333.36488.1c.
- Prime, S.L., Vesia, M. and Crawford, J.D. (2008) 'Transcranial magnetic stimulation over posterior parietal cortex disrupts transsaccadic memory of multiple objects', *J Neurosci*, 28(27), 6938 – 6949. doi: 10.1523/JNEUROSCI.0542-08.2008.
- Prins, M., Greco, T., Alexander, D. and Giza, C. C. (2013). 'The pathophysiology of traumatic brain injury at a glance', *Disease models & mechanisms*, 6(6), pp. 1307–1315. doi: 10.1242/dmm.011585.
- Purves, D., Augustine, G.J., Fitzpatrick, D., Katz, L.C., LaMantia, A-S., McNamara, J.O. and Williams, S.M. (2001). Neuroscience, 2nd edition (Chapter title: The Primary Motor Upper Motor Neurons That Initiate Complex Voluntary *Cortex: Movements*).Sunderland (MA): Sinauer Associates. Chapter available at: https://www.ncbi.nlm.nih.gov/books/NBK10962/ (Accessed :19th September 2018).
- Purves, D., Augustine, G.J., Fitzpatrick, D., Katz, L.C., LaMantia, A-S., McNamara, J.O. and Williams, S.M. (2001). Neuroscience, 2nd edition (Chapter title: The Primary Motor Cortex: Upper Motor Neurons That Initiate *Complex* Voluntary Sinauer *Movements*).Sunderland (MA): Associates. Chapter available at: https://www.ncbi.nlm.nih.gov/books/NBK10962/ (accessed 19th September 2018).
- Raghavan, P. (2015) 'Upper Limb Motor Impairment After Stroke', *Physical medicine and rehabilitation clinics of North America*, 26(4), pp. 599 – 610. doi:10.1016/j.pmr.2015.06.008.
- Rajangam, S., Tseng, P-H., Lehew, G., Schwarz, D., Lebedev, M.A. and Nicolelis, M.A.L. (2016) 'Wireless Cortical Brain-Machine Interface for Whole-Body Navigation in Primates', Scientific Reports, 22170, pp. 1 13. doi: 10.1038/srep22170
- Ramos-Murguialday, A., Broetz, D., Rea, M., Läer, L., Yilmaz, O., Brasil, F. L., Liberati, G., Curado, M.R., Garcia-Cossio, E., Vyziotis, A., Cho, W., Agostini, M., Soares, E., Soekadar, S., Caria, A., Cohen, L.G. and Birbaumer, N. (2013) 'Brain-machine interface in chronic stroke rehabilitation: a controlled study', *Annals of neurology*, 74(1), pp. 100 – 108. doi:10.1002/ana.23879.
- Ramos-Murguialday, A., Curado, M.R., Broetz, D., Yilmaz, Ö., Brasil, F.L., Liberati, G., Garcia-Cossio, E., Cho, W., Caria, A., Cohen, L.G. and Birbaumer, N. (2019) 'Brain-

Machine Interface in Chronic Stroke: Randomized Trial Long-Term Follow-up', *Neurorehabil Neural Repair*, 33(3), pp. 188-198. doi: 10.1177/1545968319827573.

- Raz, E., Cercignani, M., Sbardella, E., Totaro, P., Pozzilli, C., Bozzali, M. and Pantano, P. (2010) 'Gray- and white-matter changes 1 year after first clinical episode of multiple sclerosis: MR imaging', *Radiology*, 257(2), pp. 448 – 454. doi: 10.1148/radiol.10100626.
- Reiner, A., Dragatsis, I. and Dietrich, P. (2011). 'Genetics and neuropathology of Huntington's disease', *International review of neurobiology*, 98, pp. 325 – 372. doi: 10.1016/B978-0-12-381328-2.00014-6.
- Reis, J., Schambra, H. M., Cohen, L. G., Buch, E. R., Fritsch, B., Zarahn, E., Celnik., P.A. and Krakauer, J.W. (2009) 'Noninvasive cortical stimulation enhances motor skill acquisition over multiple days through an effect on consolidation', *Proceedings of the National Academy of Sciences of the United States of America*, 106(5), pp. 1590 – 1595. doi:10.1073/pnas.0805413106.
- Reisman, D.S., Block, H.J. and Bastian, A.J. (2005) 'Interlimb coordination during locomotion: what can be adapted and stored?', *Journal of Neurophysiology*, 94(4), pp. 2403 – 2415. doi: doi.org/10.1152/jn.00089.2005.
- Reynolds, C. and Ashby, P. (1999) 'Inhibition in the human motor cortex is reduced just before a voluntary contraction', *Neurology*, 53(4), pp. 730 735. doi: 10.1212/WNL.53.4.730.
- Richardson, A.D., Overduin, S.A., Valero-Cabré, A., Padoa-Schioppa, C., Pascual-Leone, A., Bizzi, E. and Press, D.Z. (2006) 'Disruption of Primary Motor Cortex before Learning Impairs Memory of Movement Dynamics', *Journal of Neuroscience*, 26(48), pp. 12466 – 12470. doi: 10.1523/JNEUROSCI.1139-06.2006.
- Ridding, M.C. and Rothwell, J.C. (2007) 'Is there a future for therapeutic use of transcranial magnetic stimulation?', *Nature Reviews Neuroscience*, 8(7), pp. 559-567. doi: 10.1038/nrn2169.
- Rizzolatti, G. and Luppino, G. (2001) 'The cortical motor system', *Neuron*, 31(6), pp. 889–901. doi: 10.1016/S0896-6273(01)00423-8.
- Rizzolatti, G., Fogassi, L. and Gallese, V. (2002) 'Motor and cognitive functions of the ventral premotor cortex', *Curr Opin Neurobiol*, 12(2), pp. 149 154. doi: 10.1016/S0959-4388(02)00308-2.
- Robertson, E.M., Tormos, J.M., Maeda, F. and Pascual-Leone, A. (2001) 'The role of the dorsolateral prefrontal cortex during sequence learning is specific for spatial information', *Cereb Cortex*, 11(7), pp. 628 635. doi: 10.1093/cercor/11.7.628.
- Rodseth, J., Washabaugh, E. P. and Krishnan, C. (2017) 'A novel low-cost approach for navigated transcranial magnetic stimulation', *Restorative neurology and neuroscience*, 35(6), pp. 601–609. doi: 10.3233/RNN-170751.

- Roland, P.E, Larsen, B., Lassen, N.A. and Skinhøj, E. (1980) 'Supplementary motor area and other cortical areas in organization of voluntary movements in man', *J Neurophysiol*, 43(1), pp. 118 – 136. doi: 10.1152/jn.1980.43.1.118.
- Rorden, C., Brett, M. (2000). Stereotaxic display of brain lesions. Behavioural Neurology, 12, pp. 191-200.
- Rossi, S., Hallett, M., Rossini, P. M., Pascual-Leone, A. and Safety of TMS Consensus Group. (2009) 'Safety, ethical considerations, and application guidelines for the use of transcranial magnetic stimulation in clinical practice and research', *Clinical neurophysiology: official journal of the International Federation of Clinical Neurophysiology*, 120(12), pp. 2008–2039. doi:10.1016/j.clinph.2009.08.016.
- Rossini, P.M., Burke, D., Chen, R., Cohen, L.G., Daskalakis, Z., Di Lorio, R., Di Lazzaro, V., Ferreri, F., Firzgerald, P.B., George, M.S., Hallett, M., Lefaucheur, J.P., Langguth, J.P., Matsumoto, H., Miniussi, C., Nitsche, M.A., Pascual-Leone, A., Paulus, W., Rossi, S., Rothwell, J.C., Siebner, H.R., Ugawa, Y., Walsh, V. and Ziemann, U. (2015) 'Non-invasive electrical and magnetic stimulation of the brain, spinal cord, roots and peripheral nerves: Basic principles and procedures for routine clinical and research application. An updated report from an I.F.C.N. Committee', *Clinical Neurophysiology*, 126(6), pp. 1071 1107. doi: 10.1016/j.clinph.2015.02.001.
- Rothwell, J.C., Day, B.L., Dick, J.P, Kachi, T., Cowan, J.M. and Marsden, C.D. (1987) 'Motor cortex stimulation in intact man. 1. General characteristics of EMG responses in different muscles', *Brain*, 110(Pt 5), pp. 1173 – 1190. doi: 10.1093/brain/110.5.1173.
- Rowe, J.B., Stephan, K.E., Friston, K., Frackowiak, R.S. and Passingham, R.E (2005) 'The prefrontal cortex shows context-specific changes in effective connectivity to motor or visual cortex during the selection of action or colour', *Cereb Cortex*, 15(1), pp. 85 – 95. doi: 10.1093/cercor/bh111.
- Rowland, N. C., De Hemptinne, C., Swann, N. C., Qasim, S., Miocinovic, S., Ostrem, J. L., Knight, R.T and Starr, P. A. (2015) 'Task-related activity in sensorimotor cortex in Parkinson's disease and essential tremor: changes in beta and gamma bands', *Frontiers in human neuroscience*, 9(article 512), pp 1 – 12. doi:10.3389/fnhum.2015.00512.
- Roy, F. D. and Gorassini, M. A. (2008) 'Peripheral sensory activation of cortical circuits in the leg motor cortex of man', *The Journal of physiology*, 586(17), pp. 4091–4105. doi:10.1113/jphysiol.2008.153726.
- Ruff, C. C., Driver, J. and Bestmann, S. (2009) 'Combining TMS and fMRI: from 'virtual lesions' to functional-network accounts of cognition', *Cortex; a journal devoted to the study of the nervous system and behavior*, 45(9), pp. 1043 – 1049. doi:10.1016/j.cortex.2008.10.012.
- Ruitenberg, M.F., Verwey, W.B., Schutter, D.J. and Abrahamse, E.L. (2014) 'Cognitive and neural foundations of discrete sequence skill: a TMS study', *Neuropsychologia*, 56, pp. 229 – 238. doi: 10.1016/j.neuropsychologia.2014.01.014.

- Ruohonen, J. and Ilmoniemi, R.J. (2002) 'Physical principles for transcranial magnetic stimulation', in Pascual-Leone, A., Davey, N.J., Rothwell, J., Wassermann, E.M. and Puri, B.K. (ed.) *Handbook of transcranial magnetic stimulation*. New York, US: Oxford University Press, pp. 18 – 30.
- Rushworth, M. F., Hadland, K. A., Paus, T. and Sipila, P. K. (2002) 'Role of the human medial frontal cortex in task switching: a combined fMRI and TMS study', *J. Neurophysiol*, 87, pp. 2577 – 2592. doi: 10.1152/jn.2002.87.5.2577.
- Rushworth, M.F., Johansen-Berg, H., Göbel, S.M. and Devlin, J.T. (2003). 'The left parietal and premotor cortices: motor attention and selection', *Neuroimage*, 20(Suppl 1), pp. S89 100. doi: 10.1016/j.neuroimage.2003.09.011.
- Rushworth, M.F., Krams, M. and Passingham, R.E. (2001) 'The attentional role of the left parietal cortex: the distinct lateralization and localization of motor attention in the human brain', J Cogn Neurosci, 13(5), pp. 698 – 710. doi: 10.1162/089892901750363244.
- Rushworth, M.F., Nixon, P.D., Renowden, S., Wade, D.T. and Passingham, R.E. (1997) 'The left parietal cortex and motor attention', *Neuropsychologia*, 35(9), pp. 1261 1273. doi: 10.1016/S0028-3932(97)00050-X.
- Sack, A.T. (2006) 'Transcranial magnetic stimulation, causal structure-function mapping and networks of functional relevance', *Curr Opin Neurobiol*, 16(5), pp. 593 – 599. doi: 10.1016/j.conb.2006.06.016.
- Sabouni, A., Honrath, M. and Khamechi, M. (2017) 'Thermal Effects in the Brain During Transcranial Magnetic Stimulation', Institute of Electrical and Electronics Engineers (IEEE), 8, doi: 10.1109/LMAG.2017.2759209.
- Sadato, N., Campbell, G., Ibanez, V., Deiber, M. and Hallett, M. (1996) 'Complexity affects regional cerebral blood flow change during sequential finger movements', *J. Neurosci.*, 16(8), pp. 2691 – 2700. doi: 10.1523/JNEUROSCI.16-08-02691.1996.
- Säisänen, L., Julkunen, P., Niskanen, E., Danner, N., Hukkanen, T., Lohioja, T., Nurkkala, J., Mervaala, E., Karhu, J. and Könönen, M. (2008) 'Motor potentials evoked by navigated transcranial magnetic stimulation in healthy subjects', *J Clin Neurophysiol*, 25(6), pp. 367 – 372. doi: 10.1097/WNP.0b013e31818e7944.
- Sakai, K., Hikosaka, O., Miyauchi, S., Sasaki, Y., Fujimaki, N. and Pütz B. (1999) 'Presupplementary motor area activation during sequence learning reflects visuo-motor association', *J Neurosci*, 19(10), RC1, pp. 1 – 6. doi: 10.1523/JNEUROSCI.19-10j0002.1999.
- Sakata, H., Taira, M., Murata, A. and Mine, S. (1995) 'Neural mechanisms of visual guidance of hand action in the parietal cortex of the monkey', *Cereb Cortex*, 5(5), pp. 429 – 438. doi: http://dx.doi.org/10.1093/cercor/5.5.429.
- Salatino, A., Poncini, M., George, M. S. and Ricci, R. (2014) 'Hunting for right and left parietal hot spots using single-pulse TMS: modulation of visuospatial perception during line

bisection judgment in the healthy brain', *Frontiers in psychology*, 5(article 1238), pp. 1 – 7. doi:10.3389/fpsyg.2014.01238.

- Sanes, J.N. and Donoghue, J.P. (2000) 'Plasticity and primary motor cortex', *Annual review of Neuroscience*, 23, pp. 393 415. doi: 10.1146/annurev.neuro.23.1.393.
- Sansare, K., Khanna, V. and Karjodkar, F. (2011) 'Early victims of X-rays: a tribute and current perception', *Dento maxillo facial radiology*, 40(2), pp. 123–125. doi: doi:10.1259/dmfr/73488299
- Sarter, M., Givens, B. and Bruno, J.P. (2001) 'The cognitive neuroscience of sustained attention: where top-down meets bottom-up', *Brain Res Brain Res Rev*, 35(2), pp. 146 160. doi: 10.1016/S0165-0173(01)00044-3.
- Scahill, R.I., Frost, C., Jenkins, R., Whitwell, J.L, Rossor, M.N. and Fox, N.C. (2003) 'A longitudinal study of brain volume changes in normal aging using serial registered magnetic resonance imaging', *Arch Neurol*, 60(7), pp. 989 – 994. doi: 10.1001/archneur.60.7.989.
- Schlerf, J. E., Galea, J. M., Spampinato, D. and Celnik, P. A. (2014) 'Laterality Differences in Cerebellar-Motor Cortex Connectivity. *Cerebral cortex (New York, N.Y: 1991)*, 25(7), pp. 1827 – 1834. doi:10.1093/cercor/bht422
- Schluter, N.D., Krams, M., Rushworth, M.F. and Passingham, R.E. (2001) 'Cerebral dominance for action in the human brain: the selection of actions', *Neuropsychologia*, 39(2), pp. 105 – 113. doi: 10.1016/S0028-3932(00)00105-6.
- Schluter, N.D., Rushworth, M.F., Passingham, R.E. and Mills, K.R. (1998) 'Temporary interference in human lateral premotor cortex suggests dominance for the selection of movements. A study using transcranial magnetic stimulation', *Brain*, 121(pt 5), pp. 785 – 799. doi: 10.1093/brain/121.5.785.
- Schnider, A., Hanlon, R.E., Alexander, D.N. and Benson, D.F. (1997) 'Ideomotor apraxia: behavioral dimensions and neuroanatomical basis', *Brain Lang*, 58(1), pp. 125 – 136. doi: 10.1006/brln.1997.1770.
- Schroeder, K. E. and Chestek, C. A. (2016) 'Intracortical Brain-Machine Interfaces Advance Sensorimotor Neuroscience', *Frontiers in neuroscience*, 10(Article 291), pp. 1 – 8. doi:10.3389/fnins.2016.00291.
- Schubert, D., Kötter, R., Zilles, K., Luhmann H.J. and Staiger, J.F. (2003) 'Cell type-specific circuits of cortical layer IV spiny neurons', *The Journal of neuroscience: the official journal of the Society for Neuroscience*, 23(7), pp. 2961 – 2970. doi: 10.1523/JNEUROSCI.23-07-02961.2003.
- Schubotz, R.I. and von Cramon, D.Y. (2003) 'Functional-anatomical concepts of human premotor cortex: evidence from fMRI and PET studies', *Neuroimage*, 20(Suppl 1), pp. S120 - 131. Doi: 10.1016/j.neuroimage.2003.09.014.
- Schüller, A. (1912). 'Die Schädelveränderungen bei intrakranieller Drucksteigerung'. *Deutsche Zeitschrift Für Nervenheilkunde*, 45(4-6), 287-294. doi: 10.1007/bf01629354.

- Schutter, D.J. and van Honk, J. (2006) 'An electrophysiological link between the cerebellum, cognition and emotion: frontal theta EEG activity to single-pulse cerebellar TMS', *Neuroimage*, 33(4), pp. 1227 – 1231. doi: 10.1016/j.neuroimage.2006.06.055.
- Scott, S.H. (2004) 'Optimal feedback control and the neural basis of volitional motor control', *Nature Reviews Neuroscience*, 5(7), pp. 532 546. doi: https://doi.org/10.1038/nrn1427.
- Scott, S.H. and Kalaska, J.F. (1997) 'Reaching movements with similar hand paths but different arm orientations. I. Activity of individual cells in motor cortex', *J Neurophysiol*, 77(2), pp. 826 – 852. Doi: 10.1152/jn.1997.77.2.826.
- Serrien, D.J., Strens, L.H., Oliviero, A. and Brown, P. (2002) 'Repetitive transcranial magnetic stimulation of the supplementary motor area (SMA) degrades bimanual movement control in humans', *Neurosci Lett*, 328(2), pp. 89 – 92. doi: 10.1016/S0304-3940(02)00499-8.
- Shallice, T. (1982)'Specific impairments of planning', *Philos Trans R Soc Lond B Biol Sci*, 298(1089), pp. 199 209. doi: 10.1098/rstb.1982.0082.
- Shannon, B.J., Vaishnavi, S.N., Vlassenko, A.G., Shimony, J.S., Rutlin, J. and Raichle, M.E. (2016) 'Brain aerobic glycolysis and motor adaptation learning', PNAS, 113, pp. E3782 – E3791. Doi: 10.1073/pnas.1604977113.
- Shen, L. and Alexander, G.E. (1997) 'Preferential representation of instructed target location versus limb trajectory in dorsal premotor area', *J Neurophysiol*, 77(3), pp. 1195 – 1212. doi: doi.org/10.1152/jn.1997.77.3.1195.
- Shibasaki, H. and Hallett, M. (2006) 'What is the Bereitschaftspotential?', *Clin Neurophysiol*, 117(11), pp. 2341 2356. doi: 10.1016/j.clinph.2006.04.025.
- Shibasaki, H., Sadato, N., Lyshkow, H., Yonekura, Y., Honda, M., Nagamine, T., Suwazono, S., Magata, Y., Ikeda, A., Miyazaki, M., Fukuyama, H., Asato, R. and Konishi, J. (1993) 'Both primary motor cortex and supplementary motor area play an important role in complex finger movement', *Brain*, 116(pt 6), pp. 1387 – 1398. doi: 10.1093/brain/116.6.1387.
- Shih, J. J., Krusienski, D. J. and Wolpaw, J. R. (2012) 'Brain-computer interfaces in medicine', *Mayo Clinic proceedings*, 87(3), pp. 268–279. doi:10.1016/j.mayocp.2011.12.008.
- Shima, K., Isoda, M., Mushiake, H. and Tanji, J. (2007) 'Categorization of behavioural sequences in the prefrontal cortex', 445(7125), pp. 315 318. doi: 10.1038/nature05470.
- Shukla, A. K., and Kumar, U. (2006) 'Positron emission tomography: An overview', Journal of medical physics, 31(1), pp. 13–21. doi:10.4103/0971-6203.25665.
- Shum, M., Shiller, D. M., Baum, S. R. and Gracco, V. L. (2011) 'Sensorimotor integration for speech motor learning involves the inferior parietal cortex', *The European Journal of Neuroscience*, 34(11), pp. 1817 – 1822. doi: 10.1111/j.1460-9568.2011.07889.x.

- Siebner, H. R., Hartwigsen, G., Kassuba, T. and Rothwell, J. C. (2009) 'How does transcranial magnetic stimulation modify neuronal activity in the brain? Implications for studies of cognition', *Cortex; a journal devoted to the study of the nervous system and behavior*, 45(9), pp. 1035 – 1042. doi:10.1016/j.cortex.2009.02.007.
- Silva Moreira, P., Marques, P. and Magalhães, R. (2016) 'Identifying Functional Subdivisions in the Medial Frontal Cortex', *The Journal of neuroscience: the official journal of the Society for Neuroscience*, 36(44), pp. 11168 – 11170. doi:10.1523/JNEUROSCI.2584-16.2016.
- Silvanto, J. and Muggleton, N.G. (2008) 'New light through old windows: moving beyond the "virtual lesion" approach to transcranial magnetic stimulation', *Neuroimage*, 39(2), pp. 549 552. doi: 10.1016/j.neuroimage.2007.09.008.
- Simonetta, M., Clanet, M. and Rascol, O. (1991) 'Bereitschaftspotential in a simple movement or in a motor sequence starting with the same simple movement', *Electroencephalography & Clinical Neurophysiology: Electromyography & Motor Control*, 81(2), pp. 129 – 134. doi: 10.1016/0168-5597(91)90006-J.
- Singh, R.K. and Pooley, J. (2002) 'Complete rupture of the triceps brachii muscle', Br J Sports Med, 36(6), pp. 467 469. doi: 10.1136/bjsm.36.6.467.
- Sliwinska, M. W., Vitello, S. and Devlin, J. T. (2014) 'Transcranial magnetic stimulation for investigating causal brain-behavioral relationships and their time course', *Journal of visualized experiments: JoVE*, 89, e51735, pp. 1 – 9. doi:10.3791/51735.
- Small, S.L., Buccino, G. and Solodkin, A. (2013) 'Brain repair after stroke-a novel neurological model', *Nat Rev Neurol*, 9(12), pp. 698 – 707. doi: 10.1038/nrneurol.2013.222.
- Smith P. F. (2017) 'A Guerilla Guide to Common Problems in 'Neurostatistics': Essential Statistical Topics in Neuroscience', Journal of undergraduate neuroscience education: JUNE: a publication of FUN, Faculty for Undergraduate Neuroscience, 16(1), R1– R12.
- Smith, M. A., Brandt, J. and Shadmehr, R. (2000) 'Motor disorder in Huntington's disease begins as a dysfunction in error feedback control', *Nature*, 403(6769), pp. 544 – 549. doi: 10.1038/35000576.
- Smith, M.A. and Shadmehr, R. (2005) 'Intact ability to learn internal models of arm dynamics in Huntington's disease but not cerebellar degeneration', *J Neurophysiol*, 93(5), pp. 2809 – 2821. doi: 10.1152/jn.00943.2004.
- Smith, M.J., Keel, J.C., Greenberg, B.D., Adams, L.F., Schmidt, P.J., Rubinow, D.A. and Wassermann, E.M. (1999) 'Menstrual cycle effects on cortical excitability', *Neurology*, 55(9), pp. 2069 - 2072. doi: 10.1212/WNL.53.9.2069.
- Soechting, J.F. and Ross, B. (1984) 'Psychophysical determination of coordinate representation of human arm orientation', *Neuroscience*, 13(2), pp. 595 604. doi 10.1016/0306-4522(84)90252-5.

- Soechting, J.F., Lacquaniti, F. and Terzuolo, C.A. (1986) 'Coordination of arm movements in three-dimensional space. Sensorimotor mapping during drawing movement', *Neuroscience*, 17(2), pp. 295 – 311. doi: 10.1016/0306-4522(86)90248-4.
- Sollmann, N., Bulubas, L., Tanigawa, N., Zimmer, C., Meyer, B. and Krieg, S. M. (2017) 'The variability of motor evoked potential latencies in neurosurgical motor mapping by preoperative navigated transcranial magnetic stimulation', *BMC neuroscience*, 18(5), pp. 1 – 15. doi:10.1186/s12868-016-0321-4.
- Solopchuk, O., Alamia, A. and Zénon, A. (2016) 'The Role of the Dorsal Premotor Cortex in Skilled Action Sequences', J Neurosci, 36(25), pp. 6599 – 6601. doi: 10.1523/JNEUROSCI.1199-16.2016.
- Spampinato, D.A., Block, H.J. and Celnik, P.A. (2017) 'Cerebellar–M1 Connectivity Changes Associated with Motor Learning Are Somatotopic Specific', *Journal of Neuroscience*, 37 (9), pp. 2377 – 2386. https://doi.org/10.1523/JNEUROSCI.2511-16.2017.
- Sparing, R., Buelte, D., Meister, I.G., Paus, T. and Fink, G.R. (2008) 'Transcranial magnetic stimulation and the challenge of coil placement: a comparison of conventional and stereotaxic neuronavigational strategies', Hum Brain Mapp, 29(1), pp. 82 – 96. doi: 10.1002/hbm.20360.
- Sparing, R., Hesse, M.D. and Fink, G.R. (2008) 'Neuronavigation for transcranial magnetic stimulation (TMS): where we are and where we are going', *Cortex*, 46(1), pp. 118 – 120. doi: 10.1016/j.cortex.2009.02.018.
- Spruston, N. (2008) 'Pyramidal neurons: dendritic structure and synaptic integration', *Nat Rev Neurosci*, 9(3), pp. 206 221. doi: 10.1038/nrn2286.
- Steyvers, M., Etoh, S., Sauner, D., Levin, O., Siebner, H.R., Swinnen, S.P. and Rothwell J.C. (2003) 'High-frequency transcranial magnetic stimulation of the supplementary motor area reduces bimanual coupling during anti-phase but not in-phase movements', *Exp Brain Res*, 151(2), pp. 309 – 317. doi: 10.1007/s00221-003-1490-9.
- Stinear, C.M. (2017) 'Prediction of motor recovery after stroke: advances in biomarkers', *Lancet Neurol*, 16(10), pp. 826 – 836. doi: 10.1016/S1474-4422(17)30283-1.
- Stocchetti, N. and Zanier, E. R. (2016) 'Chronic impact of traumatic brain injury on outcome and quality of life: a narrative review', *Critical care*, 20, pp. 1-10. doi:10.1186/s13054-016-1318-1.
- Stockinger, C., Focke, A. and Stein, T. (2014) 'Catch trials in force field learning influence adaptation and consolidation of human motor memory', *Frontiers in Human Neuroscience*, 8(231), pp. 1–15. doi: 10.3389/fnhum.2014.00231.
- Summers, R. L. S., Chen, M. and Kimberley, T. J. (2017) 'Corticospinal excitability measurements using transcranial magnetic stimulation are valid with intramuscular electromyography', *PLoS ONE*, 12(2), e0172152. doi: 10.1371/journal.pone.0172152.
- Sun, T. and Walsh, A. (2006) 'Molecular approaches to brain asymmetry and handedness', *Nature Reviews Neuroscience*, 7(8), pp. 655 662. doi: 10.1038/nrn1930.

- Swadlow, H.A. (1994). 'Efferent neurons and suspected interneurons in motor cortex of the awake rabbit: axonal properties, sensory receptive fields, and subthreshold synaptic inputs', *J Neurophysiol*, 71(2), pp. 437 – 453. doi: 10.1152/jn.1994.71.2.437.
- Talelli, P., Greenwood, R.J. and Rothwell, J.C. (2006) 'Arm function after stroke: neurophysiological correlates and recovery mechanisms assessed by transcranial magnetic stimulation', *Clinical neurophysiology*, 117(8), pp. 1641-1659. doi 10.1016/j.clinph.2006.01.016.
- Tanji, J. (1994). 'The supplementary motor area in the cerebral cortex', *Neurosci Res*, 19(3), pp. 251 268. doi: 10.1016/0168-0102(94)90038-8.
- Tanji, J. and Hoshi, E. (2001) 'Behavioral planning in the prefrontal cortex', *Curr Opin Neurobiol*, 11(2), pp. 164 – 170. doi: doi.org/10.1016/S0959-4388(00)00192-6.
- Tanji, J. and Mushiake, H. (1996) 'Comparison of neuronal activity in the supplementary motor area and primary motor cortex', *Brain Res Cogn Brain Res*, 3(2), pp. 143 – 150. doi: 10.1016/0926-6410(95)00039-9.
- Tankus, A., Yeshurun, Y., Flash, T. and Fried, I. (2009) 'Encoding of speed and direction of movement in the human supplementary motor area', *Journal of neurosurgery*, 110(6), pp. 1304 – 1316. doi:10.3171/2008.10.JNS08466.
- Tatemichi, T. K., Desmond, D. W., Stern, Y., Paik, M., Sano, M. and Bagiella, E. (1994)
 'Cognitive impairment after stroke: frequency, patterns, and relationship to functional abilities', *Journal of neurology, neurosurgery, and psychiatry*, 57(2), pp. 202 207. Available
 https://www.ncbi.nlm.nih.gov/pmc/articles/PMC1072451/pdf/jnnpsyc00032-0074.pdf (Accessed: 24th October 2018).
- Taube, W., Schubert, M., Gruber, M., Beck, S., Faist, M. and Gollhofer, A. (2006) 'Direct corticospinal pathways contribute to neuromuscular control of perturbed stance', *J Appl Physiol*, 2006, 101(2), pp. 420 – 429. doi: 10.1152/japplphysiol.01447.2005.
- Tehovnik, E.J, Sommer, M.A., Chou, I.H., Slocum, W.M. and Schiller, P.H. (2000) 'Eye fields in the frontal lobes of primates', *Brain Res Brain Res Rev*, 32(2-3), pp. 413 448. doi: 10.1016/S0165-0173(99)00092-2.
- Teka, W. W., Hamade, K. C., Barnett, W. H., Kim, T., Markin, S. N., Rybak, I. A. and Molkov, Y. I. (2017) 'From the motor cortex to the movement and back again', *PloS one*, 12(6), e0179288. doi:10.1371/journal.pone.0179288.
- Tendler, A., Roth, Y., Barnea-Ygael, N. and Zangen, A. (2017) 'How to Use the H1 Deep Transcranial Magnetic Stimulation Coil for Conditions Other than Depression', *Journal of visualized experiments: JoVE*, 119 e55100, pp. 1 11. doi:10.3791/55100.
- Terao, Y. and Ugawa, Y. (2002). 'Basic mechanisms of TMS', *J Clin Neurophysiol*, 19(4), pp. 322 343. doi: 10.1097/00004691-200208000-00006.
- Terao, Y., Ugawa, Y., Enomoto, H., Furubayashi, T., Shiio, Y., Machii, K., Hanajima, R., Nishikawa, M., Iwata, N.K., Saito, Y. and Kanazawa, I. (2001) 'Hemispheric

lateralization in the cortical motor preparation for human vocalization', *J Neurosci*, 21(5), pp. 1600 – 1609. doi: 10.1523/JNEUROSCI.21-05-01600.2001.

- Théoret, H., Haque, J. and Pascual-Leone, A. (2001) 'Increased variability of paced finger tapping accuracy following repetitive magnetic stimulation of the cerebellum in humans', Neurosci Lett, 306(1-2), pp. 29 32. doi: 10.1016/S0304-3940(01)01860-2.
- Thickbroom G.W., Byrnes, M.L., Archer, S.A. and Mastaglia, F.L. (2002) 'Motor outcome after subcortical stroke: MEPs correlate with hand strength but not dexterity', *Clin Neurophysiol*, 113(12), pp. 2025 2029. doi: 10.1016/S1388-2457(02)00318-8.
- Thickbroom, G.W., Byrnes, M.L, Archer, S.A. and Mastaglia, F.L. (2004) 'Motor outcome after subcortical stroke correlates with the degree of cortical reorganization', *Clin Neurophysiol*, 115(9), pp. 2144 2150.
- Thickbroom, G.W., Byrnes, M.L. and Mastaglia F.L. (1999). 'A model of the effect of MEP amplitude variation on the accuracy of TMS mapping', *Clin Neurophysiol*, *110*(5), pp. 941–943. doi: 10.1016/S1388-2457(98)00080-7.
- Tjia, M., Yu, X., Jammu, L. S., Lu, J. and Zuo, Y. (2017) 'Pyramidal Neurons in Different Cortical Layers Exhibit Distinct Dynamics and Plasticity of Apical Dendritic Spines', *Frontiers in neural circuits*, 11(43), pp. 1 – 10. doi:10.3389/fncir.2017.00043.
- Tobimatsu, S., Sun, S.J, Fukui, R. and Kato, M. (1998). 'Effects of sex, height and age on motor evoked potentials with magnetic stimulation', *J Neurol*, 245(5), pp. 256 – 261. Available at: https://link.springer.com/article/10.1007%2Fs004150050215 (Accessed: 24th October 2018).
- Tondreau, R. (1985) 'Ventriculography and pneumoencephalography: Contributions of Dr. Walter E. Dandy', *Radiographics*, 5(4), 553-555. doi: 10.1148/radiographics.5.4.553.
- Townsend D. W. (2008). 'Combined positron emission tomography-computed tomography: the historical perspective', *Seminars in ultrasound, CT, and MR*, 29(4), pp. 232–235. doi:10.1053/j.sult.2008.05.006.
- Tranel, D., Kemmerer, D., Adolphs, R., Damasio, H. and Damasio A.R. (2003) 'Neural correlates of conceptual knowledge for actions', *Cogn Neuropsychol*, 20(3), pp. 409 – 432. doi: 10.1080/02643290244000248.
- Tseng, Y.W, Diedrichsen, J., Krakauer, J.W, Shadmehr, R., Bastian, A.J. (2007). 'Sensory prediction errors drive cerebellum-dependent adaptation of reaching', *J Neurophysiol*, 98(1), pp. 54 – 62. doi: 10.1152/jn.00266.2007.
- Tubiana, M. (1996) 'A hundred years of diagnostic imaging'. World health forum, 17(1), pp. 13. Available at: http://www.who.int/iris/handle/10665/47943 (Accessed on 09 September 2018).
- Turner, D. L., Ramos-Murguialday, A., Birbaumer, N., Hoffmann, U. and Luft, A. (2013) 'Neurophysiology of robot-mediated training and therapy: a perspective for future use in clinical populations' *Frontiers in neurology*, 4(article 184), pp. 1 – 11. doi: 10.3389/fneur.2013.00184.

- Ueno, M., Nakamura, Y., Li, J., Gu, Z., Niehaus, J., Maezawa, M., Crone, S.A., Goulding, M., Baccei, M.L. and Yoshida, Y. (2018). 'Corticospinal Circuits from the Sensory and Motor Cortices Differentially Regulate Skilled Movements through Distinct Spinal Interneurons', *Cell reports*, 23(5), pp. 1286 1300.e7. doi:10.1016/j.celrep.2018.03.137.
- Ueno, S., Tashiro, T. and Harada, K. (1988) 'Localized stimulation of neural tissues in the brain by means of a paired configuration of time-varying magnetic fields', *Journal of Applied Physics*, 64(10), pp. 5862–5864. doi: 10.1063/1.342181.
- Ugawa, Y., Uesaka, Y., Terao, Y., Hanajima, R. and Kanazawa, I. (1995) 'Magnetic stimulation over the cerebellum in humans', *Annuals of Neurology*, 37(6), pp. 703 713. doi: 10.1002/ana.410370603.
- Vallar, G. and Perani, D. (1986) 'The anatomy of unilateral neglect after right-hemisphere stroke lesions. A clinical/CT-scan correlation study in man', *Neuropsychologia*, 24(5), pp. 609 – 622. doi: 10.1016/0028-3932(86)90001-1.
- Van den Berg, F. E., Swinnen, S. P. and Wenderoth, N. (2011) 'Excitability of the Motor Cortex Ipsilateral to the Moving Body Side Depends on Spatio-Temporal Task Complexity and Hemispheric Specialization', *PLoS ONE*, 6(3), e17742, pp. 1 – 10. doi: 10.1371/journal.pone.0017742.
- van Mier, H.I, Perlmutter, J.S. and Petersen, S.E. (2004) 'Functional changes in brain activity during acquisition and practice of movement sequences', *Motor Control*, 8(4), pp. 500 – 520. doi: 10.1123/mcj.8.4.500.
- Vergani, F., Lacerda, L., Martino, J., Attems, J., Morris, C., Mitchell, P., Thiebaut de Schotten, M. and Dell'Acqua, F. (2014) 'White matter connections of the supplementary motor area in humans', *J Neurol Neurosurg Psychiatry*, 85(12), pp. 1377 – 1385. doi: 10.1136/jnnp-2013-307492.
- Verleger, R., Haake, M., Baur, A. and Śmigasiewicz, K. (2016) 'Time to Move Again: Does the Bereitschaftspotential Covary with Demands on Internal Timing?', *Frontiers in human neuroscience*, 10(article 642), pp. 1 – 13. doi:10.3389/fnhum.2016.00642.
- Vernet, M., Yang, Q. and Kapoula, Z. (2011) 'Guiding Binocular Saccades during Reading: A TMS Study of the PPC', *Frontiers in human neuroscience*, 5, 14. doi:10.3389/fnhum.2011.00014.
- Verstynen, T., Diedrichsen, J., Albert, N., Aparicio, P. and Ivry, R.B. (2005) 'Ipsilateral motor cortex activity during unimanual hand movements relates to task complexity', J Neurophysiol, 93(3), pp. 1209 – 1222. doi: 10.1152/jn.00720.2004.
- Vesia, M., Henriques, D., Yan, X., Sergio, L. and Crawford, J. D. (2007) 'TMS over posterior parietal cortex disrupts the integration of initial hand position information into the reach plan', *Journal of Vision*, 7(9), pp. 293 - 293a. doi: 10.1167/7.9.293.
- Vijayakumar, A. and Vijayakumar, A. (2012) 'Comparison of hippocampal volume in dementia subtypes', *ISRN radiology*, 2013, 174524, pp. 1–5. doi:10.5402/2013/174524.

- Vingerhoets, G., Alderweireldt, A.S., Vandemaele, P., Cai, Q., Van der Haegen, L., Brysbaert, M. and Achten, E. (2013) 'Praxis and language are linked: evidence from colateralization in individuals with atypical language dominance', *Cortex*, 49(1), pp. 172 – 183. doi: 10.1016/j.cortex.2011.11.003.
- von Bonin, G. and Mehler, W.R. (1971) 'On columnar arrangement of nerve cells in cerebral cortex', Brain Res, 27(1), pp. 1 9. doi: 10.1016/0006-8993(71)90367-2.
- Wagle-Shukla, A., Ni, Z., Gunraj, C. A., Bahl, N. and Chen, R. (2009) 'Effects of short interval intracortical inhibition and intracortical facilitation on short interval intracortical facilitation in human primary motor cortex.' *The Journal of Physiology*, 587(Pt 23), pp. 5665–5678. doi: 10.1113/jphysiol.2009.181446.
- Wagner, T., Valero-Cabre, A. and Pascual-Leone, A. (2007). 'Noninvasive human brain stimulation', Annual Review of Biomedical Engineering, 9, pp. 527–565. doi: 10.1146/annurev.bioeng.9.061206.133100.
- Wang, Y., Chen, H., Gong, Q., Shen, S. and Gao, Q. (2010) 'Analysis of functional networks involved in motor execution and motor imagery using combined hierarchical clustering analysis and independent component analysis', *Magnetic Resonance Imaging*, 28(5), pp. 653 – 660. doi: 10.1016/j.mri.2010.02.008.
- Wang, Y., Isoda, M., Matsuzaka, Y., Shima, K. and Tanji, J. (2005) 'Prefrontal cortical cells projecting to the supplementary eye field and presupplementary motor area in the monkey', *Neurosci Res*, 53(1), pp. 1 – 7. doi: 10.1016/j.neures.2005.05.005.
- Wang, Y., Shima, K., Sawamura, H. and Tanji, J. (2001) 'Spatial distribution of cingulate cells projecting to the primary, supplementary, and pre-supplementary motor areas: a retrograde multiple labeling study in the macaque monkey', *Neurosci Res*, 39(1), pp. 39 – 49. doi: 10.1016/S0168-0102(00)00198-X.
- Ward, N. S., Bestmann, S., Hartwigsen, G., Weiss, M. M., Christensen, L. O., Frackowiak, R. S., Rothwell, J.C. and Siebner, H. R. (2010) 'Low-frequency transcranial magnetic stimulation over left dorsal premotor cortex improves the dynamic control of visuospatially cued actions', *The Journal of neuroscience: the official journal of the Society for Neuroscience*, 30(27), pp. 9216 9223. doi:10.1523/JNEUROSCI.4499-09.2010.
- Ward, N. S., Newton, J. M., Swayne, O. B., Lee, L., Frackowiak, R. S., Thompson, A. J., Greenwood, R.J. and Rothwell, J.C. (2007) 'The relationship between brain activity and peak grip force is modulated by corticospinal system integrity after subcortical stroke', *The European journal of neuroscience*, 25(6), pp. 1865 – 1873. doi:10.1111/j.1460-9568.2007.05434.x.
- Wardak C. (2011) 'The Role of the Supplementary Motor Area in Inhibitory Control in Monkeys and Humans', *The Journal of Neuroscience*, 31(14), pp. 5181 – 5183. doi: 10.1523/JNEUROSCI.0006-11.2011.
- Watkins, K.E., Paus, T., Lerch, J.P., Zijdenbos, A., Collins, D.L, Neelin, P., Taylor, J., Worsley, K.J. and Evans, A.C. (2001) 'Structural asymmetries in the human brain: a

voxel-based statistical analysis of 142 MRI scans', *Cereb Cortex*, 11(9), pp. 868 – 877. doi: 10.1093/cercor/11.9.868.

- Watson, J. C. and Wilson, A. M. (2007) 'Muscle architecture of biceps brachii, triceps brachii and supraspinatus in the horse', *Journal of anatomy*, 210(1), pp. 32 – 40. doi:10.1111/j.1469-7580.2006.00669.x.
- Wei, P., Bao, R., Lv, Z. and Jing, B. (2018) 'Weak but Critical Links between Primary Somatosensory Centers and Motor Cortex during Movement', *Frontiers in human neuroscience*, 12(article 1), pp. 1 – 13. doi:10.3389/fnhum.2018.00001.
- Weiermann, B., Cock, J. and Meier, B. (2010) 'What matters in implicit task sequence learning: perceptual stimulus features, task sets, or correlated streams of information?' *Journal of experimental psychology. Learning, memory and cognition*, 36(6), pp. 1492–1509. doi: 10.1037/a0021038.
- Weiler, N., Wood, L., Yu, J., Solla, S. A. and Shepherd, G. M. (2008) 'Top-down laminar organization of the excitatory network in motor cortex', *Nature neuroscience*, 11(3), pp. 360–366. doi:10.1038/nn2049.
- Wellington, S.L. and Vinegar, H.J. (1987) 'X-Ray Computerized Tomography', *Journal of Petroleum Technology*, 39(8), pp. 885 898. doi: 10.2118/16983-PA.
- Wellington, S.L. and Vinegar, H.J. (1987) 'X-Ray Computerized Tomography', Journal of Petroleum Technology, 39(8), pp. 885 – 898. doi: <u>10.2118/16983-PA</u>.
- Welniarz, Q., Morel, M.P., Pourchet, O., Gallea, C., Lamy, J.C., Cincotta, M., Doulazmi, M., Belle, M., Méneret, A., Trouillard, O., Ruiz, M., Brochard, V., Meunier, S., Trembleau, A., Vidailhet, M., Chédotal, A., Dusart, I. and Roze, E. (2017) 'Non cell-autonomous role of DCC in the guidance of the corticospinal tract at the midline', Sci Rep, 7(1), pp. 1 – 14. doi: 10.1038/s41598-017-00514-z.
- Werhahn, K. J., Kunesch, E., Noachtar, S., Benecke, R. and Classen, J. (1999) 'Differential effects on motorcortical inhibition induced by blockade of GABA uptake in humans', *The Journal of physiology*, 517(Pt 2), pp. 591 – 597. doi:10.1111/j.1469-7793.1999.0591t.x.
- Wheaton, L. A., Bohlhalter, S., Nolte, G., Shibasaki, H., Hattori, N., Fridman, E., Vorbach, S., Grafman, J. and Hallett M. (2008) 'Cortico-cortical networks in patients with ideomotor apraxia as revealed by EEG coherence analysis', *Neuroscience letters*, 433(2), pp. 87 – 92. doi: 10.1016/j.neulet.2007.12.065.
- Wheaton, L.A., Shibasaki, H. and Hallett, M. (2005) 'Temporal activation pattern of parietal and premotor areas related to praxis movements', *Clin Neurophysiol*, 116(5), pp. 1201 1212. doi: 10.1016/j.clinph.2005.01.001.
- Whishaw, I.Q., Alaverdashvili, M. and Kolb, B. (2008) 'The problem of relating plasticity and skilled reaching after motor cortex stroke in the rat', *Behav Brain Res*, 192(1), pp. 124 – 136. doi: 10.1016/j.bbr.2007.12.026.

- Whishaw, I.Q., Gorny, B. (1996) 'Does the red nucleus provide the tonic support against which fractionated movements occur? A study on forepaw movements used in skilled reaching by the rat', *Behav Brain Res*, 74(1-2), pp. 79 90. doi: 10.1016/0166-4328(95)00161-1.
- Whishaw, I.Q., Gorny, B. and Sarna, J. (1998) 'Paw and limb use in skilled and spontaneous reaching after pyramidal tract, red nucleus and combined lesions in the rat: behavioral and anatomical dissociations', *Behav Brain Res*, 93(1-2), pp. 167 - 183. doi: 10.1016/s0166-4328(97)00152-6.
- White, Y. S., Bell, D. S., and Mellick, R. (1973) 'Sequelae to pneumoencephalography', *Journal of neurology, neurosurgery, and psychiatry*, 36(1), pp. 146–151. Available at: https://jnnp.bmj.com/content/jnnp/36/1/146.full.pdf (Accessed on 13 September 2018).
- Whitlock, J.R. (2017) 'Posterior Parietal Cortex', Curr Biol, 24(14), pp. R691-R695. doi: 10.1016/j.cub.2017.06.007.
- Wiesendanger, R. and Wiesendanger, M. (1985) 'Cerebello-cortical linkage in the monkey as revealed by transcellular labeling with the lectin wheat germ agglutinin conjugated to the marker horseradish peroxidase', *Exp Brain Res*, 59(1), pp. 105 – 117. doi: 10.1007/BF00237671.
- Winer, B.J., Brown, D.R., Michels, K.M. (1991) 'Statistical principles in experimental design -3rd Edition', New York, US: McGraw-Hill, in Smith P. F. (2017) 'A Guerilla Guide to Common Problems in 'Neurostatistics': Essential Statistical Topics in Neuroscience', Journal of undergraduate neuroscience education: JUNE: a publication of FUN, Faculty for Undergraduate Neuroscience, 16(1), R1–R12.
- Winstein, C.J, Grafton, S.T. and Pohl, P.S. (1997) 'Motor task difficulty and brain activity: investigation of goal-directed reciprocal aiming using positron emission tomography', J *Neurophysiol*, 77(3), pp. 1581 – 1594. doi: 10.1152/jn.1997.77.3.1581.
- Wise, S.P. (1985) 'The primate premotor cortex: past, present, and preparatory', *Annu Rev Neurosci*, 8, pp. 1–19. doi: <u>10.1146/annurev.ne.08.030185.000245</u>.
- Wolpaw, J.R., Birbaumer, N., McFarland, D.J., Pfurtscheller, G., and Vaughan, T.M. (2002) 'Brain-computer interfaces for communication and control', *Clin Neurophysiol*, 113(6), pp. 767 – 791. doi: 10.1016/S1388-2457(02)00057-3.
- Woo, S.H., Kim, K.H. and Lee, K.M. (2009) 'The role of the right posterior parietal cortex in temporal order judgment', *Brain Cogn*, 69(2), pp. 337 343. doi: 10.1016/j.bandc.2008.08.006.
- World Medical Association. (2013) 'World Medical Association Declaration of Helsinki: Ethical Principles for Medical Research Involving Human Subjects', JAMA, 310(20), pp. 2191 – 2194. doi: 10.1001/jama.2013.281053.
- Xing, C., Arai, K., Lo, E. H. and Hommel, M. (2012) 'Pathophysiologic cascades in ischemic stroke', *International journal of stroke: official journal of the International Stroke Society*, 7(5), pp. 378 – 385. doi:10.1111/j.1747-4949.2012.00839.x.

- Xu, G. Q., Lan, Y., Zhang, Q., Liu, D. X., He, X. F. and Lin, T. (2016) '1-Hz Repetitive Transcranial Magnetic Stimulation over the Posterior Parietal Cortex Modulates Spatial Attention', *Frontiers in human neuroscience*, 10(article 38), pp. 1 – 7. doi:10.3389/fnhum.2016.00038.
- Yamagata, T., Nakayama, Y., Tanji, J. and Hoshi, E. (2012) 'Distinct information representation and processing for goal-directed behavior in the dorsolateral and ventrolateral prefrontal cortex and the dorsal premotor cortex', J Neurosci, 32(37), pp. 12934 – 12949. doi: 10.1523/JNEUROSCI.2398-12.2012.
- Yiannopoulou, K. G. and Papageorgiou, S. G. (2013) 'Current and future treatments for Alzheimer's disease. *Therapeutic advances in neurological disorders*.' 6(1), pp. 19–33. doi:10.1177/1756285612461679.
- Young, N. A., Collins, C. E. and Kaas, J. H. (2013) 'Cell and neuron densities in the primary motor cortex of primates', *Frontiers in neural circuits*, 7(30), pp. 1 - 11. doi:10.3389/fncir.2013.00030.
- Zanon, M., Battaglini, P. P., Jarmolowska, J., Pizzolato, G. and Busan, P. (2013) 'Long-range neural activity evoked by premotor cortex stimulation: a TMS/EEG co-registration study', *Frontiers in human neuroscience*, 7(article 808), pp. 1 – 18. doi:10.3389/fnhum.2013.00803.
- Ziemann, U., Tergau, F., Netz, J. and Hömberg, V. (1997) 'Delay in simple reaction time after focal transcranial magnetic stimulation of the human brain occurs at the final motor output stage', *Brain Res*, 744(1), pp. 32 – 40. doi: 10.1016/S0006-8993(96)01062-1.

Appendix 1. Ethical approval:

EXTERNAL AND STRATEGIC DEVELOPMENT SERVICES

uel.ac.uk/qa

Quality Assurance and Enhancement



13 June 2016

Dear Pegah Mohajer Shojaii,

| Project Title: | The Neural Mechanisms of Motor Learning in Healthy Adults Using Neuroimaging Technologies |
|-------------------------|--|
| Principal Investigator: | Professor Duncan Turner |
| Researcher: | Pegah Mohajer Shojaii |
| Reference Number: | UREC 1516 108 |

I am writing to confirm the outcome of your application to the University Research Ethics Committee (UREC), which was considered by UREC on Wednesday 18 May 2016.

The decision made by members of the Committee is **Approved**. The Committee's response is based on the protocol described in the application form and supporting documentation. Your study has received ethical approval from the date of this letter.

Please note the UREC Application Form for ethical approval has been revised. For future applications please use the revised application form which can be found on: https://uelac.sharepoint.com/ResearchInnovationandEnterprise/Pages/Ethics.aspx

The Committee would like to commend you on the presentation of this application for ethical approval.

Should you wish to make any changes in connection with your research project, this must be reported immediately to UREC. A Notification of Amendment form should be submitted for approval, accompanied by any additional or amended documents: http://www.uel.ac.uk/wwwmedia/schools/graduate/documents/Notification-of-Amendment-to-Approved-Ethics-App-150115.doc

Any adverse events that occur in connection with this research project must be reported immediately to UREC.



Docklands Campus, University Way, London E16 2RD Tel: +44 (0)/20 8223 3322 Fax: +44 (0)/20 8223 3394 MINICOM 020 8223 2853 Email: r.carter@uel.ac.uk EXTERNAL AND STRATEGIC DEVELOPMENT SERVICES uel.ac.uk/qa

Quality Assurance and Enhancement



Approved Research Site

I am pleased to confirm that the approval of the proposed research applies to the following research site.

| Research Site | Principal Investigator / Local Collaborator |
|---------------------------|--|
| University of East London | Professor Duncan Turner |

Approved Documents

The final list of documents reviewed and approved by the Committee is as follows:

| Document | Version | Date |
|---|---------|-------------|
| UREC application form | 2.0 | 3 June 2016 |
| Participant Information Sheet | 2.0 | 3 June 2016 |
| Consent Form | 2.0 | 3 June 2016 |
| Annexe 3 – Medical Questionnaire | 2.0 | 3 June 2016 |
| Annexe 4 - recruitment advertisement | 2.0 | 3 June 2016 |

Approval is given on the understanding that the UEL Code of Practice in Research is adhered to.

The University will periodically audit a random sample of applications for ethical approval, to ensure that the research study is conducted in compliance with the consent given by the ethics Committee and to the highest standards of rigour and integrity.

Please note, it is your responsibility to retain this letter for your records.

With the Committee's best wishes for the success of this project.

Yours sincerely,



Catherine Fieulleteau Research Integrity and Ethics Manager University Research Ethics Committee (UREC) Email: <u>researchethics@uel.ac.uk</u>

Docklands Campus, University Way, London E16 2RD Tel: +44 (0)20 8223 3322 Fax: +44 (0)20 8223 3394 MINICOM 020 8223 2853 Email: r.carter@uel.ac.uk





Appendix 2. Participant information sheet and consent form:



University of East London

Stratford Campus: School of Sport, Health and Bioscience, London, E15 4LZ

University Research Ethics Committee

If you have any queries regarding the conduct of the programme in which you are being asked to participate, please contact:

Catherine Fieulleteau, Research Integrity and Ethics Manager, Graduate School, EB 1.43 University of East London, Docklands Campus, London E16 2RD (Telephone: 020 8223 6683, Email: <u>researchethics@uel.ac.uk</u>).

The Principal Investigator(s)

1) Director of studies:

Professor Duncan Turner,

Address:

AE5.23, School of Health, Sport and Bioscience, University of East London, Stratford Campus, Romford Road, London, E15 4LZ,

(Telephone: 020 8223 4514, Email: d.l.turner@uel.ac.uk)

2) Investigator/PhD student:

Pegah Mohajer Shojaii

(Email address: p.shojaii@uel.ac.uk)

Consent to Participate in a Research Study

The purpose of this letter is to provide you with the information that you need to consider in deciding whether to participate in this study.

Project Title

The Neural Mechanisms of Motor Learning in Healthy Adults Using Neuroimaging Technologies

Project Description

<u>Research aim</u>: This research project will aim to explore the neural mechanisms of motor adaptation through a motor learning task with a computerized robot.

Procedure:

The motor learning task will require you to sit in a customised chair, facing a screen. Here, you will be asked to make arm reaching movements in a particular direction, with your arm placed in a device connected to the robot. The robot will then apply force-fields which can influence the output of your arm reaching movement.

Whilst you carry out this task, your muscle activity (EMG) and motor evoked potentials (MEPs) will be recorded through surface electrodes placed on your arm muscles, and also through a safe neuroimaging method, known as transcranial magnetic stimulation (TMS).

Risks:

There are no medical risks involved in the study:

All the methods that will be used in this research have already been established in the Neurorehabilitation Unit over the last 10 years and we do not foresee any technical or health and safety issues. We have tested over 200 subjects over 10 years with no adverse events in all studies.

Study length:

The experiment will take approximately two hours and a half to complete.

Confidentiality of the Data

As this project will have 12-15 participants, the risk of your data being identified is increased, however anonymity steps will be taken in order to ensure your privacy. For example, you will be given a participant number, and your data will be stored under that number. Your consent form and medical questionnaire will be stored separately. After the PhD project has been complete, your data will be stored in the research lab for >10 years.

Location

The research will be carried out in a Neurorehabilitation Unit (NRU), located in the Stratford campus of the University of East London, within the School of Health, Sport and Bioscience (HSB).

Remuneration

You will receive a £20 Amazon voucher after completing the experiment.

Disclaimer

You are not obliged to take part in this study, and are free to withdraw at any time during tests. Should you choose to withdraw from the programme you may do so without disadvantage to yourself and without any obligation to give a reason.

UNIVERSITY OF EAST LONDON



Consent to Participate in a Programme Involving the Use of Human Participants.

<u>The Neural Mechanisms of Motor Learning in Healthy Adults Using Neuroimaging</u> <u>Technologies</u>

Researchers:

The principle investigator: Professor Duncan Turner (BSc, PhD)

Investigator/PhD student: Pegah Mohajer Shojaii (BSc, MSc)

Please tick as appropriate:

| | YES | NO |
|---|-----|----|
| I have the read the information leaflet relating to the above programme of research in which I have been asked to participate and have been given a copy | | |
| to keep. The nature and purposes of the research have been explained to me, and I have had the opportunity to discuss the details and ask questions about | | |
| this information. I understand what is being proposed and the procedures in which I will be involved have been explained to me. | | |
| I understand that my involvement in this study, and particular data from this research, will remain strictly confidential as far as possible. Only the | | |
| researchers involved in the study will have access to the data. (Please see below) | | |
| | | |
| I understand that maintaining strict confidentiality is subject to the following limitations: | | |
| If the sample size is small, or focus groups are used state that that this may have implications for confidentiality / anonymity. | | |
| My confidentiality will be maintained unless a disclosure is made that indicates that I or someone else is at serious risk of harm. Such disclosures may be reported to the relevant authority. | | |
| I will be pseudo-anonymised in publications that will arise from the research. | | |
| I am aware that the proposed methods of publication dissemination of | |
|--|--|
| research findings, includes: | |
| - Peer reviewed iournals | |
| - Non-peer reviewed journals | |
| - Peer reviewed books | |
| - Publication in media or website | |
| - Conference presentation | |
| - Internal report | |
| - Promotional report and materials | |
| - Dissertation/Thesis | |
| Presentation to participants or relevant community group | |
| I give permission for your team to use the data in future research. | |
| | |
| | |
| I give permission to be to be contacted for future research studies by your | |
| team. | |
| | |
| It has been explained to me what will happen once the programme has been | |
| completed. | |
| | |
| Lunderstand that my participation in this study is entirely voluntary, and Lam | |
| free to withdraw at any time during the research without disadvantage to | |
| myself and without being obliged to give any reason. I understand that my data | |
| can be withdrawn up to the point of data analysis and that after this point it | |
| may not be possible. | |
| | |
| | |
| I hereby freely and fully consent to participate in the study which has been fully | |
| explained to me and for the information obtained to be used in relevant | |
| research publications. | |
| | |
| | |

| Participant's Name (BLOCK CAPITALS) |
|--------------------------------------|
| Participant's Signature |
| Investigator's Name (BLOCK CAPITALS) |
| Investigator's Signature |
| Date: |

Appendix 3. Medical questionnaire:



<u>Project title:</u> The Neural Mechanisms of Motor Learning in Healthy Adults Using Neuroimaging Technologies

Medical Questionnaire

This questionnaire is designed for the purpose of understanding if you have any conditions that may result in excluding you from the research being conducted. All the data supplied in this questionnaire will remain private and confidential.

Date: Participant Number:

Date of birth: Gender: M / F Handedness: Right / Left

Do you take any medication or recreational drugs? Y / N If yes, state below:

How many units of alcohol do you consume per week on average:

Have you had any surgeries in the past? If yes, state below:

272

Do you have a history of head or spinal injury (e.g concussion, car crash whiplash):

Do you have any chronic illness? Y / N If Yes, state below:

Do you have any neurological disorders (e.g. stroke, spinal cord injury, colour blindness, dyslexia, Parkinson's or Alzheimer's disease, epilepsy/seizures, family history of these)? Y / N If Yes, state below:

Do you have a psychiatric history (e.g. schizophrenia, bipolar disorder, depression, obsessive compulsive disorders, panic disorder, family history of these)? Y / N If Yes, state below

Do you have a cardio-respiratory Disease (asthma, angina, high blood pressure, respiratory distress)? Y / N If yes, state below:

Do you have a musculoskeletal condition: (bone fracture, muscle or ligament tear)? Y / N If yes, state below:

Do you have any metal or electrical implants (e.g. pacemakers, intracranial plates, skeletal pins, vascular clips)? Y / N If yes, state below:

If you are a woman are you pregnant or experiencing altered menstrual cycles? Y / N If yes, state below:

END OF QUESTIONNAIRE